

San Bernardino Valley Water **Conservation District and San Bernardino Valley Municipal Water** District

June 2009



Prepared in association with GEI



# Santa Ana River Groundwater Recharge Optimization Study

# **Final Report**

June 2009

Project Number 1343019

Prepared for:

San Bernardino Valley Water Conservation District and San Bernardino Valley Municipal Water District

Prepared by:

MWH 618 Michillinda Avenue, Suite 200 Arcadia, CA 91007

# **Table of Contents**

Section	Page Number					
Section 1 Introduction1-1						
1.1	Project Background					
1.2	Summary of Field Testing and Analysis					
1.3	CEQA Issues Related to the Conceptual Design					
1.4	Final Report Organization					
Section						
2.1	Conveyance of 195 cfs					
2.2	Conveyance of 300 cfs					
2.3	Conveyance of 500 cfs					
2.4	Optional Facilities					
2.5	Summary and Estimated Costs					
Section 3 Operations, Maintenance, and Monitoring						
3.1	Mechanisms Affecting Infiltration Rates					
3.2	Preventative Measures					
3.3	Best Management Practices					
3.4	Current Practices					
3.5	Recommendations					
Section	4 References					

## **List of Figures**

- 1-1 Location Map
- 2-1 Proposed Water Facilities Phases, 300 cfs
- 2-2 Proposed Water Facilities Phases, 500 cfs
- 2-3 Proposed Water Facilities Phases, 500 cfs with Optional Facilities
- 2-4 Intake and Cuttle Weir Plan
- 2-5 Intake and Cuttle Weir Plan B
- 2-6 Pond Turnout Structure
- 2-7 Intake and Cuttle Weir Plan C
- 2-8 Alternate Intake and Cuttle Weir Plan D
- 2-9 Alternate Intake and Cuttle Weir Plan E
- 2-10 Trashrack Structure Plan
- 2-11 Existing Earthen Canal Section
- 2-12 Drop Structure
- 2-13 Proposed Typical Earth Lined Canal Section
- 2-14 Greenspot Road Culvert Plan
- 2-15 Diversion Structure Modification
- 2-16 Borrow Pit Overflow

- 2-17 Pond Inlet Plan and Profile
- 2-18 Pond Overflow Plan and Profile
- 2-19 Pond 3 Discharge Structure
- 2-20 Borrow Pit Diversion Structure Plan
- 2-21 Additional Settling Facilities
- 2-22 Estimated Cost at Different Flow Rates

## Appendices

- Appendix A Technical Memorandum 1: Documentation of Existing Information and Field Work Plan
- Appendix B Technical Memorandum 2: Documentation of Field Work and Recharge Capacity Estimate
- Appendix C Cost Estimate Details

# Section 1 Introduction

The San Bernardino Valley Water Conservation District (Conservation District) and the San Bernardino Valley Municipal Water District (Municipal District), "Districts," retained MWH Americas, Inc., in association with GEI Consultants (collectively referred to as the MWH team) to evaluate the capacity of the existing Conservation District facilities along the Santa Ana River to capture and store the proposed spreading objectives identified in the Upper Santa Ana River Watershed Integrated Regional Water Management Plan (IRWMP). The IRWMP established target amounts for groundwater recharge in the project area. The maximum yearly volume identified under "Scenario A" of the IRWMP is 80,000 acre-feet. The maximum instantaneous flow rate, as defined by the Districts based on the Environmental Impact Report for the Santa Ana River Water Right Applications for Supplemental Water Supply (SBVMWD, 2007), is 500 cubic feet per second (cfs). Although it is understood that this maximum flow rate is anticipated for short periods only, it is desirable for the Districts to have the capability to accommodate this flow on a temporary basis to maximize storage of surface water when it is available.

The goals of this project are to evaluate the ability of the existing facilities to meet the recharge objective of 500 cfs, recommend and provide conceptual designs for improvements to meet the objective of 500 cfs (if needed), and provide recommendations on the operations and maintenance activities that will maximize recharge capacity.

# 1.1 PROJECT BACKGROUND

The Conservation District owns and operates surface water diversion and conveyance facilities and groundwater recharge spreading facilities adjacent to the Santa Ana River (SAR) along Greenspot Road, south and west of the Seven Oaks Dam in Mentone, San Bernardino County, California (**Figure 1-1**). The Conservation District was formed in 1932 to obtain water for conservation purposes, essentially replacing the Water Conservation Association formed in 1910 (SBVWCD, 1994). The majority of the Conservation District facilities have been in place since the 1930s for the purpose of diverting and recharging water from the Santa Ana River.

A relatively recent development in recharge operations is the construction of the Seven Oaks Dam, which is an earth and rock fill dam located approximately 1 mile upstream of the Conservation District's diversion. Construction of the dam by the United States Army Corps of Engineers (USACE), began in August 1994 and was completed in November 1999. Material was excavated for the dam from an approximately 240 acre area (herein called the "Borrow Pit") that formerly contained a series of recharge ditches and ponds operated by the Conservation District. Subsequent to the completion of the dam, three infiltration ponds were constructed at the east end of the Borrow Pit (labeled Ponds 1 through 3 on Figure 1-1).



The remaining recharge basins (Ponds 9 through 17) and the Borrow Pit are located south and west of Greenspot Road, and north of the Santa Ana River. Flow from the Santa Ana River below the Seven Oaks Dam is diverted at a concrete structure (Intake Structure) behind a small dam-like impoundment called the Cuttle Weir. From the Intake Structure, water flows through various culverts, tunnels and other structures where it can be diverted to various ponds including those in the Borrow Pit.

The Santa Ana River Groundwater Recharge Optimization Study consists of three primary tasks:

Task 1: Evaluate Existing Groundwater Recharge Spreading Facilities and Operations

Task 2: Establish Spreading Objectives

Task 3: Perform Feasibility Analysis and Prepare Conceptual Design to Meet Spreading Objectives

In April and May 2008, the MWH team reviewed pertinent existing information and prepared a workplan for collection of additional field data and reporting the results of the field data. This information was contained in Technical Memorandum 1 (TM-1, MWH, 2008). TM-1 is included to this final report as **Appendix A**. The field work was designed to identify factors which may limit the volume of water that could be recharged, including:

- Inadequate capacity of diversion or conveyance facilities
- Compaction of shallow soils in the Borrow Pit due to heavy equipment use during construction of the pit
- Presence of fine-grained material that "clog" the surface of the Borrow Pit
- Shallow groundwater which mounds to the surface
- Presence of horizontal or vertical barriers to groundwater flow, such as clay layers or faults

The field work was also designed to gather additional data on shallow soil and aquifer conditions and resolve data gaps regarding well locations and surface elevations at the site.

Results of the field investigation is provided in Technical Memorandum 2 (TM-2) which describes factors which limit recharge, and documents the recharge capacity of the existing facilities. TM-2 is summarized in the following section of this introduction, and is included in its entirety as **Appendix B**.

This final report provides recommendations for improvement of existing facilities, conceptual design of new facilities, and maintenance and operational methods to increase recharge capacity.

## 1.2 SUMMARY OF FIELD TESTING AND ANALYSIS

Field testing and analysis described in TM-2 was conducted for the purpose of quantification of the capacity of the existing diversion, conveyance and percolation facilities. This was accomplished by review of pertinent historical information and reports, preparation of a workplan for collection of field data, collection of field data, and analysis of the existing and new field data. TM-2 (**Appendix B**) presents the methods and results of collection of field data,

analysis of the data, conclusions relative to the capacity of existing facilities. The following is a brief summary of the testing activities and the conclusions of that work.

The field work was conducted during June through December, 2008 and consisted of:

- Field flow testing of the diversion and conveyance facilities
- Survey of diversion works and conveyance (measurements of dimensions and slopes)
- Soil investigation consisting of:
  - Excavation of 15 trenches.
  - Collection of 72 surface soil samples.
  - Drilling, sampling, and lithologic logging of 7 borings to a maximum depth of 157 feet.
  - Laboratory analysis of 75 samples for grain size analysis, and 16 of these samples for analysis of hydraulic conductivity.
- Construction of 6 monitoring wells and installation of automated monitoring equipment
- Several types of percolation tests at existing recharge ponds
- Physical surveys of existing well locations and elevations

Significant conclusions from the field testing and analysis include:

- The sedimentary materials underlying the recharge facilities form an unconfined aquifer consisting of permeable, coarse, sandy gravel and/or gravelly sand. No significant, laterally-continuous strata of low permeability were found that would prevent the downward percolation of recharge water.
- Some existing ponds have a thin layer of silt and/or clay derived from the introduction of turbid recharge water which limits percolation capacity.
- Faulting associated with the San Andreas Fault Zone has created a groundwater barrier which limits recharge capacity on the eastern portion of the site causing shallow groundwater that surfaces or "daylights" east (up-gradient) of this barrier.
- During high runoff periods such as those that occurred in 1980, 1993, 1998 and 2005, the regional area in the vicinity of the recharge facilities may become saturated with shallow groundwater, limiting recharge in all of the facilities. However, these events have been temporary, and may occur at a different frequency depending on the operation of the Seven Oaks Dam. This finding is consistent with regional groundwater modeling studies, which predict saturated conditions for very short durations during wet years (SAIC, 2007).
- The yearly recharge goal of 80,000 acre-feet identified in the IRWMP is possible with the construction of new infiltration, diversion, and conveyance facilities, assuming ambient groundwater levels west of the groundwater barrier are approximately 200 feet below ground level to accept this water.
- Groundwater mounding (whereby the water table rises to the ground surface) may to occur even under relatively dry conditions if a percolation rate of 500 cfs is sustained for more that approximately 30 days. (**Appendix B**).

- Downstream of the Intake Structure and Cuttle Weir, earthen canals limit the capacity of the conveyance facilities to approximately 300 cfs.
- The recharge capacity of the existing percolation ponds at the SAR recharge facility west of the groundwater barrier is approximately 195 cfs.

Existing spreading grounds operation and maintenance procedures were evaluated in comparison to "best management practices" of other agencies in California and the southwestern United States. This evaluation concluded that the Conservation District is generally following good operations and maintenance practices. However, a lack of data collection limits the ability to quantify spreading grounds performance. Recommendations are presented in Section 3 regarding improved operations and monitoring. In particular, care should be taken to avoid recharging water that has high levels of suspended solids and turbidity.

# 1.3 CEQA ISSUES RELATED TO THE CONCEPTUAL DESIGN

The expansion of additional basins and other facilities within the area known as Phases 1 through 3 (Figure 2-1) was analyzed as part of the programmatic Final Environmental Impact Report for the Upper Santa Ana River Wash Land Management and Habitat Conservation Plan (Wash Plan) project (SBVWCD, 2008), which evaluated the potential biological impacts from percolation facilities described in this report. Biological impacts were assessed by projecting the same ratio of ground disturbance for future facilities as exist presently in the Conservation District's spreading area. The current ratio of wetted and maintained area is approximately 31 percent of the overall land area. This wetted area ratio has been used in the conceptual design of additional facilities in accordance with the Wash Plan EIR.

Other impacts associated with the conceptual designs described in this report, include the areas of noise, air quality, hydrology, and traffic. The Conservation District will need to undertake tiered additional environmental review of future construction based on the project description contained in Section 2 of this report. In particular, the proposed modification of earthen canals described in this report may require additional analysis.

The conceptual design of the modification to the recharge facilities to accommodate 500 cfs described in this report constitutes a detailed enough project description for CEQA purposes. The MWH team recommends that the Districts have a legal review and finding for the Districts' adoption. This legal review should provide a statement indicating the extent to which project level environmental analysis is required in consideration of the work already completed programmatic Wash Environmental Impact Report (EIR).

## 1.4 FINAL REPORT ORGANIZATION

This final report provides conceptual design and operational recommendations to meet the IRWMP short-term maximum flow rate of 500 cfs and annual maximum volume of 80,000 acrefeet, and is organized as follows:

**Section 1** introduces the project goals, presents background information, CEQA-related issues, and provides a summary of previous reports which are included in the final report as appendices.

Section 2 provides the recommendations and conceptual design of physical improvements to meet the project goals.

Section 3 provides the recommendations of operational maintenance and monitoring methods recommended to meet the recharge goal defined in the IRWMP.

**Section 4** provides a list of the cited references.

Previous technical memoranda, Technical Memorandum-1 and Technical Memorandum-2, completed during the conduct of this work are included in Appendix A and B, respectively. Appendix C contains cost estimating details.

# Section 2 Conceptual Design

This section describes the improvements and additional facilities recommended to achieve three different flow rates of 195 cfs, 300 cfs, and 500 cfs. These discreet flow rates were selected for analysis for the following reasons:

- The flow rate of 195 cfs was selected because it is the estimated infiltration capacity of existing recharge facilities.
- The flow rate of 300 cfs was considered because it is the practical maximum flow rate of many of the diversion and conveyance facilities, in particular the earthen canals which currently convey water to the recharge ponds.
- A flow rate of 500 cfs was considered because it is the ultimate goal of the project.

**Figure 2-1, Figure 2-2 and Figure 2-3** provide an overview of the locations of facilities for which modifications are recommended. New recharge ponds will not be located east of the vertical barrier identified as part of TM-2, but within areas identified as Phase 1 through 3. More detail on physical dimensions and estimated capacities of existing facilities are given in **Appendix B**.

# 2.1 CONVEYANCE OF 195 CFS

The existing recharge facilities consist of the Borrow Pit and recharge ponds (Ponds) 9 through 17 and D west of the Borrow Pit (Figure 1-1). Estimates of percolation rates at Ponds 9 through 17 and D were obtained using the percolation testing methods as described in **Appendix B**. From these tests, the total combined recharge rate of the Ponds is approximately 145 cfs. The additional capacity of the west end of the Borrow Pit (not including Ponds 1 through 3) is estimated by the Conservation District to be approximately 50 cfs. Therefore, the current recharge capacity of the Borrow Pit and ponds is estimated to be 195 cfs (**Appendix B**).

As described in **Appendix B**, the practical conveyance limitation for most of the diversion and conveyance facilities is approximately 300 cfs. Above this flow rate, high velocity and turbulent flow in unlined channel sections is observed. However, modifications to the conveyance system will not be required to convey 195 cfs to the Borrow Pit and existing recharge ponds.







# 2.2 CONVEYANCE OF 300 CFS

A flow rate of 300 cfs is considered as part of the conceptual design based on the limitations of the existing earthen conveyance canal. At 300 cfs, the calculated canal velocities exceed the maximum recommended velocity for canals consisting of rocks and cobbles (**Appendix B**). Observations made during the flow test noted that erosion of channel and movement of boulders started at approximately 300 cfs and was pronounced at flows greater than 300 cfs. Therefore, maintaining a flow rate of less than 300 cfs is desired to prevent canal erosion. At a flow rate of 300 cfs, additional recharge ponds will be required to utilize the conveyance system to its full capacity. Modifications to the Intake Structure will also be required as described below.

#### **Modification of the Intake Structure**

The hydraulic capacity of the Cuttle Weir and the Intake Structure are interrelated. If the height of the Cuttle Weir was increased, the resulting increased water surface elevation would allow greater flow rates through the Intake Structure. At an elevation of 1954.75 feet, water levels

reached the top of the Cuttle Weir at a flow rate of approximately 300 cfs with plywood in place at the weir notch as noted in the adjacent photo. The corresponding theoretical total discharge through the six gates is approximately 362 cfs at this elevation. However, due to the accumulation of debris and boulders in front of the Intake Structure's gates, the MWH team concluded that the practical flow capacity for diversion of flow into the Intake Structure is 250 cfs. In order to meet the 300 cfs capacity, the



modifications described below are recommended to eliminate debris accumulation in front of the Intake Structure. As noted above, a flow rate of approximately 300 cfs was reached with the plywood in place at the weir notch. Therefore, no modifications to the Cuttle Weir are required to reach a flow rate of 300 cfs.

In order to prevent accumulation of boulders and debris in front of the Intake Structure, the MWH team recommends re-grading the river channel to slope away from the structure and constructing a gabion deflection wall as shown in **Figure 2-4**. The gabion deflection wall would deflect the boulders at high flows and guide them toward the river bank on the opposite side from the Intake Structure. The sloped channel should be steep enough to prevent rocks from migrating from the opposite river bank to the Intake Structure at flows higher than 250 cfs. The MWH team recommends that the river sediment and rock loading be evaluated during final design to provide design criteria for the channel re-grading, gabion wall and other related facilities. The evaluation will also provide the District's with an idea of the amount of maintenance that will be required (excavating boulders etc.).



Floating debris accumulation at the Intake Structure's gates also reduces the diversion capacity. The MWH team recommends installing a floating log boom as shown in Figures 2-4 **and 2-5**. The floating log boom is a low cost and effective solution to controlling floating debris. The log boom would be connected to the northeast corner of the Intake Structure and run diagonally toward the Cuttle Weir notch. It would be connected to the Cuttle Weir just west of the notch. This way the debris could be flushed downstream thru the notch by removing stop logs. At extremely high flows the log boom would be removed to prevent damage.

#### **Additional Facilities**

To fully utilize the conveyance capacity of the existing facilities at a flow rate of 300 cfs, additional recharge ponds west of the Borrow Pit will be required. An additional 52.1 wetted acres is recommended as part of this expansion. The additional ponds will only be a portion of the proposed Phase 2 recharge ponds as shown on Figure 2-1. The additional recharge ponds would allow for percolation of the additional 105 cfs of capacity needed to add to the existing capacity of 195 cfs, reaching a total of 300 cfs.

Because of the additional recharge ponds, modifications of the existing (earthen) Main Canal will be required. The Main Canal runs from the end of the Sand Box to the northwest corner of the Borrow Pit. At this location, the existing canal divides into two separate canals, one continuing west to Ponds 13 and 17 and one running to Pond 10. A new canal that runs west along the recharge ponds within the proposed Phase 2 recharge basins will be required and the existing canals west of the Borrow Pit will no longer be needed, as shown on Figure 2-1.

This new portion of the Main Canal will consist of pond turnout structures that would allow separate discharge to each of the proposed recharge basins; therefore, providing the capability of isolating each pond during maintenance. The structures will be constructed of concrete with Waterman C-10 canal gates, as shown on **Figure 2-6**. Each turnout structure will be sized to allow flow that is twice the percolation rate, which assumed to be 4 feet/day, and the wetted area of the recharge pond(s) fed by that turnout. The additional flow will be used to fill the proposed recharge ponds to their required depths and then the gates will be used to reduce the flow to the percolation rate of each pond. As noted in Section 3, flow measuring devices are recommended at each turnout structure to monitor flows to discreet recharge areas. The installation of flow measuring devices will allow for monitoring percolation rates and potential degradation of percolation rates, which is important information needed for adaptive management of operations and maintenance.





# 2.3 CONVEYANCE OF 500 CFS

Improvements will be required for the existing diversion and conveyance facilities to convey a flow of 500 cfs, which is the ultimate goal of the Districts. The maximum instantaneous flow rate is based on the Environmental Impact Report for the Santa Ana River Water Right Applications for Supplemental Water Supply (SBVMWD, 2007). Review of the future frequency and duration of the 500 cfs estimate was not a portion of this study. Figure 2-1 shows locations of the facilities to be improved. As stated above, the practical conveyance limitation for most of the conveyance system was reached at approximately 300 cfs. However, the Rock Structure and the Parshall Flume have the capability to convey greater than 1,000 cfs (**Appendix B**), and will not require modifications. A description of the improvements and additional facilities recommended to achieve a flow rate of 500 cfs are provided below.

#### **Modifications of Existing Diversion and Conveyance Facilities**

#### Intake Structure and Cuttle Weir

The Cuttle Weir/Intake Structure in its current configuration has an estimated capacity of 250 cfs. If the improvements made for boulder and debris handling as recommended in the 300 cfs alternative are made, then the capacity could theoretically be increased to 362 cfs at a water level elevation of 1954.75. In addition to these improvements, the water level in the intake forebay will need to be raised to an elevation of 1955.6 in order to convey 500 cfs thru the Intake Structure. In order to accomplish this, it is recommended that the Cuttle Weir be raised to an elevation of 1956.75. This provides at least 12 inches of freeboard to provide flexibility and spillage over the weir due to turbulent water. The weir would be raised using reinforcement and rock masonry (same as existing) as shown in Figure 2-5 and **Figure 2-7**. A sluice gate would be installed in the notch to allow water to be passed downstream and to control water level elevations. An access platform with handrail would be constructed over the notch to allow access to the sluice gate operator. Improvements will also be required to the existing weir access to bring it up to current OSHA standards. This includes additional handrails and ladder cage. As part of these improvements, Gate #3 at the intake would be modified to allow it to open 4 feet.

As an alternate to the sluice gate, the Districts may want to consider installing an adjustable weir as shown on **Figures 2-8** and **2-9**. This is a more expensive option than the sluice gate and will require substantial modifications to the existing weir structure. A structural engineer and geotechnical foundation engineer will be required to investigate the existing structure foundation and make recommendations for the adjustable weir foundation. The adjustable weir would be approximately 17 feet long and almost 5 feet high at maximum height. The weir operates by adjusting the air volume in a rubber bladder to raise or lower the weir plate. This air volume would be controlled from a panel and air compressor located on the top deck of the intake. The advantage of the adjustable weir over a sluice gate is that it provides more water level control and can be completely lowered to allow boulders and other debris to pass over it.







#### **Closed Conduit (Tunnel) and Sand Box**

As described in **Appendix B**, the calculated capacity of the Closed Conduit/Tunnel is greater than 500 cfs for open channel flow with the 50-foot long overflow section. Therefore, improvements to the Closed Conduit/Tunnel are not required to achieve the desired capacity of 500 cfs.

The Sand Box capacity is currently limited by the maximum gate opening at the structure's downstream end. The structure consists of two 6 feet wide bulkhead gates, each having a maximum opening of 4 feet based on thread measurements taken on each gate stem. The hydraulic model described in Appendix B suggests that at a flow rate of 500 cfs, the water depth at the Sand Box gate structure would be about 4.8 feet. Therefore, the gates will require modification to increase the gate opening heights to 5 feet in order to increase this facility's carrying capacity to 500 cfs, as shown on **Figure 2-10**.

Figure 2-10 also shows a trashrack structure downstream of the Sand Box. This structure is provided as an option to the Districts to capture any debris not caught by the log booms before it enters the earthen canal. There is an existing 50 foot overflow section in the Sand Box that will provide overflow into the Santa Ana River in the event that the trash rack gets clogged.

#### Earthen Canal

The existing Main Canal has a practical conveyance limitation of 300 cfs due to high velocity and turbulent flow (**Figure 2-11**). Therefore, improvements are required to increase the carrying capacity of the canal as well as prevent canal erosion. The canal considered for this section starts at the end of the sand box and terminates at the Phase 3 recharge ponds.

In order to reduce the velocity in the canal, the addition of drop structures (**Figure 2-12**) and a decrease in the canal slope are recommended, which will lead to changes in the canal cross section (**Figure 2-13**). The slope of the canal was evaluated using Bentley FlowMaster Version 8 software (Flowmaster). Using the following parameters, canal slopes varying from 1% to 3% were considered:

- Flow capacity of 500 cfs
- Maximum velocity of 6.5 feet per second
- Published an n-value of 0.050 (Daugherty and Franzini, 1979)
- Canal side slopes of 1.5 horizontal to 1 vertical (practical standard for earthen canals)

At a 1% slope, a canal bottom width of 12 feet is required to meet the parameters listed above. The normal water depth in the channel would be 4.41 feet deep. Therefore, the canal will be designed with a depth of 6.5 feet, which would provide 2 feet of freeboard (**Figure 2-13**). With a slope of 3%, the canal bottom width of 60 feet is required and the channel will have a normal water depth of 1.33 feet. Providing the same amount of freeboard as the 1% slope, the depth of the canal would be 3.5 feet. The canal bottom width and depth for various slopes are shown on **Table 2-1** below along with the number of drop structures required for each slope. Ten foot drop structures are recommended.









Canal Slope	Bottom Width (feet)	Normal Water Depth (feet)	Canal Depth (feet)	No. of Drop Structures	Estimated Costs
1%	12	4.41	6.5	36	\$1,850,000
1.5%	30	2.46	4.5	28	\$2,100,000
2%	40	1.91	4	20	\$2,200,000
2.5%	50	1.57	3.5	13	\$2,100,000
3%	60	1.33	3.5	5	\$2,000,000

 Table 2-1

 Estimated Costs of Drop Structures Based on Canal Slope

Canal widths of 30 feet and greater are not recommended because a uniform water depth may not be obtained throughout the width of the canal. The existing rocks and boulders in the earthen canal cause irregularities in the canal invert; therefore, a small meandering stream may develop. Subsequently, high velocities may develop causing erosion in the canal.

The estimated costs for the different canal sections are also provided in the **Table 2-1**. The 12foot bottom canal is the lowest cost alternative. The MWH Team recommends the 12-foot bottom width canal for the conveyance of 500 cfs to the existing and new groundwater recharge facilities.

## **Greenspot Road Culvert**

A point of congestion or obstruction in the conveyance system is the culvert at Greenspot Road. The hydraulic analysis determined that the maximum capacity of the culvert is approximately 372 cfs.

Recommended modifications of the culvert include replacing the 10 feet wide by 4 feet high culvert with two 6 feet wide by 6 feet high box culverts. This would increase the culvert capacity to 500 cfs. Modifications will also include a 20-foot long "broken back" concrete transition upstream and downstream of the culvert. The transitions will provide gradual change from the canal section to the box culverts (**Figure 2-14**).



#### **Diversion Structure**

The existing diversion structure consists of two 4 feet by 8 feet diversion gates mounted on a concrete structure to the west (North Diversion Structure) and one 4 feet by 8 feet gate mounted on a concrete structure to the south (South Diversion Structure). See Figure 2-15 for the layout of the existing diversion structure. As described in Appendix B, the North Diversion Structure was able to pass 300 cfs with 2 feet of free board flow testing. Theoretical calculations show that the North Diversion Structure could pass 500 cfs at approximately 3 feet of water depth. However, this was analyzed based on existing canal slopes downstream remaining at approximately 3 percent grade. The canal slope being recommended by the MWH Team downstream of the North Diversion Gate is 1 percent. Because of this, the water level in the canal creates a backwater effect in the diversion structure thus increasing the water level required to convey 500 cfs. Based on theoretical calculations, the water depth through the North Diversion Structure needs to be 4.5 feet to convey 500 cfs. Currently, the diversion structure depth is 5 feet. At this time the MWH Team does not recommend raising the diversion structure walls. However, the existing gates should be modified to increase the opening from 4 feet to 5 feet. Also, a concrete diverging cone should be constructed at the entrance to the North Diversion Structure to allow smooth water separation and reduce turbulence.

The South Diversion Structure should be modified to add a second gate as shown on **Figure 2-15**. This will require the construction of a concrete foundation and wall. Otherwise, the modifications are similar to those recommended for the North Diversion Structure.

Currently, the bottom of the diversion structure is native rock and earth. It is recommended that the existing rock and earth be excavated and replaced with grouted rock. This will provide protection to the side walls and structures against scouring due to high velocities in the channel.

#### **Additional Facilities**

Improvements recommended above will be required to convey a flow of 500 cfs to the existing and new recharge ponds in the proposed Phases 1 through 3. To fully utilize the conveyance capacity of the existing facilities at a flow rate of 500 cfs, additional recharge ponds west of the Borrow Pit will be required. An additional 151.2 wetted acres will be required as part of this expansion. The additional recharge ponds would allow for percolation of the additional 305 cfs of capacity needed to add to the existing capacity of 195 cfs, reaching a total of 500 cfs. The additional recharge and conveyance facilities needed for this flow will be similar to the additional facilities needed for the flow of 300 cfs.

Because of the additional recharge ponds, modifications of the existing earth canal, the (earthen) Main Canal, will be required. Currently, the Main Canal divides into two separate canals at the northwest corner of the Borrow Pit, one continuing west to Ponds 13 and 17 and one running to Pond 10. At this location, a new canal that runs west along the recharge ponds within the proposed Phases 1 through 3 will be required (Figure 2-3) and the existing canals west of the Borrow Pit will no longer be needed. This new portion of the Main Canal, east of the Borrow Pit, will also consist of pond turnout structures that would allow separate discharge to each recharge pond. This will provide the capability of isolating each pond during maintenance, as well as a means to monitor flow to each pond (Figure 2-6).



# 2.4 OPTIONAL FACILITIES

During the conceptual design phase, several alternatives or optional design features were identified which the Districts may wish to consider during the detailed design phase. As an option, a second canal south of the recharge basins in the proposed Phases 1 through 3 was considered during the conceptual design. This second canal, the Lower Borrow Pit Canal shown on Figure 2-1, will provide additional conveyance to the recharge ponds in the proposed Phases 1 through 3. This would provide operational flexibility and redundancy by allowing the Main Canal to be shutdown and dewatered for maintenance purposes without losing the ability to recharge. It will also provide a way to capture any overflow from the Borrow Pit and the ability to use the overflow for recharge. The proposed Lower Borrow Pit Canal will require modifications to existing facilities along with the addition of new facilities to the Borrow Pit, including Ponds 1, 2 and 3. These features are described in more detail below.

Several options including sedimentation ponds and chemical treatment were also discussed during the conceptual design. Since the completion of Seven Oaks Dam, there has been a degradation of the water quality of Santa Ana River characterized by significantly increased turbidity. The District plans to use Ponds 1 through 3 as sedimentation ponds; however, Ponds 1, 2 and 3 will not provide enough detention time to settle out the silt in the water. In order to increase the detention time, the Conservation District has the option to construct sedimentation ponds east of the Borrow Pit. These features are described in more detail below.

#### Lower Borrow Pit Canal

As shown on Figure 2-3, the Lower Borrow Pit Canal will provide the Conservation District with an additional conveyance facility to the recharge ponds in the proposed Phases 1 through 3. The Lower Borrow Pit Canal will have a flow capacity of 500 cfs. Flows to the Lower Borrow Pit Canal would be accomplished by conveying water through the Borrow Pit, which overflows into the proposed Borrow Pit Overflow and discharges into the Lower Borrow Pit Canal. The Borrow Pit Overflow is an overflow weir on the southwest corner of the Borrow Pit (**Figure 2-16**). Currently, the Conservation District discharges to the Borrow Pit for recharge and any excess flow would overflow the Borrow Pit and discharge directly into the Santa Ana River. The Borrow Pit Overflow will capture and convey any excess flow to the recharge basins.

Conveyance to the Lower Borrow Pit Canal will require modifications to the structures at Ponds 1, 2 and 3 along with new facilities in the Borrow Pit as discussed below. The Lower Borrow Pit Canal will also consist of pond turnout structures similar to the Main Canal. This would allow separate discharge to each of the recharge ponds; therefore, providing the capability of isolating each pond during maintenance.


#### Ponds 1, 2 and 3

The south gate at the Diversion Structure is used to divert water to the Borrow Pit, which includes Ponds 1, 2 and 3. If water is diverted to Ponds 1, 2, and 3, they may function as settling basins to remove a portion of the suspended sediments that contribute to basin clogging. The south gate of the Diversion Structure does not have the capacity to divert 500 cfs to the Borrow Pit. Therefore, improvements, described above, will be needed along with improvements to the facilities that convey water from the Diversion Structure to the Borrow Pit. The facilities include the open channel on the east side of the Borrow Pit to Pond 1, the inlet structure from the open channel to Pond 1, and the overflow structures at Ponds 1, 2 and 3.

The open channel that conveys water from the Diversion Structure to the Borrow Pit will be modified to have the same cross section as the Main Canal. A new Pond 1 inlet structure from the open channel to Pond 1 will be required to convey 500 cfs (**Figure 2-17**). The inlet structure will be constructed of concrete with three 42-inch Waterman C-10 canal gates. The canal gates will provide control on the discharge to Ponds 1 through 3. Three 42-inch diameter reinforced concrete pipes will discharge to an outlet at Pond 1 that consists of a buffer wall and riprap to dissipate the energy due to the high velocities and the difference in elevation from the open channel to Pond 1.

Currently, the existing pond overflow structures that allow water to flow from Pond 1 to Pond 2 and from Pond 2 to Pond 3 are not adequate for the desired capacity of 500 cfs. The new overflow structures will be required and will be designed similar to the existing structures. The structures will be box-shaped structures with one side consisting of removable five foot wide flashboards. Openings wider than five feet will make it difficult to remove the boards by hand. In order to convey 500 cfs, a total of four structures constructed side by side will be needed. Four 42-inch diameter reinforced concrete pipes will discharge into the succeeding pond that consists of grouted riprap and large boulders at the outlets to dissipate the energy due to the elevation difference (**Figure 2-18**).

#### Borrow Pit

The improvements at Ponds 1 through 3 will only allow flow from Pond 1 to Pond 2 and from Pond 2 to Pond 3. In order to convey water from Pond 3 to the Lower Borrow Pit Canal, an open ditch, the Upper Borrow Pit Canal, will be needed to deliver water from Pond 3 to the canal and the recharge area in the Borrow Pit. Existing flows from Pond 2 to the recharge area will be eliminated.

The Upper Borrow Pit Canal will consist of a discharge structure at Pond 3 and a diversion structure at the southwest end of the Borrow Pit. See Figure 2-3 for the locations of these facilities. The discharge structure at Pond 3 would provide control of water into the open ditch with three 72-inch diameter Waterman C-10 canal gates. Water will discharge through three 72-inch reinforced concrete pipes to an outlet that consists of grouted riprap to prevent erosion in the Upper Borrow Pit Canal (**Figure 2-19**).







The Borrow Pit Diversion Structure will consist of six 72-inch canal gates, three discharging to the Borrow Pit and three discharging to the Lower Borrow Pit Canal as shown on **Figure 2-20**. The gates will provide the capability to control flows to the recharge area in the Borrow Pit and the Lower Borrow Pit Canal. The structure will allow the entire flow of 500 cfs to be diverted in either direction.

Borrow Pit capacity values, provided by the SBVWCD, indicated that the recharge rate of the Borrow Pit is 50 cfs. Therefore, the Borrow Pit will not be able to recharge the desired flow of 500 cfs. As stated above, excess flow would overflow the Borrow Pit and discharge directly the Borrow Pit Overflow, where the water will be captured and conveyed along the Lower Borrow Pit Canal to the recharge ponds (Figure 2-16). Flows from the Upper Borrow Pit Canal will also discharge into the Borrow Pit Overflow before it reaches the Lower Borrow Pit Canal. As an optional item for future consideration, flows leaving Pond 3 could also be placed in a pipe and delivered by gravity to Valley District's Foothill Pump Station for delivery throughout the Valley.

#### Additional Sedimentation Ponds

Since the completion of Seven Oaks Dam, there has been a degradation of the water quality of Santa Ana River characterized by significantly increased turbidity. Several options including chemical treatment and sedimentation have been proposed (CDM 2005). Although an evaluation of pre-treatment is beyond the scope of this investigation, the following discussion indicates the potential magnitude of facilities required.

Sedimentation efficiency is a function of the surface loading rate which is the flowrate divided by the surface area of the settling basin. For typical water treatment applications where coagulants are used, the surface loading rate is 0.5 to 1.0 gallons per minute per square feet (gpm/ft<sup>2</sup>) (MWH 2005). Assuming a conservative rate of 0.5 gpm/ft<sup>2</sup> and a flowrate of 500 cfs, approximately 10.5 acres of sedimentation ponds would be required. Detention times are typically 1.5 to 4 hours (MWH 2005). At a design flowrate of 500 cfs, approximately 165 acrefeet of storage would be required to provide 4 hrs of detention time.

The Conservation District plans to use Ponds 1 through 3 as sedimentation ponds. These ponds have a wetted surface area of 7.0 acres, which is less than the desired surface area. With a desired flow of 500 cfs, the average detention time in each pond would be approximately 21 minutes. The pond detention time was determined by dividing the pond's storage volume by a design flow of 500 cfs and taking the average of the results. This would provide a total detention time of 63 minutes before water is discharged into the Upper Borrow Pit Canal. It will not be enough time to settle out the silt in the water, even with chemical coagulation. The existing configuration of the ponds is not conducive to good settling due to poor inlet design and the turbulence created as water flows from pond to pond.



In order to increase the surface area and detention time, the Conservation District has the option to construct sedimentation ponds east of the Borrow Pit, and west of Greenspot Road (**Figure 2-21**). An existing 42-inch diameter pipeline that runs south-easterly splits the proposed area into two, creating two ponds with a combined area of approximately 29 acres, more than sufficient for adequate settling. With a proposed pond depth of 10 feet, the ponds east and west of the pipeline will have a detention time of about five and two hours, respectively. This would provide a total detention time of seven hours. Ideally, the basins should be 10 to 16 feet deep, with a length to width ration of 4 to 5 to 1. The settling ponds would also need to be designed to allow periodic removal of silts without impacting recharge operations.

In order to divert flows to these ponds, a new open channel from the Main Canal to the pond east of the pipeline will be required downstream of the Parshall Flume. The open channel will need a concrete outlet structure with canal gates at the Main Canal to control flows to the proposed east pond. A new overflow structure on the south end of the east pond will be required to convey water to the west pond. Additional open channels will also be needed to convey water in the west pond to the Borrow Pit and/or to the Main Canal. The channels need to have uniform flow distribution to minimize turbulence as the water enters the basins, with an overflow structure out of the ponds back into the main channel. The conveyance to and from the sedimentation ponds can be done without the need of pumping facilities.

In addition to settling ponds, provisions may need to be made for coagulant addition. CDM report (2005) indicated that 10-15 mg/L of ferric chloride or 25-30 milligrams per liter (mg/L) of alum plus a coagulant aid may be needed to reduce turbidity from as much as 500 nephelometric turbidity units (NTU) to a range of 3-7 NTU. At a design flow rate of 500 cfs, chemical usage could be in the range of 13 to 40 tons per day. This would require significant chemical storage feed and mixing facilities prior to settling. Provisions for sludge removal, drying and disposal would also be required. Since ferric and alum sludges are quite gelatinous, dewatering may be difficult. In addition, the Regional Board may require that the settling ponds be lined to prevent iron or alum from migrating into the aquifer.

It should be noted that the use of settling ponds may not be sufficient to control turbid water and reduce clogging. A number of California agencies are investigating engineered treatment systems to pre-treat water prior to percolation to maximize infiltration rates.

# 2.5 SUMMARY AND ESTIMATED COSTS

Cost estimates for the additional and modified facilities for the three different flow rates are presented in **Figure 2-22**. The recommended modifications and additional facilities discussed above would provide the Districts with the ability to reach their established target amounts for groundwater recharge along with operational flexibility and redundancy. The District's ultimate goal is to convey a maximum flow of 500 cfs for recharge. It is understood that this flow rate is anticipated for short periods only; however, the improvements would provide the Districts with the capability to accommodate this flow on a temporary basis to maximize storage of surface water when it is available.





Figure 2-22 Estimated Costs at Different Flow Rates

As discussed above, additional and modified facilities are not required to convey 195 cfs to the existing groundwater recharge facilities. At a flow rate of 300 cfs, modifications to the Intake Structure will be required along with additional recharge ponds at a cost of \$3,600,000.

In order to convey 500 cfs to the existing and new recharge ponds in the proposed Phases 1 through 3, modified and additional facilities described in the section above would cost approximately \$9,800,000. The alternate design would cost approximately \$9,800,000. Total costs to convey 500 cfs with optional facilities would cost approximately \$18,100,000. The costs for the optional facilities are as follows:

Trashrack Structure	\$181,000
Lower Borrow Pit Canal	\$4,500,000
Sedimentation Ponds	\$3,600,000

Details of the cost estimates are provided in **Appendix C**. Figure 2-22 includes the optional facilities noted above. The intent of these estimates is to provide a budgetary estimate only for the conceptual design of this project.

# Section 3 Operations, Maintenance, and Monitoring

An important component in an overall plan to maximize spreading grounds recharge rates is an operations, maintenance and monitoring (OM&M) plan. Implementation of routine procedures that are uniformly followed will provide a means for recognizing the need for adjustments or maintenance in a timely manner to ensure optimal recharge rates are maintained. This section describes the mechanisms of clogging and key preventative measures. The combined experiences of many agencies are summarized in developing a list of "Best Management Practices" (BMPs) for recharge basin operations and maintenance. These BMPs are reviewed in light of site-specific conditions to develop a set of recommended measures for operations, maintenance and monitoring.

# 3.1 MECHANISMS AFFECTING INFILTRATION RATES

The infiltration rate is the volume of water moving into the soil or aquifer per unit of area and per unit of time. Infiltration rates are commonly expressed in units of length/time, such as inches per hour or feet per day. Infiltration rates in uniform soils without surface clogging are about equal to the vertical hydraulic conductivity of the soil (Bouwer, 1978). Hydraulic conductivities can vary from less than 4 inches per day for clay soils to more than 30 feet per day for coarse sands (Bouwer, 1999). A number of mechanisms affect the infiltration rate of a recharge facility including clogging, soil compaction, depth to groundwater, the presence of rocks, water quality vegetation, and subsurface geology including silt and clay lenses. The following briefly describes these mechanisms.

#### Clogging

The primary cause of infiltration problems for surface recharge operations is clogging. Clogging is most frequently caused by inorganic (such as silt and clay) and organic (such as algae and fine plant material) suspended solids in water accumulating on the surface of the infiltration surface. In addition to suspended matter in the water itself, additional suspended matter can be added by erosion within the basin due to excessive inflow rates, wave action and windblown dust. Biofilms forming on the basin bottom can contribute to clogging layers and reduce soil pore size (Bouwer, 1999).

Another cause of clogging is the precipitation of minerals such as calcium carbonate, and other less common mineral within the soil. This type of clogging generally occurs where the infiltration rates are low and evaporation rates are high. Source waters high in total dissolved solids can greatly increase chemical clogging. Chemical clogging is usually treated in the same way as physical clogging by the physical removal of the clogging material.

Because the clogging layers are much less permeable than the natural soil material, they reduce infiltration rates and limit the amount of water that can enter an aquifer. The thickness of clogging layers can range from 1 millimeter or less (biofilms, thin clay and silt layers) to several centimeters for thick deposits (Bouwer, 1999). Fine particles can also move downward into the soil accumulating on subsoil materials forming subsurface clogging layers.

Periodic removal of clogging layers is critical to maintaining infiltrations rates. However, cleaning activities such as disking and ripping without removal of the fine materials can drive clogging particles deeper into the soils making subsequent removal more difficult. Disking and ripping can leave the soil in a rough, loose condition creating high initial infiltration rates. However, when water is applied, the resulting soil ridges can cave and slough off, re-suspending fine particles. The muddy water redistributes and settles causing a new round of clogging (Bouwer, 1999).

The frequency in which the ponds are cleaned is dependent on the rate at which the percolation rate degrades. As noted in the following section discussing monitoring, it is important to develop information on percolation rates of the discrete basins and changes in those percolation rates with time. For example, a typical percolation rate in an unclogged pond at the recharge facilities is approximately 4 feet/day. If percolation rates are observed to be below 2 feet/day, the District may consider cleaning of the basins. However, this frequency of cleaning should be based on monitoring data collected and practical experience.

## Soil Compaction

In addition to clogging from suspended materials, soils can become compacted due to mechanical cleaning and excessive water depths in the basins. The frequent use of mechanical equipment to clean basins can lead to compaction of the underlying soils by the weight of the equipment. Both wheeled and tracked maintenance vehicles cause soil compaction; however, tracked vehicles tend to cause less compaction due to their larger surface contact area compared to wheeled vehicles. Surface compaction tends to occur in the top six inches of soil (MWH, 2003).

# Pond Depth

There is often a delicate balance between the need to store water in recharge basins during storms for later recharge and the effect of greater water depth on infiltration. Studies investigating the relationship between pond depth and infiltration rates have shown varying results. In some studies, infiltration rates increased due to the greater driving force as the water depth is increased. However, the increase is not linear (Bouwer, 1999). If the clogging layers are compressible, infiltration rates were observed to decrease with greater depth due to compaction of the clogging layers making them less permeable. If increased water depth does not result in greater infiltration rates, the hydraulic retention time in the basin (retention time = volume divided by inflow rate) will increase offering the water more exposure to sunlight and algae greater opportunities to grow. With infiltration. During photosynthesis, algae absorb carbon dioxide from the water increasing the pH and cause precipitation of dissolved calcium carbonate in the water forming a crust layer in the soil. In general, Bouwer recommends that water depths

should not exceed 1 foot to minimize algal growth and soil compaction (Bouwer, 1999). Maintaining a shallow water depth is more beneficial in basins where infiltration rates are low as a means of reducing retention time.

#### **Depth of Groundwater**

When the depth to groundwater is sufficiently below ground surface and the water is clear, flow to the underlying water table is controlled by gravity and the infiltration rate is unaffected by depth to groundwater. As the depth to the water table decreases, flow becomes more lateral until it is controlled by the gradient away from the basin. Modeling has shown the transition from gravity to gradient controlled flow occurs when the depth to groundwater is about twice the width of the recharge facility. Thus, for a 50-foot wide recharge facility, a depth of water of less than about 100 feet would tend toward gradient-controlled flow. In this case, infiltration rates tend to decrease linearly, reaching zero when the water table is the same as the depth of water in the basin. Shallow groundwater levels also inhibit the draining and drying process that is necessary to restore infiltration rates (Bouwer, 1999).

In basins where clogging is present, unsaturated flow predominates and depth to the water table does not affect infiltration rates as long as the capillary fringe above the water table is below the bottom of the basin. Since the capillary fringe may be about 1 foot thick in medium sands (more for fine soils and less for coarse), as long as the depth to the water table water is more than about 3 feet, infiltration should be unaffected by depth to water. Based on this factor, it is important to monitor groundwater depths outside of the basin area to ensure that there is adequate depth to avoid infiltration rate reductions.

#### **Presence of Rocks**

If the soil contains a significant amount of large cobbles and boulders, disking or harrowing is not possible and ripping is needed to restore infiltration rates. However, ripping also causes the upward movement of stones in the soil profile causing them to accumulate at the surface. When sufficient stones cover the surface, infiltration can occur only between the stones reducing the effective infiltration area and the net infiltration rate. Suspended material accumulates in the soil between the stones where the fine materials cannot be easily removed by drying and scraping. One solution is to remove the stones to expose as much soil as possible (Bouwer, 1999). However, in basins where subsurface cobbles dominate as with the Santa Ana River facilities, cobble removal may not be practical.

#### Vegetation

Vegetation growth has been used in some locations to improve infiltration rates but can slow the drying process and inhibit cleaning as the vegetation must be removed to effectively remove the clogging layers. Vegetation provides shade that could reduce algal growth in the water and on the basin bottom. Root activity may keep fine soils more open increasing permeability. Several successful recharge programs utilize vegetation to enhance infiltration. However, the vegetation must be able to survive repeated and extended flooding. Typical vegetation that has been used includes Bermuda grass (*Cynodon dactylon*), para grass (*Panisum* purpurascens), bulrush (*Scirpus spp.*), meadow rush (*Juncus spp.*) and cattail (*Typha spp.*) (MWH, 2003).

Excessive vegetation can create wildlife habitat that then may become difficult to eliminate, especially if endangered species are present (Zimmer, 2009). Non-native invasive species, such as arundo, salt cedar (*Tamarix*), castor bean, and pampas grass can become established. Vegetation in the basins provides increased potential for insect breeding and rodent activity. Vegetation can also increase evapotranspiration losses. When removing vegetation, it is important that it be removed and disposed off-site to avoid creating a clogging layer caused by decomposing vegetation.

# 3.2 PREVENTATIVE MEASURES

Clogging is best controlled by prevention, removing the underlying cause of the clogging. A variety of mechanisms are available to prevent or reduce the impact of clogging. For surface water infiltration systems such as the Santa Ana River Recharge Facilities, pre-sedimentation to settle clay, silt and other suspended material frequently offers the best preventive measure. The use of coagulants such as alum or polymers can provide more efficient settling. To accomplish effective settling, it is important that dedicated desilting basins be provided. Desilting basins should be designed with adequate surface area to settle out the smallest anticipated particles. Jar testing is typically performed to evaluate chemical dosage and settling characteristics. For Santa Ana River water, CDM conducted bench scale testing to determine the approximate chemical dosages required to reduce turbidities to acceptable levels (CDM, 2005). Where present, removal of nutrients can reduce growth of algae or other forms of biological clogging. Disinfection has also been used to control biological activity especially with recycled water (Bouwer, 1999). Disinfection would not be desirable where the water has significant organic content that could result in formation of disinfection by-products.

A number of approaches have been used to combat clogging. Clogging in recharge basins can be controlled by periodically drying the basins and letting the clogging layer dry out, shrink, crack and curl up. This exposes underlying soil that is more conducive to infiltration. A secondary benefit of periodic drying is the interruption of the growth cycles of vectors such as midge flies and mosquitoes and reduction in algal growth.

For most basins, it is necessary to experiment with various wet and dry cycles to determine the optimum balance that maximizes recharge. Extended wet periods allow continued recharge at gradually decreasing rates as clogging occurs. The use of wet/dry cycles prolongs acceptable infiltration rates but the wetted acreage must be increased to allow basins to drain and dry out before they are returned to service. Wet/dry cycles also interrupt vector breeding cycles. Wet/dry cycle durations range from short (1-2 days each) to long (up to continuous). The most typical wet/dry cycles are in the 5-10 day range.

If periodic drying is not sufficient to restore infiltration rates, mechanical cleaning is the most effective method for removal of clogging layers. This is accomplished by mechanical scraping the top layer of soil using front-end loaders, graders or scrapers. Following removal, the surface is typically disked or harrowed to loosen any soil compaction caused by cleaning. However, the cost of cleaning must be weighed against the benefits derived from increased infiltration. If cleaning is carried out too frequently, the value of the water that cannot be percolated while the basins are out of service may outweigh the increased benefit of a higher infiltration rate.

Soil conditioners such as polymers and gypsum have been used successfully in the agricultural industry to condition hardened soils and may have an application at spreading facilities where soil conditions need to be improved. This measure is potentially useful in fine textured soils or waters that have a high sodium absorption ratio. Consequently, soil conditioners are not likely to be useful at the Santa Ana River facilities.

Basin cleaning and grading techniques can restore and maintain infiltration rates. As discussed previously, clogging layers should be physically removed by scraping and removal of the clogging material. Ripping of the basin bottoms should be minimized and only performed in the upper six inches to reduce the potential for rocks migrating to the surface.

Several recharging agencies have successfully utilized a ridge and furrow grading technique to maintain infiltration rates. At the Leaky Acres Facility in Fresno, CA, the City of Fresno uses a system that has proven effective in keeping at least half of the basin bottom relatively free of fines and permeable. After removal of accumulated fines and ripping, motor graders are used to cut furrows that are 6-8 inches, 10 feet wide and spaced on 30-foot centers. The edge of each furrow is sloped at a 6-10 percent grade. Spreading basins are kept full to allow wind-driven wave action to move accumulated fines to the bottoms of the furrows (MWH, 2003). The narrow configuration of most SAR basins may make furrow construction difficult and minimize wave action.

The Arvin-Edison Water Storage District operates a large water spreading facility in Kern County, approximately 25 miles southeast of Bakersfield. When infiltration rates decline, the District uses motor graders to scrape fines into windrows. Then, a paddle wheel scraper removes the windrows which deposits the fines on the tops of the levees. The basins are then ripped to mitigate the compaction caused by the heavy equipment (MWH 2003). This method may not be practical at the Santa Ana River recharge facilities, because of the physical configuration of the basins and the presence of large boulders. Disposal of fines on the levees may result in reintroduction of clogging materials due to erosion from rainfall.

Vegetation can be managed by either physical removal or chemical control. Physical removal of vegetation is typically performed when basins are cleaned. Vegetation debris should be removed and disposed of rather than buried. Chemical control of vegetation using herbicides can be performed without significant environmental impact. In an investigation of recharge for LACDPW, MWH contacted an independent pest control advisor, Robert Brenton, who has worked with LACDPW in an advisory role for nearly 20 years. Mr. Brenton identified three acceptable herbicides:

Glyphosate (Trade name: Roundup, Rodeo and Accord) Imazapyr (Trade name: Stalker) Triclopyr (Trade name: Garlon and Pathfinder).

It was recommended that these herbicides are best applied prior to the rainy season and when depth to groundwater is likely to be greatest (MWH 2003).

MWH's review of basin management practices with Conservation District staff suggests that vegetative growth is not a significant factor in limiting recharge operations. Therefore, we do not recommend the routine use of herbicides at the recharge facilities. Notwithstanding Mr. Brenton's findings described above, if herbicides are contemplated, the potential environmental and water quality impacts should be carefully reviewed.

## 3.3 BEST MANAGEMENT PRACTICES

Based on the extensive collective experience gained from the evaluation of recharge basin operation by many agencies, Dr. Herman Bouwer developed the following list of design and management best practices:

- 1. Use shallow level basins and water depth not exceeding one foot.
- 2. Design inlets and banks to minimize erosion.
- 3. Pre-treat water to remove suspended solids and, if desirable, nitrogen phosphorus and organic carbon.
- 4. Develop an optimum sequence of flooding drying and cleaning of basins. Remove clogging layers before disking or ripping, and then smooth and slightly compact the soil before refilling.
- 5. Keep groundwater levels at least 3 feet below the basin bottom if infiltration is controlled by a clogging layer and the soil below the clogging layer is unsaturated, and at least one infiltration system width below the bottom some distance away from the basin if the basin bottom is clean and there is direct hydraulic connection between the basin and the aquifer
- 6. Design the system with an adequate number of basins and total surface area so that basins can be properly dried and cleaned for infiltration recovery. Base land requirements on the lowest anticipated infiltration rates (winter, rainy period, low quality water, etc.). Select sites with good soil and groundwater conditions. Test changes in basin management first on a small basin before changing the entire system (the land area limitations at the Santa Ana River recharge facilities may preclude use of this particular BMP).

Some of these recommendations may be applicable to the Conservation District's facilities while others are not. Specifically, the Conservation District operates their basins at greater water depths than the 1 foot recommended by Dr. Bouwer without adverse results. Design of banks and inlets to minimize erosion is a reasonable goal that must be weighed against land availability topography. Consequently, no changes are recommended to the existing practice.

Pre-treatment of water prior to recharge is expected to have a significant effect on infiltration rates by minimizing a significant source of clogging. As discussed in Section 2, construction of engineered desilting basins is recommended. However, adequate jar testing and evaluation must be performed prior to detailed design to ensure that the proposed design will perform as desired.

Determining the proper groundwater level criteria will require experimentation and monitoring. The Conservation District will need to implement the monitoring program as described in Section 3.5 and evaluate the results to determine specific criteria for depth to groundwater. Once these criteria are established, they can be used to control basin operations.

# 3.4 CURRENT PRACTICES

The current OM&M practices of the Conservation District consist of:

- Checking the source water for clarity The Conservation District visually observes the water clarity to determine its acceptability for recharge. This approach is very subjective, dependent on the observer and type of sediment content. However, no specific numerical objectives have been established. At times, highly turbid water has been recharged resulting in clogging.
- Periodic rotation of the basins when infiltration rates decline The Conservation District typically operates basins continuously for 3-4 weeks before allowing them to dry. Since vectors are not a significant issue more frequent drying has not been required.
- Removal of vegetation overgrowth The District periodically removes vegetation; however, vegetation growth is minimized by keeping the basins flooded for extended time periods.
- Removal of accumulated sediment The Conservation District typically removes accumulated sediment about every three years. The cost of sediment removal and disposal has impeded more frequent cleaning.
- Monitoring Total flows are monitored at the Parshall Flume. However, water quality and infiltration rates are not monitored limiting data that can be analyzed to evaluate system performance. Groundwater levels are monitored at approximately monthly intervals.

The current O&M practices are generally acceptable. However, the Conservation District should avoid recharging water that contains high levels of suspended solids and turbidity without pretreatment. In addition, the lack of sufficient monitoring activities limits the ability to evaluate the performance of the basins and the improvements that are obtained following basin maintenance. Specific recommendations are presented in the following section.

#### 3.5 **RECOMMENDATIONS**

The following daily and periodic activities are recommended to maintain the recharge goal of 500 cfs. While not absolutely required for recharge, these recommendations reflect best management practices to optimize recharge and minimize required maintenance.

#### Operations

The following recommendations apply to routine operations of the spreading facilities.

#### Water Quality

A critical factor in maintaining infiltration capacity is to recharge the best quality water. It is recommended that the following criteria be used to determine when to divert water:

- Collect samples of sediment from the watershed and prepare a set of sample bottles containing varying amounts of sediment to use as a visual reference to estimate the sediment content of the water.
- Utilize turbidity or visual comparisons to determine the acceptability of the current recharge water.
- Avoid recharging water containing more than 200 mg/L of suspended solids or having a turbidity exceeding 100 Nephelometric Turbidity Units (NTU). This is important for keeping silt and clay from depositing around the rocks and cobbles and clogging the basins.
- Bypass water not meeting this criterion or divert to settling ponds.
- Conduct jar testing of potential coagulants to determine the effectiveness of pretreatment.
- Evaluate the design parameters for desilting and evaluate the cost of treatment.

#### **Pond Depth**

Seepage into the Borrow Pit has been observed when water is recharged for extended periods of time in Basin 10. To assure the stability of the Borrow Pit levee, recharge should be discontinued in this basin if seepage occurs.

#### Wet/Dry Cycles

The following criteria should be used to determine whether particular basins should be used for recharge:

- Monitor groundwater levels and infiltration rates to determine when to initiate basin drying. Once a pattern of infiltration rate performance is established, implement a wet/dry cycle that maintains relatively high infiltration rates. Discontinue recharge and commence dewatering and drying at a point prior to significant infiltration rate decline.
- Experiment with different durations of wet and dry cycles and maintain adequate monitoring records to document infiltration performance.
- Establish an optimal wet/dry ratio based on monitoring results. Using the current infiltration rates, it is expected that a 2:1 or 3:1 cycle may be suitable. For example, recharge would take place for 2-3 weeks followed by one week to drain and dry. This may allow a monthly rotation of basin operations.

#### Maintenance

The following activities should be undertaken to maintain the recharge facilities:

- At least once a year, remove and physically dispose of vegetation along the clogging layer.
- Eliminate dense vegetation that can hide carcasses and result in the introduction of botulism into the aquatic food chain. In the event of severe botulism, drain the basin and notify the Department of Fish and Game.
- Control vegetation above the water line. Do not allow basins to become habitat for sensitive species.
- Scrape and remove the clogging layer from basins once a year or as indicated by declining infiltration rates and dispose of fines.
- Following scraping, rip top six inches to loosen deeper soils.
- Utilize tracked equipment for basin cleaning to limit the amount of soil compaction.
- Select one or two basins having adequate size to experiment with ridge and furrows to evaluate their effectiveness in extending basin operations.
- If rock cover in recharge basins becomes excessive, large rocks should be removed and transported off-site.

Physical maintenance of equipment (Cuttle Weir, diversion structure, Parshall flume, canals, tunnels and culverts) should include:

- Periodic inspection of equipment operability
- Removal of rocks, vegetation and debris Following installation of the gabion barrier and log boom, the diversion facilities should be maintained to clear any accumulated debris and rocks.

#### Monitoring

Monitoring of recharge basin performance is valuable for determining the effectiveness of operations and maintenance procedures. Data will be used to ensure that methods being employed are continuing to maintain expected infiltration rates and quality of water being diverted.

The following water quality data should be regularly monitored:

• Turbidity – Because turbidity can be easily monitored in the field it is recommended that the Conservation District purchase a field turbidity meter to measure turbidity data. The Hach Model 2100P is a battery-operated field turbidity meter that can be carried by Conservation District operating staff. Turbidity should be measured and recorded daily at the Cuttle Weir. During storm events turbidity should be monitored hourly. In the absence of treatment, highly turbid water should be bypassed down the river channel.

- Suspended solids Suspended solids is measured by laboratory analysis of water samples. Since suspended solids results cannot be obtained immediately for operation decision making, it is not useful for routine operations. However, a correlation can be made between suspended solids and turbidity so that turbidity can be used for operations. As discussed previously, sample bottles containing varying amounts of suspended solids can also be used as a visual indicator of water quality. Suspended solids should be visually checked and recorded daily.
- Data logging Water quality data should be stored in a database or spreadsheet to allow evaluation in comparison to infiltration rates. Note the flow rate, source of water (dam release, storm flows, and imported flows), date, time, and weather conditions (raining, drizzle, clear).

The following recommendations are made regarding monitoring of infiltration rates:

- The Conservation District should consider the installation of V-notch or other weirs appropriate for the anticipated flow ranges at each basin inlet and outlet to allow measurement of flow rates into recharge basins to compute infiltration rates.
- Infiltration rates should be calculated based on the difference between basin inflows and outflows divided by the wetted area of the basin. Infiltration rates should be computed and recorded for each operating basin at weekly intervals.
- Infiltration rates should be reviewed to determine when to discontinue recharge at specific basins and commence drying.
- The Conservation District should evaluate the correlation between turbidity data and infiltration rates annually.

High groundwater levels due to groundwater mounding can significantly reduce infiltration rates. The Conservation District should monitor groundwater levels and note when the elevation of the water table approaches typical ground surface elevations. There are two general locations where mounding is expected to occur; east of the fault identified in TM-2 (Appendix B), and west of the fault. For monitoring of groundwater levels east of the fault, the Conservation District should monitor MW-1, MW-3, SB-1, and SB-2 which were installed during this study. All of these monitoring wells were fitted with automated dataloggers and transducers by the MWH Team.

For monitoring of groundwater elevations west of the fault, the Conservation District should monitor wells SB-3 and MW-2 (also fitted with automated dataloggers and transducers), as well as SBVWCD #3 and SBVWCD #2 as shown on Figure 3-1 of TM-2 (Appendix B).

When groundwater elevations approach approximately 3 feet below the ground surface elevation, recharge operations should be moved to locations where the depth to groundwater is greater. As record keeping of flows and groundwater elevations are developed, it will be possible to predict the rate of groundwater mounding and the approximate flows which will result in groundwater levels at or near the ground surface.

#### **Anticipated Performance**

Implementation of the recommendations made in this section is expected to improve the performance of the recharge basins. However, it is not possible to estimate the improvement that would occur with implementation. MWH recommends that the Conservation District and Valley District commence monitoring of basin performance as soon as practicable to establish a baseline for measuring performance. As individual measures are implemented, improvements in performance can be documented and operations and maintenance methods modified accordingly.

- Asano, Takashi 1985. *Artificial Recharge of Groundwater*. Butterworth Publishers. Stoneham, MA.
- Bouwer, Herman, 1978. Groundwater Hydrology. McGraw-Hill, New York, NY.
- Bouwer, Herman , 1999. Chapter 24 Hydraulic Design of Infiltration Systems for Artificial Recharge of Groundwater and Water Reuse. *Hydraulic Design Handbook*. Larry W. Mays, Editor-in-Chief. McGraw-Hill, New York, NY.
- CDM, 2005. Seven Oaks Dam Water Impacts Study. Prepared for the Upper Santa Ana Water Resource Association in conjunction with East Valley Water District. December 8, 2005.
- MWH, 2003. Percolation Optimization Study Final Report. Prepared for Los Angeles County Department of Public Works. November 2003.
- MWH, 2005. Water Treatment Principles and Design. Second Edition. John Wiley & Sons.
- MWH, 2008. Santa Ana River Groundwater Recharge Optimization Study, Technical Memorandum 1 – Documentation of Existing Information and Field Work Plan – June 16, 2008. (Note: This memorandum is included in this document as Appendix A)
- MWH, 2009. Santa Ana River Groundwater Recharge Optimization Study, Technical Memorandum 2 – Documentation of Field Work and Recharge Capacity Estimate – Draft, January 6, 2009. (Note: This memorandum is included in this document as Appendix B)
- Science Applications International Corporation (SAIC), 2007. Technical Memorandum: Seven Oaks Dam Economic Analyses, December 2007.
- SBVMWD and WMWD, 2007. Santa Ana River Water Right Applications for Supplemental Water Supply: Community Report. January 2007.
- SBVWCD, 2008. Final Environmental Impact Report for the Upper Santa Ana River Wash Land Management and Habitat Conservation Plan, #SCH2004051023. November 2008.

Zimmer, Ken, 2009. Personal communication. April 3, 2009.

# Appendix A

Technical Memorandum 1: Documentation of Existing Information and Field Work Plan



# Prepared for:

San Bernardino Valley Water Conservation District And San Bernardino Valley Municipal Water District

By:

MWH Americas, Inc. GEI Consultants





**Revision 1 -- June 16, 2008** 

# **Table of Contents**

## **Section Name**

# Page Number

Section 1	l - Introduction	1-1
1.1	Background	1-1
1.2	Purpose And Scope	1-1
1.3	Workplan Organization	
Section 2	2 - Summary of Existing Information	
2.1	Conveyance and Diversion	
2.	1.1 Construction History And As-Built Information	
2.2	Site Testing and Observations	
2.3	Initial Percolation Testing	
2.4	Summary of Existing Data On Hydrogeologic Conditions	
2.	4.1 General Geology	
2.	4.2 Anecdotal Observations	
2.	4.3 Existing Subsurface Information	
2.5	Field Testing Goals	
G		
Section 3	- Proposed Field Testing Program	
<b>Section 3</b> .1	Sequence of Work	<b>3-1</b>
3.1 3.2	Sequence of Work	<b> 3-1</b> 3-1 3-2
3.1 3.2 3.3	Sequence of Work Shallow Soil Sample Collection From Hand-Dug Pits Trenching	<b>3-1</b> 3-1 3-2 3-3
3.1 3.2 3.3 3.4	Sequence of Work Shallow Soil Sample Collection From Hand-Dug Pits Trenching Sonic Drilling	<b>3-1</b> 3-1 3-2 3-3 3-3
3.1 3.2 3.3 3.4 3.5	Sequence of Work Shallow Soil Sample Collection From Hand-Dug Pits Trenching Sonic Drilling Monitoring Well Installation	<b>3-1</b> 3-1 3-2 3-3 3-5 3-5
3.1 3.2 3.3 3.4 3.5 3.6	Sequence of Work	<b>3-1</b> 3-1 3-2 3-3 3-5 3-5 3-6
3.1 3.2 3.3 3.4 3.5 3.6 3.7	Sequence of Work	<b>3-1</b> 3-1 3-2 3-3 3-5 3-6 3-6
3.1 3.2 3.3 3.4 3.5 3.6 3.7 3.8	Sequence of Work Shallow Soil Sample Collection From Hand-Dug Pits Trenching Sonic Drilling Monitoring Well Installation Percolation Testing Interaction with Districts Staff Mapping and Surveying	<b>3-1</b> 3-1 3-2 3-3 3-5 3-5 3-6 3-6 3-6
3.1 3.2 3.3 3.4 3.5 3.6 3.7 3.8 3.9	Sequence of Work	<b>3-1</b> 3-1 3-2 3-3 3-5 3-6 3-6 3-6 3-6 3-7
3.1 3.2 3.3 3.4 3.5 3.6 3.7 3.8 3.9 3.10	Sequence of Work Shallow Soil Sample Collection From Hand-Dug Pits Trenching Sonic Drilling Monitoring Well Installation Percolation Testing Interaction with Districts Staff Mapping and Surveying Sample Numbering Laboratory Soil Analysis	<b>3-1</b> 3-1 3-2 3-3 3-5 3-5 3-6 3-6 3-6 3-7 3-7
3.1 3.2 3.3 3.4 3.5 3.6 3.7 3.8 3.9 3.10 3.11	Sequence of Work Shallow Soil Sample Collection From Hand-Dug Pits Trenching Sonic Drilling Monitoring Well Installation Percolation Testing Interaction with Districts Staff Mapping and Surveying Sample Numbering Laboratory Soil Analysis Clean Up and Investigation-Derived Waste	<b>3-1</b> 3-1 3-2 3-3 3-5 3-5 3-5 3-6 3-6 3-6 3-6 3-7 3-7 3-7 3-8
3.1 3.2 3.3 3.4 3.5 3.6 3.7 3.8 3.9 3.10 3.11 Section 4	<ul> <li>Sequence of Work</li></ul>	<b>3-1</b> 3-1 3-2 3-3 3-5 3-5 3-5 3-6 3-6 3-6 3-6 3-7 3-7 3-7 <b>3-7</b> <b>3-8</b> <b>4-1</b>

# Section 1 Introduction

The San Bernardino Valley Water Conservation District (SBVWCD), in association with the San Bernardino Valley Municipal Water District (SBVMWD), "Districts" retained MWH Americas, Inc. (MWH) and GEI Consultants (collectively referred to as the MWH team) to evaluate the Districts recharge facilities adjacent to the Santa Ana River in San Bernardino County, California. The general purpose of this work is to determine the need for and technical viability of additional or modified facilities to capture native and imported surface waters for groundwater recharge to meet the recently-adopted Upper Santa Ana Integrated Regional Water Management Plan (IRWMP). This work is being completed under a contract for the *Santa Ana River Groundwater Recharge Optimization Study* dated April 16, 2008 between MWH Americas, Inc. and the San Bernardino Valley Water Conservation District.

## 1.1 BACKGROUND

SBVWCD owns and operates surface water diversion and conveyance facilities and groundwater recharge spreading facilities at the project site (define the project site). The SBVWCD recharge facilities have been in place since the 1930s, for the purpose of diverting and recharging water from the Santa Ana River. The SBVWCD was formed in 1932 to obtain water for conservation purposes, essentially replacing the Water Conservation Association formed in 1910 (SBVWCD, 1994). A relatively new development in recharge operations is the construction of the Seven Oaks Dam, approximately 1 mile upstream of SBVWCD's diversion. Construction of the dam began in August of 1994, supervised by U.S. Army Corps of Engineers (USACE). The dam is an earth and rock fill type, and during construction of the dam, material was excavated for the dam from an approximately 120 acre area that formerly contained recharge ditches operated by the SBVWCD, referred to herein as the "Borrow Pit". The dam was completed in November of 1999. Subsequent to that time, three infiltration ponds were constructed at the east end of the Borrow Pit.

The remaining recharge basins and the Borrow Pit (Figure 1) are located west of Greenspot Road and north of the Santa Ana River. Flow from the Santa Ana River below the Seven Oaks Dam is diverted at a structure where water flows through an underground box culvert (Intake Structure). This culvert connects to a concrete "sandbox" structure which allows for the expulsion of excess debris or sand in the water. Flow then crosses beneath Greenspot Road where flow can be measured at the Parshall Flume. Water can then be diverted to various basins including the Borrow Pit.

#### **1.2 PURPOSE AND SCOPE**

The MWH team is completing a number of tasks to assess the current capabilities of diversion, conveyance, and recharge facilities and determine what improvements, if any, are required to

achieve the recharge goals of the IRWMP. This assessment includes extensive field work. The purpose of this document is to summarize existing information gathered to date, and to provide a work plan for field investigation efforts planned in the near future based on that information. Initial efforts of the MWH team have consisted of:

- Meeting with SBVWCD and SBVMWD staff to review the nature and amount of existing data about the site
- Site reconnaissance of diversion, conveyance, and recharge facilities
- Review of data housed in SBVWCD's extensive library
- Review of data and meeting with Mr. Wes Danskin, of the U.S. Geological Survey (USGS)
- Review of data on the Borrow Pit from the USACE
- Requests for geologic data from the California Department of Water Resources (DWR) data still pending
- Interviews with facilities operators and staff on field conditions and typical recharge operations and field observations
- Completion of diversion and conveyance flow testing
- Completion of initial percolation testing and analysis

The purpose of the workplan is to summarize this existing data to the extent that it guides future field testing and data collection efforts. More detailed evaluation of this data will be provided in subsequent reports. The field work is designed to evaluate the ability of the existing facilities to meet the recharge objectives identified in the IRWMP and resolve currently-existing data gaps regarding site conditions.

#### **1.3 WORKPLAN ORGANIZATION**

This workplan is organized as follows:

- Section 1 introduces the project goals and provides background information;
- Section 2 provides a summary of the existing information on conveyance and diversion construction, flow analysis, geological conditions, and the field testing goals;
- Section 3 discusses the proposed field testing program;
- Section 4 describes the scheduling and reporting of field work; and
- Section 5 provides references utilized during development of this work plan.

A Health and Safety Plan is provided as Appendix A.



This section provides a summary of existing information on the SBVWCD recharge facilities, description of field testing to date, summary of existing geologic data, and outlines the goals of further field testing designed to fill data gaps in existing information.

#### 2.1 CONVEYANCE AND DIVERSION

Prior to development of this data summary and work plan, water needed to be released from Seven Oaks Dam, providing the opportunity to conduct flow testing of the diversion and conveyance works at relatively high flow rates. After this testing, water released from the dam at lower flow rates provided an opportunity to conduct initial percolation testing of existing percolation ponds and ditches. This section discusses historical documentation of conveyance and diversion facilities, as well as the results of flow testing of both the diversion and conveyance facilities, and existing percolation ponds and ditches.

#### 2.1.1 Construction History and As-Built Information

The SBVWCD library contains a variety of historical information and drawings of existing facilities. The most relevant of these are summarized in Table 1.

Title	Date	Description
Borrow Pit Topo	4/18/2008	DWG file, ESRI ArcMap Document
IRWM Plan	11/2007	Upper Santa Ana Watershed Integrated Regional Water Management Plan
USA IRWMP Highlights	11/21/2007	Highlights of the Upper SA Watershed Integrated Regional Water Management Plan
Optimization Study	4/23/2008	JPG file, Basin 3; Diversion Gates; SWP pipe
Observation Well MPE		SBVWCD
WCD Santa Ana Spreading		From 1912 to 2008
Basin 10 Head Wall 1		
Basin 10 Head Wall 2		
Conduits Plans Santa Ana Weir	5/1/1930	Design
Cone Camp Hydrographs		June 1998 ~September 2007
District Spreading Basin Storage Capacity	1/8/2008	
Diversion Tunnel Profile and Outlet Work details		SBVWCD
Draft 2008 EI	2007-2008	Engineering Investigation Bunker Hill Basin

 Table 1 - Summary of Pertinent Information in SBVWCD Files

Title	Date	Description
Existing facilities	12/1/2007	SA River Spreading Grounds Existing Facilities and Improvement Project
Facilities Overview	1/1/2008	Diversions and Canals Leading into SBVWCD Facilities
General Layout Santa Ana Weir	5/1/1930	
Headworks Plan -1930	5/1/1930	
Headworks Plan -1930- Alternate	5/1/1930	
Parshall Flume Readings from SA River	1/14/2008	From 1938 to 2007
Map Archive Index		
Monitoring Wells DTW		Observation Data
Operations Manual	3/1/1994	Operational Management Manual SBVWCD
Overview of Diversion Points		Overview of SBVWCD Diversion Points
Profile of Alternate Conduit 1930		
Proposed Build Out of the Borrow Pit	10/18/2007	Design
Santa Ana River Historical Spreading		From 1938 to 2007
SBVWCD Basin Storage		SBVWCD Basin Storage Capacity for SAR and MC
SBVWCD 4 Well Log	4/19/1974	One Driller's Lithology Log to 240 ft bgs
SAR1_SAR2_Lithology		Driller's log and E-log for SAR1 and SAR2
Schematic of Water Control Features		Figure 2.2-1 Schematic of Water Control Features and Gages in the SA River Canyon
Topo West of Borrow Pit		Scale 1'' = 200'
Venturi Flume 1933	3/1/1933	Design
Weir Structure Topo 2000	2/9/2000	by Schmidt Geomatic Mapping

# 2.2 SITE TESTING AND OBSERVATIONS

A flow test was conducted by the Districts on March 27, 2008, in coordination with a variety of other agencies. This testing allowed the MWH team to observe flow conditions in the conveyance facilities between the Diversion Works and the SBVWCD recharge facilities. Interaction between the MWH team and the Districts staff during the preparation and conducting of the field test allowed for an exchange of operational knowledge of the conveyance system.

During the flow test, MWH and Districts staff observed the physical condition of diversion and conveyance facilities; photographs were taken and recorded. at the following facility locations (Figure 1):

- Pool behind Seven Oaks Dam
- Plunge pool outlet release point from Seven Oaks Dam
- Cuttle Weir intake structure with six gates
- Conduit tunnel between the river intake structure and Sandbox
- Sandbox Sandbox structure with two outlet gates
- Rock Structure Rock structure near northfork outlet, downstream of Sandbox
- Bridge over Canal downstream of Rock structure
- Greenspot Road culvert under Greenspot road
- Parshall Flume flow measurements were automatically recorded
- Diversion Structure (Borrow Pit gate was opened when flow reached ~320 cubic feet per second (cfs) at the Parshall Flume)

#### Dam Release Rate and Flow Measurements

The flow test involved communication between Districts personnel and dam operators such that various release rates from the dam could be tested. The dam tenders (reservoir regulation staff) provided the requested flow by executing the settings needed to obtain the release flow rate using a rating curve and reservoir water was released into the plunge pool below the dam. The resulting flow rate was measured downstream from the point of release by the United States Geological Surve (USGS) gage number 11051499 (Mentone Station), located immediately upstream of the Cuttle Weir. Flow was also monitored throughout the test at the Parshall Flume located downstream of the Greenspot Road culvert.

Observations were made of the conveyance facilities during five requested dam release rates of: 150, 200, 250, 325, and 400 cubic feet per second (cfs), as shown in Table 2 (Eftehari, 2008).

Date	Time	Reservoir Level (Staff)	Reservoir Level (Digital)	Release (cfs)
03/27/2008	0730	2259.50	2259.31	3.0
	0828	-	2259.26	150.0
	0928	-	2259.17	200.0
	1028	-	2259.90	250.0
	1200	-	2258.98	325.0
	1239	-	2258.98	400.0
	1315	-	2258.62	100.0
	1350	2259.00	2258.58	100.0

## Table 2- Seven Oaks Dam Release Summary

Release rates at the Seven Oaks Dam, as well as flow measured at the USGS Mentone Station (Agajanian, 2008), and measured at the Parshall Flume (Flordelis, 2008) are summarized in Figure 2. Figure 2 also shows the times of key observations during the flow test (which are discussed in more detail below).

Following the dam release setting of 400 cfs, the temporary sand bag dam that increased the height of the Cuttle Weir gave way at a measured flow of 390 cfs; a peak flow of 414 cfs was measured at the USGS gage prior to the end of the test. A peak flow of 367 cfs was measured downstream at the Parshall Flume that corresponded to approximately the same time as the USGS gage peak flow.

Districts staff provided a summary of photos taken at the Cuttle Weir and the Greenspot Road culvert for each of the flow release set points. The MWH team collected all field notes recorded for each observation point and have consolidated them into an Excel file for record keeping. Time and flow were written on corresponding representative photos of the observation points as a visual reference for the flow test. Preliminary hydraulic calculations were made by the MWH team that utilized the recorded flow test information and the as-built design drawings provided by SBVWCD.



Figure 2 - Flow Measurements During the March 27 Flow Test

Key observations of the diversion works and conveyance facilities during the March 27 flow test follows include:

- The silt and sand content of water at the diversion structure was high the first hour to hour and a half. The color initially was "chocolate milk", but eventually cleared to "milky green".
- The "practical" flow limit for most of the conveyance system was reached at approximately 300 cfs due to high velocity and turbulent flow in unlined channel sections. The "practical" flow limit is loosely defined as the flow at which field observations suggest a high flow may cause conveyance failure such as overflow or erosion. Erosion of channel and movement of boulders started at about 300 cfs and was pronounced at flows greater than 300 cfs. Locations where rocks were heard during observations (as the flow reached 300 cfs and above) were downstream of the Sandbox, upstream and downstream of the Rock structure, the bridge over the canal, near Greenspot Road, upstream of the Parshall Flume, and upstream of the Diversion Structure.
- At the Diversion Structure, the canal gate to the spreading grounds below the Borrow Pit was limited to roughly 300 cfs and the South Gate structure was opened for flows above 300 cfs.
- The field observations combined with a reported weakness located at a sharp curve in the canal downstream from the Diversion Structure indicated flow in the canal to the west spreading grounds would not be advisable over 300 cfs.
- The intake structure above the Cuttle Weir was affected by trash and debris during the flow testing which "backed up" water and caused higher water levels behind the Cuttle Weir (contributing to failure of the weir). Potential short term solutions to this may be to remove some sediment behind weir, increase the height of weir structure, and/or remove the I-beams at the weir so that debris passes through the diversion structure.
- The Greenspot Road culvert had little freeboard room in case of any debris clogging part of the opening. Only 3-4 inches were left on the upstream opening at the highest flow for the test, which was reported to be 414 cfs at the USGS Mentone Station and 367 cfs at the Parshall Flume.

## Subsequent Flow Measurements

On April 18, 2008, water level and structural measurements were made on the conveyance and diversion system to better appreciate the capacity of the SBVWCD facilities. The results of these flow measurements are currently being evaluated. Water levels at the flow rate of 140 cfs were taken at the flowing areas:

- The Cuttle Weir
- The Sandbox
- The Rock Structure below the Sandbox
- The channel below the Rock Structure
- The Bridge over Canal

- The Greenspot Road culvert
- The Parshall Flume
- The diversion structure below the Parshall Flume

#### 2.3 INITIAL PERCOLATION TESTING

After completion of the high flow test, estimates of percolation rates at selected ponds downstream of the Borrow Pit were obtained using the percolation testing methods described in the following sections. The percolation tests generally consisted of a constant flow test for SBVWCD recharge ponds 9 through 17, and D west of the Borrow Pit, and specific percolation testing of Pond 17 and Borrow Pit Pond 1 using a falling-head test. From March 28 to April 21, the flow rate into the recharge pond west of the Borrow Pit was approximately 140 cfs, or 280 acre-feet/day (AF/day). Recharge pond capacity values provided by the SBVWCD indicate that the total wetted area for ponds 9 through 17, and D is about 56.7 acres. This acreage does not include conveyance and diversion facilities between individual ponds. No outflow occurred from the last pond (D), suggesting a percolation rate for the system during this time period was about 4.9 feet per day (ft/day, 280 AF/day divided by 36.7 acres). No outflow occurring from Pond D suggests that the wetted area could still have been increasing during this period and that the total wetted area could have been less that the estimated 56.7 acres. Because the wetted area does not include conveyance and diversion facilities, the percolation rate is likely less on a per acre basis.

#### Percolation Testing Pond 17

A falling-head percolation test was performed on Pond 17 using a pressure transducer and remote telemetry data logger that transmitted water level data via a modem. For these falling-head tests, Pond 17 was filled by Districts staff and allowed to percolate during which time water levels were recorded every ten minutes. The rate of decrease in water level provides an estimate of the percolation rate. A falling-head percolation test is a very useful and reliable method for obtaining percolation rates, particularly when inflow and outflow from the test pond can not be reliably measured, or then the wetted area is not accurately known. Because the recharge ponds are designed to operate at full water levels, and percolation were used to estimate percolation rates. After about eight hours, percolation rates at Pond 17 decreased due to decreases in area and in water level head. Plots of water level over time produce a linear water level decline which is a function of percolation rate and evaporation rate. Evaporation rates over an eight hour period with percolation rates greater than one foot per day are considered negligible. A plot of the first test on Pond 17 is provided in Figure 3 as an example.



Figure 3 - Percolation at Pond 17

Percolation rates at Pond 17 ranged from about 3.83 to 4.19 ft/day and averaged about 4 ft/day.

## Percolation Testing at Borrow Pit

Percolation testing at Borrow Pit Pond 1, was in progress during writing of this document. The procedure for percolation rate testing at Borrow Pit Pond 1 is the same as that used in Pond 17. Results from percolation testing of Borrow Pit Pond 1 produced an estimated percolation rate of about 1.2 ft/day. This percolation rate is much slower than the 4 ft/day estimated at Pond 17 and the estimated 4.9 ft/day for the SBVWCD basin system, excluding conveyance and diversion facility areas. The slower percolation rate may be partially due to the design of Borrow Pit Pond 1. Long narrow deep ponds like Pond 17 tend to have a higher percolation rate than do large rectangular shallow ponds like Borrow Pit Pond 1, potentially due to "edge effects" of water leaking laterally as well as horizontally.

#### Observations of Rising Groundwater during Borrow Pit Pond 1 Testing

Rising groundwater was observed and photographed flowing into Borrow Pit Pond 3 during the installation of the telemeter equipment in Borrow Pit Pond 1 on May 10. The source of this groundwater is apparently from the east or north of Pond 3. The rising water started in Borrow Pit Pond 3 before the filling of Borrow Pit Pond 1, meaning that it is not completely associated with the testing of Pond 1. However, during a second visit to Pond 3 on May 19, during the testing in Borrow Pit Pond 1, the rate of rising water in Pond 3 did appear to be higher. This suggests that some leakage from Pond 1 was occurring. The location of the rising groundwater in Borrow Pit Pond 3 is at the northeastern end of the pit at its lowest level. Large areas of rising groundwater were also noted and photographed in the main Borrow Pit area near the center.

This rising water then flows west to the end of the main Borrow Pit where a large pond has developed.

#### 2.4 SUMMARY OF EXISTING DATA ON HYDROGEOLOGIC CONDITIONS

The following section provides a summary of the general geologic conditions that exist at the site, observations noted by field personnel, and a summary of available information gathered to date.

#### 2.4.1 General Geology

The Santa Ana River recharge facilities are located in the northeastern portion of the Bunker Hill groundwater basin near the boundary of relatively impermeable basement rocks of Mesozoic age and relatively recent fluvial and alluvial deposits which fill the groundwater basin. This boundary is formed by active faulting along the San Andreas Fault zone. The San Andreas Fault zone consists of two nearly parallel branches through the most of the San Bernardino area. In the vicinity of the recharge facility, the fault zone trends in a trends southeast-northwest direction, with the main San Andreas fault crossing near the northeast extent of Greenspot Road, and a splay of the fault crossing through the northeastern portion of the Borrow Pit (Figure 4).

The Quaternary age alluvium was formed as the ancestral Santa Ana River migrated across the alluvial fan, variously incising channels or depositing sediments in a braided stream environment through Quaternary time as the San Andreas Fault went through periods of activity and relative quiescence. The USGS has identified at least five separate alluvial sequences that are present near the recharge facilities. The youngest of these are very young Holocene surficial deposits consisting of sediment transported and deposited in channels and washes, or surfaces of alluvial fans and alluvial valleys. These are typically very coarse and permeable, with no soil development.

The older alluvial deposits consist of sedimentary units that are slightly to moderately consolidated and slightly to moderately dissected. The upper surfaces of these deposits are commonly capped by slight to moderately developed soil profiles.

Underlying the Holocene deposits near the surface, are yet older alluvial fan deposits of mid to late Pleistocene age. These deposits would be expected to be slightly more consolidated, and also contain soil horizons on their upper surfaces.

The significance of this geologic environment is that all of the sediments underlying the recharge facilities would be expected to have relatively high hydraulic conductivity and percolation capacity. The most permeable materials would be expected to be in proximal (toward Seven Oaks Dam) portions of the Santa Ana River alluvial fan(e.g., the Borrow Pit), becoming slightly finer in the distal (southwestern – away from the apex of the fan) portions of the fan. Although no thick laterally-extensive clay layers are expected (or have been reported), the soil zones developed on the surface of older alluvial fan deposits may contain clay or silt materials produced by mineral weathering in a soil zone that could locally retard downward movement of water. These older soil zones would not be expected to be laterally continuous because they

would have been eroded by active stream channels through time as the thalweg of the Santa Ana River migrated across the fan. (USGS, 2003a and b).



#### 2.4.2 Anecdotal Observations

As previously noted, the SBVWCD recharge facilities have been in operation since the 1930s. During this time, information on the total amount of water percolated at the facilities was recorded at the Parshall Flume. However, because it is unclear which individual ponds were used during this time, and the total acreage over which recharge occurred, it is difficult to reconstruct what the percolation rate was for a particular area. Records are not detailed enough to enable calculation of percolation rate in feet per day.

However, several of the SBVWCD staff have many years of experience operating the conveyance and recharge system, and their expertise provides valuable information which will be used to guide future field work. On May 2, the MWH Team met and discussed observations noted in the field by SBVWCD personnel. The following are some key observations that were noted in this discussion and on previous field visits:

- Prior to construction of the Borrow Pit, percolation rates at ponds formerly in the area that was to become the Borrow Pit were significantly higher than observed in the Borrow Pit today.
- During relatively wet periods, it is common to observe water "daylighting", or surfacing groundwater along the north wall of the Borrow Pit. This appears to be independent of use of the Borrow Pit for recharge.
- Because of the low percolation rates in this area, shallow bedrock is suspected towards the east end of the Borrow Pit.
- Relative to when Ponds 1, 2, or 3 are used in the Borrow Pit:
  - a. When Ponds 1, 2, or 3 are filled, water daylights near the middle of the main area of the Borrow Pit west of the Ponds.
  - b. Borrow Pit Pond 1 will percolate approximately 3.5 cfs on a continuous basis. If the amount of water added is increased to about 5 cfs, water typically spills into the next pond within 2 days, but will not spill into Pond 3.
  - c. Ponds below the Borrow Pit (Ponds 10-12) appear to percolate at a higher rate after a period of approximately 2 to 3 days.
  - d. Ponds 10-12 would take approx 40 cfs (as measured at the Parshall Flume) for weeks and will not spill.
  - e. Borrow Pit Pond 2 is thought to have the best percolation rate of the three ponds in the Borrow Pit.

#### 2.4.3 Existing Subsurface Information

Compilation of existing data on the subsurface conditions of the recharge facilities is critical to guide field efforts. In general, all of what is known about the subsurface hydrogeologic conditions at the site is derived from drilling (well construction) or trenching information.

The MWH team compiled subsurface information from a variety of sources, including the Districts files, the USGS and USACE. Approximately 35 wells have been identified within a mile of the SBVWCD recharge facilities. However, the location of many of these wells is uncertain, and it is possible that many of them have been abandoned or destroyed. The MWH team contacted the California Department of Water Resources (DWR) and requested well completion reports for that 2 mile area of the Borrow Pit. However, this information was not available at the time of production of this workplan and therefore will be included in subsequent reports.

The MWH team also contacted Mr. Robert Kwan, of the USACE, who provided data from investigations of the Borrow Pit area conducted in 1986 in preparation for construction of the Seven Oaks Dam. Field tests for determining the characteristics of the borrow area were developed specifically to evaluate the large size and percentage of rocks present in the borrow area; no borings were conducted. Five test pits were excavated in the SBVWCD recharge area using a Catepillar 235 backhoe to depths of 18 to 24 feet. Tests were conducted for gradation analysis, triaxial compression, permeability, and dry density. The gradational analysis indicated that the amount of the material larger than three inches ranged from 49 to 59 percent of the total sample. The maximum size of the rock in the samples ranged from 20 to 36 inches in diameter, although rocks up to 72 inches in diameter were observed. It is notable that virtually no fine material that might retard downward percolation was found. In all cases, fine material (passing U.S. Standard sieve number 200) was less than five percent. Permeability values from samples from one of the pits ranged from approximately 23 to 315 ft/day.

A key aspect of evaluating recharge rates at the site is the potential for groundwater mounding which would reduce recharge rates. MWH reviewed available recent groundwater level measurements from the Districts and the USGS and developed a map of equal groundwater elevation. This data indicates that in May of 2008, the water table is close to the surface in the vicinity of the Borrow Pit. However, it is difficult to resolve the exact depth to water because the topographic information at the relatively new Ponds 1, 2, and 3 in the Borrow Pit is not available.

## 2.5 FIELD TESTING GOALS

Review of existing data, coupled with the overall goals of the project highlight some specific goals of further field efforts. Available geologic data (Section 2.3.1) and soil testing by the USACE in the Borrow Pit area (Section 2.3.3) suggest that percolation rates in the Borrow Pit should theoretically be equal or greater than those in the western percolation ponds. However, comparative testing and anecdotal information indicates that percolation rates are much lower in the Borrow Pit.

In addition, although much subsurface information has been found in the vicinity of the SBVWCD recharge facilities, the exact location of existing wells is uncertain.

Based on these and other issues identified during the review of previous information, the general goals of the field investigation include:

- Evaluate the extent to which the shallow soils in the Borrow Pit have been altered by deposited fine materials (silts and clay) or algae.
- Evaluate the extent to which compaction of shallow soils in the Borrow Pit by heavy equipment has caused abnormal compaction which would slow percolation rates.
- Determine the presence or absence of shallow (less than 50 feet) layers of clay or other low-permeability material that would retard downward percolation of water. In particular, determine the cause of surfacing groundwater in the west-central portion of the Borrow Pit and the north face of the Borrow Pit.
- Determine the presence or absence of shallow bedrock on the eastern edge of the Borrow Pit that would limit the transmissivity of alluvial materials.
- Determine the presence or absence of shallow groundwater in the immediate vicinity of the Borrow Pit.
- Collect soils samples for laboratory gradation analysis to quantify the soil parameters at various locations inside and outside of the Borrow Pit, and allow for interpolation of patterns in shallow soils based on quantitative and visual descriptions of the soils.
- Locate and determine the coordinates of existing wells in the alluvial materials north of the Santa Ana River and east of Cone Camp Road.
- Conduct percolation testing in the Borrow Pit Pond 1 after cleaning of the bottom of the pond.
- Determine the elevation of the bottoms of Ponds 1, 2, and 3 in the Borrow Pit, and reference point elevations of selected wells where depth to water measurements are taken.

No additional flow testing goals have been identified for the conveyance and diversion system. Enough information was obtained during the March 27, 2008 release flow test and subsequent measurements that the system can be fully evaluated as to its capacity, limitation, and expansion needs.

# Section 3 Proposed Field Testing Program

This section describes the proposed field testing program designed to accomplish the goals outlined in the previous section, and to fill critical data gaps associated with the understanding of the recharge facilities.

## 3.1 SEQUENCE OF WORK

The fundamental rationale for sequencing of field work is to begin with an extensive shallow soil sampling effort, followed by a focused sampling effort using methods that allow deeper exploration. The proposed sequence of field work is as follows:

- Shallow surface sampling of soils, followed by
- Trenching, followed by
- Sonic borings, followed by
- Monitoring well installation, followed by
- Percolation testing

Based on the information gained in preparation of this technical memorandum Table 3 provides a comparison of the changed to the contract submittal and this technical memorandum.

Contract Scope of Work	TM-1
65 soil samples from shallow pits and trenches	More soils samples will be collected and stored (80+, 65 soil samples to be submitted for analytical analysis.
Up to 10 trenches	Same
5 Sonic borings to 50 feet	5 Sonic borings to 50 feet, completed as a peizometer.
1 MWl to 300 feet	2 MW to 150 feet

Ait percussion or rotary/foam drilling method with geophysics	Sonic drilling without geophysical
40 survey points in the Borrow Pint, MW locatons, and conveyance faacities	40 survey points focused on Borrow Pit topography and MWs
3 (2- <sup>1</sup> / <sub>2</sub> acre pits side by side) test pits to be constructed	Test western ditches, and Pond 1 in the Borrow Pit before and after cleaning. Additional testing may be recommended based on soil sampling.

The following is a description of the field activities. Proposed sampling locations are shown on <u>Figure 5</u>. The sampling program targets those areas of concern as noted in Section 2 to better identify and understand the hydrogeologic conditions in the vicinity of the Borrow Pit. It is important to note that although preliminary sample locations are identified in this work plan, the field program is intended to be adaptive in that the site geologist may modify sample locations based on what is discovered during the field program.

## 3.2 SHALLOW SOIL SAMPLE COLLECTION FROM HAND-DUG PITS

The purpose of shallow soil sampling is to characterize the very shallow (less than 12 inches) layer of the bottom of existing recharge facilities where clogging is most likely to occur. The intent of this sampling is to characterize a wide area for relative comparison. Samples will be collected on a grid rectangular grid system at approximately 500 foot centers in the Borrow Pit. Additional samples will be taken in western pond bottoms for comparison. During collection of the samples, the on-site geologist will observe:

- Depth of various soil horizons
- Algal growth
- Soil type
- Visual classification of gradation
- Color
- Apparent moisture content

Samples will be described in accordance with the American Society for Testing and Material ASTM) D2487-06 Standard Practice for Classification of Soils for Engineering Purposes

(Unified Soil Classification System). Samples will be collected from hand dug pits using a shovel.

Samples will be collected in one quart Ziplock freezer bags and retained at the SBVWCD offices. Selected shallow soil samples will be sent to a soils laboratory for analysis, as described in Section 3.10.

## 3.3 TRENCHING

Up to ten trenches will be excavated in the existing percolation basins to a maximum depth of five feet at each location. Figure 5 identifies the preliminary locations of the trenches, which may be modified by the field geologist based on observations from shallow soil sampling or previous trenches. The trench side wall will be logged under the oversight of a Professional Geologist registered in the State of California. A graphic depiction of the trench wall will be sketched on the trench log (Attachment 1) by the on-site geologist showing the thickness, depth, and relationship of various soil layers, and the presence of oversize material . Particular attention will be focused on evidence of shallow clogging by fine material and/or algal growth. Sample locations and depths will be noted on the trench log and photographs of the trench wall will be taken.

Representative samples of each soil type observed in the trench will be collected in one quart Ziplock freezer bags and retained at the SBVWCD offices. Selected soil samples from the trenches will be sent to a soils laboratory for analysis as described in Section 3.10.

The soils at the project site are known to be loose and unconsolidated, and contain a high percentage of oversize material. These conditions may make trenching unpractical or unsafe. Under no conditions will trenching be continued if the conditions appear unsafe, and no trench will be completed deeper than five feet. If conditions are deemed unsafe, trenching will be terminated immediately.



# 3.4 SONIC DRILLING

Sonic drilling is a relatively new exploration technique that has some particular advantages for investigation of the Santa Ana River site. The method provides continuous (disturbed) samples in a wide range of soil types, including soils with large particles that preclude sampling by many other techniques. The drill stem and sampler barrel are vibrated vertically at frequencies between about 50 and 180 Hz (hence the name sonic) such that the sampler barrel normally advances by slicing through the soil. The sampler can often cut through large soil particles with the resulting sample providing a valuable view of the soil stratigraphy that other techniques could not provide. This data will be carefully noted by the onsite geologist on a detailed boring log (<u>Attachment 1</u>).

A total of seven sonic borings will be conducted along the long axis of the Borrow Pit. Preliminary locations of the borings are identified on <u>Figure 5</u>. Soil samples will be collected and described as discussed in Section 3.1 at each major change in stratigraphy. The field team will pay particular attention to stratigraphic layers of low permeability, and the existence of older soil horizons, based on grain size, color, and evidence of weathering and staining. Sonic borings will be advanced to a maximum depth of 50 feet, or to refusal. An exception to this is that two of the borings will be converted to monitoring wells, as discussed below.

## 3.5 MONITORING WELL INSTALLATION

Existing information indicates that the regional groundwater table is relatively close to the surface in the vicinity of the Borrow Pit. Surfacing groundwater has been noted in the west-central portion of the Borrow Pit, the northern face, and in Borrow Pit Pond 3. However, existing well information is not of great enough resolution to accurately determine the groundwater elevation in the Borrow Pit to discern if the surfacing groundwater is perched water or representative of the regional water table. Therefore, it is recommended that two of the sonic borings be converted to monitoring wells -- one well would be located midway between Borrow Pit Pond 2 and 3, and the other would be located on the west end of the northern edge of the Borrow Pit.

The monitoring wells will be constructed using the sonic drilling method, as described above, to provide information on the stratigraphy of the basin at depth, and will be used later for documentation of the water table elevation underlying the sites during recharge operations.

Cuttings or cores from the boring excavated for installation of monitoring wells will be logged by the onsite geologist as described above. Because the sonic method allows for close observation of the soil stratigraphy, geophysical logging will not be necessary. The wells will be completed with a 2 inch PVC casing, surface seal and protective barrier or flush mount completion. The monitoring wells will be drilled to a depth of at least 25 feet below the water table, or to 150 feet, whichever is shallower.

#### **3.6 PERCOLATION TESTING**

As previously noted, percolation testing has been completed in the western recharge ponds outside of the Borrow Pit, and at Pond 1 inside the Borrow Pit. Because preliminary information suggests that the shallow layer of the ponds within the recharge pit are clogged (Section 2.4), the concept is to perform percolation tests before and after cleaning of the ponds to evaluate the effectiveness of removing the clogged layer.

Percolation testing of Pond 1 in the Borrow Pit will be conducted using similar methods, as described in Section 2.2, after the SBVWCD has had the opportunity to remove the shallow surface layer from Pond 1. About 65 acre-feet of water will be required for testing Pond 1 after cleaning. The MWH team will provide guidance on the cleaning of Pond 1 depending upon the results of the soil sampling conducted in the pond.

If additional information from drilling and sampling identifies a more favorable area in the Borrow Pit, it may be desirable to construct a new percolation test pond to see if percolation rates in the Borrow Pit could be improved. The most favorable area of Borrow Pit should be chosen for the pond and  $\frac{1}{2}$  to 1 acre test pond constructed. The falling-head test procedure used in the current testing of Borrow Pit Pond 1 should be used.

## 3.7 INTERACTION WITH DISTRICTS STAFF

The MWH team will maintain communication with Districts personnel regarding scheduling, sampling activities, and needs. During periods of field activity, the field team will complete daily field reports describing the work completed each day, special challenges or issues, and planned activities for the next day. MWH will provide oversight and direction to the contractor regarding the location of sampling sites and depth of investigation.

In general, the MWH team will not require extensive assistance from Districts staff during completion of the field work, with the possible exception of percolation testing, which will require diversion of water in a controlled fashion to Borrow Pit Pond 1. The MWH team will request access to the sites during field work.

Visitors to field activities are, of course, welcome. However, it should be recognized that the onsite geologist will be focused on sampling tasks at hand and direction of excavation contractors and, therefore, may not have time for extended discussions with Districts staff. It would be advisable to arrange field trips to review site activities in advance so that the MWH team can have adequate personnel on site to review field activities and to maintain safe working conditions.

#### 3.8 MAPPING AND SURVEYING

All locations of surface soil sampling, trenches, sonic borings, and monitoring wells completed by the field team will be surveyed using a hand-held global positioning system, with a minimum horizontal accuracy of five feet. In addition, the location will be sketched on an aerial photograph of the site. These methods will also be used to locate all existing, and suspected, wells that are believed to be at the site. . The surface completion, condition of the casing, and any identifying markers will be noted on field logs.

More precise surveying by a professional surveyor will be completed at selected locations at the site to determine the topographic elevation and existing wells. Up to 40 survey points will be selected by the field team during the sampling activities. The points are expected to include:

- Elevation of the lowest portion(s), and transects of Borrow Pit Ponds 1, 2 and 3
- Elevation of the reference point for depth-to-water measurements on the new wells and selected existing wells

# **3.9 SAMPLE NUMBERING**

All samples will be marked in the field immediately after collection using a permanent marker. An example of the sample numbering system is as follows:

T-1 /5'-A

Where "T" denotes "trench"; the number after the T indicates the trench number (numbered sequentially, i.e., Trench 1, 2, ...); and the number after the slash indicates the depth (in feet) below ground surface of the sample; the letter after the depth would be used if more than one sample was collected at a particular depth. Other sample types will include:

"S" for surface soil samples "B" for sonic boring samples "MW" for monitoring well samples

All samples collected will be noted on field logs, boring logs, and trench logs, with notation as to the total number of samples collected at each location.

## 3.10 LABORATORY SOIL ANALYSIS

Representative soil samples of each distinct soil horizon will be collected by the onsite geologist at all shallow surface, trench, and boring locations such that each soil type at the particular location is represented. These samples will then be stored at the SBVWCD offices for comparison and cataloging.

After collection of all samples, a subset of approximately 65 soil samples will be selected from this group of samples to be transported to the laboratory for gradation analysis, in accordance with ASTM D-4411 and ASTM D-4464, by a qualified laboratory.

Ten of the 65 soil samples will also be tested for saturated hydraulic conductivity using ASTM-5084. These samples will be transported to:

PTS Laboratories, Inc. 8100 Secura Way Santa Fe Springs, CA 90670 USA (562) 907-3607 (562) 907-3610 (Fax)

# 3.11 CLEAN UP AND INVESTIGATION-DERIVED WASTE

Subsurface exploration at the site will generate soil spoils from trenches and borings. Because the site has been protected from development and is actively used for recharge operations, no contamination of this soil is expected. In general, after a trench or boring is completed, soils will be backfilled into the hole from which they originated. However, if significant fine material, or other material of low permeability is encountered, the field crew will attempt to segregate this soil for movement to a berm or to other areas designated by the Districts.

Waste from the monitoring well will not be returned to the hole from which it was originated. These soils will also be moved to a berm or other area designated by the Districts. Proposed field sampling activities for the SAR Optimization Study will commence upon approval by the SBVWCD and the SBVMWD of this work plan.

As previously noted, field activities will be sequential, such that each activity can take advantage of the information acquired from the previous activity. The estimated duration of each activity is shown in Table 4; however, the actual timing will be dependent on the availability of contractors and equipment. The MWH team will advise the Districts of significant changes in schedule.

Activity	Duration (Weeks)	Approximate Start	Approximate Finish	Notes
Surface Sampling	1	16-June-08	20-June-08	Assumes workplan accepted by June 9
Trenching	1	23-June-08	27-June-08	
Sonic Boring and MW Installation	2	30-June-08	4-July-08	Dependent on availability of drilling rig
Percolation Testing	2	27-June-08	4-July-08	Dependent on completion of cleaning Pond 1 in the Borrow Pit and availability of SA River Water

 Table 4 - Estimated Schedule of Field Work

[use US dating above – June 16?]

A description of the field activities and associated sampling results will be presented in the TM-2 [TM?). The TM-2 will be submitted for review and comment by the SBVWCD and theSBVMWD. This document will present the findings of the field work and an analysis of the percolations tests and recharge capacity estimates. In addition, the TM-2 will describe methods and implementation of the field program.

- ACOE, 2003. Water Control Manual, Seven Oaks Dam & Reservoir, Santa Ana River, San Bernardino County, California. September.
- Agajanian, Jefferey (U.S. Geological Survey), 2008. Data on flow at the Mentone gauging station (USGS No. 11051499) during the March 27 field test. E-mail correspondence.
- Danskin, Wes (U.S. Geological Survey), 2008. Personal communication regarding the geology of the Santa Ana river area near Cone Camp.
- Efthari, Aram (Orange County Resource Development and Management Department), 2008. Data on release rate from the Seven Oaks Dam during the March 27 field test. E-mail correspondence.
- Flordelis, Cisco (SBVMWD), 2008. Data on flow at the SBVWCD Parshall Flume during the March 27 field test. E-mail correspondence.
- Kwan, Robert (U.S. Army Corps of Engineers), 2008. Personal communication regarding geotechnical testing of the Borrow Pit site for the Seven Oaks Dam. E-mail correspondence.
- SBVWCD, 1994. Operational Management Manual of the San Bernardino Valley Water Conservation District. March
- USGS, 1975. Artificial Recharge in the Upper Santa Ana River Area, San Bernardino County, California. USGS Water-Resources Investigations 15-75. 27 p.
- USGS, 2003a. Geologic Map of the Redlands 7.5' Quadrangle, San Bernardino and Riverside Counties, California.
- USGS, 2003b. Geologic Map of the Yucaipa 7.5' Quadrangle, San Bernardino and Riverside Counties, California.

# ATTACHMENT 1 Trench Log and Boring Log

				Plan View-Site Location	(Provide Sketch)	+				EXPLANATION	SOIL TYPE CONTACT (SHARP)	<ul> <li>OTHER CONTACT</li> <li>(AS INDICATED ON LOG)</li> </ul>	ANALYTICAL SAMPLE LOCATION (WRITE SAMPLE NUMBER OUT TO SIDE)	GEOTECHNICAL SAMPLE LOCATION (WRITE SAMPLE NUMBER OUT TO SIDE)	SHADING TO DENOTE	TT BASE OF EXCAVATION	SHOW LOCATIONS AND TYPES OF ALL MAJOR	
C													×	U				
																		_
		ss th =			SW													_
(J)	d	If Ye Dep	No		Dne) W SE													
LOC					n (Circle 0 NE N													-
СН		Circle One No	cle One) No	de Debris	nch Show E W		_											_
REN	Sheet	efusal? (C Yes	hoto? (Cii Yes	6 Man-Ma	/all of Trei J S													_
DTF		<u>~</u>	<u> </u>	~	5 2													
IEL																		_
ш				l Auger														
				Hano	-													_
		eq	ñ															
		: Complet	of Sample															_
	tion	and Time	Number o	0														
	Loca	Date	Total	Drive														
																		_
				de		(F												
				Gr	-	(FEE												_
						GTH												
	ш	arted		Bulk		LEN												_
	n and Dat	d Time Sta	pth		t by/Date													
	Elevatio	Date and	Total De	No. of Samples	Checkec													_
																		_
																		_
																		_
	ber		Itation	£														
	oject Num	erator	anch Orier	anch Widt														_
	Pro	ð	Tre	Tre														_
				ength														
				Trench L														
					ist/Date													
		lier			Irogeologi													
Name	Number	tent Supp	ient Type	Width	list or Hyd		ALE SALE											
Project	Trench	Equipn	Equipn	Bucket	Geoloc		Т Ш М	$\frac{1}{1}$										_

Site Sketch Map										Boring #:       MW#:       Sheet       1       of         Project: SAR Groundwater Recharge Optimization Study       Job #: 1343019       Site:         Job #:       1343019       Site:       Site:         Logged By:       Reviewed By:       Drilling Contractor: Boart Longyear         Drill Rig Type/Method: Sonic Drilling       Drillers Name:       Borehole Diam./Drill Bit Type:       Total Depth         Ref. Elev.       Sampler Type:       Sampler Type:       Sampler Type:							
Dep	th to 1	st Wa	ater ( <u>S</u>	Z):			Time	e/Dat	e:	Drill Start Time/Date: Drill F	inish	n Time	e/Date	:			
Dep	th to V	Vater	After	Drillin	ng (🔽	):	Time	e/Dat	e:	Well Completion Time/Date:							
Dep	th to c	other \	Water	Bear	ing Zo	ones:			1	Soil Boring Backfill Time/Date:							
	الا	(	6 in.	halysi	Size			ě		F		Estimated % Of					
	ntervé	∋d (in.	Ints /	for Ar	ype &	Filler	set)	il Typ					Sand				
/OVA	nple li	overe	v Cot	ained	ing T	ulus I	oth (Fe	CS Sc	Soil Descr	intion	vel	Irse	dium	0	clay		
DIA	San	Rec	Blov	Ret	Cas	Ann	Dep	nSt		iption	Gra	ů C	Mec	Fine	Silt/		
							1 —										
							2 —										
							3 —										
							4 —										
							5 —										
							6 —										
							/										
							8 —										
		<u> </u>					9 —										
					[		10 —										
							11 —										
					·												
			•	•	•		12 -					•		•	QA/QC		

Borir	ng #:		MV	V#:			Project:			Sh	eet		of	
		pe	unts	for	ype			ii			Estin	nated % Sand	6 Of	
PID/OVA	Sample Interval	Recovere (in.)	Blow Cou / 6 in.	Retained Analysis.	Casing T & Size	Annulus Filler	Depth (Feet)	USCS So Type	Soil Description	Gravel	Coarse	Med.	Fine	Silt/Clay
							2 -							
							3							
							4							
							5 —							
							6							
							7-							
							8							
							9							
							0							
							1 –							
							· · · ·							
							2 —							
							3 —							
							4 —							
							5 —							
							6 —							
							7 —							
							8 —							
							9 —							
				+										
							0 -							
				+										
							1 —							
				+										
							2	l	1					QA/QC

# Appendix B

Technical Memorandum 2: Documentation of Field Work and Recharge Capacity Estimate



**Prepared for:** 

San Bernardino Valley Water Conservation District and San Bernardino Valley Municipal Water District

By:

MWH Americas, Inc. in association with GEI Consultants

June 2009

# **Table of Contents**

# Page Number

Executiv	e Summary	1
Section 1	l Introduction	1-1
1.1	Background	1-1
1.2	Purpose and Scope	1-3
Section 2	2 Summary of Field Investigation – Methods and Results	2-1
2.1	Survey of Diversion and Conveyance Facilities	
2.2	Soil Investigation Methods	
2.3	Soil Investigation Results	2-13
2.4	Monitoring Well Construction	
2.5	Percolation Testing and Conveyance Testing	
2.6	Physical Surveys	
Section 3	3 Aquifer Characteristics	
3.1	Geologic Setting	
3.2	Groundwater Migration Conditions	
3.3	GroundWater Barrier	
3.4	Aquifer Characteristics	
Section 4	4 Analysis of Diversion and Conveyance Capacity	
4.1	Intake Structure and Cuttle Weir	
4.2	Discharge Conduit	
4.3	Canal Sections	
4.4	Rock Structure	
4.5	Greenspot Road Culvert	
4.6	Parshall Flume	
4.7	Diversion Structure	
4.8	Summary	
Section 5	5 Analysis of Percolation Capacity	
5.1	Surface Clogging by Fine Particles	
5.2	Compaction of Soils	
5.3	Presence of Vertical or Horizontal Groundwater Movement	
5.4	Groundwater Mounding Limitations	
5.5	Estimation of Percolation Capacity of Existing Facilities	
Section 6	6 Summary of Conclusions	6-1
6.1	Capacity of Existing Diversion, Conveyance, and Recharge Facilities	6-1
Section 7	7 References	7-1

# List of Tables

Table 2-1 Summary of Field Testing Activities	
Table 2-2 Summary of Borings and Depths	
Table 2-3 Sample Particle Size Summary	
Table 2-4 Unified Soil Classification System	
Table 2-5 Average Median Grain Size and Weight Percentage of Silt/Clay	
Table 2-6 Summary of Laboratory Analysis Results	
Table 2-7 Manual Groundwater Level Measurements in Monitoring Wells	
Table 2-8 Summary of Pond 17 Percolation Testing	
Table 2-9 Summary of Pond 1 Percolation Testing (not cleaned)	
Table 2-10 Summary of Pond 1 Percolation Testing (cleaned)	
Table 2-11 Summary of Pond 1 Linear Trench Percolation Testing	
Table 3-1 Summary of Aquifer Test Data	
Table 3-2 Hydraulic Conductivity for Selected Unconsolidated Sediments	
Table 4-1 Maximum Conveyance of Groundwater Recharge Facilities	
Table 5-1 Maximum Infiltration Capacities	

# List of Figures

- 1-1 Location Map
- 2-1 Site Plan of Diversions and Conveyance Facilities
- 2-2 Profile Locations
- 2-3 Profiles of Conveyance Facilities
- 2-4 Detail of Selected Facilities
- 2-5 Surface Soil Sample, Trench, and Monitoring Well Locations
- 2-6 Contours of Equal Median Grain Size Surface Soil Samples in the Borrow Pit
- 2-7 Contours of Equal Fine Particle Percentage Surface Soil Samples in the Borrow Pit
- 2-8 Calculation of Daily Percolation Rates
- 2-9 Topography of Borrow Pit
- 3-1 Geologic Map with Additional Fault Locations from Various Sources
- 3-2 Profile A-A'
- 3-3 Profile B-B'
- 3-4 Groundwater Elevations for Well SBVWCD #3
- 3-5 Hydrograph of Cone Camp Wells
- 3-6 Groundwater Contours August 1, 1998 Wet Year
- 3-7 Groundwater Contours December 1, 2004 Dry Year
- 3-8 Depth to Groundwater Contours
- 4-1 Cuttle Weir Typical Section
- 4-2 Conservation District Diversion Structure Discharge Conduit Backwater Curve
- 4-3 Conservation District Diversion Structure Collection Channel Behind Gates Backwater Curve
- 4-4 Present Calculated Flow Rate
- 4-5 Conservation District Diversion Structure Discharge Conduit Backwater Curve
- 4-6 Conservation District Diversion Structure Discharge Conduit – Backwater Curve
- 5-1 Change in Groundwater Elevation Assuming Various K Values
- 5-2 Capacity of Diversion and Conveyance Facilities
- 5-3 Capacity of Existing Percolation Ponds
The San Bernardino Valley Water Conservation District (Conservation District) in association with the San Bernardino Valley Municipal Water District (Municipal District) retained MWH Americas, Inc., in association with GEI Consultants to evaluate the Conservation District's recharge facilities adjacent to the Santa Ana River in San Bernardino County, California. The Agencies are seeking to evaluate the capability of the existing Conservation District facilities along the Santa Ana River to capture and store the proposed spreading objectives identified in the Upper Santa Ana River Watershed Integrated Regional Water Management Plan (IRWMP). Should the existing facilities be inadequate, MWH will prepare conceptual designs for the improvements necessary to meet the proposed spreading objectives. The maximum instantaneous flow rate, as defined by the Districts and based on the Environmental Impact Report for the Santa Ana River Water Right Applications for Supplemental Water Supply (SBVMWD, 2007), is 500 cubic feet per second (cfs). The maximum yearly volume identified under "Scenario A" of the IRWMP is 80,000 acre-feet.

This Technical Memorandum (TM) covers the first phase of the study involving quantification of the capacity of the existing diversion, conveyance and percolation facilities. This was accomplished by review of pertinent historical information and reports, preparation of a workplan for collection of field data, collection of field data, and analysis of the existing and new field data. This report presents the methods and results of collection of field data, analysis of the data, conclusions relative to the capacity of existing facilities.

The field work was conducted during March through December, 2008 and consisted of :

- Field flow testing of the diversion and conveyance facilities
- Survey of diversion works and conveyance (measurements of dimensions and slopes)
- Soil investigation consisting of:
  - Excavation of 15 trenches
    - Collection of 72 surface soil samples
    - Drilling, sampling, and lithologic logging of 7 borings to a maximum depth of 157 feet
    - Laboratory analysis of 75 samples for grain size analysis, and 16 of these samples for analysis of hydraulic conductivity.
- Construction of 6 monitoring wells and installation of automated monitoring equipment
- Several types of percolation tests at existing recharge ponds
- Physical surveys of existing well locations and elevations

Major conclusions of the study are:

• The sedimentary materials underlying the recharge facilities form an unconfined aquifer consisting of permeable, coarse, sandy gravel and/or gravelly sand. No significant, laterally-continuous strata of low permeability are present that would prevent the downward percolation of recharge water.

- Some existing ponds have a thin layer of silt and/or clay derived from the introduction of turbid recharge water which limits percolation capacity.
- Faulting associated with the San Andreas Fault Zone has created a groundwater barrier which limits recharge capacity on the eastern portion of the site, causing shallow groundwater that surfaces or "daylights" east (upgradient) of this barrier.
- During high runoff periods such as those that occurred in 1980, 1993, 1998 and 2005, the regional area in the vicinity of the recharge facilities may become saturated with shallow groundwater, limiting recharge in all of the facilities. For example, groundwater elevations at Well SBVWCD #3 came within 40 feet of the ground surface and groundwater elevations at the Cone Camp well came within 25 feet of the ground surface during some of these periods. In areas of lower topographic location that these wells the groundwater elevation was probably at or near the surface. However, these events have been very temporary, and may occur at a different frequency depending on the operation of the Seven Oaks Dam.
- The yearly recharge goal of 80,000 acre-feet identified in the IRWMP is possible with the construction of new infiltration, diversion, and conveyance facilities, assuming ambient groundwater levels are low enough to accept this water.
- Groundwater mounding (whereby the water table rises to the ground surface) may to occur if a percolation rate of 500 cfs is sustained.
- The current intake capacity of the Intake Structure without modification is approximately 150 cfs.
- Downstream of the Intake Structure and Cuttle Weir, earthen canals limit the capacity of the conveyance facilities to approximately 300 cfs.
- The recharge capacity of the existing percolation ponds at the SAR recharge facility west of the groundwater barrier is approximately 145 cfs. With additional capacity of 50 cfs in the eastern portion of the Borrow Pit, the total estimated recharge capacity of existing facilities is 195 cfs.

The final report will include recommendations and conceptual design of any physical improvements or change in maintenance and operational methods required to meet the recharge goal defined in the IRWMP.

# Section 1 Introduction

The San Bernardino Valley Water Conservation District (Conservation District), in association with the San Bernardino Valley Municipal Water District (Municipal District), "Districts," retained MWH Americas, Inc., in association with GEI Consultants (collectively referred to as the MWH team) to evaluate the capability of the existing Conservation District facilities along the Santa Ana River to capture and store the proposed spreading objectives identified in the Upper Santa Ana River Watershed Integrated Regional Water Management Plan (IRWMP). Should the existing facilities be inadequate, MWH will prepare conceptual designs for the improvements necessary to meet the proposed spreading objectives. The maximum instantaneous flow rate, as defined by the Agencies and based on the Environmental Impact Report for the Santa Ana River Water Right Applications for Supplemental Water Supply (SBVMWD, 2007), is 500 cubic feet per second (cfs). The maximum yearly volume identified under "Scenario A" of the IRWMP is 80,000 acre-feet.

## 1.1 BACKGROUND

The Conservation District owns and operates surface water diversion and conveyance facilities and groundwater recharge spreading facilities adjacent to the Santa Ana River along Greenspot Road, south and west of the Seven Oaks Dam in Mentone, San Bernardino County, California (**Figure 1-1**). The Conservation District was formed in 1932 to obtain water for conservation purposes, essentially replacing the Water Conservation Association formed in 1910 (SBVWCD, 1994). The majority of the Conservation District facilities have been in place since the 1930s for the purpose of diverting and recharging water from the Santa Ana River. A relatively recent development in recharge operations is the construction of the Seven Oaks Dam, which is a earth and rock fill dam located approximately 1 mile upstream of the Conservation District's diversion. Construction of the dam, supervised by the United States Army Corps of Engineers (USACE), began in August 1994 and was completed in November 1999. Material was excavated for the dam from an approximately 240 acre area (herein called the "Borrow Pit") that formerly contained a series of recharge ditches and ponds operated by the Conservation District. Subsequent to the completion of the dam, three infiltration ponds were constructed at the east end of the Borrow Pit (labeled Ponds 1 through 3 on Figure 1-1).

The remaining recharge basins (Ponds 9 through 17) and the Borrow Pit are located south and west of Greenspot Road, and north of the Santa Ana River. Flow from the Santa Ana River below the Seven Oaks Dam is diverted at a concrete structure ("Intake Structure") behind a small dam-like impoundment called the Cuttle Weir. From the Intake Structure, water flows through various culverts, tunnels and other structures where it can be diverted to various ponds including those in the Borrow Pit.



## **1.2 PURPOSE AND SCOPE**

The Santa Ana River Groundwater Recharge Optimization Study consists of three primary tasks:

Task 1: Evaluate Existing Groundwater Recharge Spreading Facilities and OperationsTask 2: Establish Spreading ObjectivesTask 3: Perform Feasibility Analysis and Prepare Conceptual Design to Meet SpreadingObjectives

**Appendix A** includes the approved scope of work and schedule for the Santa Ana River Optimization Study. Task 1 includes a review of the pertinent historical information and reports, preparation of a workplan for collection of field data, collection of field data, and reporting the results of that field data regarding the diversion, conveyance, and infiltration capacity of the existing facilities.

The MWH team reviewed pertinent information and prepared a work plan which was contained in Technical Memorandum 1 (TM-1, MWH, 2008). The field work was designed to identify factors which may limit the volume of water that could be recharged, including:

- Inadequate capacity of diversion or conveyance facilities
- Compaction of shallow soils in the Borrow Pit due to heavy equipment use during construction of the pit
- Presence of fine-grained material that "clog" the surface of the Borrow Pit
- Shallow groundwater which mounds to the surface
- Presence of horizontal or vertical barriers to groundwater flow, such as clay layers or faults

The field work was also designed to gather additional data on shallow soil and aquifer conditions and resolve data gaps regarding well locations and surface elevations at the site.

The purpose of this document (TM-2) is to document the results of the field investigation, describe factors which limit recharge, and document the recharge capacity of the existing facilities.

The final report will involve recommendations for improvement of existing facilities, conceptual design of new facilities, and maintenance and operational methods to increase recharge capacity.

# Section 2 – Summary of Field Investigation Methods and Results

The field work performed at the SAR recharge facilities included:

- Field flow testing of the diversion and conveyance facilities
- Survey of diversion works and conveyance
- Soil investigation consisting of:
  - o Trenching
  - Collection of surface soil samples
  - Borehole drilling
- Monitoring well construction
- Percolation testing
- Physical surveys

The bulk of the field investigation occurred between March and August 2008, with the specific testing milestones shown below in Table 2-1.

Date	Activity
March 27, 2008	Flow testing to observe flow conditions in the conveyance
	facilities between the Intake Structure and the Conservation
	District recharge facilities
March 27-May 9, 2008	Percolation testing of Ponds 9 through 17, and Dike D
May 10-22, 2008	Percolation testing of Borrow Pit, Pond 1
June 18-June 26, 2008	Surface soil sampling conducted of Borrow Pit and adjacent ponds
June 19-20, 2008	Trenching (Larry Jacinto Construction, Inc.) and sampling
June 23-July 10, 2008	Sonic Drilling and monitoring well installation (Boart Longyear
	Company)
July 14-August 1, 2008	Percolation testing of Borrow Pit, Pond 1 after approximately 1
	foot of basal material was removed
July 31, 2008	Samples submitted to PTS Laboratories, Inc. for grain size
	distribution and hydraulic conductivity analysis
August 7-8 2008	Survey conducted (Calvada Surveying, Inc.)
August 22, 2008	Installation of transducers at monitoring wells
September 19, 2008	Field visit and survey of existing well locations with the
	Conservation District
October 13-24, 2008	Percolation testing of linear trench excavated in Pond 1
December 5, 2008	Transfer of transducer data logging equipment to the Conservation
	District

Table 2-1Summary of Field Testing Activities

These events are described in more detail in the following sections.

# 2.1 SURVEY OF DIVERSION AND CONVEYANCE FACILITIES

A flow test was conducted on March 27, 2008 that allowed the Districts and the MWH team to observe flow conditions in the diversion and conveyance facilities between the Seven Oaks Dam and the Conservation District recharge facilities. A summary of the flow test and key observations was provided in TM-1 (MWH, 2008). Subsequent to that test, more detailed physical surveys of the diversion and conveyance facilities were performed.

The following is a summary of the physical survey of the diversion and conveyance facilities to collect the data needed to calculate each facility's current capacity. The locations of the diversion works and conveyance facilities are shown in **Figure 2-1**. Diversion and conveyance dimensions were compiled from record drawings and/or field measurements to produce channel cross-sections at the locations shown on **Figure 2-2**. The cross-sections are shown on **Figure 2-3** (these field measurements were used to calculate theoretical capacities for typical sections). Details of selected diversion and conveyance facilities are shown on **Figure 2-4**. Photos taken during the physical survey are included as **Appendix E**.

The record drawings of the recharge facility were reviewed, and field measurements of the facilities were taken to supplement information that was lacking in the record drawings provided by the Conservation District and the Municipal District. Information gathered included canal channel and side slopes as well as maximum gate openings at control structures. Because earthen canal sections along the conveyance system varied, several canal cross-sections were surveyed.

Provided below is a description of information on the diversion and conveyance facilities collected during the field investigation. Key features are listed on Figure 2-1.

## Intake Structure

Flow to the recharge basin conveyance system is initiated by diverting water released from the Seven Oaks Dam into the Intake Structure (Figure 2-1). The diversion occurs by water backing up behind the Cuttle Weir, which is a dam-like structure perpendicular to the river channel. The Intake Structure is composed of six gates, each 6 feet wide. The gates are raised to allow flow into a concrete channel that slopes to a tunnel. The intake structure record drawings show the general dimensions and elevations of the structure. Verification of the elevation difference between the gate sills (elevation 1952.0 feet per record drawing) and the top of the Cuttle Weir was made during field surveys conducted on September 30 and October 7, 2008.

Key information that was not shown in the records drawings was the maximum height at which each gate could open. The maximum height each gate could open was determined by measuring the thread lengths of each gate stem. With Gate 1 being the gate closest to the tunnel entrance and Gate 6 being the furthest, Gates 1 and 2 can be opened by about 4.5 feet, Gate 3 by 2.25 feet, and Gates 4 to 6 by 4.0 feet.









The Gate 3 stem appears to have been altered, which limits its opening height in relationship to the other gates. The difference in elevation between the gate bottom sill and the Cuttle Weir was measured and compared to the spot elevations shown on the record drawings. The field survey data revealed that the bottom elevation of the gates was 2 feet 8 inches lower than top of the lowest section the Cuttle Weir with an elevation of 1954.7. Pictures taken during the field survey show some large boulders directly in front of the intake structure which could obstruct water from flowing freely through the gate openings (Appendix E).

## **Closed Conduit (Tunnel) and Sand Box**

Water enters the Intake Structure and discharges into a sloped concrete channel that varies in depth from 7 to 13 feet below the Intake Structure's bottom sill. The discharge end of the channel is connected to a reinforced concrete structure made up of two closed conduit sections with a slope of 0.5%. The first is a rectangular 6 feet by 9 feet, approximately 200 feet long section. The second is a rectangular 8 feet by 7 feet, approximately 100 feet long section. These structures discharge into the Sand Box, an initially 8 feet wide open channel with 11 feet high concrete walls (except at the overflow section which has a 7 feet wall section). The Sand Box is approximately 110 feet in length and expands to a width of 14 feet just downstream of a 48 inch diameter sluice gate that could be used to flush water into the Santa Ana River.

The Sand Box structure contains two 6 feet wide bulkhead gates at its downstream end, after which it discharges into an earthen canal. Physical measurements were taken at the 6 feet gate structures by measuring the gate stems' threads to determine their opening height. The measurements taken suggest that both gates can open to a maximum of 4 feet. The slope of the closed conduit and Sand Box were taken from record drawing information.

## Canal Section between Sandbox and Rock Structure

Beyond the Sandbox gates is an earthen canal. The typical canal section shown on record drawings for "Conduit Santa Ana Weir," dated May 1930, is not representative of the existing canal. Therefore, cross-sections and slopes of the existing canal were measured to more accurately determine its capacity. Slopes were measured using a level and rod. To find the representative slope of a particular reach of canal, the elevation difference of two points 100 feet apart was measured. Canal cross-sections were obtained by measuring the elevation differences at the bottom corners and top edges of the canal with the level and rod. The horizontal lengths of the cross-section were then measured by hand with a measuring tape. This procedure was used to obtain slopes and cross-sections of earthen canals throughout the conveyance system.

The representative slope of the earthen canal measured between the Sandbox and Rock Structure is 0.033 vertical foot per horizontal foot (unitless). The cross-section taken for this canal section was at a point where the banks of the canal appeared to be at its lowest. The canal contains various sized boulders with high and low points due to uneven erosion.

## **Rock Structure**

Between two sections of the earthen canal is a structure built out of rocks that has been mortared together to form a rectangular channel (Rock Structure). The Rock Structure measures 14 feet wide by 6 feet tall. On the structure's north side is an inlet with a gauging station. On the south side is a gate that opens into the Santa Ana River. At the Rock Structure's downstream end there are two 7 feet wide gate openings. No record drawings of the Rock Structure are available, so physical measurements were taken of the Rock Structure as shown on Figure 2-4.

## Canal Section between Rock Structure and Bridge

After the Rock Structure, water flows into an earthen canal. The beginning of this canal appears to have eroded approximately 1 to 1.5 feet below the invert at the inlet of the Rock Structure. Two cross-sections were surveyed between the Rock Structure and the bridge as shown in Figure 2-3. The representative slope measured through this reach was 0.023.

## Bridge

A deteriorated wooden bridge with steel I-beams spans across the canal downstream of the Rock Structure. No record drawings of the bridge are available. A vertical distance of 8 feet was measured from the bottom of the bridge to the canal invert. The approximate widths of the top and bottom of the canal at the bridge are 32 and 14 feet, respectively.

#### **Canal Section between Bridge and Greenspot Road**

The concrete abutments and pier to an old bridge are present in the canal approximately 200 feet upstream of Greenspot Road. The slope of the canal measured along this reach is 0.024. The approximate locations of the cross-sections are located in Figure 2-2.

#### **Greenspot Road Culvert**

The culvert at Greenspot Road is 10 feet wide by 4 feet high (Figure 2-4). The top (road) is 1 foot 10 inches thick. The concrete on the downstream end of the culvert is eroded and washed out, leaving a depression in the canal. This depression indicates that water flows through this culvert at high velocities.

#### **Canal Section between Greenspot Road and Parshall Flume**

The slope of the earthen canal between Greenspot Road and the Parshall Flume was measured to be 0.018 feet (Figure 2-3). The approximate location of the cross-section is shown on Figure 2-2.

## Parshall Flume

Figure 2-4 shows the Parshall Flume's overall dimensions taken from a record drawing. The structure is 97 feet long with an initial width of 28 feet with 6 feet high concrete walls. The

Parshall Flume necks down to 18 feet before it expands to a width of 32 feet at its downstream end.

## **Canal Section between Parshall Flume and Diversion Structure**

The slope of the earthen canal between the Parshall Flume and the Diversion structure is 0.035. The location of a cross-section measured at the Canal Section is shown on Figure 2-2, while the cross-section measurements that were taken at the bank's lowest points are shown on Figure 2-3.

## **Diversion Structure**

The Diversion Structure consists of three gates: two gates side by side on the north and one gate on the canal's southerly side. Record drawings of this structure are not available, so field measurements were recorded. The gate to the south is used to divert water to the Borrow Pit, which includes Ponds 1, 2, and 3, while the north gates discharge water to the spreading grounds west of the Borrow Pit. The width of the canal measured upstream of the diversion structure is 14 feet. Each gate is 8 feet wide by 3 feet tall, and opens to a maximum height of 4 feet (Figure 2-4).

## 2.2 SOIL INVESTIGATION METHODS

In June and July 2008, MWH conducted a soil investigation as described in the field work plan (MWH, 2008). The soil investigation consisted of surface soil sampling, trenching, and sonic drilling. The investigation was performed under the supervision of a registered Professional California Geologist in accordance with the American Society for Testing and Materials (ASTM) Method D 2487-06 and the Unified Soil Classification System (USCS). The various sample locations are shown on in **Figure 2-5**. Trench logs and boring logs are included as **Appendix C and D**, respectively. Selected photos of the soil sample, trenching and sonic drilling taken during the soil investigation are included in Appendix E.

## Surface Soil Sampling

Seventy-two surface soil samples were collected at a depth of approximately one foot along a rectangular pattern within the Borrow Pit and from Ponds 10 through 17 to the west of the Borrow Pit (Figure 2-5). The soils encountered during this sampling were visually classified using ASTM methods, and selected samples were sent to a soils laboratory for more quantitative testing. Visual soil descriptions noted during surface soil sampling are provided in **Appendix D**.

## Trenching

Ten trenches were proposed in the field work plan (MWH, 2008). However, to provide better definition of the area and to investigate specific areas of interest, a total of 15 trenches (T1 – T15) were excavated at 12 locations in the study area as shown on Figure 2-5 and described below:



- Two of the trenches (T1 and T2) were in Pond 1 to investigate the soils in this pond and to attempt to observe evidence of a splay of the San Andreas Fault system which was shown to transect Pond 1 in existing literature (USGS, 2003b).
- A trench was excavated in both Pond 2 (T3) and Pond 3 (T4) to characterize surface and shallow sediments in these ponds.
- Three trenches (T5 through T7) along the north wall of the Borrow Pit were excavated to evaluate the cause of surfacing groundwater historically observed on the north wall.
- A trench (T8) was located in the southwest corner of the Borrow Pit where water was often observed to surface or "daylight" by the Conservation District staff.
- A trench (T9) was located in the southeast corner of the Borrow Pit to characterize soils at that location.
- A trench (T10) was located outside the Borrow Pit at Pond 10 to compare the sediments with those located within the Borrow Pit.
- Four trenches were excavated in the center portion of the Borrow Pit to investigate the surfacing groundwater noted during field efforts (T11A through T11D).
- A final trench (T12) was located northwest of T8 to confirm soil conditions where water historically ponds.

All trenches were excavated using a backhoe operated by Larry Jacinto Construction, Inc. under contract to MWH to an approximate depth of 5 feet. An on-site geologist logged the side wall of the trenches to observe stratification, clay seams, or other pertinent features (Appendix C). At the location of T11, 4 trenches were excavated (T11A, T11B, T11C, and T11D). Water was observed to percolate into the trenches at depths of 2.5, 2.9, 3.0 and 4.5 feet, respectively. Water was not seeping into the trenches at any other locations. Soil samples were collected at depths of approximately 1 foot and 5 foot or just above water. Once logged and sampled, the trenches were backfilled.

## Sonic Drilling

Five borings and two monitoring wells were proposed in the field work plan (MWH, 2008). However, based on the field observations of MWH geologists, and to provide more data for soil and aquifer analysis, a total of seven borings were completed, six of which were completed as monitoring wells. Drilling was conducted using the sonic drilling method as described in TM-1 (MWH, 2008), by Boart Longyear Company, under contract to MWH. The rationale for location of the borings and monitoring wells is described below.

- Three sonic borings were excavated along the long axis (east-west) of the Borrow Pit (SB-1, SB-2, and SB-3) to evaluate changes in soil properties from east to west. These borings were completed as shallow monitoring wells for future monitoring to evaluate aquifer conditions and depth to groundwater.
- One sonic boring was excavated between Ponds 2 and 3 in the Borrow Pit (MW-1) and completed as a monitoring well for observation of changes in groundwater levels during

subsequent percolation testing.

- One sonic boring was excavated on the northern boundary of the Borrow Pit (MW-3) and completed as a monitoring well to assess why groundwater surfaces along the north wall of the Borrow Pit.
- One sonic boring was completed (MW-2a) on the upper elevations of the perimeter access road on the southern edge of the Borrow Pit to evaluate soil and aquifer conditions on the southern edge of the Borrow Pit. This boring met refusal at a depth of 90 feet. To complete a monitoring well at this location, a second borehole was completed at lower elevation on the southern edge, and within the Borrow Pit (MW-2).

The sonic boring method consists of a 6 inch diameter barrel with a 4 inch diameter barrel sampler and a cutting bit at the lower end. MWH's on-site geologists logged the samples and transferred the samples into a sample box for storage. Per the direction of the Conservation District, the samples were stored at a facility owned by the Conservation District located at the Seven Oaks Dam. The boring information is summarized in **Table 2-2**. More detail on construction of the monitoring wells is provided in Section 2.4 below.

Boring/Monitoring Well Designation	Date Installed	Depth	Completed as a Monitoring Well?
Wen Designation		(leet below ground surface)	Womtoring wen.
SB-1	7/3/2008	50	Yes
SB-2	7/3/2008	50	Yes
SB-3	7/7/2008	50	Yes
MW-1	6/30/2008	150	Yes
MW-2a	6/26/2008	90	No
MW-2	6/26/2008	102	Yes
MW-3	7/9/2008	157	Yes

Table 2-2Summary of Borings and Depths

## Sample Selection for Laboratory Analysis

MWH selected representative soil samples for laboratory analysis and submitted seventy-five soil samples for grain size analysis distribution by ASTM Method D422/4464M. Sixteen of the 75 samples were also submitted for hydraulic conductivity analysis by ASTM Method D5084. Among the 75 samples selected for analysis, 43 of them were from surface soil sampling locations, 10 were from the trenches, and 22 were from boreholes. The rationale for sample selection for laboratory analysis was to focus on locations where groundwater had historically surfaced in the Borrow Pit, to focus on the three existing ponds in the Borrow Pit and the western end of the Borrow Pit, and to provide a reasonable geographic and vertical distribution of samples.

# 2.3 SOIL INVESTIGATION RESULTS

Sample identification, depth, median grain size, and particle size distribution in weight percent are summarized in **Table 2-3** for all samples submitted for laboratory analysis.

			Median	Particle Size Distribution, wt. percen		ercent		
	Depth	Description	Grain Size	Gravel	5	Sand Size	· •	Silt/Clav
Sample ID	(foot)	USCS <sup>(1)</sup>	(mm)		Coarse	Medium	Fine	j
MW-1-52	52	Medium sand	0.924	17.73	13.25	44.41	16.94	7.67
MW-1-57	57	Gravel	N/A	71.59	8.58	10.01	7.54	2.28
MW-1-72	72	Coarse sand	0.899	27.34	6.9	32.23	24.55	8.97
MW-1-75	75	Gravel	5.699	53.74	15.96	24.16	5.7	0.43
MW-1-91	91	Medium sand	0.586	6.63	4.67	46.34	33.78	8.58
MW-2-3	3	Coarse sand	0.696	26.23	9.26	22.82	26.14	15.55
MW-2-32	32	Gravel	0.904	34.12	5.43	23.96	24.81	11.67
MW-3-102	102	Medium sand	0.661	8.62	12.54	41.69	28.57	8.57
MW-3-104	104	Medium sand	0.637	3.49	12.66	46.07	28.56	9.22
MW-3-116	116	Medium sand	0.67	7.95	8	47.77	26.99	9.29
MW-3-30	30	Fine sand	0.297	0	2.88	36.81	42.11	18.2
MW-3-38	38	Gravel	0.667	33.99	2.02	24.16	30.63	9.2
MW-3-43	43	Medium sand	0.701	5.38	13.16	49.23	27.77	4.45
MW-3-58	58	Coarse sand	1.625	31.63	14.03	31.93	16.42	5.99
MW-3-91	91	Medium sand	0.589	0.67	12.43	48.52	31.36	7.02
SB-1-40	40	Gravel	3.772	46.79	11.89	27.66	12.28	1.38
SB-1-47	47	Gravel	14.125	61.49	7.84	17.46	10.72	2.48
SB-1-5	5	Gravel	2.184	42.31	8.54	18.7	19.51	10.94
SB-2-18	18	Medium sand	1.191	12.48	19.15	46.95	15.75	5.68
SB-2-3	3	Gravel	2.596	43.6	9.3	23.39	17.33	6.38
SB-3-43	43	Medium sand	0.725	21.65	11.83	28.32	27.15	11.05
SB-3-7	7	Gravel	1.395	38.77	6.76	27.21	21.19	6.07
SS-01	1	Gravel	1.984	39.95	9.91	27.68	16.41	6.05
SS-03	1	Gravel	1.321	30.83	11.26	31.04	19.53	7.32
SS-05	1	Coarse sand	1.114	24.8	13.5	31.31	21.67	8.72
SS-07	1	Medium sand	0.754	12.14	15.21	37.68	25.49	9.49
SS-09	1	Medium sand	0.912	15.66	14.91	39.44	22.32	7.68
SS-11	1	Gravel	1.683	37.62	9.71	28.33	18.84	5.49
SS-13	1	Medium sand	0.843	19.45	12.47	35.36	23.97	8.75
SS-15	1	Medium sand	1.085	14.32	18.26	43.35	15.8	8.28
SS-17	1	Medium sand	0.837	14.05	14.25	39.57	24.02	8.11
SS-19	1	Gravel	7.249	56.55	9.29	18.44	11.24	4.49
SS-21	1	Medium sand	0.979	13.23	18.72	39.31	21.19	7.55
SS-23	1	Coarse sand	1.051	23.43	11.58	39.48	19.65	5.85
SS-25	1	Coarse sand	1.062	22.04	14.25	37.61	20.49	5.61
SS-27	1	Coarse sand	1.628	31.93	14.05	28.84	20.93	4.26
SS-29	1	Coarse sand	1.358	26.51	15.05	33.25	21.86	3.33
SS-31	1	Coarse sand	1.177	19.29	18.69	36.03	20.85	5.14
SS-32	1	Coarse sand	1.097	24.22	13.75	33.62	20.01	8.39

Table 2-3Sample Particle Size Summary

			Median	Part	icle Size	Distributio	n, wt. pe	ercent
	Depth	Description	Grain Size	Gravel		Sand Size		Silt/Clay
Sample ID	(foot)	USCS <sup>(1)</sup>	(mm)		Coarse	Medium	Fine	
SS-33	1	Coarse sand	1.099	22.21	15.57	34.22	21.26	6.74
SS-34	1	Coarse sand	0.934	23.49	9.77	35.88	21.4	9.46
SS-35	1	Coarse sand	1.104	26.17	11.26	34.43	20.8	7.34
SS-36	1	Gravel	2.783	43.27	11.16	27.9	13.6	4.06
SS-38	1	Medium sand	1.036	20.74	14.45	32.31	22.74	9.76
SS-40	1	Medium sand	0.757	13.73	14.2	34.34	25.19	12.53
SS-42	1	Medium sand	0.777	13.09	16.11	34.5	27.93	8.37
SS-44	1	Coarse sand	1.138	24.98	14.45	31.77	21.55	7.25
SS-46	1	Medium sand	0.795	20.58	10.6	34.16	24.49	10.18
SS-47	1	Medium sand	0.438	4.25	4.37	43.17	39.64	8.58
SS-49	1	Medium sand	0.438	3.96	7.86	39.52	37.35	11.31
SS-50	1	Medium sand	0.71	5.24	11.6	48.23	30.06	4.88
SS-51	1	Medium sand	0.59	0.78	9.77	51.84	32.91	4.7
SS-52	1	Medium sand	1.133	14.83	18.21	46.63	19.63	0.69
SS-53	1	Fine sand	0.05	0	2.03	16.58	20.66	60.73
SS-54	1	Medium sand	0.329	6.83	7.13	30.72	29.88	25.43
SS-55	1	Medium sand	0.423	6.09	10.82	33.21	27.01	22.88
SS-56	1	Medium sand	0.652	17.52	7.89	36.5	25.38	12.71
SS-57	1	Gravel	0.926	30.55	6.84	31.7	23.31	7.6
SS-58	1	Coarse sand	1.191	26.8	10.42	43.37	15.55	3.85
SS-62	1	Coarse sand	1.176	24.57	10.55	44.53	16.55	3.79
SS-64	1	Gravel	0.996	32.78	4.17	34.27	22.71	6.07
SS-66	1	Gravel	2.419	32.29	26.21	23.1	9.58	8.82
SS-68	1	Medium sand	0.881	16.02	6.69	55.17	16.39	5.72
SS-71	1	Medium sand	1.033	13.51	16.03	47.39	19.18	3.9
SS-72	1	Medium sand	0.786	14.58	13.35	39.69	24.46	7.91
T1-1	1	Coarse sand	1.009	29.49	8.75	29.47	18.94	13.34
T11C-1	1	Gravel	2.242	39.35	12.29	23.69	16.96	7.7
T11C-3	3	Gravel	N/A	74.72	5.31	16.18	3.3	0.5
T1-5	5	Coarse sand	1.654	31.03	13.28	46.7	7.8	1.18
T2-1	1	Medium sand	0.862	19.88	10.52	39.07	25.45	5.08
T2-5	5	Coarse sand	1.384	17.91	20.4	47.65	12.4	1.64
T3-1	1	Gravel	14.397	62.01	4.41	18.39	10.55	4.64
T3-5	5	Coarse sand	1.553	22.96	17.37	49.17	9.11	1.39
T4-1	1	Medium sand	0.914	14.5	14.95	41.77	21.82	6.96
T4-5	5	Gravel	1.782	36.64	11.1	34.99	14.74	2.54

<sup>(1)</sup>Unified Soil Classification System

For comparison to field results, standardized USCS size ranges are provided in Table 2-4.

Classification	Size Range
Cobbles	Above 3 inches
Gravel	3 inches to No. 4 sieve
• Coarse	• 3 inches to <sup>3</sup> / <sub>4</sub> inch
• Fine	• <sup>3</sup> / <sub>4</sub> inch to No. 4 sieve
Sand	No. 4 to No. 200 sieves
• Coarse	• No. 4 to No. 10 sieves
• Medium	• No. 10 to No. 40 sieves
• Fine	• No. 40 to No. 200 sieves
Fines (clay or silt)	Below No. 200 sieve (no minimum size)

Table 2-4Unified Soil Classification System

Forty-eight (43 from the Borrow Pit and 5 from outside the Borrow Pit) surface soil samples and trench samples collected and analyzed at a depth of one foot, and were classified as follows:

- 10 are classified as gravel
- 14 are classified as coarse sand
- 23 are classified as medium sand
- 1 is classified as fine sand

**Figure 2-6** illustrates the estimated contours of equal median size based on samples from a depth of 1 foot in the Borrow Pit. The contours suggest that grain size is finer at the western edge of the Borrow Pit, where water has historically ponded. The 3 surface soil samples from Pond 1 and 3 also have relatively fine grain size distribution. Excluding the 2 abnormally greater median grain size of samples SS-19 and T3P2, the remaining 41 samples from the Borrow Pit have an average median grain size of 1.02 millimeter (mm), and the average median grain size for the 5 samples from the ponds outside the Borrow Pit is 1.33 mm (31 mm coarser). Contours of equal fine particle (silt and clay) percentage of surface soil samples in the Borrow Pit are illustrated in **Figure 2-7**, showing that most areas of the Borrow Pit have a fine particle percentage of less than 6 percent. However, along the western end where water has historically ponded, two anomalously high fine particle percentage ranges (as high as 60 percent) exist. On average, samples from the Borrow Pit contain 9.32 percent of fine particles, while those from the outside the Borrow Pit may be responsible for the low percolation rate as compared to ponds outside the Borrow Pit.





Sample Location	Number of Samples	Average Median Grain Size (mm)	Average Weight Percentage of Silt/Clay (%)	
Borrow Pit	43	1.02*	9.32	
Outside Borrow Pit	5	1.33	5.65	

Table 2-5Average Median Grain Size and Weight Percentage of Silt/Clay

NOTE \* Average of 41 of the 43 surface soil samples, where the samples were taken at a depth of approximately one foot. Samples SS-19 and T3 have been excluded because they have an anomalously high median grain size.

The four soil samples taken at a depth of five feet in trenches have been classified as either coarse sand with a median grain size of 1.654 mm, 1.384 mm, and 1.553 mm, or gravel with a median grain size of 1.782 mm. Of the remaining 22 sonic drilling samples, nine (9) are classified as gravel, 3 coarse sand, 9 as medium sand, and only 1 fine sand.

**Table 2-6** summarizes median grain sizes, silt and clay weight percentage, and hydraulic conductivity of the 16 samples collected from trenching and sonic drilling. Laboratory-measured hydraulic conductivity ranges from 0.00274 to 19.7 feet/day.

Table 2-6 also illustrates that the hydraulic conductivity is more dependent on the percentage of fine particles than median grain size and overall grain size distribution. For example, sample SB-2-3 contains 6.38% silt and clay, but has a relatively large median grain size and is classified as gravel. However, this sample has a much lower hydraulic conductivity than sample SB-2-18, even though SB-2-18 has a smaller median grain size and is classified as medium sand.

In general, samples from Pond 2 have larger median grain size, less fine particles (surface samples) and higher hydraulic conductivity (samples at a depth of 5 feet) than those from Ponds 1 and 3. This result is consistent with the Conservation District staff's observation that Pond 2 has the fastest percolation rate of the three ponds in the Borrow Pit.

Sample ID	Depth (feet)	Description (USCS)	Median Grain Size (mm)	Silt /Clay wt. percent	Hydraulic Conductivity (feet/day)
T1-P1-1	1	Coarse sand	1.009	13.34	0.066
T1-P1-5	5	Coarse sand	1.654	1.18	7.26
T2-P1-1	1	Medium sand	0.862	5.08	0.374
T2-P1-5	5	Coarse sand	1.384	1.64	19.7
T3-P2-1	1	Gravel	14.397	4.64	0.0346
T3-P2-5	5	Coarse sand	1.553	1.39	1.01
T4-P3-1	1	Medium sand	0.914	6.96	0.283
T4-P3-5	5	Gravel	1.782	2.54	6.72
T11C-1	1	Gravel	2.242	7.70	0.0788
T11C-3	3	Gravel	N/A	0.50	15.1
SB-2-3	3	Gravel	2.596	6.38	0.0955
SB-2-18	18	Medium sand	1.191	5.68	0.349
MW-1-72	72	Coarse sand	0.899	8.97	0.113
MW-3-30	30	Fine sand	0.297	18.20	0.0371
MW-3-38	38	Gravel	0.667	9.20	0.00274
MW-3-102	102	Medium sand	0.661	8.57	0.00371

Table 2-6Summary of Laboratory Analysis Results

Note - Hydraulic conductivity values are obtained from laboratory analysis on disturbed samples which may differ from in-situ conditions.

# 2.4 MONITORING WELL CONSTRUCTION

As previously noted, 6 of the 7 sonic borings were completed as monitoring wells for future monitoring of groundwater levels. Deeper monitoring wells were installed to a depth of approximately 150 feet or to the water table and identified as MW-1 through MW-3. Shallower monitoring wells were installed to a depth of 50 feet below ground surface and identified as SB-1 through SB-3 (Figure 2-5) as designated in the work plan (MWH, 2008).

Monitoring wells were constructed using 2 inch PVC casing with a 0.010 inch slotted screen. Wells were screened from the bottom of the borehole to a depth approximately 15 feet below ground surface. A blank casing was installed from 15 feet below ground surface to either ground surface (flush-mounted) or 3 feet above ground surface. The annular space between the borehole wall and the screened casing was packed with #2 Monterey Sand from the bottom of the borehole to a depth of approximately 10 feet below ground surface. Bentonite clay was used to seal the casing above the sand to ground surface. The surface seal used for MW-1 and MW-3 was concrete and completed as flush-mount. SB-1, SB-2, SB-3 and MW-2 were completed with a riser to approximately 3 feet above ground surface, enclosed in a 6 inch steel pipe with cap and surrounded by four protective bollards (4 inch steel pipe sealed on top with concrete). Well completion logs are provided in **Appendix F**.

At each new well location, groundwater levels were measured on July 8, 2008 using a manual electronic sounding tool. On August 22, 2008, MWH completed a second round of manual groundwater level measurements and installed six transducers to automatically record groundwater level measurements. The transducers installed are Level Troll® 100 models

manufactured by In-Situ Inc., of Fort Collins, Colorado. Results of the two manual measurement events are summarized in **Table 2-7**.

Well	Surface Point Elevation <sup>1</sup>	Depth to Groundwater on July 8, 2008 <sup>2</sup>	Groundwater Elevation on July 8, 2008 <sup>1</sup>	Depth to Groundwater on August 22, 2008 <sup>2</sup>	Groundwater Elevation on August 22, 2008 <sup>1</sup>
MW-1	1765.74	67.3	1698.4	75.99	1689.75
MW-2	1700.38	DRY		DRY	
MW-3	1729.03	53.0	1676.0	64.50	1664.53
SB-1	1718.17	23.7	1694.5	35.56	1682.61
SB-2	1673.77	2.8	1671.0	7.37	1666.40
SB-3	1632.80	DRY		DRY	

Table 2-7Manual Groundwater Level Measurements in Monitoring Wells

1 - feet above mean sea level

2 - feet below measuring point

Hydrographs taken from the transducers for these locations are given in Appendix G.

## 2.5 PERCOLATION TESTING AND CONVEYANCE TESTING

The Conservation District percolated water in Ponds 9 through 17, and Dike D west of the Borrow Pit from March 27 to May 9, 2008. During this period, the percolation rate of the ponds and the conveyance system were recorded and subsequently analyzed. A separate percolation rate test was conducted on Pond 17 from April 26, 2008 to May 6, 2008.

On May 9, 2008, the release from Seven Oaks Dam was decreased to conserve water for percolation testing of Borrow Pit Pond 1, which occurred from May 10, 2008 to May 22, 2008. At the completion of this initial percolation test, Pond 1 was cleaned by removing approximately 1 foot of basal material. The pond was then retested using imported water from the State Water Project from July 14, 2008 to August 1, 2008.

The initial percolation rate of Pond 1 was approximately 1.2 feet/day. After cleaning, the percolation rate increased to about 2.6 feet/day. However, this rate was still significantly slower than the percolation rate at Pond 17 of approximately 4.0 feet/day. To evaluate the effect that pond dimensions have on percolation rate (rectangular, as opposed to more linear), a third test was conducted on a portion of Pond 1. For example, Pond 1 is a large rectangular pond with an average water depth of about 4 feet, whereas Pond 17 was constructed by excavating a 15 feet deep linear trench. The third test on Pond 1 consisted of constructing a 6 feet wide, 4 feet deep linear trench that was approximately 550 feet long. Percolation testing of the Pond 1 linear trench was from October 13, 2008 to October 24, 2008.The following sections describe the percolation testing methods and details the percolation testing results on the Conservation District ponds and conveyance system, Pond 17 and Pond 1.

## **Percolation Testing Methods**

Two testing methods were used for estimating percolation rates: (1) falling-head percolation tests and (2) constant-head or constant-rate percolation tests.

## **Falling-Head Percolation Tests**

In falling-head percolation tests, ponds are filled and then the water supply is shutoff. The water levels are allowed to drop due to infiltration and evaporation. Evaporation is considered minimal with infiltration rate greater than 1 foot/day and is generally within the error of the analysis. The percolation rates are estimated by plotting water levels over time. The benefits of a falling-head percolation test are:

- 1. Water supply inflows in to the test pond do not need to be accurately measured. Flow meters generally produce large errors when used to calculate recharge rates in small test basins. This is due to inaccuracies at low flows when a constant water level is being maintained.
- 2. Tests generally require less water for testing than constant rate tests generally because they are run for a shorter time period.
- 3. This method works well on small, irregularly shaped test basins where the wetted area may be difficult to determine.
- 4. The wetted area does not need to be determined with great accuracy.

During the testing, water levels were measured and recorded electronically using a pressure transducer and data logger. Percolation rates were calculated over a 24-hour period using water level measurements. In the test ponds, water level measurements were recorded using this method every 10 minutes for a 24 hour period. Because the wetted area of the test ponds tended to decrease with time, the first 5 to 7 hours of water level measurements were used to estimate the percolation rate for the 24 hour period (**Figure 2-8**). The initial measurements represent pond percolation rates at a near constant-rate with maximum wetted area. The process of filling and percolating is then repeated to obtain a consistent estimate of the percolation rate. A consistent rate is determined when the running average and the daily average are nearly the same over several days.

## **Constant-Head Percolation Tests**

Constant-head or constant-rate percolation tests utilize water levels that are maintained at a consistent level. Water inflows are measured daily, and there is no outflow from the area. The wetted area for each of the ponds must be known and is used to calculate the percolation rate for the area.

Errors in water supply measurements, wetted area, and changing percolation rates produce errors in the calculation of the daily percolation rates. However, for a large test area, a constant-rate test over a long period of time generally produces reliable rate results.



#### Water Level Measurements and Analysis Methods for Calculating Percolation Rates

Test pond water levels were determined with the use of a 15 pounds per square inch (psi) MiniTroll pressure transducer manufactured by In-Situ, Inc. The MiniTroll was installed in 2 inch diameter perforated polyvinyl chloride (PVC) pipe. The PVC pipe was placed vertically in the recharge basin, with the pressure sensor positioned near ground level to give a zero water level reading at zero depth of water. The transducer reads water levels as the height of water column above the pressure sensor read every 10 minutes.

The MiniTroll transducer was connected via SD12 cable to a Campbell Scientific, Inc. CR510 Datalogger. The CR510 Datalogger provided data storage and transmitted water level data to the MWH Team using a modem. The CR510 Datalogger and modem were powered by a 12 volt battery and solar panel. The telemetry station, consisting of datalogger, modem, battery, solar panel, and modem antenna, were housed within a <sup>1</sup>/<sub>4</sub> inch steel-cased protective box.

Water level measurements are used to calculate the percolation rate. Changes in this wetted area or precipitation cause a corresponding change in the percolation rate. Test ponds generally have a consistent wetted area until a particular water level is reached. For Pond 17, the wetted area remained fairly consistent for about 8 hours of percolation, and then decreased. Percolation rates and the wetted area of Pond 1 were consistent for approximately the first 5 hours of percolation, and then decreased.

#### **Summary of Percolation Testing**

The following sections summarize the percolation testing results conducted at the recharge facilities. A complete set of water level plots for the tests are included in **Appendix B**.

#### Percolation Testing at Ponds 9 through 17, Dike D, and Conveyance System

A constant-head flow test was conducted on Ponds 9 through 17, Dike D, and the conveyance system. From March 31, 2008 to April 17, 2008 the flow rate as measured at the Parshall flume into the recharge area west of the Borrow Pit was approximately 145 cubic feet per second (cfs), or 288 acre feet per day (acre-feet/day). Recharge pond capacity values provided by the Conservation District indicate that the total wetted area of Ponds 9 through 17, and Dike D is about 57.5 acres. This wetted area does not include conveyance and diversion facilities between individual ponds. No outflow occurred from Dike D, which could suggest that the wetted area was still increasing during this period or that the inflow equaled the percolation rate. Based on this information, the percolation rate during this time period was about 5 feet per day (288 acrefeet/day divided by 57.5 acres). Because the wetted area does not include the conveyance and diversion facilities, the percolation rate is likely less on a per acre basis for the ponds alone. A rough estimate of the conveyance and diversion facilities used during the test is about 3.1 acres (about 34,000 ft total length by 4ft width). If you assume a percolation of 4 ft/day for the conveyance and diversion facilities this would account for about 12.5 acre-feet/day, and suggest that the ponds recharged at a rate of about 4.8 ft/day ((288-12.5) acre-feet/day divided by 57.5 acres).

Four falling-head percolation tests were performed on Pond 17. Pond 17 is typical of the ponds located west of the Borrow Pit area and was selected for testing because the water flow to the pond could be stopped without affecting other ponds. Pond 17 has a more linear pond design, which is conducive to greater percolation rates, than those ponds located within the borrow pit (Ponds 1-3). This is probably due to "edge effects" of water leaking laterally as well as horizontally.

The results of the four falling-head tests are presented in **Table 2-8** below. As noted previously, the wetted area of Pond 17 remained generally consistent for about the first 8 hours of percolation test, and these measurements were used to estimate the percolation rate of the pond. Based on these tests the percolation rate for the ponds outside the Borrow Pit area appears to be about 4 feet/day.

Test #	Date	Test Drawdown Time (hours)	Recharge Rate (feet/day)	Running Average (feet/day)
1	April 26, 2008	7.7	4.02	4.02
2	April 29, 2008	7.7	3.83	3.92
3	May 2, 2008	7.5	3.92	3.92
4	May 6, 2008	7.7	4.19	3.98

Table 2-8Summary of Pond 17 Percolation Testing

## **Percolation Testing at Pond 1**

Pond 1 was chosen for percolation testing due to its proximity to available water and it is representative of three ponds located at the east end of the Borrow Pit. Pond 1 was tested "as is," meaning that no cleaning of the bottom of the pond was done before testing. The ponds in the Borrow Pit have been used for recharge, and a noticeable layer of fine white clay and silt size material, locally up to 1/16 inch thick, was observed to be common on the floor of pond.

During the first five tests of Pond 1, some leakage out of the pond occurred between the wooden boards used to prevent water from flowing to Pond 2. The leakage was visually estimated at approximately 0.5 cfs and later adjusted to 0.55 cfs to better approximate the later test results. The pond did not dry between tests.

Based on the test results, Pond 1 had a percolation rate of about 1.2 feet/day in its initial condition. Results from the percolation testing are provided in **Table 2-9**, and plots of water level over time are included in Appendix B.

Test #	Date	Test Drawdown Time (Approx. hours)	Comments	Recharge Rate (feet/day)	Running Average (feet/day)
1	May 10, 2008	5	Leakage to Pond #2	1.37	1.37
2	May 12, 2008	5	Leakage to Pond #2	1.25	1.31
3	May 13, 2008	5	Leakage to Pond #2	1.13	1.25
4	May 14, 2008	5	Leakage to Pond #2	1.18	1.23
5	May 15, 2008	5	Leakage to Pond #2	1.18	1.22
6	May 19, 2008	5	No Leakage to Pond #2	1.12	1.17
7	May 20, 2008	5	No Leakage to Pond #2	1.34	1.19
8	May 21, 2008	5	No Leakage to Pond #2	1.20	1.19
9	May 22, 2008	5	No Leakage to Pond #2 Rain Events	1.21	1.20

Table 2-9Summary of Pond 1 Percolation Testing (not cleaned)

Pond 1 was retested by the falling-head method after cleaning. Cleaning consisted of the removal of approximately 1 foot of material from the bottom of the pond and stockpiled outside of the pond in the Borrow Pit area. Pond 1 had a percolation rate of about 2.6 feet/day after cleaning. This is an increase of over 110 percent from the Pond's initial percolation rate of 1.2 feet/day. Results from the percolation testing of the cleaned pond are provided in **Table 2-10** below, and plots of water level over time are included in Appendix B.

Table 2-10Summary of Pond 1 Percolation Testing (cleaned)

Test #	Date	Test Drawdown Time (Approx. hours)	Comments	Recharge Rate (feet/day)	Running Average (feet/day)
1	July 14, 2008	5	Dry from May 27. Cleaned	2.76	2.76
2	July 16, 2008	5	1.3 ft of water at filling	2.62	2.69
3	July 18, 2008	5	Dry for 12 hr before filling	2.56	2.65
4	July 21, 2008	5	Dry for 35 hr before filling	2.51	2.61
5	July 23, 2008	5	Dry for 10 hr before filling	2.62	2.62
6	July 25, 2008	5	Dry for 5 hr before filling	2.66	2.59
7	July 28, 2008	5	Dry for 27 hr before filling	2.53	2.58
8	July 30, 2008	5	Dry for 3 hr before filling	2.51	2.57
9	Aug. 1, 2008	5	Dry for 4 hr before filling	2.53	2.56

## Water Levels during Percolation Testing of Pond 1 (cleaned)

During the percolation testing of Pond 1 (clean), groundwater levels in the newly constructed MW-1 were measured. On July 10, 18 and 20, 2008 water levels were measured by hand and were relatively consistent at 67.3, 66.5, and 66.1 feet below ground surface, respectively. From July 20, 2008 to August 22, 2008, groundwater levels in MW-1 were measured every 30 minutes using a temporary MiniTroll pressure transducer. Depth to groundwater during testing of Pond 1 (clean) from July 14, 2008 to August 1, 2008 generally ranged between 66 and 67 feet below ground surface. However, the groundwater level in MW-1 rose in response to recharge at Pond 1. After pond testing, groundwater levels continued to decline and measured at 76 feet below ground surface on August 22, 2008 when the MiniTroll pressure transducer was removed. On August 22, 2008 the MiniTroll pressure transducer was replaced with the permanent transducers (Level Troll 100).

## **Pond 1 Linear Trench**

A linear trench measuring 6 feet wide by 4 feet deep, and running approximately the length of the Pond 1 was constructed to test if pond design contributed to the slower percolation rate of Pond 1 compared to Pond 17. Falling-head percolation rate testing results indicate that pond design may play a significant role in percolation rates. The percolation rate of the linear trench was about 8 feet/day compared to the cleaned Pond 1 which was about 2.6 feet/day. This is an increase of over 200 percent, and 100 percent better than Pond 17 at 4 feet/day. The higher percolation rate is most likely due to increased lateral infiltration in the trench. Percolation testing results for the linear trench are provided in **Table 2-11**, and plots of water level over time are included in Appendix B.

Test #	Date	Test Drawdown Time (Approx. hours)	Comments	Recharge Rate (feet/day)	Running Average (feet/day)
1	October 13, 2008	7.2	Dry - Newly Constructed	10.84	10.84
2	October 14, 2008	8.3	Dry 14.5 hours	9.43	10.14
3	October 15, 2008	8.3	Dry 13.2 hours	9.29	9.85
4	October 16. 2008	8.7	Dry 13.7 hours	9.06	9.66
5	October 17. 2008	8.8	Dry 13.7 hours	8.76	9.48
6	October 20. 2008	8.2	Dry 62.8 hours	9.30	9.17
7	October 21. 2008	8.3	Dry 11.2 hours	8.45	8.97
8	October 22. 2008	8.7	Dry 11.3 hours	8.16	8.84
9	October 23. 2008	9.7	Dry 11.3 hours	7.88	8.60
10	October 24. 2008	7.5	Dry 9.0 hours	7.88	8.50

Table 2-11Summary of Pond 1 Linear Trench Percolation Testing

## **Observations of Rising Groundwater During Pond 1 Testing**

Rising groundwater was observed flowing into Pond 3 on May 10, 2008 prior to the first percolation testing at Pond 1. The source of this water is not believed to be directly related to the percolation testing, because the presence of rising water occurred in Pond 3 before the filling of Pond 1. However, during a second visit to Pond 3 on May 19, 2008 during the testing in Pond 1, the flow rate of the rising water in Pond 3 did appear to be higher. The location of the rising groundwater in Borrow Pit Pond 3 is at the northeastern end of the pit at its lowest level. The fact that this location is the lowest elevation of Ponds 1 through 3 is consistent with a rising groundwater source, but this cannot be confirmed based on monitoring well data because the monitoring wells were not installed yet. Large areas of rising groundwater were also noted in the main Borrow Pit area near the center during May 2008. This rising water then flows west along the ground surface to the end of the main Borrow Pit where a large pond has developed.

No rising groundwater in Pond 3 was observed during the subsequent testing (clean and linear trench) at Pond 1 in July and October of 2008. However, hydrographs of existing wells and monitoring wells installed during this study indicate that groundwater levels were generally declining from March 2008 into the Fall of 2008. During the testing in July and October of 2008, the lower ambient groundwater levels probably prevented groundwater from rising to the surface during percolation testing performed in the Fall.

## 2.6 PHYSICAL SURVEYS

On August 7, 2008, Calvada Surveying, Inc., under contract to MWH conducted a survey of the new monitoring wells installed as part of the SAR optimization investigation. Additional surveys were completed in Ponds 1 through 3 in the Borrow Pit, and an area in the north-central portion of the Borrow Pit where water has been observed to surface (this area is visible as green vegetation on aerial photographs (e.g. Figure 2-7). The survey information was combined with existing topographic data available (**Figure 2-9**). On September 19, 2008, MWH and the Conservation District staff conducted a field survey of surrounding wells in the vicinity of the Borrow Pit (**Figure 3-1** in the following section). These locations were surveyed using a hand held global positioning system (GPS) and the common names used for each well were confirmed with the Conservation District staff.


# Section 3 Aquifer Characteristics

Understanding of the characteristics of the aquifer underlying the SAR recharge facilities is critical to the evaluation of the facilities ultimate recharge capacity. This section describes how the physical characteristics of the aquifer underlying the SAR recharge facilities relate to percolation of water at existing recharge facilities. The understanding of aquifer conditions at the recharge facilities is based on review of existing literature and data, as well as new information collected during field work described in the previous section.

# 3.1 GEOLOGIC SETTING

**Figure 3-1** is a geologic map of the project area compiled from mapping performed by the U.S. Geological Survey (USGS, 2003a, and 2003b). Figure 3-1 also shows mapped or suspected faults in the area (USGS, 2005), known existing wells in the vicinity of the recharge facilities, and the locations of geologic profiles developed during this study. Each of these is described in more detail below.

Figure 3-1 also illustrates the distribution of two general rock types which occur in the vicinity of the SAR recharge facilities. The two general rock types are unconsolidated Quaternary deposits and Mesozoic bedrock material. Bedrock materials consisting of older (Mesozoic) igneous and metamorphic rocks are located in the topographic highlands northeast of the site. These materials are relatively impermeable and do not percolate or store appreciable amounts of water.

The other general rock type is relatively recent unconsolidated alluvial material where the existing recharge facilities are located, that forms the aquifer that receives, stores, and transmits recharge water. The USGS identified at least five separate alluvial sequences that are present near the recharge facilities. These separate alluvial sequences are identified on Figure 3-1 using the symbol convention "Q" (for Quaternary) followed by an abbreviation of a description of the deposits, such as very young wash deposits (Qvyw). The youngest of these is a Holocene surficial deposits consisting of sediment transported and deposited in channels and washes (Qvyw) of the Santa Ana River, or surfaces of alluvial fans (Qvyf), and alluvial valleys (Qvya). These deposits are typically very coarse and permeable, with no soil development. Within and surrounding the Borrow Pit, Qvyw consists of cobble-boulder-gravel and poorly-sorted gravelly sand that contains boulders as much as one meter or more in diameter.

Older surficial deposits consist of sediment that is slightly to moderately consolidated and slightly to moderately dissected. These alluvial valley deposits are exposed at the ground surface within and in the vicinity of the Borrow Pit. The upper surfaces of these deposits are commonly capped by slight to moderately developed soil profiles.

MWH



The presence of the San Andreas Fault Zone in the vicinity of the recharge facilities is an important component not only in the physical configuration of the unconsolidated deposits, but also in the groundwater flow regime. The main San Andreas Fault Zone consists of two nearly parallel branches through most of the San Bernardino area. In the vicinity of the recharge facility, the fault zone trends in a southeast-northwest direction, with the main San Andreas Fault (San Bernardino Strand) crossing near the northeast extent of Greenspot Road, and a splay of the fault (Mission Creek Strand) crossing through the northeastern portion of the Borrow Pit (Figure 3-1).

The San Andreas Fault is one of the most significant geologic features in California in terms of displacement and recent activity. It is a tectonic plate boundary in the earth's crust between the North American and Pacific Plates. It is termed a "fault zone" because it consists of numerous faults, or "splays" or strands, instead of a single discreet linear fault. Several faults associated with the San Andreas Fault System have been mapped or postulated in the vicinity of the SAR recharge facilities, including the Bryn Mawr Fault, the Oak Glen Fault, and the Greenspot Fault, as shown on Figure 3-1. Based on groundwater data collected during this study, there is evidence of yet another splay which trends northwest and dissects alluvial deposits under the Borrow Pit (discussed in more detail below).

The MWH team reviewed well information for approximately 85 wells in the vicinity of the recharge facilities and noted (where available) lithologic and/or borehole geophysical information from these wells. These data, combined with surface geologic mapping and drilling data collected during this study, were used to develop two geologic cross-sections (A-A' and B-B') near the Borrow Pit. The locations of the cross-sections are shown on Figure 3-1, and the profiles are given in **Figures 3-2** and **3-3**. Based on these cross-sections and supporting data, there are several key observations that can be made about the aquifer underlying the SAR recharge facilities:

- The alluvial materials underlying the Borrow Pit, and much of the area surrounding it, consist of coarse, sandy gravel and/or gravelly sand. This material is expected to be highly permeable, and is consistent with an active alluvial fan environment.
- No significant, laterally-continuous strata of low permeability are present that would prevent the downward percolation of recharge water. Relatively fine deposits of clayey sand are reported in deeper portions of the Corps of Engineers Well No. 1 near the mouth of the SAR, and at the USGS Cone Camp wells southwest of the Borrow Pit, but they are not observed to be continuous from well to well, and are expected to be limited in areal extent, which is also consistent with an active alluvial fan environment.
- Faulting associated with the San Andreas Fault Zone dissects all but the most recent of alluvial deposits.





# **3.2 GROUNDWATER MIGRATION CONDITIONS**

Groundwater elevation data is available from existing wells and new monitoring wells constructed as part of the SAR Recharge Optimization field program. **Figure 3-4** shows a hydrograph (record of groundwater elevation measurements) for well SBVWCD #3 from 1980 to present, while **Figure 3-5** shows hydrographs from June 1998 to June 2008 of the USGS Cone Camp multiplezometers which are completed at depths of 65 to 75, 91 to 101, 124 to 144, 280 to 300, 500 to 520, and 770-790 feet below ground surface (bgs). These representative hydrographs suggest two key observations about groundwater conditions in the vicinity of the SAR recharge facilities:

- During periods of high rainfall and runoff (which also coincide with periods of high recharge natural and artificial recharge activity) such as in 1980, 1993 and 1998, groundwater levels approach the ground surface to the southwest and west of the recharge facilities. For example, groundwater elevations at Well SBVWCD #3 came within 40 feet of the ground surface and groundwater elevations at the Cone Camp well came within 25 feet of the ground surface during some of these periods. In areas of lower topographic location that these wells the groundwater elevation was probably at or near the surface. However, these events have been very temporary, and may occur at a different frequency depending on the operation of the Seven Oaks Dam.
- The USGS multiplezometers indicate a consistent pattern in which the groundwater elevation in deeper plezometers is lower than the groundwater elevation in shallower plezometers. This indicates a downward vertical gradient representative of recharge occurring at the surface and flowing to deeper portions of the aquifer where the ground water elevation potential is lower.

Groundwater elevation data from these and other wells in the vicinity of the recharge facilities have been utilized to construct regional contours of equal groundwater elevation for the "wet" (high runoff) and "dry" (low runoff) periods of 1998 and 2004, respectively (identified as green circles on Figure 3-4). These contours of equal groundwater elevation are shown in **Figures 3-6** and 3-7. The contours of equal groundwater elevation indicate that groundwater in the study area flows from the north and east to the west-southwest, similar to the flow path of the Santa Ana River.

#### **3.3 GROUNDWATER BARRIER**

The installation of monitoring wells during this study allows more detailed examination of groundwater elevations in the vicinity of the Borrow Pit. The geologic cross-sections shown in Figures 3-2 and 3-3 show groundwater elevation data from the recently-installed monitoring wells on July 8, 2008. These data confirm a generally westward groundwater flow direction, but also show something else that is not evident from the regional data: a rapid decline in groundwater elevation between monitoring wells SB-3 and MW-2, and new wells located northeast of these two wells. This relatively sharp decline in groundwater elevation over a short horizontal distance suggests the presence of a barrier to groundwater flow which trends to the northwest.









Given that no significant laterally-extensive clay layers were encountered during the field investigation that would impede groundwater flow, the groundwater barrier in the western portion of the Borrow Pit is interpreted as a splay of the San Andreas Fault, shown as a dashed green line on Figures 3-6 and 3-7. The orientation of the barrier is consistent with it being a previously unmapped splay of the San Andreas Fault Zone. Although the nature and extent of the fault are not completely known, groundwater movement across the fault in the unconsolidated sediments may be impeded due to the presence of clayey fault gouge, cementation along the fault zone, or displacement of permeable strata. A fault trace was not observed at the surface, suggesting concealment by recent surficial deposits. The depth at which the fault becomes an effective groundwater flow barrier is unknown, but it is clear that groundwater elevations are significantly higher to the northeast of the suspected fault trace.

**Figure 3-8** depicts estimated contours of equal depth to water which illustrate that groundwater is much shallower northeast of the groundwater barrier. This figure was developed by mathematically subtracting a grid of estimated groundwater elevations using new monitoring wells and existing wells on July 8, 2008 from a grid of topographic elevation contours from the field survey and USGS digital terrain data. From this data, it is clear that groundwater is much shallower in the eastern portion of the Borrow Pit upgradient of the groundwater barrier.

The shallow groundwater northeast of the suspected fault explains surfacing groundwater historically observed by the Conservation District staff in Pond 3, the north wall of the Borrow Pit, and the north central area of the Borrow Pit visible as green vegetation on aerial photographs (e.g. Figure 2-7), all of which are upgradient of the groundwater barrier.

# **3.4 AQUIFER CHARACTERISTICS**

As previously noted, groundwater in the vicinity of the Borrow Pit occurs primarily in the unconsolidated deposits which form an unconfined aquifer. Among many hydraulic parameters, hydraulic conductivity (ease of which groundwater can flow through these deposits) are most important because it determines the height and extent of how groundwater rises, or "mounds" during recharge. Provided below is a review of collected hydraulic conductivity data obtained either from studies in and surrounding the SAR recharge facilities.

Well completion reports provided by the Conservation District include pumping test data for Wells SAR 1 and SAR 2 (locations are shown on Figure 3-1). The hydraulic conductivity was estimated from continuous pumping data using the Cooper-Jacob aquifer test analysis method. The hydraulic conductivity is averaged for aquifers at depths from 215 to 490 feet bgs. SAR 1 is located at a distance of approximately 2,110 feet to the west-northwest of the Borrow Pit, whereas SAR 2 is approximately 8,900 feet to the west-southwest of the Borrow Pit. These data are summarized in **Table 3-1**.



Well Number	Depth (feet)	Perforated Interval(s)	Static Water Level (May 2005, feet bgs)	Hydraulic Conductivity (feet/day)
SAR 1	500	260 to 285, 305 to 330 415 to 440, 465 to 490	217	7 to 33
SAR 2	420	215 to 245, 340 to 410	213	52 to 160

Table 3-1 Summary of Aquifer Test Data

For reference, previous U.S. Geological Survey investigations (USGS, 2005) conducted approximately 11.5 miles to the west of the Borrow Pit along with published data for gravelly sand and sandy gravel are summarized in **Table 3-2**. With some exceptions (surface soil samples and three samples at MW-3 at 30, 38, and 102 feet bgs), field program data indicate that measured hydraulic conductivity values generally fall within the range from 0.113 to 19.7 feet/day. The laboratory results of hydraulic conductivity analysis collected during this study is generally lower than field measurements reported by Conservation District and USGS. Several of the samples were taken from areas where ponding water deposited silts, which are not representative of deep aquifer conditions. It is therefore concluded that a typical hydraulic conductivity value for the aquifer underlying the recharge facilities is on the order of 50 feet/day and may range from a few feet to several hundred feet/day.

 Table 3-2

 Hydraulic Conductivity for Selected Unconsolidated Sediments

References	Materials	Hydraulic Conductivity (feet/day)
USGS, 1975	Upper unconsolidated water-bearing sediments	120
USGS, 2005	Upper unconsolidated water-bearing sediments, Newmark area	40 to 100
Walton, 1988	Sand and gravel mixtures	13 to 668
Davis, 1966; page 164	Clayey sands to clean gravel	$10^{-3}$ to $10^{5}$

# Section 4 Analysis of Diversion and Conveyance Capacity

This section describes the analysis of the maximum capacity of the diversion and conveyance facilities, based on the flow testing and physical survey described in Section 2. Parameters (channel slope, roughness coefficient, etc.) gathered from as-builts and the flow test were used in the hydraulic analysis to calculate maximum flow capacities for discrete conveyance facilities. Field data from the flow testing was used to reconcile actual conditions with calculated conditions. Each structure's maximum capacity were calculated, and a description of their respective limitations is provided next in the following order:

- Intake Structure and Cuttle Weir
- Discharge Conduit (Closed Conduit/Tunnel and Sandbox)
- Canal Sections
- Rock Structure
- Greenspot Road Culvert
- Parshall Flume
- Diversion Structure

# 4.1 INTAKE STRUCTURE AND CUTTLE WEIR

#### Hydraulic Analysis

The hydraulic capacity of the Cuttle Weir and the Intake Structure are interrelated. If the height of the Cuttle Weir were increased, the resulting increased water surface elevation would allow greater flow rates through the Intake Structure. Similarly, if constrictions or limitations to flow carrying capacity at the Intake Structure (due to gate opening height, number of gates opened, or debris accumulation) were to occur, then the water level at the structure's entrance would need to rise to provide the head needed to overcome these limitations.

A hydraulic analysis was performed to confirm flow test conditions observed in the field and to extrapolate to greater flow volumes. The hydraulic analysis included determination of the flow characteristics of the following three components when passing various flows:

- Discharge conduit (or Closed Conduit/Tunnel and Sand Box)
- Collection channel (concrete channel on downstream side of Intake Structure)
- Intake structure (and influence from Cuttle Weir elevation)

The flow rates selected for analysis included: 500 cfs and the rate of flow corresponding to the existing facility's capacity.

The analysis began by determining the backwater curve along the conduit (downstream of the

gates at the Intake structure) and determined the theoretical flow depth where the collection channel feeds the tunnel (**Figure 4-1**). For a flow rate of 500 cfs, the estimated water depth at the upstream end of the tunnel is approximately 7.5 feet). Head loss at the tunnel entrance (k = 0.5 and velocity head of 2 feet, k=head loss coefficient) equals one foot, resulting in a depth of flow upstream of the tunnel entrance (at the downstream end of the collection channel) of roughly 8.5 feet (**Figure 4-2**).

Because flow into the Intake Structure discharges perpendicularly to the collection channel axis, this structure acts as a typical side-channel spillway collector. Therefore, water depth at each end must differ significantly to provide the momentum needed to move flow along the collector. Using a downstream depth of 8.5 feet or water surface elevation of 1947.5 feet where the flow enters the tunnel, the backwater calculation yields a corresponding upstream water profile of elevation 1949.6 feet (**Figure 4-3**). The water level profile represents the tailwater depths on the downstream side of the six gates.

The Intake Structure analysis determined the discharge capacity of the six gates, considering the upstream water level (formed by the pool behind the Cuttle Weir) and the downstream or tailwater levels (formed by flow in the collection channel). Each gate's hydraulic performance is governed by the gate opening, a discharge coefficient, and the net head between upstream and downstream water levels. When a gate is opened to a point above the upstream pool's water level it discharges as a weir, whereas a partially opened gate can be inundated and would function according to orifice flow and a differing set of hydraulic criteria. The MWH team developed a spreadsheet model that accounts for both flow regimes and includes definitions of: sill elevation, gate opening, shape of the discharge passageway (in this case it is a flat broad-crested weir as opposed to an ogee, a double curve in the shape of an elongated 'S', or stepped overflow), and the head- and tailwater levels.

# **Observed Capacity and Limitations**

Historical diversion operations, as discussed with facility operators indicated that a flow capacity of 150 cfs could be diverted at the Cuttle Weir by placing stop logs in the weir notch.

The hydraulic analysis of the Intake Structure determined that in order to achieve a total flow of 500 cfs through all six gates, a headwater elevation of approximately 1949.6 feet would be required. This headwater level is approximately 5.1 feet below the top of the lowest section of the Cuttle Weir (Figure 4-1). Gates 1, 2, 4, 5 and 6 were found to have discharge capacities of about 85.1 cfs. Gate 3 discharge would reduce to about 73.9 cfs since it can only be opened 2.25 feet (Figure 4-3).

From field observations during the flow test and the above analysis, the current configuration of the Cuttle Weir and Intake Structure is not adequate to convey the requisite 500 cfs. To analyze the theoretical capacity of the currently configured Intake Structure and Cuttle Weir, a water level elevation of 1954.7 ft, which corresponds to the top of the lowest section of the Cuttle Weir, was used (Figure 4-1). With gates opened to at least the headwater level or each gate's maximum (maximum opening of each gate is limited and differs), the corresponding theoretical total discharge through the six gates is approximately 362 cfs (**Figure 4-4**).









Review of the flow test field notes and time stamped photos reveal that water levels reached the top of the Cuttle Weir at a flow rate of around 300 cfs with the plywood in place at the weir notch. The difference in the theoretical capacity and observed capacity appears to be due to the accumulation of debris and boulders at the Intake Structure's bar screens. Therefore, in order to meet the theoretical capacity, a means of eliminating material accumulation on the gates' upstream side would be required. Given the potential for material accumulation and its impact on the flow characteristics, the MWH team conclude that the practical flow capacity for diversion of flow into the Intake Structure, given the current configuration, is 250 cfs.

# 4.2 DISCHARGE CONDUIT

# Hydraulic Analysis

Due to the interrelationship between the Cuttle Weir and the Discharge Conduit (Closed Conduit/Tunnel and Sand Box) the hydraulic analysis for the interrelated components is presented above.

# Capacity and Limitations

The capacity of the Closed Conduit/Tunnel was determined by calculating the maximum free water surface in the structure. This free water surface would correspond with open channel flow. Additional capacity could be realized by allowing flow in the Closed Conduit/Tunnel to be under pressure. However, if this were the case, the transition from open channel to closed conduit flow would be unstable, causing considerable variations in water surface elevations within and upstream of the structure. The calculated capacity of the Closed Conduit/Tunnel is therefore 550 cfs for open channel flow (**Figure 4-5**).

The Sand Box capacity is currently limited by the maximum gate opening at the structure's downstream end. The record drawings did not indicate the maximum height each gate can be opened. Thread measurement taken on each gate stem revealed that the gates have a maximum opening of 4 feet. The hydraulic model showed (**Figure 4-6**) that at a flow rate of 362 cfs the corresponding water elevation upstream of the Sand Box gate structure would be 4 feet. For a flow rate of 500 cfs, the water depth at the Sand Box gate structure would be about 4.8 feet (Figure 4.2). Therefore, the gates would need to be modified to increase this facility's carrying capacity to 500 cfs. Without the gate limitations, the Sand Box has a capacity of 550 cfs which is greater than the desired capacity of 500 cfs. With modifications to the 50-foot long overflow section, in order to eliminate this overflow, the capacity could be increased to 825 cfs.





# 4.3 CANAL SECTIONS

#### Hydraulic Analysis

Because the typical canal section shown on the record drawing Plan for Conduit Santa Ana Weir, (May 1930) is not representative of the existing canal, a survey was conducted to determine the typical parameters of the various canal sections (Section 2). Representative sections of the canal were known, so Manning's equation was used to determine the capacity.

Manning's equation: (Q=1.486/n \* AR^2/3 \* S^1/2) Q=Flow, A=Area, R=Hydraulic Radius, S=Slope, and n=roughness coefficient

One parameter required to apply Manning's equation that cannot be physically measured is Manning's roughness coefficient, n. There are published guidelines as to the range of typical nvalues based on the canal physical properties. Due to the variation of the canal cross-section, the information obtained during the flow tests is insufficient to assess the n-value. During future flow conditions it is advisable that velocity measurements be made in the channels. With the velocity measurements and the documented canal cross-sections, more representative n-values could be determined.

Using published data an n-value of 0.050 (Daugherty and Franzini, 1979) was estimated. The estimated n-value and the typical canal sections and slopes obtained from the field survey were then used to estimate the capacity of the existing canal using Bentley FlowMaster Version 8 software (Flowmaster).

#### **Capacity and Limitations**

The calculations showed that most of the canal sections had banks high enough to convey the desired maximum flow rate of 500 cfs; however, the high velocities reached in many sections of the canal caused significant potential for channel erosion. The calculated velocities for 300 cfs were in excess of 6.19 feet/second. At a rate of 500 cfs the calculated velocities exceeded 7.73 feet/second with some sections near 10 feet/second. The maximum recommended velocity for canals of this type (cobbles and shingles) is 6.5 feet/second (King, 1939), whereby velocities exceeding this amount will likely cause canal erosion. Correspondingly, observations made during the flow test noted that rocks and boulders could be heard moving in the canal starting at a flow rate of approximately 300 cfs.

Due to its slope, the existing canal has a practical conveyance limitation of 300 cfs. Future improvements to increase the carrying capacity of the facilities could include: (1) adding drop structures and decreasing the channel slope, (2) improving the channels, and (3) increasing the channel width. Detailed recommendations for proposed improvements will be provided in the final report.

### 4.4 ROCK STRUCTURE

#### Hydraulic Analysis

Data obtained from a flow test and the dimensions of the Rock Structure were used to determine its capacity. At a flow rate of 241 cfs, the water depth in the structure is approximately 1.6 feet. The measured dimensions of the structure are 14 feet wide and 6 feet high. Using a slope of 0.032, an n-value of 0.029 was determined. This value corresponded with the grouted rip rap value given in FlowMaster's library of roughness coefficients.

### **Capacity and Limitations**

Utilizing the above roughness coefficient, the capacity of the structure was determined for an assumed maximum water depth of 5 feet, which is equal to the height of the gate openings. The structure's capacity was estimated to be 1,268 cfs.

# 4.5 GREENSPOT ROAD CULVERT

### Hydraulic Analysis

A clear choke point, or point of congestion or obstruction, in the conveyance system is the culvert under Greenspot Road. During the March 27, 2008 flow test it was observed that for a flow rate of approximately 350 cfs there was only 4 to 6 inches of freeboard available. To further analyze the culvert, its dimensions were measured. The culvert was measured to be 10 feet wide and 4 feet and 2 inches high. A roughness coefficient of 0.015 was selected for this concrete structure. The FlowMaster program was then used to calculate the culvert's slope by using the normal depth of 2.1 feet that was measured at a flow rate of 141 cfs at 9:05 a.m., on March 27, 2008. A slope of 0.00273 was then calculated. The FlowMaster program was run again with the calculated slope and roughness coefficient of 0.015 to solve for a normal depth at 330 cfs (flow that was measured at 12:45 pm). The "normal depth" is the depth of flow in the channel or culvert when the slope of the water surface and channel bottom is the same and the water depth remains constant. Normal depth occurs when the gravitational force of the water is equal to the friction drag along the culvert and there is no acceleration of flow. The calculated normal depth was 3.8 feet and the observed normal depth was recorded as 3.6 feet. The depth measured during the flow test and the calculated depth generally agree. Therefore the roughness coefficient and calculated slope appear to be reasonable.

#### **Capacity and Limitations**

Once the slope and roughness coefficient were checked and confirmed, the normal depth was solved using a desired flow rate of 500 cfs. The normal depth at 500 cfs was calculated to be 5.18 feet, which is higher than the available 4 feet 2 inches. Therefore, it is not possible to convey 500 cfs through this culvert without the water hitting the top slab. The maximum capacity of the culvert was then solved using a normal depth of 4.16 feet, and the maximum capacity at this depth is 372 cfs. This result is confirmed by observations from the flow test. The culvert was near capacity at a flow rate of 367 cfs, which was measured just downstream of the culvert at the Parshall Flume.

### 4.6 PARSHALL FLUME

#### Hydraulic Analysis

The original design drawings for the Parshall Flume (*Venturi Flume, March 1933*) contain a table based on the equation  $Q=4WH_a^{1.522W^{0.026}}$  (Q=Flow, W=width, H=Height) that provides a maximum flow through the Parshall Flume of 1,405 cfs. This value assumes that the flow would be at the top of the structure. If a freeboard of 6-inches is used, the capacity would be 1,214 cfs.

#### **Capacity and Limitations**

The maximum capacity of the Parshall flume is 1405 cfs. A capacity of 1,214 cfs is provided when a free board of six inches is used.

#### 4.7 **DIVERSION STRUCTURE**

#### Hydraulic Analysis

During the March 27, 2008 flow test the flow through the Diversion Structure north gates located northeast of the Borrow Pit to the existing westerly spreading ponds was limited to roughly 300 cfs. This upper limit was based on historical operator knowledge of the downstream conveyance system limitations. When the flow rate reached 300 cfs, the southerly gate was opened and flow was discharged into the Borrow Pit. There was 2 feet of freeboard on the channel walls prior to the southerly gate being opened.

Preliminary analyses indicate that 500 cfs could be directed through the northerly gates to the existing westerly spreading grounds. Ultimate configuration of the spreading ponds will determine how the water will be directed through the diversion structure to the spreading facilities.

#### **Capacity and Limitations**

The channel has the requisite capacity for conveyance of flows to 500 cfs; however, modifications to the gate structure may be required to reduce hydraulic losses along the center pier. Due to the complexity of this facility, additional analysis will be required once the ultimate configuration of the spreading facilities and the flow distribution requirements are known. The distribution of flow will be dependent on the ultimate spreading ground and conveyance system layouts to be provided in the final report.

#### 4.8 SUMMARY

**Table 4-1** shows the maximum conveyance capacity for each of the Conversation District's conveyance facilities.

Description	Qmax		
Intake Structure	150 cfs <sup>(1)</sup> 250 cfs <sup>(2)</sup> 362 cfs <sup>(3)</sup>		
Cuttle Weir	362 cfs <sup>(4)</sup>		
Tunnel	550 cfs		
Sandbox	825 cfs <sup>(5)</sup>		
Earthen Canal	300 cfs		
Rock Structure	1268 cfs <sup>(5)</sup>		
Greenspot Road Culvert	372 cfs		
Parshall Flume	1214 cfs <sup>(6)</sup>		
Diversion Structure North Gate	500 cfs		
Diversion Structure South Gate	253 cfs		

Table 4-1Maximum Conveyance of Groundwater Recharge Facilities

(1) Existing capacity using stop logs instead of plywood at weir notch.

(2) Practical capacity given the potential for material accumulation and its impact on the flow characteristics.

(3) Theoretical capacity through the six gates to maximum gate height.

(4) Theoretical capacity without the use of sandbags.

(5) Requires modifying the 50-ft long overflow section, otherwise practical capacity is 550 cfs with minor gate modifications.

(6) Capacity corresponding to a water depth of 5.5 ft, thus allowing 5 inches of free board.

(7) Capacity dependent upon maintaining existing slope of canal downstream of diversion structure.

From the analyses described above the MWH team concludes that the Cuttle Weir will need to be raised to increase the capacity to 500 cfs. Although it appears that modifications to the Cuttle Weir could be made inexpensively, additional analyses are required to determine whether it can be raised without causing environmental impacts or structural instability to the existing weir dam.

The existing gates at the Intake Structure have the capacity to move 500 cfs and greater flows. The Cuttle Weir height and Intake Structure are interdependent. So, although the Intake Structure has substantial excess capacity, the only way to realize this capacity would be by raising the Cuttle Weir. Managing and controlling debris at the gate structure or downstream will be addressed in the final report.

The Intake Conduits, Sandbox, Rock Structure and Parshall Flume all have sufficient capacity to convey flows of 500 cfs and greater. Modifications of the gates and possibly improvements to the overflow section of the Sandbox are the only recommended future improvements for these facilities.

The existing earthen channels generally experience velocities greater than recommended values at flows of 300 cfs and greater. Future improvements could include the installation of drop structures and lessening of the existing slopes or widening of the existing channels. Detailed recommendations for improvements will be addressed in the final report.

The Diversion Structure appears to have capacity to convey flows of 500 cfs, however, additional analysis will be done once the configuration and flow regime of the ultimate spreading facilities is known.

# Section 5 Analysis of Percolation Capacity

The field work was designed to identify the relative significance of factors which may limit the percolation capacity of existing recharge ponds at the SAR recharge facility. Potential limiting factors identified the planning phase of the study included:

- Presence of fine-grained material that "clog" the surface of the Borrow Pit
- Compaction of shallow soils in the Borrow Pit due to heavy equipment use during construction of the pit
- Presence of horizontal or vertical barriers to groundwater flow, such as clay layers or faults
- Shallow groundwater which mounds or rises to the surface

Based on the field work described in Section 2, more is known about the significance of these potential limiting factors, as described in the following sections.

# 5.1 SURFACE CLOGGING BY FINE PARTICLES

Shortly after the completion of the Seven Oaks Dam, it was reported that silt-laden water was delivered to the Borrow Pit. Fine particles of silt and clay may have settled out of this water and formed a thin layer of relatively impermeable soil which significantly reduces infiltration rates. Field data collected during this study suggest that this was indeed a factor in reducing infiltration rates in the Borrow Pit. Observations that support this conclusion include:

- Visual observations during physical surveys suggest that at locations where water ponds, a fine silt/clay layer is observed which is much finer than that of adjacent native soils is present.
- An anomalously high percentage of silt or clay was noted in grain size analyses from sample locations where water historically ponds on the western edge of the Borrow Pit. Whereas a typical fine particle percentage for most areas of the Borrow Pit is less than 6 percent, areas where water ponds have fine particle percentage as high as 50 percent. Comparison of hydraulic conductivity and fine particle percentage data collected during this study indicate that fine particle percentage is a very significant factor limiting hydraulic conductivity.
- Tests at Pond 1 in the Borrow Pit where the shallowest soils were removed resulted in a significantly higher percolation rate.

# 5.2 COMPACTION OF SOILS

Soil compaction can occur when the weight of machinery or other forces compress the soil causing it to lose porosity, leaving little space for air and water. One hypothesis is that during

construction of the Seven Oaks Dam, the very heavy earthmoving equipment used for excavation in the Borrow Pit may have compacted the shallow soils, thereby reducing their infiltration capacity. The degree of compaction caused by heavy equipment is dependent on the weight of the equipment and the prevailing soil moisture content. Contact pressure of the equipment is determined by the overall weight of the vehicle and its footprint. The greater the contact pressure and/or the more frequently the vehicle passes over a particular area in the field, the greater and deeper the resulting compaction. The soil moisture content at the time the equipment is driven over the soils is also a factor, whereby the greatest amount of compaction occurs when the soil is wet. However, details regarding the type of equipment and frequency of use of equipment used during construction of the Seven Oaks Dam is generally not available, so that estimates of the amount of compaction that occurred cannot be reliably made.

Although no quantitative data on soil density was collected during the field effort, during shallow soil sampling and trenching, no visual evidence of severe compaction such as loss of granular soil structure or crushed or hardened soils was observed. If compaction did occur, it is not expected to be significant more than a few feet below the surface, particularly because of the coarse nature of native soils (which are less compressible). Based on these factors, soil compaction by heavy equipment is not considered to be a significant factor in reducing infiltration rates in the Borrow Pit. The potential effects of soil compaction in the Borrow pit may be somewhat of a moot point if shallow fine-grained soils need to be removed in any case.

### 5.3 PRESENCE OF VERTICAL OR HORIZONTAL GROUNDWATER MOVEMENT

Two types of barriers to groundwater flow may be present at the SAR recharge facilities; horizontal barriers that retard downward flow of groundwater, or vertical barriers that retard lateral or horizontal flow of groundwater. The evidence for the presence or absence of each of these types of barriers is discussed in the following sections.

# **Horizontal Barriers**

As noted in Section 2, there is no evidence of a laterally-continuous strata of low hydraulic conductivity which would inhibit infiltration of recharge water regionally. This is consistent with the active alluvial fan depositional environment of the unconsolidated deposits underlying the recharge facilities. The depositional environment also suggests that discontinuous layers of silt and clay are present locally, and that the horizontal hydraulic conductivity is probably one or more orders of magnitude higher than the vertical hydraulic conductivity. The discontinuous layers of fine material could be cause by soil zones developed on older fan deposits or overbank deposits along active alluvial channels.

Thus, it is concluded that the presence of a horizontal barrier to groundwater flow is not a significant limitation to recharge at the SAR facilities.

# Vertical Barriers

As noted in Section 2, groundwater elevation data collected during this study provides good evidence for the presence of a vertical barrier to groundwater flow which dissects the Borrow Pit

and trends in a northwesterly direction parallel to the San Andreas Fault. This vertical barrier is interpreted as being a previously unmapped splay of the San Andreas Fault. During the Spring of 2008, this groundwater barrier impeded the westward flow of groundwater enough to cause groundwater to rise to the surface in Pond 3 and the central section of the Borrow Pit immediately northeast of the fault. Based on observations of the Conservation District staff and the presence of semi permanent vegetation in these areas, this is a relatively common occurrence.

In the Spring of 2008, infiltration occurring in unlined canals and the Santa Ana River east of the barrier during recharge operations provides enough groundwater to "fill" the aquifer east of the fault. This is evidenced by the fact that groundwater surfaces even before water was added to Ponds 1, 2, or 3 in the Borrow Pit. Because the monitoring wells east of the barrier are new, there is no data to evaluate how often this has occurred in the past, but it is believed to be relatively common. The net effect of this phenomenon is that the entire eastern portion of the Borrow Pit (Phase 1 Area) has a very low percolation rate and is of minimal value for additional percolation ponds, although other facilities such as settling basin could be located in the area.

# 5.4 GROUNDWATER MOUNDING LIMITATIONS

During artificial recharge operations where a high volume of water is recharged to a finite region, the groundwater rises and forms a mound beneath the region or recharge area. If ambient groundwater levels are relatively high, the artificial recharge may cause groundwater to rise to the ground surface, thus reducing infiltration capacity. For this reason, it is generally advisable to avoid this situation in order to optimize recharge operations. During the testing of Pond 1, groundwater mounding was observed in MW-1, and, although the groundwater did not rise to the surface underlying the pond, recharge in Pond 1 caused rising groundwater in Pond 3 and the central portion of the Borrow Pit.

Southwest of the groundwater barrier however, the depth to the ambient groundwater elevation is much greater, which is a more desirable condition in order to prevent the groundwater mound from reaching the surface (Figure 3-8). Long-term hydrographs of well SBVWCD 3 and the Cone Camp wells indicate a typical depth to water of approximately 200 feet in the area southwest of the barrier, and during the summer of 2008, depth-to-groundwater ranged from 100 to 160 feet.

The method of Hantush (1967) has been used to estimate the rate of groundwater mounding at the highest point on the mound. The input variables in the Hantush model include the prerecharge thickness of the aquifer, specific yield, aquifer hydraulic conductivity, length and width of the recharge ponds, and average recharge rate. The length and width of the recharge facilities was estimated using the total area available for recharge west of the fault in the Borrow Pit. The average rate was derived by dividing 500 cubic feet per second (desired maximum recharge rate) by the total area (resulting in an average rate over the entire area). This average rate over the total area is less than what would occur in individual ponds, but simulates the infiltration of 500 cubic feet per second (cfs) over the total area. The following inputs to the model where used, based on aquifer conditions described in Section 3:

Initial thickness of aquifer	500 feet
Specific Yield	15 percent
Length of Recharge Pond	6,494 feet
Width of Recharge Pond	3603 feet
Average Recharge Rage	1.85 feet per day

The IRWMP "Scenario A" calls for a total of 80,000 acre-ft in one year. From a groundwater mounding perspective, the worst-case scenario would be that this volume of water was recharged at a continuous maximum rate of 500 cfs, which equates to 81 days. Therefore, the Hantush model was run for a period of 81 days at 500 cfs. The largest uncertainty using this model is the average hydraulic conductivity of the aquifer. As noted in Section 3, a typical hydraulic conductivity value for the aquifer underlying the recharge facilities is estimated at 50 feet/day, but may range from a few feet to several hundred feet/day. **Figure 5-1** depicts the calculated maximum rise in the water table based on the Hantush method using a range of assumed hydraulic conductivities.

Figure 5-1 Change in Groundwater Elevation Assuming Various K Values



The Hantush model indicates that using the best estimate of hydraulic conductivity of 50 feet per day, the infiltration of 500 cfs will not cause groundwater to reach the surface for approximately 30 days of continuous recharge (assuming a starting groundwater depth of 200 feet). If the average hydraulic conductivity of the aquifer is higher than 50 feet per day, this time period would be longer, and if it is as high as 200 feet per day, it would not reach the surface at all. These calculations probably underestimate potential mounding because the Hantush method assumes idealized aquifer conditions which do not include the boundary conditions present at the site. However, it does suggest that if the starting groundwater level is below approximately 200 feet, it is possible to recharge 80,000 acre-feet in one year. However, continuous percolation at

the maximum rate of 500 cfs may cause groundwater mounding, which limits percolation.

A more realistic number of days for the recharge of 80,000 acre-feet is over a longer period of approximately nine months. This would mean that the average percolation rate would drop to 147 cfs, or 0.54 feet per day. Change in groundwater elevation at this percolation rate is also depicted in Figure 5-3. The Hantush method suggests that at this lower average percolation rate, groundwater will not reach the surface assuming the average hydraulic conductivity of 50 feet/day or higher.

It is very important to note, however, that in very high runoff periods, the entire Phase 1, Phase 2, and Phase 3 area west of the barrier may be saturated because of the combined effect of natural and artificial recharge along the Santa Ana River. Hydrographs of well SBVWCD 3 and the Cone Camp wells indicate that in 1980, 1993, 1998, and 2005, the groundwater came to within approximately 50 of the surface for short periods of time. During these periods, recharge may be limited by groundwater mounding even west of the barrier which dissects the Borrow Pit. Under these saturated conditions, it is certain that a percolation rate of 500 cfs will cause groundwater mounding that will limit percolation rates. This is consistent with regional groundwater modeling studies, which predict saturated conditions for very short durations during wet years (SAIC, 2007).

### 5.5 ESTIMATION OF PERCOLATION CAPACITY OF EXISTING FACILITIES

In 1975, the USGS reported that an average long-term infiltration rate in the range of 0.7 feet (with silt clogging) to as high as 10 feet/day (for well maintained surface) could be expected for the SAR recharge facilities. In addition, the USGS reported that on average, typical long-term infiltration rates are on the order of 3 feet/day. Relatively short-term testing conducted during this study indicated rates of approximately 4 feet/day for the area west of the groundwater barrier. Therefore, a conservative range of typical infiltration rates for well-maintained existing or new ponds west of the Borrow Pit barrier would be 3 to 4 feet/day. The maximum infiltration capacity of the existing pond facilities outside the Borrow Pit can be estimated based on various infiltration rates as shown in **Table 5-1**.

Testing of the Pond 17 west of the Borrow Pit resulted in a calculated infiltration rate of 4 feet per day. Combining this calculated rate with the total area of Ponds 9-17 and Dike D estimated by the Districts results in a calculated percolation capacity of 116 cfs. However, it was noted that during testing of Ponds 9-17 and Dike D in April of 2008 that 145 cfs could be added to these ponds without spilling water from Dike D. This apparent discrepancy might be explained in several ways:

- 1. Significant percolation occurs in unlined conveyance facilities not accounted for in the pond area calculations.
- 2. The average percolation rate of Ponds 9-17 and Dike D is higher than 4 feet/day measured in Pond 17, and could be as high as 5 feet/day.
- 3. During testing in April of 2008, some water was still being added to storage in the soil zone and/or the wetted area was increasing, suggesting a percolation rate higher than would occur during long-term operation.

Existing Percolation Pond	Wetted Area (feet <sup>2</sup> )	Percolation Capacity in cfs (Rate of 3 feet/day)	Percolation Capacity in cfs (Rate of 4 feet/day)	Percolation Capacity in cfs (Rate of 5 feet/day)	Notes
1	94,633	3.3	4.4	5.5	May not be usable due to high groundwater
2	99,616	3.5	4.6	5.8	May not be usable due to high groundwater
3	110,152	3.8	5.1	6.4	May not be usable due to high groundwater
9	104,264	3.6	4.8	6.0	Outside of Phase 1, 2, and 3 areas
10	258,916	9.0	12.0	15.0	
10-S	42,343	1.5	2.0	2.5	Outside of Phase 1, 2, and 3 areas
11	282,525	9.8	13.1	16.3	
11-S	15,322	0.5	0.7	0.9	Outside of Phase 1, 2, and 3 areas
12	97,038	3.4	4.5	5.6	
13	298,083	10.4	13.8	17.3	
14	186,068	6.5	8.6	10.8	
15	203,764	7.1	9.4	11.8	
16	65,692	2.3	3.0	3.8	
17	213,808	7.4	9.9	12.4	
Dike D	735,002	25.5	34.0	42.5	Outside of Phase 1, 2, and 3 areas
Total	2,807,227	97.5	130.0	162.5	
Pond 9-17, Dike D Only Total	2,502,826	86.9	115.9	144.8	

Table 5-1Maximum Infiltration Capacities

Notes: Total acres = 64.4

Total acres of Pond 9-17 and Dike D only = 57.5

Given that the testing occurred over a period of over 2 weeks, factors 1 and 2 appear most significant, and it seems the a combination of the two is most probable. Therefore, it is estimated that the total long-term percolation capacity of the existing ponds and unlined conveyance system outside the Borrow Pit (Figure 5-2) is approximately 145 cfs (Figure 5-3), and an additional 50 cfs of long-term percolation capacity is estimated for the west end of the Borrow Pit, being the total percolation capacity to 195 cfs, except during very high runoff periods when the regional area experiences shallow groundwater conditions for temporary periods.




# 6.1 CAPACITY OF EXISTING DIVERSION, CONVEYANCE, AND RECHARGE FACILITIES

Based on previously-available data and the results of field work conducted as part of this study, the estimated capacity of the SAR recharge facilities can be compared to the short-term instantaneous flow rate goal identified in the IRWMP of 500 cubic feet per second (cfs). The following is concluded about physical conditions and capacity of both the diversion and conveyance facilities, and the percolation ponds at the SAR recharge facilities:

# **Diversion and Conveyance**

- The current intake capacity of the Intake Structure without modification is approximately 150 cfs.
- Downstream of the Intake Structure and Cuttle Wier, earthen canals limit the capacity of the conveyance facilities to approximately 300 cfs.

# **Percolation Ponds**

- The is no evidence of a continuous strata of low hydraulic conductivity that would limit downward percolation of recharge water.
- Percolation in portions of the Borrow Pit is currently limited by the presence of a thin layer of silt and clay, probably deposited when silty water was added to the Borrow Pit.
- The presence of a vertical barrier to groundwater flow (probably a fault), causes groundwater to rise in the Borrow Pit and limits the recharge capacity of the western 2/3<sup>rd</sup> of the Borrow Pit.
- The recharge capacity of the existing percolation ponds at the SAR recharge facility west of the fault is approximately 145 cfs. The recharge capacity of the borrow pit is approximately 50 cfs for a total current recharge capacity of 195 cfs.

# Section 7 References

- Daugherty and Franzini, 1979. Water Resources Engineering, 3rd ed., McGraw-Hill, New York, New York.
- Davis, S. N., 1966. Hydrogeology, John Willey & Sons, Inc.
- Hantush, M.S., 1967. Growth and Decay of Groundwater-Mounds in Response to Uniform Percolation, Water Resources Research, vol. 3, no. 1, pp. 227-234.
- King, Horace Williams, 1993, Handbook of Hydraulics, McGraw-Hill, New York, New York.
- Muckel, G.B. (Editor), Understanding Soil Risks and Hazards, Using Soil Survey to Identify Areas with Risks and Hazards to Human Life and Property. United States Department of Agriculture, Natural Resources Conservation Service, National Soil Survey Center, Lincoln, Nebraska. Online Publication.
- MWH, 2008, Santa Ana River Groundwater Recharge Optimization Study, Technical Memorandum 1 Documentation of Existing Information and Field Work Plan June 16.
- SBVMWD and WMWD, 2007, Santa Ana River Water Right Applications for Supplemental Water Supply: Community Report. January 2007.
- SBVWCD, 1994. Operational Management Manual of the San Bernardino Valley Water Conservation District. March 1994.
- SBVWCD, 2007. Miscellaneous documents related to hydrologic investigations related Wells SAR1 and SAR2, undated.
- Science Applications International Corporation (SAIC), 2007. Technical Memorandum: Seven Oaks Dam Economic Analyses, December 2007.
- USGS, 1975. Artificial Recharge in the Upper Santa Ana River Area, San Bernardino County, California. USGS Water-Resources Investigations 15-75. 27 p.
- USGS, 2003a. Geologic Map of the Redlands 7.5' Quadrangle, San Bernardino and Riverside Counties, California.
- USGS, 2003b. Geologic Map of the Yucaipa 7.5' Quadrangle, San Bernardino and Riverside Counties, California.

- USGS, 2005. Hydrology, Description of Computer Models, and Evaluation of Selected Water-Management Alternatives in the San Bernardino Area, California. USGS Open-File Report 2005-1278.178 p.
- Walton, WC, 1988. Practical Aspects of Groundwater Modeling. National Water Well Association, Worthington, Ohio. 588 pps.

Appendix A Water Level Plots

# Attachment 1.

# Santa Ana River Groundwater Recharge Optimization

# Scope of Work

The scope of work consisting of the following main Tasks:

- Task 1: Evaluate Existing Groundwater Recharge Spreading Facilities and Operations
- Task 2: Establish Spreading Objectives
- Task 3: Perform Feasibility Analysis and Prepare Conceptual Design to Meet Spreading Objectives
- Task 4: Draft and Final Report
- Task 5: Project Management

The work that the MWH will complete for each of these tasks is described in detail in the following sections.

# Task 1: Evaluate Existing Groundwater Recharge Spreading Facilities and Operations

The Task 1 objective is to evaluate existing spreading facilities (diversion works, conveyance and basins) along with the current maintenance and operations practices to determine their maximum estimated capacity. Work under this task will consist of performing the field tests and quantitative analyses necessary to achieve this objective.

MWH team will evaluate the recharge facility capacity by addressing each component of the facility individually. Four steps will be generally included for evaluation of each component of the facility:

- Data compilation and review
- Field data collection
- Field flow verification
- Capacity estimation

The individual tasks to be performed using this general approach will be as follows:

# Task 1.1: Data Compilation and Review

This task will consist of compilation of information from previous investigations and construction activities in the vicinity of the SAR recharge facility. MWH will review data from the USGS, U.S. Army Corps of Engineers (USACE), and additional studies, reports, and data available from SBVWCD and SBVMWD. MWH will establish a secure project FTP site which will serve as a data repository for all information and

data used by the team. MWH team will compile these data into a Technical Memorandum (TM) for future use and reference.

This task will also involve meetings with SBVWCD and SBVMWD staff to obtain records and verbal information on past recharge operations. MWH anticipates that much of the operations information obtained will not be documented, but will include important observations by the operators regarding system response to historic recharge practices. The MWH team will document this anecdotal information for use in future phases of the work. The MWH team will document and present the data compiled for this project and approach to field data collection before field work begins.

A description of data compilation and review activities anticipated for each facility component is provided below.

<u>*Task 1.1.1: Diversion Works and Conveyance*</u> – The MWH team will compile and review the following information:

- Available record drawings and related information for all diversion and conveyance facilities
- Existing elevations for key points of diversion and along the conveyance route
- · Operations information obtained for diversion records and interviews with operations staff

Task 1.1.2: Recharge Basins – The MWH team will compile and review the following information:

- Available record drawings and related information for the recharge basins
- · Operations information obtained for spreading records and interviews with available operations staff
- · Current and past groundwater elevation records for wells in the project area
- Lithologic logs for wells in the project area
- Shallow borings performed by USACE prior to construction of the Borrow Pit (if available)
- Groundwater modeling assumptions by Geoscience and USGS
- Existing elevation records for key locations at the current and potential future locations of recharge basins in the project area
- Previous hydrogeologic investigations in the area to obtain information on subsurface conditions that govern recharge rates (i.e. soil grain size distribution, total alluvial thickness, vertical and horizontal hydraulic conductivity of soil and aquifer materials, aquifer transmissivity, location and thickness of aquitards)

Task 1.2: Completion of TM-1 – Documentation of Existing Information and Field Work Plan

This Technical Memorandum will consist of:

- Summary of data compiled and reviewed including a catalog of material referenced
- Work plan for the acquisition of subsequent field data documenting:

- o the rationale, location and methodology for field data collection
- o the schedule for field data collection
- o coordination needed with SBVWCD and SBVMWD staff during completion of the field work
- Health and Safety Plan to be followed by field personnel

## Task 1.3: Field Data Collection

A description of proposed field data collection activities for each facility component is provided below.

Task 1.3.1: Diversion Works and Conveyance – The MWH team will obtain the following field information:

- Inspect, photograph and document physical condition of diversion and conveyance facilities
- Physical measurement of diversion works and conveyance facilities (i.e. length, width, height, channel condition, debris and obstructions)
- Survey selected locations to verify latitude, longitude and elevation (to nearest 0.1 foot for horizontal and 0.01 of a foot for vertical control). For cost estimating purposes, it is assumed that 20 points will be required.
- Profile of canal facilities with cross-sections at each major change in channel geometry

The MWH Team will conduct field flow testing of the diversion works when water is available. Focus will be on the potential flow limitation at the Cuttle Weir, which is now choked with sediment. During the long-term percolation testing described below, observations will be made of the diversion and conveyance works for a variety of flow rates.

Task 1.3.2 - Recharge Basins - The MWH team will perform the following field activities:

# **Basin Physical Survey**

- Inspect, photograph and document the physical condition of each recharge basin noting type and thickness of clogging layer (if present), vegetation, algae, evidence of side wall erosion, and other features pertinent to recharge capacity.
- Physical measurement of recharge basins (i.e. length, width, height) or estimation from existing maps.
- Measurement of depth to water in project and private wells identified near the project facilities (for budgeting purposes, we have assumed five wells).
- Survey of key elevations at recharge basins and existing monitoring well locations to verify latitude, longitude and elevation (to nearest 0.1 foot for horizontal and 0.01 of a foot for vertical control). For cost estimating purposes, it is assumed that 20 points will be required.

# Shallow Soil Sampling and Quantitative Soil Characterization

Soil samples will be collected for quantitative laboratory analysis of soil grain size gradation. These samples will be collected in the following manner:

## Trenching

Up to 10 trenches will be excavated in the existing percolation basins to a maximum depth of 5 feet at locations described in the field work plan (TM-1). The side walls of the trenches will be logged by a professional geologist (PG). Types of soil information will consist of:

- Depth of various soil horizons
- Soil type
- Visual classification of gradation
- Color
- Apparent moisture content

Sample descriptions will be in accordance with the American Society for Testing and Materials (ASTM) D2487-06 *Standard Practice for Classification of Soils for Engineering Purposes* (Unified Soil Classification System).

#### Areal Shallow Soil Sampling

Shallow soil samples will be collected by the on-site geologist at selected locations identified in TM-1 and described using ASTM D2487-06.

Up to 65 soil samples from trench and shallow soil sample locations and borings will be analyzed for gradation analysis in accordance with ASTM D-422 and ASTM D-4464 by a qualified laboratory. Ten (10) of the 65 soil samples collected will also be tested for saturated hydraulic conductivity using ASTM-5084.

Gradation analysis will be collected and analyzed at the test pit locations described below. Two (2) samples will be conducted at each test pit.

#### Sonic Drilling

Five sonic borings will be conducted along the long axis of the borrow pit. This type of drilling allows for representative core sampling in rocky terrain. Five borings will be completed to a maximum of 50 feet below ground surface for deeper correlation of sediments below the borrow pit.

#### **Construction of Monitoring Wells**

MWH will install one (1) monitoring well (MW) in the immediate vicinity of existing and potential basins to the water table (assume 300 feet below ground surface). The MW will be located in areas that will not affect recharge operations such as berms and sides of existing roads. Data will provide information on the stratigraphy of the basins, and will be used later for documentation of the water table elevation underlying the sites during recharge operations. The MWs will be installed using air percussion or rotary/foam methods to a maximum depth of 300 feet.

The cuttings from the borings excavated for installation of monitoring wells will be logged by a Professional Geologist using ASTM D5434-97 (2003) *Standard Guide for Field Logging of Subsurface Explorations of Soil and Rock*. In addition, the boreholes will be logged using geophysical methods of spontaneous potential, long and short normal resistivity, and point resistivity. These geophysical methods will provide a quantitative assessment of the resistivity of the materials underlying the basin, which can be correlated to the lithology and relative hydraulic conductivity of the materials. The wells will be completed with 2-inch PVC casings, surface seals, and protective barriers or flush completions as appropriate.

#### Percolation Testing and Conveyance Testing

The MWH team will design and site three test pits. Each of these test pits will consist of a pair of half acre pits side-by-side, one containing existing material at it's below bottom, the other "cleaned", with fine shallow material removed. It is assumed that the District will provide conveyance of State Water Project water for these tests.

The MWH Team will identify the locations to be tested and the detailed testing methods in TM-1, after consultation with the Districts. SBVWCD will construct and operate the diversion facilities and conduct flow measurement during the testing. The MWH team will be responsible for developing the detailed construction drawings, testing methods, and supervision of collection of field data during the tests. Locations, construction designs, testing methods and data compilation needs will be defined in TM-1.

<u>Task 1.3.3 Capacity Estimation</u> – A description of work proposed to estimate the capacity for each facility component is provided below.

#### Diversion Works and Conveyance

The MWH team will:

- Prepare channel cross sections and estimate roughness
- Make estimates of channel losses (percolation)
- Estimate the capacity and operational limitations based on theoretical capacity (using percolation tests as calibration) and an evaluation of historical data

#### **Recharge Basins**

The field data collected in Task 1.3 will allow MWH to evaluate each one of these potential limitations. The analysis of this data will consist of the following:

- Estimates of aquifer characteristics will be made by development of two geologic cross-sections, one perpendicular to the SAR, and another parallel to the long axis of the recharge facilities. These sections will include the following information as available:
  - o Ground surface elevation

- o Groundwater surface historic high and low
- Lithology thickness and extent of high permeability and low permeability units based on information obtained during drilling and trenching
- o Bedrock depth (from lithologic logs)
- o Mounding estimates in response to recharge
- o Groundwater flow direction
- Well locations
- Total aquifer flow estimates will be prepared for the perpendicular section described above.
   Estimates will be made using Darcy's Law and these flow estimates will be compared with numerical estimates that may be generated using the Geoscience model described in the RFP.
- Piezometric surface maps of groundwater elevations for project area will be constructed for both wet and dry initial conditions. These initial conditions are critical to the mounding effects and ultimate capacity of the facilities.
- Mounding estimates based on aquifer properties and calculated diversion and conveyance capacity will be completed. Experience has shown that mounding needs to be considered on time steps as short as one week. The impact of mounding is more critical in Borrow Pit grounds than for the surface spreading grounds because of the pit's lower elevation. The MWH team will use the Hantush analytical method for estimating mounding beneath the spreading basins. Mounding estimates will be illustrated on the sections described above. Decay rates of mounds will also be estimated.
- Total theoretical capacity will be estimated by comparison of measured percolation rates during field testing to total area of similar shallow soil characteristics. This will be compared to historical maximum percolation rates based on observations of operators.
- Estimates of mounding impacts will be used to estimate groundwater elevations under "wet" scenarios. This data will be used to evaluate potential off-site impacts to mining operations and other stakeholders.

#### TM-2 Recharge Characterization and Capacity Summary

The results of field surveys and analytical evaluation methods will be documented in TM-2. This document will contain the results of field work and recharge capacity estimates. The information included in this document will consist of:

#### • Field documentation

- Results of physical surveys (maps, observations, and photographs)
- o Trench logs

- Maps showing soil sampling locations
- o Laboratory results and tabular summaries
- o Monitoring well lithologic and geophysical logs
- o Monitoring well completion as-builts
- o Observations of testing of conveyance and diversion tests
- Results of percolation testing

## • Analysis of Existing Spreading Capacity

- Surface clogging
- o Diversion or conveyance limitations
- o Groundwater mounding limitations
- Evaluation of off-site impacts (e.g. gravel mining)
- o Relationship of initial groundwater elevation conditions to spreading capacity
- Estimate of the range of maximum percolation rates as a function of groundwater depth

TM-2 will be provided in draft form for review and comment by SBVWCD and SBVMWD. The MWH team will meet with representatives of these agencies to discuss the results of the field work and analysis. MWH assumes a time period of two weeks for the District's review. After incorporation of the comments, the TM will be produced in final form.

#### Task 2: Establish Spreading Objectives

The objective of Task 2 is to compare the capacity of the existing spreading grounds (evaluated in Task 1) to the amount of water set forth as spreading objectives.

Objectives to be evaluated were established in the IRWMP and are as follows:

- IRWMP Baseline Scenario Total water needed to be spread in the SAR spreading grounds for this scenario (ultimate demands for the Valley). The annual flow data are presented in Exhibit A to the RFP.
- IRWMP Scenario A Total water needed to be spread in the SAR spreading grounds to achieve the objectives for this scenario (additional yield of 40,000 acre-feet). These data is provided in Exhibit B to the RFP.
- Maximum Instantaneous Spreading Scenario Instantaneous flow rate of 500 cubic feet per second (cfs) for a limited duration.

Task 3: Perform Feasibility Analysis and Prepare Conceptual Design to Meet Spreading Objectives

The completion of Task 2 will allow for a comparison of the three refined spreading objectives to the capability of the existing facilities documented in TM 2. Each of the three scenarios identified in the RFP will be evaluated. Outcomes possible for each scenario are illustrated below:



Maintenance, and Monitoring Recommendations Our preliminary evaluation of the recharge facilities indicates that improved operation, maintenance, and monitoring methods will substantially improve the capacity of the existing facilities. MWH will build upon the collective experience of our work at 27 groundwater recharge facilities operated by Los Angeles County Department of Public Works, and 11 outside agencies polled during the Percolation Optimization Investigation to

Operation,

recommend Best Management Practices to SBVWCD. MWH will also use the experience of Dr. Herman Bouwer, to assist with these recommendations. MWH will make these recommendations regardless of the estimated capacity of the outcome of the comparison of existing capacity to spreading objectives.

## Conceptual Designs

If it is determined that new construction is required to meet the objectives of any of the three scenarios, MWH will develop conceptual designs for both short-term and long-term facilities for each of the three scenarios.

Conceptual designs will show the limits of work, primary improvement requirements and approximate facility sizes. The conceptual designs will provide a clear understanding of the required improvements and will be used as the basis for estimating construction improvement quantities and associated construction costs opinions and will be appropriate for use for the succeeding environmental work. For work within the Borrow Pit, the background information from the District's drawing "Proposed Build Out of the Borrow Pit" will be evaluated and used to the extent appropriate.

USGS mapping information, supplemented with field measurements by the MWH team, will be used for other conceptual drawing backgrounds. MWH will also explore and use U.S. Army Corps of Engineers asbuilt data. MWH assumed that adequate topographic mapping is available to the project team, and that the area will not require re-mapping.

Conceptual designs will be prepared to show improvements to fix the existing conveyance system's choke points and to strengthen the canal in areas where erosion has occurred. The MWH team assumed that the conceptual designs will also show:

- New basins in the Borrow Pit
- Conveyance system in Borrow Pit area for basin flow management
- · Basin overflow and flow control facility details
- Extent of removal of fine material in the Borrow Pit
- · Typical sections of the basin dikes
- Dike profiles
- · Extent of material excavation and disposition at Cuttle Weir
- · Construction of a new higher capacity box culvert at Greenspot Road
- Construction of an intake and pipeline from Cuttle Weir to the Foothill and Santa Ana River crossing pipelines downstream. This will allow SBVMWD to deliver water to other spreading basins and/or treatment plants in the area

Conceptual design requirements will be determined based on Task 1 and 2 activities. The following discussion provides background on our project approach for all potential conceptual design items.

Improvements to the diversion works would be dependent on the interaction of the existing weir, channel geometry and terrain, gate structure capacity and conveyance system inlet capacity. Should improvements be needed, a site plan and plan drawings of the proposed improvements will be prepared. More detailed drawings,

as needed by the MWH Team to clearly define the work, will also be included. For example, if channel improvements were determined to be required, the conceptual drawings will define the limits of the work and the new and existing cross sections. These items will be used to calculate the required earthwork quantities.

Conveyance capacity improvements may consist of expansion of the existing channel prism, improvements to improve channel roughness coefficients (such as lining or use of alternative materials) or replacement or improvements to the existing transition structures. Improvements to the transition structure at Greenspot Road are anticipated due its planned realignment. An increase to the transition structure's width will increase its flow carrying capacity. Improved transitions into and out of the box channel will decrease the erosion occurring on the downstream end.

Conceptual design for conveyance system improvements will include a general site plan identifying the existing facilities that will highlight the planned areas of improvement. Specific cross sections of proposed channels and box structures will be prepared. Drawings showing modifications or improvements to control structures (such as the addition or replacement of gates) will be added as needed by the MWH team to clearly define the work.

Spreading ground drawings will show the overall spreading ground layout, proposed basin limits and interior dike requirements. Typical dike cross sections will be prepared to show the maximum height, side slope and roadway width. Details of flow control structures will be prepared. These structures could vary from simple overflow weirs to more complicated weir boxes depending on operational and flow variation requirements.

Flow control structures within the spreading facilities will also be detailed. These facilities will allow refinement of flows into the spreading ground to bypass and isolate basins for cleaning and system management. The number and type of facilities needed will be dependent on operational criteria established for the work. With development of the conceptual plans for spreading facilities an approximation will be prepared of the additional land owned or controlled by SBVWCD that is available for construction of spreading facilities.

#### **CEQA-Compliant Project Description**

Although preparation of CEQA documents is not included in the Scope of Work, MWH will supply a separate project description for each of three scenarios that are appropriate for use in subsequent CEQA documents. MWH will also identify potential environmental issues with construction that become apparent, or based on our experience with these types of projects.

#### Cost Estimates (Opinion of Probable Construction Cost)

The Opinion of Probable Construction Cost to be developed for necessary facility improvements under each of the three scenarios will be prepared in accordance with the cost estimate classes defined by the Association for the Advancement of Cost Engineering for Conceptual Design Phase - "Class 5" level estimates. The MWH Team will use the services of experienced cost estimators working within the construction group of MWH (MWH Constructors) to insure realistic estimates of construction costs.

#### **Task 4: Draft and Final Report**

The MWH Team will prepare a draft report for review by SBVMWD and SBVWCD staff. The report will include:

- · Operations, maintenance, and monitoring recommendations
- Conceptual design of recommended additional facilities needed under each of the three Scenarios
- CEQA-compliant project descriptions for the three scenarios
- · Cost estimates for the facilities necessary for each of the three scenarios

Technical Memoranda 1 and 2 will be included as an appendix to the report, as will meeting minutes from the Workshop held in Task 2 to refine spreading objectives.

#### **Task 5: Project Management**

Mr. Victor Harris will be the primary point of contact with SBVMWD and SBVWCD staff. Mr. Harris will ensure the completion of the following during execution of the work:

- At the commencement of the project, Mr. Harris will facilitate a "kickoff" meeting to discuss the scope and parameters of the project. Specific goals and milestones required accomplishing the project, and details of project communication will be reviewed and refined.
- Mr. Harris will develop a project execution plan which details specific requirements of the project, points of contact, schedule, required documentation, and roles and responsibilities.
- Mr. Harris will be responsible for monthly reports to SBVMWD and SBVWCD that will detail the
  activities completed during that month, anticipated activities for the next month, unique challenges or
  problems with recommended actions, and schedule and budget tracking.
- Mr. Harris will facilitate periodic project meetings to be held with SBVMWD and SBVWCD staff for the duration of the project. These project meetings will be used to coordinate and guide conduct of the work, and necessary coordination with other agencies. These meetings will also be the forum for review of interim deliverables, concepts, drawings, and maps such that materials are previewed and accepted prior to formal submittal. MWH assumed that 12 meetings will be required over the course of the project.



Santa Ana Rive	er Groundwate	er Recharge Optimizat
Start	Finish	Task Name
Fri 4/4/08	Fri 4/4/08	Notice to Proceed
Fri 4/11/08	Fri 4/11/08	Kick-Off Meeting
Fri 4/11/08	Thu 5/8/08	Data Compilation an
Fri 5/9/08	Thu 5/29/08	TM 1- Data Collectio
Thu 5/29/08	Thu 5/29/08	Submit TM 1
Fri 5/30/08	Thu 6/26/08	Diversion Works and
Fri 6/27/08	Thu 7/17/08	Trenching, Sample C
Fri 6/27/08	Thu 7/10/08	Construction of Moni
Fri 7/18/08	Thu 11/6/08	Percolation Testing
Fri 11/7/08	Thu 12/18/08	Capacity Estimation
Fri 12/19/08	Thu 1/8/09	TM 2-Capacity of Ex
Thu 1/8/09	Thu 1/8/09	Submit TM 2
Fri 5/9/08	Thu 5/22/08	Establish Spreading
Fri 1/9/09	Thu 4/2/09	Feasibility Analysis a
Fri 4/3/09	Thu 4/23/09	Write Draft Report
Fri 4/24/09	Thu 5/7/09	<b>Client Review</b>
Fri 5/8/09	Thu 5/21/09	Write Final Draft
Fri 5/22/09	Mon 5/25/09	<b>Print Final Report</b>
Mon 5/25/09	Mon 5/25/09	Submit Final Report
Mon 5/5/08	Mon 4/6/09	<b>Project Managemer</b>
Mon 5/5/08	Mon 5/5/08	Progress Report/
Mon 6/9/08	Mon 6/9/08	Progress Report/
Mon 7/7/08	Mon 7/7/08	Progress Report/
Mon 8/11/08	Mon 8/11/08	Progress Report/
Mon 9/8/08	Mon 9/8/08	Progress Report/
Mon 10/13/08	Mon 10/13/08	Progress Report/
Mon 11/10/08	Mon 11/10/08	Progress Report/
Mon 12/8/08	Mon 12/8/08	Progress Report/
Mon 1/12/09	Mon 1/12/09	Progress Report/
Mon 2/9/09	Mon 2/9/09	Progress Report/
Mon 3/9/09	Mon 3/9/09	Progress Report/
Mon 4/6/09	Mon 4/6/09	Progress Report/
Project: SanBernardino	SantaAnaRive	ask
Uate: Mun 4/∠ I/∪o	<i>.</i>	olit

Appendix B Water Level Plots
































































Appendix C Trench Logs

				Plan View-Site Location	(Provide Sketch)	*		EXPLANATION	SOIL TYPE CONTACT (SHARP)	OTHER CONTACT AS INDICATED ON LOG)	ANALYTICAL SAMPLE LOCATION (WRITE SAMPLE NUMBER OUT TO SIDE)	GEOTECHNICAL SAMPLE LOCATION WRITE SAMPLE NIIMBER	OUT TO SIDE)	- //// SHADING TO DENOTE STAINING	 SHOW LOCATIONS AND TYPES OF ALL MAJOR DEBRIS
RENCH LOG	Sheet of	Refusal? (Circle One) If Yes Yes (No) Depth =	Photo? (Circle One) No. Yes/ No	% Man-Made Debris	Wall of Trench Shown (Circle One) N S E W NE NW SE SW										
FIELD T	Location $\rho_i + 1$	Date and Time Completed $6 - 19 - 0\%$	Total Number of Samples	Drive N. A. Hand Auger										Image <th< td=""><td></td></th<>	
rap Optimization Studu	C Elevation and Datum	Clark Bate and Time Started	5 Total Depth 5 1 tot ASC 8 4	No. of Bulk Grab 2 Samples 2 Bulk Grab 2	6-19-08 Checked by/Date	LENGTH (FEET)					Wy SX & Chy, his 4 to card		ou d'aux care art a d'a d'a		





				Plan View-Site Location	(Provide Sketch)			EXPLANATION Soll TYPE CONTACT (SHARP)	OTHER CONTACT (AS INDICATED ON LOG)	FILL/NATIVE BOUNDARY	ANALYTICAL SAMPLE LOCATION (WRITE SAMPLE NUMBER OUT TO SIDE)	GEOTECHNICAL SAMPLE LOCATION (WRITE SAMPLE NUMBER OUT TO SIDE)	STAINING TO DENOTE STAINING	TTTTT BASE OF EXCAVATION	SHOW LOCATIONS AND TYPES OF ALL MAJOR DEBRIS
RENCH LOG	Sheet   of /	Refusal? (Circle One) If Yes Yes (N) Depth =	Photo? (Circle One) No. Vesy No	% Man-Made Debris	Wall of Trench Shown (Circle One) N S E Ŵ NE NW SE SW										
FIELD TI	Location $\mathcal{V}; \mathcal{F}$	Date and Time Completed $6 - 19 - 0S$	Total Number of Samples	Drive $\mathcal{N}\mathcal{A}$ Hand Auger 19											
cheres () ptimization Studu	G Elevation and Datum	C/CLK Date and Time Started	Total Depth	No. of Samples $\mathcal{X}$ Bulk Grab $\mathcal{P}$	6-19-09 Checked by/Date	LENGTH (FEET)						wh bravel, dry light brand	1912 13 Slighty dedices Slit.	A CURRENT OF CONTRACT OF CONTR	





				Plan View-Site Location	(Provide Sketch)	26	T S T	· · · · · · · · · · · · · · · · · · ·	EXPLANATION	SOIL TYPE CONTACT (SHARP)	OTHER CONTACT (AS INDICATED ON LOG)	FILL/NATIVE BOUNDARY	ANALYTICAL SAMPLE LOCATION (WRITE SAMPLE NUMBER OUT TO SIDE)	GEOTECHNICAL SAMPLE LOCATION (WRITE SAMPLE NUMBER OUT TO SIDE)	STAINING TO DENOTE STAINING	TTTTT BASE OF EXCAVATION	SHOW LOCATIONS AND TYPES OF ALL MAJOR DEBRIS
TRENCH LOG	Sheet of	Refusal? (Circle One) If Yes Yes (No) Depth ≃	Photo? (Circle One) No.	% Man-Made Debris	Wall of Trench Shown (Circle One) N S E W NE NW SE SW												
	Location BOMON PIL	Date and Time Completed $f - fq - O C$	Total Number of Samples	Drive N. C. Hand Auger		-											
ptimischim Fl. I		Date and Time Started $6 - 19 - 05$	Total Depth 6 4	No. of Bulk Grab 2	Checked by/Date								The second secon	15 2 2 CHERER - MURERENT			
1 ochece 0)	2/9	Clark			iau 6-19-0								21 + 44 Dredel 20	412 5 12 12 12 12 12 12 12 12 12 12 12 12 12			



				Plan View-Site Location	(Provide Sketch)	24		A P		EXPLANATION	SOIL TYPE CONTACT (SHARP)	OTHER CONTACT (AS INDICATED ON LOG)	FILL/NATIVE BOUNDARY	ANALYTICAL SAMPLE	WRITE SAMPLE NUMBER OUT TO SIDE)	GEOTECHNICAL SAMPLE LOCATION	(WRITE SAMPLE NUMBER OUT TO SIDE)	SHADING TO DENOTE	TTTTT BASE OF EXCAVATION	SHOW LOCATIONS AND TYPES OF ALL MAJOR	DEBRIS
	BORD Pit Sheet I of 1	bate and Time Completed Refusal? (Circle One) If Yes $\mathcal{L} - 19 - 0S$ Depth = $\mathcal{R}$	otal Number of Samples Photo? (Circle One) No.	Drive N.A Hand Auger % Man-Made Debris	Wall of Trench Shown (Circle One) N S E W NE NW SE SW		-														
arge UD+1 M1227101 Strety	430ig	C/a $f$ $f = (q - 0)$	1-5 Total Depth 6	C / No. of Bulk Grab 2	Trad 6-19-cg Checked bylDate									a agular de rours de restantes		1835 Silt / Werne Strender Citensep					





	╶┟╾┾╾┾╾╊╌┾╼┿╼╋╼┽╼╋╼┽╾╋╼┽╸╋╺╪╗╴╋╌╋╸╋╴┥╸╋╸┥╸╋╴┥╸┥╸┥╸┥╸┥╸┥╸┥╸┥╸┥╸	
	╋╋╪╪╪╪╪╋╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗╗	
	· <mark>┟╶┼╶┼╶┤╶┥╶┥╴┥╸┥╸┥╸┥╸┥╸┥╸┥╸┥╸┥╸┫╸┥╴┥╸┥╸</mark> ┥╴ <mark>┙╴</mark> ╱┥╸┥╸┥╸┥╸┥╸┥╸	
		┼╉┾╝┼┝╈╂┼┼┼┼╂┼Ŵ┝╔╂┾╔┼┾┼╂┼┼┼┼╂┼┼┼┼╂┼┼┼┼┨
		┼╂┡┡┼╅┠┼┼┼┼╂┼╬┼└╂╠┽┼┼╂┼┼┼┼╂┼┼┼┼┨┼┼┼┤┨
ा ज । । । ज		
	╶┠╌┼╌┽╌┽╌╉╌┼╌┼╌╄╌┼╌┽╌╫╌╢╧╏╱╎╌┼╌┼╴┨╴┤╱╎╴┽╴┼╴┼╴┥╴╁╴┠╶┼╌┼╌┼	
- 19      >		
	┢╩┈┿┿╋┿┿╋┿╋┿╋╋╋	
	┣╸┣━╺┽┿┥┫┿┽┽┽┽┥╄┊┼╎┽┽┥┫┽┽┽┥┫┿┽┽┽╋┿┥┥╋┥┥┥	
3    ごまや		
	┣╴Ш╶┼┼┨┼┼┼┼┼┼┼┥┥┥┑╴╴	
	+ °C	
	┣╤┿┿╋┿┿┿╋┿┿╋┿╋┿╋	
	· ┃- <mark>●┙</mark> ╶┽╌┽╴┫╼┼╌┽╼┼╾┫╼┼╌┽╼╉╼┼╶┽╼┼╌┨╼┽╼┼╼┼╼┼╼┼╼┼╼┼╼┼╼┼╼┼	
	╞╸ <u></u> <u>┲</u> ╺┽┽┨┽┽┽┽┨┽┽┽┽┨┽┽┽┥┨┽┽┽┽┨┽┽┽┥┨┥┥┥╸┝╖┠┿╶┿┥	
	┟╴╠┷╺┉┽┥┨┝┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥┥	
	그는 그는 것을 가지 않는 것이 같아. 이 가지 않는 것이 가지 않는 것을 많이 않는 것을 많이 많다.	

				Plan View-Site Location	(Provide Sketch)	*	000	EXPLANATION Soil TYPE CONTACT (SHARP)	OTHER CONTACT (AS INDICATED ON LOG)	FILL/NATIVE BOUNDARY	ANALYTICAL SAMPLE LOCATION (WRITE SAMPLE NUMBER OUT TO SIDE)	GEOTECHNICAL SAMPLE LOCATION (WRITE SAMPLE NUMBER OUT TO SIDE)	SHADING TO DENOTE STAINING	TTTTT BASE OF EXCAVATION	SHOW LOCATIONS AND TYPES OF ALL MAJOR DEBRIS
<b>FRENCH LOG</b>	Sheet / of /	Refusal? (Circle One) If Yes Yes (No) Depth =	Photo? (Circle One) No. Ges No	% Man-Made Debris	Wall of Trench Shown (Circle One) (i) S E W NE NW SE SW	-									
FIELD	Location Burlow P.F	Date and Time Completed	Total Number of Samples	Drive NA Hand Auger		-									
arge Optimization Stude	Elevation and Datum	STK Date and Time Started	Total Depth 6.1	No. of Bulk Grab Samples 2 Bulk Grab	とし、しつく Checked by/Date	LENGTH (FEET)					B. 1 Sub and what to reward a colling	ables Ruch bur der with 2 25			







	HMM	Plan View-Site Location (Provide Sketch)	Da St Joint Joint	EXPLANATION Soil TYPE CONTACT (SHARP)	OTHER CONTACT (AS INDICATED ON LOG) FILL/NATIVE BOUNDARY	ANALYTICAL SAMPLE LOCATION (WRITE SAMPLE NUMBER OUT TO SIDE)	GEOTECHNICAL SAMPLE LOCATION (WRITE SAMPLE NUMBER OUT TO SIDE)	STAINING TO DENOTE STAINING	SHOW LOCATIONS AND TYPES OF ALL MAJOR DEBRIS
FIELD TRENCH LOG	Location ACTU Retursal? (Circle Cne) If Yes Date and Time Completed たっノーの名 Total Number of Samples Total Number of Samples Mes) No.	Drive Man-Made Debris % Man-Ma	And Soft And	Carded Standed My 11:5 4 1 5 and Clark	CC (1 3 1ain 5 1k 2 Change to less silt / while 9 Key calm C 115 1 1 1 1 2 Change to 10 1 5 1 1 / while 9 Key calm	Same as in the second s	) Sand with share 1, dr 1, 1 but having (104R),	C 15 501 528 CUE 18 Vounded.	C(15 C) C C C C C C C C C C C C C C C C C C
Project Name SAR Low Adwater Rocharge COAL wisation Studie	Trench Number TIA, $B$ , $C$ , $D$ Project Number Trench Number TIA, $B$ , $C$ , $D$ Project Number Trench Operator Project Number TA, $B$ , $C$ , $D$ Project Number TA, $B$ , $C$ , $D$ Project Number TA, $B$ , $C$ , $D$ Date and Time Started Date and Time Started Determine Type Back, $h$ , $P$ , $h$	Bucket Width Trench Length Trench Width VOI No. of Bulk 6145 15 20 3 Bulk 6145 3 Bulk 6155 3 Bulk 6155 3 Bulk 6155 3 Bulk 6155	LENGTH (FET)						

			Plan View-Site Location	(Provide Sketch)	NW (1001)	14 TI2	EXPLANATION Soil TYPE CONTACT (SHARP)	OTHER CONTACT (AS INDICATED ON LOG)	FILL/NATIVE BOUNDARY	ANALYTICAL SAMPLE LOCATION (WRITE SAMPLE NUMBEF OUT TO SIDE)	GEOTECHNICAL SAMPLE LOCATION (WRITE SAMPLE NUMBEF OUT TO SIDE)	SHADING TO DENOTE STAINING	SHOW LOCATIONS AND TYPES OF ALL MAJOR DEBRIS
TRENCH LOG	Sheet of Refusal? (Circle One) If Yes Yes (jo, Depth =	Photo? (Circle One) No. (Ŷes No	% Man-Made Debris	Wall of Trench Shown (Circle One) (N) S E W NE NW SE SW	-								
FIELD	Surruw Pit Date and Time Completed K-2 2 0 S	Total Number of Samples	Drive N.J. Hand Auger		-								
et, Ur Study Televation and Datum	Date and Time Started	Total Depth $\frac{1}{6}$ $\frac{1}{2}$ $\frac{1}{1}$	No. of Samples 2 Bulk Grab 2	Checked by/Date	LENGTH (FEET)					12 4 May + manual ( 20 GW	s and the second of the second		
or Recharge OPH w 2 Project Number	13413019 Operator Dan Clarlo	Trench Orientation	Trench Width ( U 1	04119 Siac 6-20-08	- - -					e pure 51/7 mush check la well freshed with frend	mm/ +1/15 mm - 0600 - 1/2/		
AR CRUNNUM	oment Supplier	pment Type back hup	trench Length $30^{\circ}$	poist or Hydrogeologis/Date	-					A RAZINE CARACTERIZE CONTRACTOR			

Appendix D Boring Logs

	A	Ð,	M۱	NH	I						Boring #: SB-1 MW#:	She	eet	1	of	3
-					-						Project: SAR Groundwater Recharge	e Opt	imiza	tion S	tudy	
											Job #: 1343019	S	lite.			
			/	3			Λı	VII			Logged By: (_) Revi	ewed	By:			
				A	•	X	1/(				Drilling Contractor: Boart Longyear					
			$\langle -$	E			00				Drill Rig Type/Method: Sonic Drilling					
			•		~	~	•••••		o		Drillers Name: JUSE				<u> </u>	
							r h	34 117	0 0 0 0	6-056' 7-050'	Borehole Diam./Drill Bit Type: 4 'C ÚTTING 131T	To Re	otal De ef. Ele	epth ev.		
					Site	Sketc	h Map	<u> </u>	71	FT	, Sampler Type:					
	Dep	th to	1st W	ater ()	<b>V</b> ):			Tim	ie/Da	te:	Drill Start Time/Date: 7-2-08 Drill	Finish	n Time	e/Date	=: 7- 11	-2-01
	Dep	th to	Water	After	Drillir	ng (🔽	():	Tim	e/Da	te:	Well Completion Time/Date: 7 - 3	- 00	8	7=3	lo	
	Dep	th to	other	Water	<sup>.</sup> Bear	ing Z	ones:				Soil Boring Backfill Time/Date:			•		
				Ŀ	alysis	Size							Esti	matec	1%0	f
		erval	(in.)	ts / 6	or Ana	De & S	ller	et)	Type					Sand		
	AVO	ole Int	vered	Coun	ned fo	g Typ	lus Fi	(Fee	Soil			-	e	E		J A
	PID/O	Annuti Depth							nsce	Soil Descr	iption	Grave	Coars	Mediu	Fine	Silt/cla
	O A Stu LIGH								Su	LIGHT	BROWN WELL GRADEP	20	20	2,5	20	r
								1	1	DISNU W	TIF BUCKINERS				<u> </u>	-
	2							 2_								
								<u> </u>								
			<u> </u>					з —	1							
									-							
			74	/				4 —								
			1													
								5 —								
								6 —								
		Z	N/					7								
		7	$\Lambda$					•	Gw	BROWN	L. LIGHT GRAY	2r	20	20	11-	10
								8 —		WELL G	RADEDGRAVEL	23		~	13	
			••••					~ ~ .		WITH C	OBBIES BOULDERS					
F		-						9 —								
ŀ				•												
ŀ								10 —								
	-									**********	λ					
			513	tr				11		····						
ĺ			<u></u>					10								
C								12					·			QAVQC




	4	Ď	M	NΗ							Boring #: 58 - 2- MW#:	Sh	neet	1	of	3
	V	ν									Project: SAR Groundwater Recharg	e Op	timiza	ation S	Study	
			,								Job #: 1343019		Site:			
			/	12	)			۸n	nl		Logged By: C J Revi	ewe	d By:			
				K	1×			YV	U (		Drilling Contractor: Boart Longyear					
				1/2							Drill Rig Type/Method: Sonic Drilling					
		1							٠.	.01. arg	Drillers Name: JOSE					
		•						N	74	2 - 2541	Borehole Diam./Drill Bit Type:	Т	otal D	epth		
								そ	117	07-413	4" CUTTING	R	ef. El	ev.		
					Site	Sketc	h Map	) =	(	643	Sampler Type: SOLIC (D)	NSI	eri	Th	VE	
ſ	Dept	h to	1st W	/ater (	<b>∑</b> ):			Tir	me/D	ate:	Drill Start Time/Date	Finis	h Tim	e/Dat	e 7-	-3-0
Ī	Dept	h to	Wate	r After	Drillir	ng (🔽	·):	Tir	ne/Da	ate:	Well Completion Time/Date:			<u>,</u>	<u> </u>	:01
ľ	Dept	h to	other	Water	r Bear	ing Zo	ones:	8		3	Soil Boring Backfill Time/Date					
ŀ	Ì		<u></u>	i -	ysis	ze							Fet	imater	1%0	
		rval	ju.)	s / 6 ii	Anal	& Si	2	_	AD6					Sanc	 	1
	A)	e Inte	ered (	ounts	ad for	Typ∈	s Fill	(Feet)	Soil T						-	┨
	0/0	ample	ecove	ow C	etaine	asing	ทุกม	epth (	SCS (	Soil Descr	ption	avel	arse	∋diur	je	t/clay
╞	Ē	ů Č	TA A		ď	Ö	Ā	ă			the contract of a	ğ	<u>  8</u>	ž	<u>ل</u> تً	<u></u>
ŀ		0						-	->1	BROWN	WELL GIFADED SAND	25	25	2	10	5
			++-			·····		1 -		WI H G	KAVEL, 10 (HE 1407764					
			†- <b>†</b>	• • • • • • • •				-	-1	ROULDE	P IECS SAD AND					
Ī			1					2 -		FINES	RICESS JIME MINE			-		1
			-1-	31				3-								
			D	<b>}</b>				J -	674	BOULDE	RS WITH MINOR SANDS	45	2.0	10	1	ł
ĺ								4 -	_			~		19		ļ
-			-					-	-							
-								5 -	_							
-		6							-							
-		6	V	1.				6 -	on	SAME	M Angelt					
ŀ		ž	1' <del>7</del>	<i>y</i>						<i>714.4</i>	<u></u>	60	25	10	2	
F		2	$\Lambda$					7 -	5w	TOPIF	T SW	25	35	22	10	(-
			-					0			·····				C.	-7
	T							0	R4	2FTB	OULPERS					
						-	(1105)	9 —			, .					
-			<b> </b>						8							
-								10 —								
4.			+						qu	BOULVI	143 G RAVELS SANDA	60	25	ic.	5	
-			$\mathbf{H}$					11 —	-	MUNICH	E. MINOR FINES					
-					•••••											
L	l	1	I	L				12 —	.1	1		l	l	l		QA/QC

$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Boring #: 9B-2 MV	V#:	P	Project:	:		SI	neet	2	of	3
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	ered Counts	ied for sis. g Type	s		Soil			Esti	mated Sand	<u>% Of</u>	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	PID/O Samp Interv Recov (in.) Blow (	Retain Anatys Casin & Size	Annult Filler	Depth (Feet)	USCS Type	Soil Description	Gravel	Coarse	Med.	Fine	Silt/Cla
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	g1-1			/2							
$\begin{array}{c c c c c c c c c c c c c c c c c c c $				13-				<u> </u>			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $							<b> </b>	<b> </b>		 	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $				14-	2.5				ļ		
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $					SM	SILTY SAND WITH	<b>y</b> c	30	20	10	10
$\begin{array}{c c c c c c c c c c c c c c c c c c c $				(5-		BOULDERS GRATE					<b> </b>
$\begin{array}{c c c c c c c c c c c c c c c c c c c $					Gui	INFIL GRADED CAND	2.0	25	22	60	P
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $				16-		WITH CORPER GOALES	->	23	50	/~	12
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	12 1			 2						+	
$\frac{1}{10} = \frac{1}{10} $	12			17-	SP	TOP 3'9" BROWN	15	35	25	10	5
$\frac{1}{19} = \frac{1}{19} $	·			18	-	SAND WITH COBBLES		1			
$\frac{1}{3} + \frac{1}{9} + \frac{1}{19} +$						VERT TOP 1' PODRLY					
$\begin{array}{c c c c c c c c c c c c c c c c c c c $				19-	SW	GORADED. TO THE BOTTOM					
$\begin{array}{c c c c c c c c c c c c c c c c c c c $				`		MOREGRAVEL COBBLES	30	25	28	<u>F</u>	5
$\begin{array}{c c c c c c c c c c c c c c c c c c c $			"	20-		FIND FINES					
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	14										
12 - Gw $12 - Gw$ $13 - W$ $14 - W$ $14$				21-	5M	SILTY SAND WITH	15	20	25	20	20
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$						GRAVELL COBBLES		~			
$\frac{23}{24}$ $\frac{23}{24}$ $\frac{23}{24}$ $\frac{24}{24}$ $\frac{24}{54}$ $\frac{24}{54}$ $\frac{24}{54}$ $\frac{24}{54}$ $\frac{25}{54}$ $\frac{25}{3'9''}$ $\frac{25}{27}$ $\frac{27}{4}$ $\frac{27}{$					GW	BROWN, WELL GRHPEP	40	25	25	10	
$\frac{29}{24} \sqrt{\frac{14}{50}} \sqrt{\frac{14}{50}} \sqrt{\frac{14}{50}} \sqrt{\frac{12}{50}} \sqrt{\frac{16}{50}} \frac{16$			^	3-		WITH COBRLES				*****	
$\frac{27}{24} \sqrt{24} \sqrt{324} 324$					0x	9" LONG ROCK CORE					
$\frac{27}{3} \frac{1}{9} \frac{1}{1} \frac{1}{10} 1$			2	24-		BROWN LIGHT GRAT					
$\frac{25}{3'9''}$ $\frac{25}{3'9''}$ $\frac{25}{16}$ $\frac{27}{3'9''}$ $\frac{17}{16}$ $\frac{17}{57}$ $\frac{1}{57}$ $\frac{1}{5$	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~				511	BROWN WELL GRADEP					
$\frac{3'9''}{1}$ $\frac{27}{7}$ $\frac{1}{7}$				25 -		SAND COARSE SAND			2.00		~
$\frac{27}{1}$	2'97					AD COLK COP EN	<u>15</u>	30	28	45	<u>.</u>
$\frac{27}{1}$ $\frac{1}{1}$ $1$			7	26-		AND LUCE CONC.					
$\frac{1}{10} \frac{1}{10} \frac$	$2\overline{\chi}$		· · · ·								
28- 500 28- 500 29- 29- 500 29- 29- 29- 29- 29- 29- 29- 29-			L		SP [	BROWN POCKLY GRADED	20	60	15	5	
30 - 31 - SM SILTT SURD WITH GRAVEN 20 20 25 20 15 31 - SM SILTT SURD WITH GRAVEN 20 20 25 20 15 30 - 31 - SM SILTT SURD WITH GRAVEN 20 20 25 20 15 COBBLES AND MORE 29 - 30 - 31 - SM SILTT SURD WITH GRAVEN 20 20 25 20 15 COBBLES AND MORE 29 - 30 - 31 - SM SILTT SURD WITH GRAVEN 20 20 25 20 15			ב 🖳	28	k	COARSE GAND WITH GRAVEL					
29- 29- 29- 29- 29- 29- 29- 20- 20- 20- 20- 20- 20- 20- 20					ςω	LIGIHT GRAY WELL GRAM	25	30	25	15	5
30- 30- 30- 30- 31- 51- 51- 51- 51- 51- 51- 51- 5				29 -	ŀ	SAND WITH FIRAVELS					
30- 31- 51- 51- 51- 51- 51- 51- 51- 51- 51- 5		•••••			C	COBBLES AND MORE		·  -			
31-SM SILTY SUBS WITH GRAVEN 20 20 25 20 15				0-	ŀ	FINES					
51 SM SILTY SCAR WITH GRAVEN 20 20 25 20 15					ŀ						
OF CATRICE OULAND TO			- 3	51-0	SM	SILTY SCAR WITH GELD	20	20	25	20	15
	10'4+''		'2	[		CABIES GLADED TO	>		• • • • • • • • • • • • • • • • • • •		·***



									· · · · ·	ŕ – – – – – – – – – – – – – – – – – – –				_	
Л		Ň/J 1								Boring #SB - 3 MW#:	S	heet	1	of	3
V	ע									Project: SAR Groundwater Recl	narge O	ptimiz	ation	Study	
										Job #: 1343019		Site:			
		1	-		******			. ,	7	Logged By: CJ	Reviewe	ed By:			
	1	1 in	ļ	$\square$			1	$\Lambda$ [		Drilling Contractor: Boart Longy	ear				
	ĺ	X	É							Drill Rig Type/Method: Sonic Dr	lling				
	1		Ľ				v	~ ~		Drillers Name: TOSE					
						~			1	Borehole Diam./Drill Bit Type:	E	Fotal [	Depth		
							NB	4	06.040,	4" EUTTING		Ref. E	lev.		
				Site	Sketr	t h Mai	$\mathcal{M}_{\mathcal{H}}$	<b>२</b> ०	04.533	Sampler Type: C.GALSECI	37° n 3	- 5	OA.	<u> </u>	<u> </u>
Dent	h to 1	lst W	later (				<u></u> т;	me/D	ate:	Trill Stort Time /Deta		~ ~ +		10	30
Dent	th to V	Nate	r After	r Drilli	ina ( <b>T</b>			D	ate:	Well Completion Time/Date: 7:30		sn Im		. <u>e:</u> `≯∕	7/08
Dept	h to c	thor	Wate	r Boo	19 ( <b>X</b>	<u>.).</u> onoo:		ne/Da		Seil Device Destrict Time/Date:	6-15		- /	- 58	<u> </u>
Dept				i Deg			Τ		Τ	Soli Boring Backfill Lime/Date:					
	val	) ('i	/ 6 in	Analy:	& Siz			,pe				Est	umate	a % C	и Т
A	Inter	red (i	ounts	d for ,	Type	Fille	<sup>=</sup> eet)	ioi T					Sand	ב 	-
D/O/	mple	cove	N CC	taine	sing	nulus	pth (F	ics s	Soil Descri	ption	lave	arse	dium	e	/clay
Ē	S Sa	¥ Re	ă	<u> </u>	Ca	An	Ď	S		•	und not	<u> </u>	₹ ¶		Silt
		4				4		Si	LIGHT	BRANN VELL GRAI	1-12	5 71	0 70	20	10
					+	<u> </u>	1 1		SAND L	TH CUISIBLES GRAV	the so				
		+,		<u> </u>			.	-	RNJ 1500	COR ICTO LENC	7				
		6,6					2.	-	Noor C				-		
				1			2								
				l				-					- > >		
				ļ	<u> </u>		4 -	_ SP	B0770.M	1'E" POORLT GIRA	DEA 13	4.5	150	20	
	7						-	-	BROWN	COARSE SAND WIT	H. 4	35	25	<b>1</b>	F
	2	V <sub>f</sub>		<b> </b>	+		5 -	- (L.)	LIKAUE	LS R R M LI TH A DA			-		<b> </b>
	.>	71. 1 11	17				-	·Pw	SARD .	TH ARAMA X	<sup>(1)</sup> 2(	21	20	20	10
		1	1				6 -	1	COBBLE	5		+	+		<u> </u>
	7	Ţ				•	-	-							
	7	<u>۸</u>					/-	Siv	Top 1'2	" SAME AS ABOVE	25	25	20	20	10
							8 -	_	FOLLOWE	OBY BOULDERS		ļ			1-
	4.						-	-	CROCK C	PRE TO 9" WY LENGIT	)				
					<b> </b>		9 -	-							
				•	4		-								
					<u>├</u>		10 -	SW	BOTTOM	SET COLOUR CAL	<u>_</u>	<u> </u>			
									BROWN	THE GEARTIC AND	125	3	25	20	5
							11 -		COBBLEC	I'M MININELY (MUS		<u> </u>			
	+ -	- 1		}	t			1				1		i	

l

Boring #: SB - 3 MW#:	Proje	ect:		SI	heet	2	of	3
A red a for	Type	oi			Estir	mated Sand	% Of	
PID/OV Sample Interval Recove (in.) / 6 in. Retaine Analysis	Casing & Size Annulus Filler	(reet) USCS Si Type	Soil Description	Gravel	Coarse	Med.	Fine	Silt/Clay
51811	12							
13 V	] 3		0.0 (1/					
			ROCK CORE AND CUTTING					
f. 1	<b>1 1 1</b>							
62	/5	- Cal	Roma dell'una	1î.			.0	
			FICAUFI WITH COLRIFS	) To	23	20	10	<u>)</u>
	/6		BOULDERS LIGHTBROW					
	J7	SW	LIGHT GRAY WELL	15	20	25	30	10
·····	×8		GRADEP COARSE SAND	<u>/3</u>	<u>,,,,</u>	43		19
			CITH GRAVELS, COBBLES					
	/9		MINOR FINES, DAMP					
	>0							
····								
	21							
	22-							
		-		•••••				
	23-	RX	BOTTOM 2 FT BOULDER					
	24-		ROCK COLE AND POURER					
25 4	25-	-						
25 (1)		.Ġw	GRAY GRAVELS WITH	70	15	10	5	
		-	COTSELES AND BOULDERS					
27 V	27 -							
23		Sw	TOP (8" BROWN					
	28-		SAND WITH GRAVELS	15	25	25	20	15
	29-		COBBIES AND SILT/CLAY					
		-	TO THE BOTTOM ALDEE	15	207	ur l	2-	IT
	30-		FINE GFAT			2	->	
	31-	4  -						
	)2			l	L			DAVOC

Boring #: 56-3 MW#:		Projec	t:		S	heet	3	of	3
A for d for	, Type		Ĩõ		<u> </u>	Est	mated Sand	% Of	].
PID/OV Sample Intervai Recove (in.) / 6 in.	Casing Casing & Size Annulus Filler	Depth (Feet)	USCS S Type	Soil Description	Gravel	Coarse	Med.	Fine	Silt/Clav
21-11		7 72-							
2		2	- 1						 
		73-			1				1
34 1						1			
34 1		]24_	]ML	TOP - FT GRAY CLATE	r		1	10	91
		25-		SILT WITH MINOR SAMP		1			· • • • •
ci g'		7°		CEARSE DOWN GRADED					
		- 2 -	_	TO SILTT SANDAND		]			
		, <b>7</b> °	Gu	GRAVEL BOTTOM 2'5"	60	15	jo	10	5
37 1		27-		COBBLES IMA BOULDERS					
37/1		.[7'	Sh	JOP IFT, GRAY WITH	25	25	20	20	10
		- 28-		GRAVELS					
		. ×	Gw	GRAT WITH COBBLES	60	15	10	(0	5
		- <u>5</u> 0-		AND BALVERS					
			SW	BROWN WELLGRADEP	25	3.5	20	15	5
		u-		SANDWITH GRAVELS	ļ				
		Г°		AND COBBLES	<b> </b>				
		K1-	-						
		· · ·	-						
		¥2-	-				2		
		d							
		¥3-							
		4-	4.1	To shell on the					~
			ML	101 14 GRAT SILT				10	97
<u> </u>		¥5-		WITH MINOR CLAT AND					
				SAUD, FISHT SMELL					
		<b>4</b> 6-	con	UKT, HARP			7		
	•		36	LI GUIT GRAY WELL GRADES	0	25	35	20	10
$+ \mathcal{R}' / \mathcal{H}$		<b>Q</b> 7-		MEDIUM SAND COARSER					
···	·			POWNWARD EZRADED					
		<u>4</u> 8	ŀ	TO SANDY GRAVEL	30	25	25	20	
								·	
		49-	600	GOA	/-				
			ye	AND POLIDIA	00	15.	19	10	
		50-	F	TV-V BUULVERS					
	+		ŀ						
		1	F						
	+		ŀ			•••••			
		2							DA/OC

	(		Boring #: / MW#: /	Sheet	1 o	r 8
-			Project: SAR Groundwater Recharg	e Optimiza	ation Stu	dy
$\bigcirc$		<u> </u>	Job #: 1343019	Site:		
		f(f)	Logged By: CT Revi	iewed By:		
		VVI	Drilling Contractor: Boart Longyear			
			Drill Rig Type/Method: Sonic Drilling			
	Λ	534° 0E.074'	Drillers Name: <i>J</i> Cらを			
	C	0117 06.021	Borehole Diam./Drill Bit Type:	Total D	épth	
			4 CUTTING BIT	Ref. El	ev.	
	Site Sketch Map	1 761	Sampler Type:	<del>-</del>		
	Depth to 1st Water ( $\mathbf{\nabla}$ ):	Time/Date:	Drill Start Time/Date: C6-30 Drill	Finish Tim	e/Date:	
	Depth to Water After Drilling ( $\mathbf{Y}$ ):	Time/Date:	Well Completion Time/Date:			
	Depth to other Water Bearing Zones:		Soil Boring Backfill Time/Date:	·		
	6 in.	Φ		Esti	mated %	6 Of
	nterve ed (in for Ar	eet)			Sand	
	VOVA nple 1 v Cou w Cou ained ained	in the SS SS Field	a ti a u	le le	ш	lay
	PID Sar PID Rec Cas		ріюп	Grav Coa	Med	Fine Silt/c
		SW WELLGE	ADED SARD 1-174	10 20	20	r r
		1- GIRALEL	, TOP 3" MORE GRAVER	1 20	20	→ <u></u>
		GW SAND 70	1 LOUCHED BY 14"	20 20	30 1	0 10
-		2- DRY	FICH GIRAVELLI SAND			
		3				
ŀ	····					
ŀ		4-61-00-				
ŀ	·····	qui VR1 C	TRAT WELL GRADEP	50 20	20	FE
ŀ	2-11	5- GRAVE	L			
ŀ						
		6				
-		7				
	·····  <del>3</del> .  \.	GRAY	. DRY WELL GRADED	60 20	1r	ζ
$\vdash$		8- GRAVEL	WELL GOUNDED		6	
		5015 AU (90	JLAR			
F		9				
	84	10				
<u>_</u>						
		11				
-						
F		12				QAVQC







Borin	g #:	·	MV	V#: [			Projec	t:		S	heet	5	of	8
ج ا	Ø	ared	ounts	s. dfor	Type	6		lio	7		Es	timated Sand	% Of	
PID/O/	Interva	Recove (in.)	Blow C / 6 in.	Retaine Analysi	Casing & Size	Annulus Filler	Depth (Feet)	USCS S Type	Soil Description	Gravel	Coarse	Med.	Fine	Silt/Clay
	72						72-	50	TOP 14" CLATERS AND	5	25	30	245	15
							73-	4	WITH GRAVEL					
-		5.5	<u>!</u>					GN	MIPPLE 211, COARSER	yc	20	Zi	10	re
							74-	-	WITH ROIK COLES UN	-				
	75	7/				•••••		sc	ROTTON DU A COE FUTE	10	1.8	 b.•		17
	75						<b>√</b> 5 −	ETh	TOPIN' COARGER THAN	40	20	20	10	10
		24	()				26-	RX	ABOVE BOTTOM ROLL					
		•-+					70	ł	CORES NE"LONG					
	77	N					77-	Dr	Darry Carry And I DR					
								Gru	BUDDED COD (E) AT THE	- <b></b>			+	
		1, 11	<b></b>				78-	-( -	BOTTOM K'	40	20	20	10	10
	29						70-			••••••	1		<u> </u>	
;	29	1					χ9 	Gu	GRAY WELL GRAPTO	40	20	20	10	10
		1					80-		ETRAVEL WITH BOVIDERS				ļ	ļ
<b></b>		121			•••••		<u> </u>							
		$\uparrow$					81-							
Ś	52	J					ŵ					<b> </b>	<u> </u>	
8	+2	$\uparrow$					<i>0</i>	9w	Top 2'6'	40	20	20	10	10
		1	<del>,</del>				83-	SW	MIDPLE 7", LESS WARKE	15	35	3.5	10	5
	····-	26	·					Cr	LESS FINE	7-	·····	2		
							84-	RX	VERY BOTTOM POIN COR	2	25	>0	25	/ 3
8	5						V <sub>5</sub>		NOOL OK					
8	5	1					<i>n</i>	Su	TUP 15'SW GRAPED	5	40	30	20	5
		184					86-		TO GW TO THE BOTTOM					
C	>2							GW.		30	25	3	20	
		*					87-	swt	TOP 11 (-1) CW GRADTO	1	1/ 1	20	20	$\overline{\chi}$
							6		TO AW TO THE BOTTOM	30	28	27	70	<u>, , , , , , , , , , , , , , , , , , , </u>
		1					8°-	GWI			-)			
	<del>_</del>	119					89-	·						
F		++-						-					<b></b>	
		+				'	70-	-						
g	<u>U</u>	177											• • • <del> </del>	
9	11	licy				•••••		$\mathbf{q} \mathbf{w}$	TOP1'2" (1W 3	5 2	20	20 2	20	5
9	2				S	(	j <sub>2</sub> [	ex	BOTTOM ROLK CORE					04/00

....

.







			ΛH							Boring #: 2 MW#: 2	SI	neet	1	of	6
										Project: SAR Groundwater Recharge	ge Op	timiza	ation S	Study	
		/	, ,							Job #: 1343019		Site:			
-								λ.	n Al	Logged By: CJ Rev	iewe	d By:			
		(	K						/	Drilling Contractor: Boart Longyear					
				A				V		Drill Rig Type/Method: Sonic Drilling	3				
			Ċ	للمستك		×				Drillers Name: JOSE					
						-1	N	34	05-814	Borehole Diam./Drill Bit Type:	Т	otal D	epth		
							6	117."	07.0951	4" /CUTTING BIT	R	ef. El	ev.		
				Site	Sketc	h Map	c	15	lo2FT	Sampler Type: CONSECUTI	VF-				
	Depth t	o 1st W	'ater (	<b>∑</b> ):			Tin	ne/Da	ite:	Drill Start Time/Date: 40-24-08	Finis	h Tim	e/Dat	<u>م.</u>	
	Depth t	o Wate	r After	Drillin	ng (🔽	<b>(</b> ):	Tin	ne/Da	te:	Well Completion Time/Date:			<u>, Dui</u>	<u> </u>	
	Depth te	o other	Wate	r Bear	ing Zo	ones:				Soil Boring Backfill Time/Date:					
			Ŀ	lysis	ize		<u> </u>				1	Esti	mate	1%0	)f
	levie	(in.)	s/6	r Ana	e & S	er	<b>•</b>	Type					Sand	 1	
	VA le Inte	/ered	Count	led fo	g Typ	lli Su	(Fee	Soil -							-
	D/OIC	- Jecor	3low (	Retair	Casin	Innul	Depth	ISCS	Soil Descr	iption	Iravel	oarse	lediur	ne	ilt/cla
ŀ	0	「木		<u> </u>	<u> </u>			-éc	CURPOU	NOFO TO LIFEL RUILADED			2		S
								SW	SANDS	GRAVELS AND FINES	20	25	30	20	5
		.						_	7 HE BUT	70M 2 FEET CONTAINS		1	1		
╞		<u> </u>			ļ		2 –	_	MORE E	FRAVELS PEPHLES			ļ		
ŀ		- a'r	Y					-	AND RO	LK PIECES					
╞		+0					3 –	-	0.44.0				ļ		
ŀ								Gw	ROLP 1	CLES GKAVELS	50	15	15	10	10
F							4 —		LINES	SANVS (HNV) (INK					
ŀ	5								. <u></u>						
	.5	1					5	Gu	SAME	AS ABOVE	11. 5		ih		1
		11	ų				6	Ţ			50	B	15	15	7
		0 <i>1 (</i> 													
F		V.					7 —	0							
-			<b></b> .					th	SAME	A3 ABOVE	50	15	15	IS	5
-		┼┠┤					8 —					1			
-	* * * * *	- + + - +													
F							9 —		* <u>-</u>						
[		51	r (+- }				40								
<u>.</u>		1					10								
											1			,	
9							11 —	-							
<b>,</b>							11 — 	-							



Boring #: MW#: Project: Sheet of 3 Blow Counts / 6 in. Casing Type & Size Retained for Analysis. Estimated % Of Recovered (in.) USCS Soil Type PID/OVA Sand Sample Interval Annulus Filler Depth (Feet) Silt/Clay Coarse Gravel Med. Fine Soil Description 32 PX TOP HATEF FOUT. RELK SW PIECE, FOLLOWED BY LAW GRAPED SANDWITH ETRAVELS AND COBBLES BROWN WET. AT 36' Z DRISSLED. SATURATED 6-26-08 14:42 33-33 34-35-2,6 -37 37-RX UP TO 15" ROCK CORE 27 GIRANTEID - 05-27-08 7:15 AM Ĩ 28-CT 39 <u>7</u>9-RX UPTOIC'CORE ERANITOID 39 ×0-191 41-2'3" £2-42 RX UP TO 9" CORE 43 - RX TOP 214" ROCKS UP 43 6-27-8 4:18 4- SW MIDDLE 2'2' SAND 30 30 20 10 10 65 WITH GRAVELS AND ¥5-COBBLES Botion 1'9" ROCK RX | 46-47 47- RX TOP 3 TO 4". FOLLOWED BY SILTY CLATEY 10 30 20 10 SAND, FISHY SMELL PARK GRAY, WET 8 20 THE LEFT IS WELL GRADED 40 20 20 15 5 SAND WITH GRAVEL COMME 56 ¥8— -<u>[]</u> 50-AND ROCK PIECES BROWN DRY 51-52 2 QA/Q(





Boring #:	T	MV	V#:	L th	1	Project	t:		SI	neet	6	of	6
PID/OVA Sample Interval	Recovered (in.)	3low Count: / 6 in.	Retained for Analysis.	Casing Type & Size	Annulus Filler	Depth (Feet)	SCS Soil ype	Soil Description	ravel	Estir esuse	nated ° Sand	<u>% Of</u>	į
92	$\overline{\Lambda}$				<b>X</b>	92-	SP	son Description	15	0 3. <b>5</b>	∑ 2€	Ē 10	-
						93-	4	14" SP. ROCK PIECES			3*		
						94-	SP	16'' SP					
	74	:Ľ				 G 5		3FTY ROCK VIECES MINOR SP					
	-+					' <sup>-</sup>	-						
92													
97	¥ 			•		97-		NO SAMPLE					
						98-		RECEVERED					
						9 <sub>9</sub> -							
						100-							
						 /01							
								TD= 102					
							-						
							-						
						4	-						
						5							
						6 —	_						
						7 —	-						
						8 —							
						9							
• • • • • • • • • • • • • • • • • • • •													
						1							
	l		l	l		2 🗆							

										Boring #:	MW#: -3	Sł	neet	1	of	C
	W									Project: SAR Grou	ndwater Rechar	ge Op	timiza	tion S	study	,
					×					Job #: 1343019			Site:			
			F							Logged By:	Rev	lewe	d By:			~
				17	7			i) il	Λ	Drilling Contractor:	Boart Longyear					
								M	U	Drill Rig Type/Meth	od: Sonic Drilling	a				
				Ĺ	S			U ·	1	Drillers Name: J	ose				57.	
										Borehole Diam./Dri	Il Bit Type:	Т	otal D	epth		
							N	34	06.185	"CUTTIN	5	R	lef. Ele	ev.		
				Sito	Skoto	h Mar	- WI	17°	07.288'		-/					
De	onth to	1 et M	lator (			in wa	т.,		, 	Sampler Type:	13:12		·····			<del></del>
	onth to	Moto		<u>×</u> ).		·	111 T:			Drill Start Time/Date	e:	Finis	h Tim	e/Date	ə:	
		vvate	r Alter	Driilir	ng ( 🔽	.):	i in	ne/Da	ate:	Well Completion Tir	ne/Date: 7 -	9 - 0	8/	13 =	10	
De	pin to	otner	vvate	r Bear		ones:	T	1		Soil Boring Backfill	Time/Date:	1				
	<u>ឆ</u>	(;	6 in.	nalys	s Siz			e e					Esti	mateo	1 % 01	f
_	Interv	ed (in	unts /	l for A	ype (	Filler	eet)	i T <sub>VI</sub>						Sand		
10/1	nple	over	N Co	ained	sing T	snIns	th (F	)S SC	Soil Deser	intion		je	rse	ium		
B	San	Hed	Blo	Ret	Cas	Ann	Dep	- SO	Soli Descr	puon		Grav	Coa	Med	Fine	
	0	$ \uparrow$					-	Su	TOP 4	FT TAN	WELLERAD	Đ_			-	
	_	<u>    .</u>		<u> </u>			1 -		Strip W	TH GRAVE	S, CORBINS	15	2)	25	20	
							-	-	NDMIN	OR BOULDER	>					
		┝╌┠╌					2 -									
		<u> </u> <b> </b> -∙					-	-								
		+					3 -									
	• • • • • • • •	╞╌╁╌				• • • • • •		-								
			7				4	R	BOILDE	RS TOTAL	2FT	<u> </u>				ļ
		7						-	THICK		·					
							5									
							6 -									
							· · ·	GP	GRAVEL	5, COBBLES	MINOR	30	2.0	0		
	t	¥					7	•	BOUDE	2		10		<u> </u>		
	7	/Λ.						KX.	707 10"	BOULDER						
	+					]	8 —	<b>e</b> 0	Decliy	·						
								۲r	MORLI (	ILADED TAU	COARSE	30	45	20	5	
		51					9 —	00	GRAV WI	LH CTRAVEL	A110- A	-				
		×						eq	LICHUEL ADDINT	S.CV. FILLES,	MINOR	75	20	5		
		-++					10 —		prolott	7		1				
				•••••	• •											
	t	-++					11 —		#**********							
		1			- E	1					1	\$	1	1		

Boring #:		M۷	N#:->	>		Projec	t:		S	sheet	2	of	9
≤	red	ounts	d for	Туре			oi			Es	imated Sand	% Of	
PIU/OV Sample Interva	Recove (in.)	Blow Co / 6 in.	Retaine Analysis	Casing & Size	Annulus Filler	Depth (Feet)	USCS S Type	Soil Description	Gravel	Coarse	Med.	Fine	Silt/Clay
12	$  \wedge$					12-	56	TOP IFT TAK LIGHT	15	3.	F 24	120	1.5
		1					-1	GRAY WELL GRADFO		L #		+	
	$\square$					13-	-	NOAASE SAUD WITH	-			<u>+</u>	
	1-4						-1	GRAVELS FOLLOWED				+	•••••
	51	9				4 -	SC	BY S' SILT SANDI SILT	1	+	1	35	65
	(j=- C	1					GP	GRAVELS COBBLES AND	70	20	5	5	+
		]				(5-	] '	BOULDERS	1			1	1
						10-				1			
					 	10							
17_	X					17-							
17	<u>/</u> ]\						<b>₽</b> X	TOP 18' UP TO 10'LONG					
						18-		ROCK CORE, BOULDERS					
	- <b> </b>						Sh	TAN LIGHT GRAT	15	25	25	25	10
						19-	4	WELL GRADED SAUP		ļ			
								WITH GRAVELS COPPLES		 			
	c 14	11				<u> </u>	4	MINOR BOULDERS	ļ	ļ	ļ		
	64	<b>-</b>				L °			<b>.</b>				
	- 1					21-	-						
						22-	-						
			•••••	• • • • • • • •									
	_					Z <sup>3</sup> —							
2.5													
24						24-	Gal	1 DIL GARDER CALE	10	سراذ		• 7	
μΨ				· <b></b>			-cw	WELL GRAVES CIENT	60	<u>15</u>	15	10	
						25-		ROLLAZOS LES INNI	,				
	2' 8			• • • • • • •				WULLIC LICTHI ARA			•••••		
	1					26-							
27	57										•••••		
27	~					27-	Gu	"GRAVEL FOLLOWED					
	•••••						M)	BY GRAY SANDY MAYEY	4		5	10	G.A
								GILT DOY FLUTY SUFIL				· · ·	74
							a	GIRADED TO SANDY SITT		8			
						~	m	BROWN SILTY SAND		-			
						201	50			5	15	20	60
						<u>ッ</u> ー							-
						2.				• • • •			
						7'	9P 1	GRAUFL (OBLES					
						20	ſ	BOUL DT-RS					

	Boring	#:		M۱	N#: -	4	_	Projec	t:		5	Sheet	3	of	9
	۲,	17	red	ounts	d for	Type			oit			Esti	imated Sand	% Of	T
(	VO/DI4	Interva	Recove (in.)	Blow Co / 6 in.	Retaine Analysi:	Casing & Size	Annulus Filler	Depth (Feet)	USCS S Type	Soil Description	Gravel	Coarse	Med.	Fine	Silt/Clay
			6'					<b>3</b> 2-	1						
		33	V					3-						+	• • • • • • •
		33	<u>/</u>		 			/ · · ·	N.	TOP DUE(1) FOUT ROCK					
			/					34-	0.0	CORE + CUTTINGS				ļ	
			le tr	-+;				·	Gu	TAN WELL GRADED	40	20	20	10	10
			44					35-	1					<u> </u>	
ľ			·					. 0				.			
ĺ								<b>7</b> 6-			+	1		<u> </u>	
		1	V					27-							
-		32	4					<i></i>	ML	GRAT SMIDY CLAYEY					
╞								<u>}</u> 8−	-	SILT, TRACE CIFALELS			5	1리	85
ŀ			+						•	RRY HARD					
F			. /	, tr				39-	40	GRADED TO SUTT SMD	F	10	76.	11	2.5
ŀ			4-1	(						FOR 6" TOLLOUTORY	2.	[	10	<u>-&gt;</u>	<u>30</u>
								40-	RX	BULLDERS	1	+			
ļ								<b>U</b> 1							
F		· 2						۲' 							
		-	¥ A					<b>4</b> 2 —	Gil	LICON CANY ALAN		0.12			
ł		•				·			12	LIGHT CTRAT GRAVELS	60	15	15	10	
F			3					<b>4</b> 3 –		BULDIO				+	
	¥	X						112					•••••		
	4	F	1					Ψ <sup>4</sup>	Gw	SAME AS ABOVE	60	15	15	10	
-			rig	1				45-		-					
-			· )- /		• • • • • •				-						
┢			∦					46-	Sea.	1/1/11/2 OUT DAVIS					
ŀ			· · · ·	•••••					200	WATHER DU, DRILLER					
F								47-	ŀ	REFUND STAD					
								 	-			• • • • • •		•••••	
			1		·			4º _	ļ.						
			<b> </b>					$\psi_9$	Ļ						
			<b>+</b> +.	·   -					-				.		
$\vdash$			+					50-	F						
F-			•		•••••	•••••			+ -			••••	.		
			1,	,				S1-							
			5	,			1	G							
								) -				·····			4/00











Boring	#:		MV	V#: ~	3		Project	: 5BV	WCP	Sh	neet	4	of	9
٨ .		pe	ounts	d for	Type			5		-	Estin	nated Sand	% Of	ŕ
PID/OV Sample	Interval	Recover (In.)	Blow Co / 6 in.	Retaine	Casing & Size	Annulus Filler	Depth (Feet)	ype	Soil Description	gravel	oarse	Aed.	eu	
-	-	1					2-	Gri	177 110		0	~	L.	+
								γw	(FI) CIGHI GRAY	60	20	15	3	+-
	-	+-					3-	Ser	CURRCES. BOUDERS	20	2.4			-
								7~	CONTRACT STRAT	20	20	20	20	+
-	-	+					4		GRAIN SIERS VARY	<u> </u>				╀
		·							FRAM FINE TO CARALE	<b>\$</b>				╞╴
-		+					5 —							┝
														╞╴
-							6 —							┝
1	37	V												<u>-</u> -
-	71	2	-				7 —							┢
		×												ŀ-
							8 —							F
														<b> </b> -
							9							
_														
							0	Ī						
							4							
	]									·				
							2_							
							3 -	ļ						
								ļ						
							4	ļ						
		·						ļ						
							5 —	-						
					·			ļ						• -
							6	ŀ						
					·									
							7	F						<u></u>
					·			ŀ			••••+			
							8	-						
•••+••								-  -			· • • • • • • •		····-	
							9 —	ŀ						
+								ŀ					•••••	
							0 -	┝						
•••								-  -				·		
							1-	$\vdash$						
			•••••	·	• • • • • <b> </b> •			-  -						

	Boring #: MW#: 之 Sheet 1 of 5								
	Project: SAR Groundwater Recharge Optimization Study								
	Job #: 1343019 Site: 多乃								
	Logged By: C, J/AL Reviewed By:								
Borrowfit	Drilling Contractor: Boart Longyear								
	Drill Rig Type/Method: Sonic Drilling								
	Drillers Name: JOSK								
	Borehole Diam./Drill Bit Type: Total Depth								
	4 / CUT BIT Ref. Elev.								
Site Sketch Map	Sampler Type: Consecutive								
Depth to 1st Water ( $\mathbf{\nabla}$ ): Time/Date:	Drill Start Time/Date: 6-23 - 7 Drill Finish Time/Date:								
Depth to Water After Drilling ( <b>y</b> ): Time/Date:	Well Completion Time/Date: ABAN DONED								
Depth to other Water Bearing Zones:	Soil Boring Backfill Time/Date: 6-26-08 9=10 AM								
e Size	Estimated % Of								
d (in /pe & literva	Sand								
OVA OVA overði Ing T, SSo SSo	a a a a a a a a a a a a a a a a a a a								
PID Pep Num Sam PID									
Gu EVE	LL GERADED BROWN SANDY								
m 1 - Circ	WEL, GRAVEL ~ 3"-4" 20 25 30 24 8								
Mor	EFINE THE TOP 1~2'								
	THEERD GRAVEL, DKI								
CUA CUA	RT7. FELD: PAR, LESS								
	UOR MAPPICMINERALS								
to the the the test of tes	CE MECA								
5 - 5									
	40 25 15 15 5								
7 6 6									
ZA Gui she	LE AS ABOVE, MORE TITT								
8- <u>B</u> OU	LDERS PEBBLES CORDIES 45 30 10 10 5								
9-									
12 KX E.5"	GRANCID								

Boring #:	1	M١	N#: 🗍	2		Project	: SB	VMWP	S	heet	2	of	5
A	þe	ounts	dior	Type			E			Est	limated Sand	% Of	
VOV mple		. vo E	ainec	sing . ize	nulus	pth eet)	S S S S S S S S S S S S S S S S S S S		rei	lse	-		ļ
PIC Sa	E E	Blo / 6	Ret An	0 8 0 8	Anr	a ij	USC Typ	Soil Description	Gra	Coa	Mec	Fine	100
12	- 12	9	1			12-	Fili	SAME M NAME PARK	us	20	10	10	(*
12	-12	·	+	+			DY.	Deconord	$+\tau_{1}^{2}$	12.1	+/		+>
17						13-		104/1ER	$+ \mathbf{x}$				
					<b></b>			SAME AL ABOUE	+='}	¥			
			<u> </u>		L	14-		MORE GRAIELS. CORBIES					_
								ANT BOULDERS	<b>                                     </b>				L
	31	5 Y				15-			50	25	10	10	5
						13		20 C	1				
	1									1			
						)6-					+		+
73	<u>-+-</u> };		+				1					+	•
17	- 4		<u> </u>			17-		CAAR AL MODILE	115	1.	10	1	+
	`- <u> -</u> {{	×						SPYOLE AS ABOVE	4.)	2		<i>1.</i> 0	12
						18-	-		<u> </u>		+	11	
			ļ						- <u> </u>	╎╴╻┨╴		<b>  </b>	<u> </u> ./
						10-							
						19							
					-						T	17	
						20-	1						i f
			•								+-+-	<u>∤-</u> <u></u>	
						21		······································			+	┝╋─	$\left  \right $
		517			• • • • • • •						┝╌┠╴	<b>∤-                                    </b>	-+
	- <del>X</del> -4	5				ı2 —					$\vdash$	<b> </b>	┞──┦
											<i>-</i>	-	
						23-		- <i>i</i>	V	V	V	V	V
						-	SW	PREEL GRADED SAND WITH	15	20	30	20	11
	1							CLAY AND GRAVEL BROWN					
						24		MOREFINE			1		
										- <b>1</b> /-	10	5	t
						25 -	CX	DOCKS BOTTOD S !! AL PE	HC I	2	10	ic.	1-
21	517								2	-19			7
24						26 —	$G_{\mu\nu}$	DEPEN Off Mane Fuel	20			17-	14
20	18	· <b>/</b>					cin	173 TTOM & MORE FINE	70	20	20	12	15
27		<u></u>				27-	au	2					~
2 X	44.						RX .	BOULDER GRAVEL SANA	50	20	10	Ø	Ľ
						78-	Gil	AND CLAT MIXTURE					
							Jw [						
		,		[			ſ				-		
	1			1		29-	F						
							F			• • • • •			
	++					30-	F						
2-1													
7		-				31-	ov F	The life from and the second					
	44						2	INUNUATES OF ISULDERS	30	20	13	10	15
						12		MAN KOCK POWDER					







	Sol X								i Sector Sector	Project: SAR Groundwater Recharge Optimization Study					
						501 /		5	~	Job #: 1343019 . 0/0/	S	ite: S	oil	ß	
					/	//		1p		Logged By: A. Lecvitt Rev	ewed	By:			
					1			12		Drilling Contractor: Boart Longyear	л	nw	14		
								3		Drill Rig Type/Method: Sonic Drilling	r h	and	54	. <i>p</i>	
										Drillers Name: A.Leavi	++	-			
								$\cup$		Borehole Diam./Drill Bit Type:	То	tal De	epth		
											Re	ef. Ele	ev.		
				Site	Sketc	h Map	ว			Sampler Type: hand sam	pla	و		1-1-1-1	
Dep	oth to	1st W	ater (	☑):		Rest	Tim	e/Dat	te: 🛏	Drill Start Time/Date: Drill	Finisł	n Time	e/Date	-/ ( a:	
Dep	oth to	Water	After	Drillir	ng (🔽	<u>()</u> : -	- Tim	e/Dat	e: 🛥	Well Completion Time/Date:	Gitzger				
Dep	oth to	other	Water	Bear	ing Z	ones:		~		Soil Boring Backfill Time/Date: 6	-18	0	8		
			.Ľ	alysis	Size							Esti	nated	1%	
	erval	l (in.)	lts / 6	or Ané	be & (	ller	at)	Type					Sand	 I	
AVC	ple In	verec	Cour	ned f	IVI Br	lus Fi	h (Fe	S Soil			5	e	Ę	Ī	
PID/	Sam	Reco	Blow	Reta	Casil	Annu	Dept	nsc	Soil Descr	iption	Grave	Coar	Medi		
	$\mathbb{N}$	$\overline{\mathbf{N}}$	[					er	tan 1br	own, dry, poorly	1.0	~		17	
	$  \land$	$\square$	ļ		ļ		• • •		comente	d, loose, Enartz/feldsp	15	20	3		
									micay	rucle chips tope					
						<u> </u>	2		uptu	<i>Б</i> ″.				-	
	·														
				-			3-		CPS: 0	DOI - N 34º 06.142'				┢	
					· · · · · ·					W117°06.90/	-				
														Γ	
				\$			5 —		· ·					ļ	
			^												
							6							-	
		+													
			:	·····			7 —		~						
							8								
							9 —							ļ	
							10 —							-	
		+			•••••		~ ~ .				,				
							11 —								
							12								
							1 6								
[										r					
----------	------------	---------	-----------------	-------------	---------------	---------	--------	-------------	------------------------	--	--------	---------	--------	---------	----------
		VIV	VL							Boring #:55-52 MW#:	She	et	1	of	1
	<b>y</b> "		•				1	T		Project: SAR Groundwater Recharge	e Opt	mizat	ion St	udy	
						1				Job #: 1343019	S	ite: Sa	oil	bor	nu
							+1	b		Logged By: A. Lean H Revi	ewed	By:			
						5	502 1	7		Drilling Contractor: Boart Longyear	Λ	14	٢H		
										Drill Rig Type/Method: Sonic Drilling	- h	9.20	ls	em/	olin
							C			Drillers Name: A. Lear!	++				
										Borehole Diam./Drill Bit Type:	То	tal De	pth	1	(
										<b>Barran</b> ia	Re	ef. Ele	v.	distop.	
				Site S	Sketc	h Map	)			Sampler Type: hand Sam	101	ine	ż		-
Dep	th to 1	st Wa	ater ( <u>Y</u>	<b>Z</b> ):		Silvers	Tim	e/Dat	e: 💳	6-18-08 Drill Start Time/Date: Drill	Finisł	n Time	/Date	-18-	08
Dep	th to V	Vater	After	Drillin	ng ( <b>V</b>	():	Tim	e/Dat	e: -	Well Completion Time/Date:					
Dep	th to c	other \	Nater	Bear	ing Zo	ones:		•	· ·	Soil Boring Backfill Time/Date: 6-	8-0	28			
<u> </u>			Ŀ.	ysis	j							Estir	nated	l % Of	 F
	ral	(ju	s / 6 i	Anal	8 S S	ъ		ype					Sand		
A/	e Inte	ered	ounts	ed for	Type	s Fill	(Feet	Soil 7					L C		
)0/0	ample	ecove	ow C	etaine	asing	nInuu	epth	SCS	Soil Descr	iption	ravel	oarse	ediun	ne	It/clay
<u> </u>	S.		, <u>,</u>	Ť	0	Ā	Ō	) ) )	Col		σ	Ŏ	Σ	Γ	- iii
	X	.X						pw	Jando Va	rell graded in 199 gravel,	15	20	25	30	10
	/'						1		any le	1 feldsofer Impa mick					<u> </u>
					·	4			chips h	pto 6", rounded to					
							2-		angular,						
					ļ		3 –					ļ			
									- fi31	3ª surface with					
							4 —	-	Plant	roots					
								1	<u> </u>	and he die a let					
•				•			5		+ 3 a/+	doce hard compacted					<b> </b>
					·				<i>w</i> 1, ( <i>c</i>	neser F prand					
			·····				6		GPS "C	202 - N 340 06.086					
								1		W 117°06.904					
							8 —		*	• .					
							9 —			• • • • • • • • • • • • • • • • • • • •					
					<u> </u>					·····					
<u> </u>							10 —								
 		 .:													
							11								
							12								
	musiannamm						· ••••								QA/QC

_		`								Device # (Cra7 NAME	0			- 6	4
(H	Ð	VIV	VH							Builing #		mizet	ion St	udv	1
									Ē I	Froject. SAR Groundwater Recharg				auy .	
						/	/			JOD #: 1343019	<u> </u>		211	wurn	060
								[		Logged By: A. LENT Rev	iewed	By:			
				and the second se	and the second s	and the second se		1		Drilling Contractor: Beart Longyear	N				
				- and the second second	and the second se					Drill Rig Type/Method: Sonie-Drilling	r h	and	54	mp	ling
						650	3 🖌			Drillers Name: /+. Leavit	<u>+</u>				
						_	~			Borehole Diam./Drill Bit Type:	10	tal De	eptn	1	
						(					Re	91. EIG	۷.	Distance	
				Site	Sketc	h Map				Sampler Type: hand Samp	<i>el:</i> 1	5	k -	16-0	<u>5</u> 0
Dept	h to 1	st Wa	ater (	<b>∀</b> ):			Tim	e/Dat	ie: 🛰	Drill Start Time/Date: Drill	Finisł	n Time	e/Date	);	
Dept	h to V	Vater	After	Drillir	ng (👿	(): 🛰	- Tim	e/Dat	e: <b>`</b>	Well Completion Time/Date:	California				-
Dept	h to c	other V	Nater	Bear	ing Zo	ones:		•	-	Soil Boring Backfill Time/Date: 6 -	18-0	18			
			3 in.	alysis	Size			6				Estir	mated	% Of	
	iterva	d (in.)	nts / (	for An	/pe &	iller	iet)	Typ					Sand		
AVO	iple Ir	overe	/ Cou	ined 1	ng Ty	ulus F	th (Fe	S Soi	0.15		ē	Se	ium		lay
/ald	Sam	Rec	Blow	Reta	Casi	Anni	Depi	nsc	Soil Descr	iption	Grav	Coai	Med	Fine	Silt/c
	$\mathbf{N}$	$\boldsymbol{\lambda}$						su	Sand v	ell graded with gravel,	2	24	2-	25	10
	$\bigtriangleup$	( )					1 —		diy, Ye	Marish biow 1, Wose,	10	10			: 
								-	guartz/	feldspar (mice, rock					
					-		2 —		Chips u	prio 7ª, argular to					
									TOUNACO		-				
							3 —	1	-first 3	3" surface with		<u> </u>			
								]	Planti	route					
					 				-surfa	ce hard compacted					
							5 —		vina	lesert plants					
							Ţ -		00000	12 1/20006 026	-				
							6 —		~p> 0	1 - 1170 0 6 904		<u> </u>			
				<b> </b>	<b> </b>					V 11/ UB. (01					
				· · · ·		<u> </u>	7 —				-				
							Q								
						<b> </b>	9 —				<u> </u>	ļ	ļ		
				 	<b> </b>						-				
							10 —								2.00
					<b> </b>										
1	1	1		<u> </u>		<u>  </u>	11 —								
											1			. 1	
											1				

<b>Г</b>										T	1				
		N <i>I</i> N						$\cap$	\	Boring #:5504 MW#:	She	et	1	of	1
	יש	WIW	WBB					P		Project: SAR Groundwater Recharge	e Opti	mizat	ion St	udy	
								0		Job #: 1343019	Si	te:So	il	Sorr	ซป
							K	N		Logged By: A. Lean't + Revi	ewed	By:			
							ľ	d		Drilling Contractor: Boart Longyear	м	wi	4		
								3		Drill Rig Type/Method: Senie Drilling	ha	nd :	5e.~~	plin	۱ş
			-	القتابيين والقاري	Contraction of the local division of the loc		2	/		Drillers Name: A. Leavit	han				<i></i>
										Borehole Diam./Drill Bit Type:	То	tal De	pth	11	
							550	,4		-	Re	f. Ele	v.	<del>-</del>	
				Site	Sketc	h Mac	XX	-		Sampler Type: head Same	21:1	15	á.		
Dep	oth to 1	st Wa	ater (*	☑):		•	 Tim	e/Dat	e: -	Drill Start Time/Date: Drill	Finish	Time	6-1 Date	8-0	r fr
Den	oth to \	Nater	After	Drillir	na ( 🐨	·):	. Tim	e/Dat	e: 🕳	Well Completion Time/Date:			· Daile		
Den	oth to c	other \	Water	Bear	$\frac{1}{100}$	nes.			<u> </u>	Soil Boring Backfill Time/Date:	5-0	5			
				sis	8						s	Fetiv	nated	1% 01	
	val	(i	/ 6 in	Analy	& Siz	_		Vpe					Sand	/o Ui	 
A	Inter	red (i	ounts	d for	Type	Fille	Feet)	Soil T					Sanu	<u> </u>	
No c	mple	cove	N N N	taine	sing	snlnu	pth (I	CCS S	Soil Desci	ription	avel	arse	dium	e	t/clay
ā	S	Å,	ă	Ъ <sup>в</sup>	<del>S</del>	An	ă				ອັ	ပိ	ž		1 I I I I I I I I I I I I I I I I I I I
	$ \chi $	[ <u>)</u>			<b></b> -			hr	Sand W	ell graded with gravely	25	20	20	25	10
	$  \land$	$( \land )$					1	-	ary, per	1 Seid cases 1 20050,					
<b> </b>									Chills 1	in to Y" angular to					
							2	1	wunde	1					
							 o 2		*						
								].	-fiat	3" surface with					
							4		plant	ruuts				ļ	
							·		-curfa	re hard compacted					
							5 —	4	with	desert plants				<u> </u>	
<u> </u>						<u> </u>	6 —		-c PC:	ADY NOU DE ICI					· .
<b> </b>									<u> </u>	V117º06.888					
<b> </b>	<b> </b>						7 —	1	· .						
<u> </u>						1	·		· · · · · · · · · · · · · · · · · · ·						
							o —	]							
ļ						ļ	9-							ļ	
<b> </b>	<b>.</b>				<b> </b>										
<b> </b>				<u> </u>		<u> </u>	10	-	۳.					<b> </b>	
				<b> </b>						· · · · · · · · · · · · · · · · · · ·					
	<u> </u>						11 —	-							
<b> </b>															
<u> </u>	L	L	L	I	I	I	12 —	1		· · · · · · · · · · · · · · · · · · ·	L	L	I	L	QA/QC

(		<b>VIV</b>	VH							Boring # Boring # MW#:	She	et	1	of	1
V	<b>•</b>							0	Δ.	Project: SAR Groundwater Recharg	e Optii	mizat	ion St	tudy	
	/			5	¢		$) \cap$	11	1) 1.	Job #: 1343019	Si	te: Se	<i><b>sil</b></i>	burr	w
1				1	$\frac{1}{2}$	A	11			Logged By: A. Leavitt Revi	ewed	By:			
1				5	2 (	$\Pi$	1 /(			Drilling Contractor: Beart Longycar-	N	14	14		: 
				)	K.	et	Ľ	V	K .	Drill Rig Type/Method: Sonic Drilling	- he	ind	sa.	mpl.	'nz
1					X					Drillers Name: A. Leavit	ł				
/	$\langle \rangle$			5	555					Borehole Diam./Drill Bit Type:	To	tal De	pth	1.1	1
			Teacher States and States and		~						Re	f. Ele	V.	~	
				Site	Sketcl	h Map	)			Sampler Type: hand Sam	pl:	29			
Dep	th to 1	1st Wa	ater (	<b>Z</b> ):		<b>6</b> 0000	Time	e/Da	te: -	Drill Start Time/Date: Drill	Finish	Time	6- Date	-18 <sup></sup> 9:	08
Dep	th to V	Nater	After	Drillir	ng (👿	): *	∽ Time	e/Dat	le: -	Well Completion Time/Date:	-				
Dep	th to c	other \	Water	Bear	ing Zo	ones:		-		Soil Boring Backfill Time/Date: 6 -	18.	-08	1		
			. <u>c</u>	alysis	Size							Estir	natec	1%0	1
	terval	1 (in.)	its / 6	or An	oe &	ller	et)	Туре					Sand		
AVC	ini əlc	verec	Cour	ned fi	IVT Br	Ius Fi	h (Fet	S Soil			<del>a</del>	es	Ę		ay
PID/C	Sam	Reco	Blow	Retai	Casir	Annu	Deptl	USC(	Soil Desci	ription	Grav	Coar	Medi	Fine	Silt/cl
	$\mathbf{N}$	$\mathbf{N}$						hn	Sandw	ell graded with gravel,	20	20	25	25	1
	$\square$	$\square$					1 -		dry yel	la ish brown, loose,					
									2Martz	foldsparlmica, rock	i a disperi				
							2 —		Chips u	pto 1, angular to					
									rounde	2 <b>271</b>			19. s. 3		
							3-		~first	2" Surface with					
									0151t	roots.					
							4-		-surfe	ce hard compacted					
						i jan Kasi	5 —		wim	desert plants					
										ga <sup>1</sup> 1 ga o ga ta ta ga ya tao 1999 a. Manazarta					
							6-		A 01:	N - 1/2 0					
									6-ps.	1 1170 DL 901					
							7-			~ <i>11)</i> 00, 199					
					na iv Vesi		8-								
							9								
							10					ga Sa Marata Marata			
							11								
	1.000	1					12		11월 22일 - 김 사는		1	l de	1.1.1	1	

	D .		VL							Boring #: \$\$ 06 MW#:	She	eet	1	of	
U	ש	IVE W						$\sim$	2 2	Project: SAR Groundwater Recharge	e Opt	imizat	ion S	tudy	
	/			~	<u>, 1. 1.</u>	میں میں	A	$\left( \right)$	////	Job #: 1343019	S	ite: <b>S</b> a	sil L	sorr	00
1	(				3	(	(	][	1(-)	Logged By: A.Lean'H Revi	ewed	By:		· · · · · · · · · · · · · · · · · · ·	
1					107	((	Y I	/	$\mathcal{M}$	Drilling Contractor: Beart Longycar	м	w	4		
				1					$\overline{}$	Drill Rig Type/Method: Sonic Drilling	ha	nd.	sa ~	rpl	ing
					J č.	2.1			1	Drillers Name: A. Leari	14				
				J		566				Borehole Diam./Drill Bit Type:	То	tal De	epth	t '	¢
	$\bigcirc$				$\lambda$		$\leq$		A		Re	ef. Ele	V.		<u> </u>
				Site	Sketc	h Map	) )			Sampler Type: hand Sam	pl	ng	<u> </u>		
Dep	th to 1	1st Wa	ater ( <u>)</u>	<b>Z</b> ):			Tim	e/Da	te: 🔶	Drill Start Time/Date: Drill	Finist	n Time	e/Date	78- ):	08
Dep	th to V	Nater	After	Drillir	ng (👿	): <u> </u>	Tim	e/Da	te: 🖌	Well Completion Time/Date:			197 		
Dep	th to c	other \	Water	Bear	ing Zo	ones:		<u> </u>		Soil Boring Backfill Time/Date: 6-	18-	.08			
			6 in.	lalysic	Size			Ð				Esti	mated	1%0	f j
	nterve	d (in.	ints /	for Ar	ype &	Filler	set)	il Typ					Sand	T .	
AVO/	nple I	covere	w Cot	ained	sing T	l sulur	oth (F	CS Sc	Soil Descr	intion	vel	arse	mil		clay
<u> </u>	Sar	Це Ч	Blo	Ret	Cai	Anr	Del	ns N			Gra	Ö	Me	iii iii	Silt
	-X-							bu	Scridel	1 graded with gravel,	25	20	20	25	D
		$\land$					1		aug jelle	Fold-Bar Imira court					
									Chips no	to 2" angular to				nter Anton Anton Antona	
									rounde	d - very compacted					
							3-		and har	I to get through				n de la constante de la consta	
										en la constante de la constante La constante de la constante de		- 42			
						전 24년 2년 - 일문	4		- Surta	ace is gravelly, hard					
	522224		ander Aprili						Costp	cycu <sub>4</sub>					
							5-								
							6-		LPS: C	06 - N34005,926					
				دینی: معالمات		1) 1) 1) 1) 1) 1) 1) 1) 1) 1) 1) 1) 1) 1				W 1170 07,002					
							7-						1944 1947 1947		
		<u>1997)</u> Sec. 3		1 5 4  						<u>2011 – Paulos Princerse, se Consel</u> La se a districto da sub substantes					
							8 —								11.545 1.658
							9								an a
				eren er Sterenen			10					ļ			
							11								
							1000 <u>88</u>   10088			er en					
신문화제	the second s	and the state of the	1	1.2 1.1 1.1		L - 1 - 1 - 1 - 1		1 - 11 11-5	<ol> <li>Contraction (1997)</li> </ol>		1 C - 1 C - 1	1	1.1	1 1.1.1	

G		MV	VH							Boring #:	<u>&gt;&gt;U/N</u>	лW#:		She	et	1	01	
	<b>V</b>							T	ΝΛ	Project: S	AR Ground	dwater Re	echarge	Opti	mizati	on St	udy 0	
1						X	af			Job #: 13	43019		- 1.5 	Si	te: Se	211	201	run
						X	I	V	(/)	Logged B	<u>y:</u> /4. <i>le</i>	avitf	Revie	ewed	By:		<u>.</u>	<u></u>
						X			-	Drilling Co	ontractor:	Boart-Lon	gyear	MI	NH			
						x 5	507		)	Drill Rig T	ype/Metho	od: <del>Senic</del>	Drilling	ha.	nd s	6	rp4	ng
					1	X				Drillers Na	ame: A	. Lea	.v.; t	-1-				a gʻalat Afrika A
					'n	$\sum_{i=1}^{n}$		/		Borehole	Diam./Drill	Bit Type:		То	tal De	pth	<u>. 1 '</u>	/
					_	$\geq$	$\bigcirc$						<u> </u>	Re	f. Ele	V.	~	
				Site	Sketc	h Map	)		4 4	Sampler 7	Type: ha	nd S	amp	oli	ng		- 2-2	
Dep	th to	st W	ater (	⊻):			Tim	e/Da	te: ∽	Drill Start	Time/Date	-18-08	Drill F	-inish	Time	/Date	:	08
Dep	th to V	Nater	After	Drillir	ng (🔽	):	Tim	e/Dat	e: 丶	Well Com	pletion Tin	ne/Date:						- ( 
Dep	th to o	other '	Water	Bear	ing Zo	ones:		<u></u>		Soil Borin	g Backfill T	Time/Date	: <b>6</b> - (	8-	08	N., Maria		
			in.	alysis	Size			6							Estir	nated	1%0	
	terval	d (in.)	nts / 6	or An	pe &	iller	et)	l Type								Sand		
ANO	ple In	vere	Cour	ined f	ng Ty	Ilus F	h (Fe	S Soi						0	es	Ę		lay
PID/	Sam	Reco	Blow	Reta	Casi	Annt	Dept	nsc	Soil Desci	iption				Grav	Coar	Medi	Fine	Silt/c
	$\overline{\mathbb{N}}$	ſγ						a	sand ne	11 grade	2dwis	harco	el,	15	25		3	ia
	IД	$\square$							dry, Ye	110013	n brun	1/91	ey 133	ØØ	20	15		
									nown	Noose	, 2 na	1721	/					
							2		feldsp	hr / m	1,22	<u>rvck c</u>	his	,				
									up to	i jon	palsa	to rov	ndel					2 
							3-		Ladia	Than	<u> </u>					i de Para Referencia		
i u sruči Solovija Solovija									<u> </u>	.1.11.2	<u> </u>							
							4-	]										
							5 —		-Snrf	rce h	ard li	comp.	ci fel					
											49 							
							6		0 10	~ ~ ~	A /	- 911	11	1.				
	-								10-12 ic	10/ - /	1/11-0 -	rs $ls$	b //					
							7 —				<u>v 11 / 0</u>	1.01	na se	<u>an sta</u> National				
				<b> </b> -						alde Judeeld Saa	· · · · · · · · · · · · · · · · · · ·						1	
							8-											
																	i an chuir Tha chuir Al chuir an chuir Tha chuir an chuir Tha chuir an chuir	
							10 —					1						
		1000																
		11 1 1 1 1 1 1 		1	1												-	
							11 —				s	n Na al Tari	jač L					

1	M -									Boring #:55 - 08 MW#:	Sh	eet	1	of	
IJ	<b>V</b>	IVI V	VFI							Project: SAR Groundwater Recharg	le Opt	imizal	tion S	tudy	
	F			1999-140 - 140 - 140 - 140 - 140 - 140 - 140 - 140 - 140 - 140 - 140 - 140 - 140 - 140 - 140 - 140 - 140 - 140			~ ~			Job #: 1343019	S	ite: 🖌	oil	Bo	m
	1						$\wedge *$	Λ	A AL	Logged By: A. Lean H Rev	iewed	By:			
	1					~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	XX			Drilling Contractor: Boart-Longyear-	· N	nw	H		
						o~- ۲۰۰0	<b>6</b> 87 Y			Drill Rig Type/Method: Sonie Drilling	ha-	d s	Sm	pli	15
	1						ĹΧ	V (	VV	Drillers Name: A. Leavit					
	l						Y			Borehole Diam./Drill Bit Type:	Тс	otal De	epth	1'	
	``										R	əf. Ele	€V.		
				Site	Sketc	h Mar	)			Sampler Type: Land San	rol	115			
Dep	oth to	1st W	ater (	Z):		·	Tim	e/Da	ite: -	Drill Start Time/Date: 06-(8-08	Finis	ר ר Time	e/Date	-18- 9:	-20
Dep	oth to '	Water	After	Drillir	ng (👿	(): ~	- Tim	e/Da	te: 🔦	Well Completion Time/Date:					
Dep	th to (	other	Water	Bear	ing Zo	ones:				Soil Boring Backfill Time/Date: 6-	18-	08			
			Ŀ	lysis	eze							Esti	mated	1%0	f
	arval	(iu)	s/6	r Anal	e & S	er	()	Type					Sand		
٨	le Inte	ered	Count	led for	g Typ	III SL	(Feel	Soil -				Ð	E		2
	Samp	Reco	Blow	Retair	Casin	Annul	Depth	nscs	Soil Descr	iption	Grave	Coars	Mediu	Fine	Silt/cla
	Ι <u>Υ</u>	$ \chi $						Sh	Sand	ell staded with stall	15	75	25	25	105
	$\frac{1}{1}$	<u> </u>					1		dry, fel	lowish bown / lighthown,					
	<b> </b>								luck ch	narte/telds/ar/mica,					
							2 —		LU ON	10 up to 1, ungular					
				ini ini Segre								·			
							3		-first 2"	3" plant routs		·		- 144 144	1 
							4 —		-surfac-	e have compacted					
									wim.	desert plants					
							5 —								
									C DC' MU	8. N/74° A6 12421					
							6 —		0-12-0-	W 1170 07. DI91					
							8 —							1947) 	
					1										
	• • • • • • • • • • • • • • • • • • •				and an a		9 —								
		1.122	10 C 10 C							<u>na para series de la companya de la compa</u> nya de la companya de la companya de la companya de la companya de la Esta de la companya d					
				iridi Secol								•	1 1 1 1 1 1 1	1.12.12.11	1. 1. 1.
				1111 <u>Anii</u> 1111			10 —								
							10								
							10 — 11								

						<u>,</u>			· ·						
		MV	VH							Boring #: >>=O (MVV#:	Sne	er	1		<u> </u>
				Sector and the sector of		g karang sa karang k	and a second			Project: SAR Groundwater Recharge	e Opti	mizati		udy	
						X	er l	$\int \Lambda$	$  \cap  $	Job #: 1343019	Si	te:Sø		Buri	2
				53	509,	X	Mil			Logged By: A.Lew. H+ Revie	ewed	By:			
						X	(14)	)V		Drilling Contractor: Boart Longycar-	N	w	H		
		4				X	$\sim \iota$	/		Drill Rig Type/Method: Senie Drilling	ha	nd	68-	pli	rj
						×			/	Drillers Name: A. Leavit,	4				
						X				Borehole Diam./Drill Bit Type:	То	tal De	pth		
											Re	f. Ele	v.	-	
				Site	Sketcl	h Map	)			Sampler Type: hand Sam	.pl:	ns			-
Dep	th to 1	st W	ater ( <u>)</u>	<b>Z</b> ):		COMPANY.	Tim	e/Dat	te: -	Drill Start Time/Date: Drill F	- Finish	Time	/Date	18-0	9E
Dep	th to V	Vater	After	Drillir	ng ( 👿	): 🛰	- Tim	e/Dat	e: -	Well Completion Time/Date:					
Dep	th to c	other '	Water	Bear	ing Zo	ones:	5			Soil Boring Backfill Time/Date: 6-1	8-0	8			·
			. <u>c</u> i	lysis	lize							Estir	nated	% Of	:
	erval	(in.)	s/6	r Ana	8	er	¢	Type					Sand		
٨٨	e Inte	ered	Count	ed fo	g Typ	III sr	(Fee	Soil			_	Ð	ε		ay
0/01	ampl	Recov	Blow (	letain	Casing	Innul	Depth	ISCS	Soil Descr	iption	àrave	Coars	Aediu	ine	silt/cla
	$\overline{\mathbf{n}}$	<u> </u>	ш			4		<u> </u>	Cand 1-P	Il cre let i do con el	0	0	2	<u>u</u>	0)
	Ď	-X						Par	der Voll	and brown Sclichtly	15	25	52	25	/8
	,	<u> </u>	·				1-		damp 6	"down loose, guarty					
							2_	].	Feldspa	rs/mila, ruck chips					
		_'				 			up to 1	", angular to subrande	,				
							3 -								
									-surf	rce compacted thed					
							4		win (	algert plands					
						<b> </b>									
							5 —		P-Pero	NG - N34° DG DG9'					
										w 1/7°07,00)'		-			
							6-								
							7_	]							
							8 —			· · ·					
						 	9 —								
								•							
							10								
								1 :							
							11 —	4							
				•••••	*******	*******	12	*****	*****						04/00

đ	Ð	VI V	٧H							Boring #; ్/ల MW#: Project: SAR Groundwater Recharg	She e Opti	eet imizal	1 tion S	of tudy	1
						××××××	sic IC	)0	$\mathcal{D}$	Job #: 1343019 Logged By: A. Learitt Rev Drilling Contractor: Boart Longyear Drill Rig Type/Method: Sonie Drilling Drillers Name: A. Learitt Borehole Diam./Drill Bit Type:	S iewed N I N 4 Z To Re	ite:Se By: Au ad otal De	epth	Borr npl. 1'	ing
				Site	Sketc	h Map				Sampler Type: hand Sam	rels	In	56	-18-	58
Dep	th to 1	st Wa	ater (	<u>Z):</u>			Tim	e/Da	te: 🔽	Drill Start Time/Date: Drill	Finish	n Time	e/Date	9:	
Dep		vater	Mator	Drillin	ig ( 🕎	):	l Im	e/Da	le: <b>-</b>	Vvell Completion Lime/Date:	- U 13	TC -			
Peb				is is				<u>~</u>			8-0	<b>Ø</b> Esti	mater	1%0	f
	arval	(in.)	s / 6 ir	r Analy	e & Si	er	ĥ	Type			· · · · · · · · · · · · · · · · · · ·		Sanc	<u> </u>	1
PID/OVA	Sample Inte	Recovered	Blow Count	Retained for	Casing Typ	Annulus Fill	Depth (Feel	USCS Soil -	Soil Desci	ription	Gravel	Coarse	Medium	Eine	Silt/clay
	Х	X						sv	Sand (gra Grave	relly) wells raded with	30	25	20	20	5
		 					 2 		Louse, 9 Gray Lar angular	ruck chips up to 7"					
							3  4		-surfa with	ce hard (compactal, desort plantscm.nor)					
							5		A muc large g	h more sravel, especk revel, Man Die rest	φ				
							6 —  7 —			<i>,                                    </i>					
							 8		6-P5:01	0 - N 39°06,145' V 117°06,991'					
							9 —								
							10 —								
							11 — 								

	1					
	Boring #: 55 // MW#:	Shee	et 1	0	of	[
	Project: SAR Groundwater Recharge	e Optin	nizatio	n Stu	ıdy	
× 55.11 +	Job #: 1343019	Sit	e:Soi	:I Ø	orr	<u>v</u> ~
	Logged By: A.Lean! H Revie	ewed E	By:			
	Drilling Contractor: Boart Longyoar	Мц	νH	!		
	Drill Rig Type/Method: Senic Drilling	hane	d sa	. m/	olN	25
X	Drillers Name: A. Leav. +	F				
	Borehole Diam./Drill Bit Type:	Tota	al Dep	oth	1	
		Ref	. Elev		-	
Site Sketch Map	Sampler Type: Usad Samp	olin	٢			
Depth to 1st Water (文):	6-18-08 Drill Start Time/Date: Drill I	Finish	/ Time/l	6-1 Date:	8-0	S
Depth to Water After Drilling ( <b>V</b> ): — Time/Date:	Well Completion Time/Date:					
Depth to other Water Bearing Zones:	Soil Boring Backfill Time/Date: 6-1	8-0	8			
		-	Estim	ated	% Of	
(in.)			S	and		
A A A A A A A A A A A A A A A A A A A		-		г		ح
Description SC and an Urr as a start of C A A A A A A A A A A A A A A A A A A	ription	ravel	oarse	lediur	ne	ilt/cla
	Il a lade the an as	0	0	$\frac{2}{25}$	<u>п</u> 25	<u></u>
X X ALL DU STAR WE	11912 deal 10 gravely	20	25	Se l	14e	5
1 anertz	1 foldstars / m22, cock				·····	
Chips u	nto 6", angular to					
- Founda	A					
- surf	ace hard / compacted					
4	minur desert plents					
5- 6-15:0	11 - N 34806,146'					
	V 117° U7.059'					
7						
┠┞┞┞┞┞  ·  ·						
8-						
┠╌╍╍╄╍╍╍╄╍╍╍╄╍╍╍╄╍╍╍╿───┤ ┝╍╍╍╍╍╍╍		.				
9-1						
┠┠┠┠┠┠┣	· · · · · · · · · · · · · · · · · · ·					
			·			
·   ·   ·   ·   ·   ·   ·   ·   ·   ·						-
						QA/QC
·	· · · · · · · · · · · · · · · · · · ·		And South States			
	an a		. 1 . 30			

(	i) I	<b>M</b> V	VH		Baikuta					Project: SAR Groundwater	Recharge	e Opti	mizat	ion S	tudv	
		<b>()</b>	960 E	<b>1</b> 47				$\overline{\gamma}$	$\overline{}$	lob #: 13/2019		C	ito: C	a. 1	Bar	(1)h
	1				X	1	211	111		Job #. 1343019	1 Davi	0 	D			/ <b></b>
	1			55-	12p	Ű	(J)	$\left \right $	$\mathcal{I}$		* Hevi	eweq	Бу: с //			
					Ţ		Ň		)	Drilling Contractor: Boart L	ongyear	Mh	14	1 -		<u></u>
					5					Drill Rig Type/Method: Sen	e Drilling	- n 1	and	n si	any	Min
					7	ζ,				Drillers Name: 17 - 64	carit	7			. /	r r
					/	R.3				Borehole Diam./Drill Bit Ty	<b>De:</b>	To Re	f. Ele	v.	<u> </u>	<u>- 1877</u> - 1873
		$\searrow$		Site	Sketcl	h Map				Sampler Type: hand	samp	lin	ý			
Dep	oth to	st W	ater (	<b>∑</b> ):		Correction,	Tim	e/Da	te: -	Drill Start Time/Date:	08 Drill I	Finish	Time	/Date	-18. 9:	-08-
Dep	oth to \	Vater	After	Drillir	ng (👿	):	Tim	e/Da	e: ゝ	Well Completion Time/Date	):					
Dep	th to c	other	Water	Bear	ing Zo	ones:				Soil Boring Backfill Time/D	ate: 6-4	8-0	28		en de la composition de la composition En esca esca de la composition de la comp	2012 2013 2014 - 2015
			<u>.</u>	ysis	eze					• •			Estir	nated	1%0	If
	rval	in.)	s / 6 ii	Anal	& Si	3r	_	ype						Sand		
۲A	e Inte	ered (	ounts	ed for	Type	s Fill	(Feet	Soil 7								
PID/O	Sample	Recov.	Blow C	Retain	Casing	Annulu	Depth	USCS	Soil Des	cription		Gravel	Coarse	Mediur	Fine	Silt/cla
	$\lambda$	X						sv	send w	ellgraded withgra	rel,	20	20	25	25	10
	$\mu$	( )					1-		aug te	1 Coldelore / airo	<u>5x,</u>					
					;-				chips.	pt 2" snewker	r ceg					
						<u>1912 - 19</u> 1914 19	2 —		round	el je signed	0				<u> </u>	25
										<b>F</b>						
				1997) 1998					-Shrefa	ce hardand com	Acctes				19,79	
							4 —						ing di Disent			
							912, 33 71, 32, <u>94</u> 71, 32, 94, 94									
							5 —		C-PS.C	NZ N34 06,0	289					
					na de persona de Caractería de Caractería Caractería de Caractería de					w11/00,c	56					
							6 —						ng Parto sa			
						1993 -	7 —									
				linin Oya d												
					1 11 - 1 - 1		8									
							° –									
							J 									
							10 —									
							11						the second s Second second s Second second s			
					<u></u>											
				1			10		は影響になる人							

G		VIV	VH							Boring #: <u>513</u> MW#:	She	eet	1	of	)
			10000000000000000000000000000000000000					- and a state of the		Project: SAR Groundwater Recharge				uay Cara	
	$\left( \right)$				2	5	Cal	$\sum$	501	JOD #: 1343019	5	ITE: 24	9/1	Dor	~0~
						٢	All	Λ		Drilling Contractor: Boart Lenguear	ewea N	By:	11		
				Ŷ	-13)	Ŕ	a			Drill Big Type/Method: Sonic Drilling	- h	end	57 Ea	. 01.	n c
					2	<				Drillors Name: A. L. P. L. P. L.	1		200	100	2
					Ż	ς ς				Borehole Diam./Drill Bit Type:	То	tal De	epth	11	
						, 02				enter and a second s	Re	ef. Ele	v.		
			1000 Carlos C	Site 9	Sketc	►∕ h Mar	1			Sampler Type: Jac of Co.	ol Co	e			
Dep	th to 1	st Wa	ater (N	<u>√):</u>			Tim	e/Da	te:	Drill Start Time/Date: Drill	Finish	J Time	6. /Date	-18-	- 58
Dep	th to V	Vater	After	Drillir	na (🔽	·): •	Tim	e/Dat	te:	Well Completion Time/Date:	1 11101	1 1 1110	, Daic		
Dep	th to c	other	Water	Bear	ing Zo	ź ones:			~	Soil Boring Backfill Time/Date: 6-0	8.0	54			
· · ·			-	ysis	ize					L	ľ –	Estir	mated	1%0	f
	erval	(in.)	s/61	r Anal	e & S	ler	÷.	Type					Sand		
A V	le Inte	/ered	Count	led fo	g Typ	us Fil	(Fee	Soil			-	0	Ε		ay
D/OII	Samp	Recov	Blow (	Retair	Casin	Annul	Depth	nscs	Soil Descr	iption	Grave	Coars	Mediu	Fine	Silt/cl
	$\overline{\mathbf{N}}$							su-	send u	ell graded wongravel,	15	20	25	30	6
	$ \Delta $	Д.			<b> </b>		1-		dry, Yell	10-134 brown, 10050,	Ľ				
									quartz, f	eldspar, mica, ruck					
		<u> </u>					2		Valina la	prog, angaro de					
									<u>nowner</u>						
							3	]	-surf	nel herd but not					
							4		as con	npactod as the other					
									holes 1	mus far (rows 1, 2, s)					
							5 —		Metis	+ holes in rows).					
					<b> </b>										
							6-		CPS: 0	13 - N24° 66,030'					
							7_			W117 07,067'					
<b> </b>					L			-							
							8-								
					<u> </u>	<b> </b>				·····					
					<b> </b>		9-	-			<u> </u>	+	<u> </u>		
					<u> </u>					·····					
							10	]							
<u> </u>							11								
					<b> </b>										
				L	L		12	L				 	L	l <u>.</u>	OA/OC

	۱ U	M	VL							Boring #;5519 MW#:	She	eet	1	of	<u> </u>
U	יש									Project: SAR Groundwater Rechar	ge Opt	imizat	ion S	tudy	
		/				٢		$\cap I$	171	Job #: 1343019	S	ite:Se	pi l	Bur	v
	/				X	ŕ	ffil	1	$\mathcal{N}$	Logged By: A. Leanst Re	viewed	By:			
	1				λ		M			Drilling Contractor: Boart Longyear	1	nu	1H		
	1				2	K cc.				Drill Rig Type/Method: Sonic Drillin	g kas	nd	se	NO	1,19
					X	53-	-17			Drillers Name: A. Leavis	-}		i USBN		
					X	- - -	/			Borehole Diam./Drill Bit Type:	Тс	otal De	epth	1'	
					-						R	əf. Ele	₽V.		
	$\sim$			Site	Sketc	s h Map	) )			Sampler Type: hardsen	Mi.	25			
Dep	th to 1	lst W	ater (	Z):	. 1992	<u> </u>	Tim	e/Da	te: —	Drill Start Time/Date: Dri	/	Time	6-1 -/Dat	8-0	58
, Dep	th to V	Nater	After	 Drillir	na (👿	): •	— Tim	e/Da	e:	Well Completion Time/Date:			<i></i>		
Dep	th to c	other	Water	Bear	ing Zo	ones:			an a	Soil Boring Backfill Time/Date:	18-1	oy			
			i e	ysis	ize				an an tao an tao an Tao amin'ny faritr'o amin'ny faritr'o amin'ny faritr'o amin'ny faritr'o amin'ny faritr'o amin'ny faritr'o amin'			Esti	mateo	1%0	f Cal
	ival	(in.)	s/6i	r Anal	e & S	e	C C	Type					Sand	1	
٨A	e Inte	ered	Count	ed fo	g Typ	III St	(Feet	Soil -				0	ε	Ī	- - -
0/01	Sampl	Recov	Slow C	Retain	Casing	Vnnult	Depth	ISCS	Soil Desc	ription	aravel	Coarse	Aediui	ine	Silt/cla
	$\overline{\Lambda}$	5		199 <b>14</b> -999 1989				5	Sand	ell aradel with read					
	X	X							dry Yel	low of bourn, louse,	9.15	25	25	25	10
							1-	]	guar to	2/ feldspar / mila, rock					
							2		CLips 1	pto 2", angular to					یے۔ 12 میں
							÷		round	ed	-				
							3	4						-	
								-	Surfa.	ce is hard and comfact	e/,				
							4		<u>un's</u>	one absert plants					
	<u>Pidiri</u> Ingled	<u>aladi</u> Mari								<u>de l'Estato de l'Arte d'Arte d'Arte (1997)</u> A l'Arte de la Constante de la C	-				
							5 —		C.PS:	014 - N34°m5,9170'					
										W11700,075"					
							6	]							
							7-								
											-				
							8 —							<u> </u>	
										<u>en e Berlen de Berlen.</u> En en en Berlen de B	- - - -				
				janes Sata			9 —								
											-				
							10 —	1	in the second						
	17 Augusta	i. Esta ess													
													1.1.1		1

Л	M.	<b>N/1</b> 1/			inger (	4 (44 11 (44)				Boring #:55-15 MW#:	She	eet	1	of	1
V	יש	IVIV	a Li			RR				Project: SAR Groundwater Recharg	e Opt	imizat	ion S	tudy	
	1					<u> </u>		~	$\overline{n}$	Job #: 1343019	S	ite: Se	>11	Bor	m
	1					s -	(fr	$\square$	()()	Logged By: A. Lear. 7+ Rev	iewed	By:			
					1	$\overline{\Lambda}$		V	VVJ	Drilling Contractor: Boart Longyear	Mu	」好	n Kiron Alan N		
					<	Х	14	Z	/	Drill Rig Type/Method: Senic Drilling	ha.	rd s	59 ~	npl.	ing
					/	x		1		Drillers Name: A. Leavi	r7				
					A 7	<b>K</b> SS <	-15	/		Borehole Diam./Drill Bit Type:	To R	otal De ef. Ele	epth		1
	$\checkmark$			Site	Sketc	h Mar				Sampler Type: head Se	PL	16			
Den	th to <sup>-</sup>	1st W	ater (				Tim	a/Da	he: 🖛	Drill Start Time/Data	Finia	J.	6-	18-	- 4
Den	th to V	Water	After	<del>ير ان</del>	na ( <b>w</b>	<u> </u>	Time	a/Da	···	Well Completion Time/Date:		a (IM)	Jate	*•	••
Den	th to c	other	Water	Bear	ina 70	ones:		., u		Soil Boring Backfill Time/Date:	18-	08			
				/sis	en N					Loon Doning Daokim Title/Dale. 0		Feti	mater	1%0	f i se
	rval	in.)	; / 6 ir	Analy	& Si.	<del>۲</del>		ype			-		Sand		•
۲¥	e Inte	ered (	ounts	ad for	Type	s Fille	(Feet)	Soil T							
	Sample	Recove	Blow C	Retaine	Casing	Annulu	Depth	nscs	Soil Descr	ription	Gravel	Coarse	Mediun	Fine	Silt/clay
	Х	X						50	Sand well	Usraded with sravel, (lowish brown, louse,	N	25	25	25	10
							 		quartz chips of	1 feldspar 1 mica, rock					
									ronnde.					12 July -	
							3-		surfac	ce hard/compacted,					
							4								
							5 —		0-85'01	15 - N74° DE 909'					
·					1 1					W117º 07.082					
							6-								
							7 —								
			n Ar Beler Arton A									۰ ۲۰۰۱ میں ۲۰۰۱ میں ۲۰۰۱ میں			
				ander de Steansel			8 —		na na serie de la composición de la com Esta de la composición						
							9-								
							10								
							10								
							11 —								
							1. 								
1656	and the second	1. A 199		<ul> <li>A 10 A 10 A 10 A 10 A</li> </ul>	<ul> <li>A - 25</li> </ul>	a		10 A. S.A.	10.1 March 10.1 Mar		4.15	a + 4 + 4 + 4 + 4	<ul> <li>1.141</li> </ul>	<ul> <li>• • • • • • •</li> </ul>	

	M.		N/L						Boring #:55-16 MW#:	Sh	eet	1	of	
U	יש	IVI V				יי ייצרי			Project: SAR Groundwater Rechar	ge Op	limiza	tion S	tudy	
		/			<u> </u>	<u></u>		0	Job #: 1343019	5	Site:5	sil	Barr	v
	1				5	Λ K	Ri	1/1	Logged By: A. Leant Re	viewec	l By:			
	1				X	۱. م	AN	V	Drilling Contractor: Beart Longyea	M	WH	•		
					$\lambda$		d a	8	Drill Rig Type/Method: Senie Drillin	g h	and	59-	npl	iA
	1				λ	l Selation La Selation		/	Drillers Name: A. Lear	;+,	L			
			٩	5-11	5 ¥				Borehole Diam./Drill Bit Type:	To	otal D	epth	1	<u>.</u>
	$\sim$				<u></u>	/				R	ef. Ele	€V.		
				Site	Sketc	h Map			Sampler Type: NANA SAN	apl;	ns	- 6-	16	est
Dep	th to	1st W	ater ()	<b>V</b> ):		<u>.</u>	Tin	ne/Da	Drill Start Time/Date: Dr	ll Finis	h Tim	e/Date	• • •	
Dep	th to \	Nater	After	Drillir	ng (🔽	):	- Tin	ne/Dat	:: - Well Completion Time/Date: -					<u></u>
Dep	th to c	other	Water	r Bear ∣º	ing Zo	ones:		-	Soil Boring Backfill Time/Date: 6	18-	08			
	폐	(1	6 in.	nalys	& Size			be			Esti	mated	1%0	)f
4	Interv	ed (in	unts /	d for A	Type &	Filler	reet)	oil Ty				Sand		
NO/al	ample	ecovei	low Co	etaine	asing	nnulus	epth (F	scs s	Soil Description	ravel	oarse	edium	ne	
٩	s N	<u>۳</u>	m		U U	Ā	۵	$\frac{3}{0}$	conderellered with one	0		Σ		
	X	X					-	ľ	dry, followish brown, louse,	-15	52	52	25	1
									Quartz/feldspar/mica, rick	-				
							2 -		chips upto 2", angular to					
									runned					
							3 -		surface hard / compactal.					+
							- -		WM Some desert plants					
							-							
							5 -	-	COCINIL CONPOCE	14 <u>1</u> 4 14 • 1				
							-		WID 010 - N 39 03.831					
							6 -		(11 01 1030					
							7 -							
								-						
1998 3499							8 -	-						
							9 - -							
							10 -	-						L
									<u>er de la Calendaria de Carence.</u> Sector de la Calendaria de C					
							11 -		na transmana a stranda filmana a sing Halibba sa sing Manifesta di santa si kara sa sing sa sa sing s			-		
			the second second	2 1 11			and the second second				1	1 1 1 1 1 1	1. A.	

Sheet Boring #: (54> MW#: of 1 🕀 MWH Project: SAR Groundwater Recharge Optimization Study Ry Site: Soil Borrow Job #: 1343019 Logged By: A. Lean't Reviewed By: Drilling Contractor: Boart Longycar MWH Drill Rig Type/Method: Sonic Drilling band Sampling Drillers Name: A. Learitt Total Depth Borehole Diam./Drill Bit Type: 11 Ref. Elev. Sampler Type: hand camplins Drill Start Time/Date: Drill Finish Time/Date: Site Sketch Map Depth to 1st Water ( $\mathbf{\nabla}$ ): Time/Date: 1 Depth to Water After Drilling (T): Time/Date: 🖌 Well Completion Time/Date: Soil Boring Backfill Time/Date: 6-18-05 Depth to other Water Bearing Zones: **Retained for Analysis** Estimated % Of Casing Type & Size Blow Counts / 6 in. USCS Soil Type Sample Interval Recovered (in.) Sand Annulus Filler **Depth** (Feet) PID/OVA Medium Silt/clay Coarse Gravel Soil Description Fine Sh Sand well graded with soit " dry, yellwith brown, loose Que, tz/feldsper/Miza, rak Chills up to 3", crighter to 25 20 20 5 30 1 -15 25 25 25 10 2 runded - -3 --- After @ 6" decrease in Scavel 4 -- --surface hard (compacted, lots of gravel 5 -6 -6PS:017 - N3405.844' - -W1707.148' 7 -8 -9 -10 -- -11 -12 -QA/Q

G	D I	VIV	VН							Project: SAR Gro	undwater Re	charge	Opti	mizat	ion St	udy	
_	-				C	4				Job #: 1343019		<u> </u>	S	ite:Sc	sil i	Berr	っし
	/					/			N	Logged By: A.	Leaviti	L Revie	wed	Bv:			
					$\frac{\lambda}{\lambda}$		fel	χ	U	Drilling Contracto	r: Beart Long	lyear	N	14	14		
					X		le			Drill Rig Type/Me	thod: Senie I	 Brilling	50	; <del> -</del> @-	<del></del>		151
					X	Cic		/		Drillers Name:	A.Lea	~: t	4				
ł				/	X	<i>518</i>				Borehole Diam./I	Drill Bit Type:	<b>`</b>	To Be	tal De	pth v.	11	
				Site	Sketcl	h Map	)			Sampler Type:	hand s	and	e ( 2	>			
Dep	th to 1	st Wa	ater (	Z):		(Carlos and Carlos and	Tim	e/Dat	e:	Drill Start Time/D	6-18-09 ate:	Drill F	inist	Time	6-6 Date	8-0	sV
Dep	th to V	Vater	After	Drillir	ng ( <b>T</b>	): `	– Tim	e/Dat	e:	Well Completion	Time/Date:						
Dep	th to c	other \	Nater	Bear	ing Zo	ones:		<u>~</u>		Soil Boring Backf	ill Time/Date	: 6-19	§-6	-32			
			<u>c</u>	lysis	ize								**	Estir	nated	% O1	f
	erval	(in.)	ls / 6	r Ana.	e & S	ler	()	Type							Sand		
AV	le Inte	/ered	Count	ned fo	g Typ	us Fil	I (Fee	Soil					Ŧ	e	E		av
PID/C	Samp	Reco	Blow	Retair	Casin	Annul	Depth	nsce	Soil Descr	iption	•		Grave	Coars	Mediu	Fine	Silt/cl
	$\overline{\mathbf{n}}$	$\overline{\mathbf{V}}$						Si	Sand us	ellgradad v	Marak	æ];	1			25	
	$\square$	$\Delta$					1-		dry, yel	lang hour	, louse,		15	25	LS	25	10
									Quartz	/feldspar	5 /mile	×					
							2-		rock M	ips up to	34, 9-19	nla					
									40 100	ndou							
	· · · · ·						3-	1	surface	e hard/1	unpact	al,					
					, ,				withse	Ma desert	plants	/					
							4				, 						
					<u> </u>		5-	4			0						
									LPS: O	18 - N34	- 05 ,11. 07 14	2					
							6	1		~ ~ ///*	0 1419	(					
					h			1									
					<b>_</b>	<b></b>	/-										
					ļ		8-	. 									
					<b> </b>								13				
							9 —	-									
								-	, 	· · · · · · · · · · · · · · · · · · ·							
							10	1		,							
					t	1		1	~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~								
							44			۵							

	Boring #: JS-1? MW#: Sheet 1 of
	Project: SAR Groundwater Recharge Optimization Study
r4	Job #: 1343019 Site: Soil Borrou
X DOD	Logged By: A. Lew, 22 Reviewed By:
× 1/00	Drilling Contractor: Boart Longyear MWH
X	Drill Rig Type/Method: Conte Drilling hand sampling
X X	Drillers Name: A. Leavitt
	Borehole Diam./Drill Bit Type: Total Depth / t
X	Ref. Elev.
Site Sketch Map	Sampler Type: hand Sampling
Depth to 1st Water ( <b>V</b> ): - Time/Date: -	Drill Start Time/Date: Drill Finish Time/Date:
Depth to Water After Drilling ( <b>y</b> ): Time/Date: -	Well Completion Time/Date:
Depth to other Water Bearing Zones:	Soil Boring Backfill Time/Date: 6-18-58
	Estimated % Of
val (in.)	Sand
A A ounts of freed (Feet Solid Solid Solid Solid A Sol	
lioS SCS epth Ulu of C	Description is a set of the set o
	li pillere la manuel
XX	Vollaris La 10000 (2.54) 10 25 25 30 10
Feld	sog-s/ponice, cock chips
	to 4". angular to rounded
3	face hard/compacted,
	M some desert plants
	-019 - N 34 03,976
5-	WII 07.191
6-	
8-	
9-	
<u>↓</u> ↓	

a		MV	VH							Boring #:5520 N	1W#:	She	et	1	of	
~	<i>y</i>									Project: SAR Ground	lwater Recharg	e Opti	mizat	ion S	tudy	i (j. Na
	/				X			$\mathcal{T}$	KM	Job #: 1343019		S	ite:Se	sil i	Bori	~U
1	/				X	•		1/1	()))	Logged By: A. Le	aritt Revi	iewed	By:		- 11	
1					×.	55-2	۵	νı		Drilling Contractor:	oart Long <del>yca</del> r	MI	SH			
					$\sum_{i=1}^{n}$					Drill Rig Type/Metho	d: <del>Sonie Drillin</del> g	ha	nd.	50-	Pl.	ng
					X					Drillers Name: \Lambda .	Leavit	4				9
					X	_				Borehole Diam./Drill	Bit Type:	То	tal De	epth	1'	
										<u> </u>	a digit. Ang ang ang ang ang ang ang ang ang ang a	Re	ef. Ele	ev.	Copin-	
				Site	Sketc	h Map	<b>)</b>			Sampler Type: 🔪	ind sa-	npl	ins		1.4 11.4 14.4	
Dep	th to	1st W	ater (	☑):			Tin	ne/Da	te: 🖌	Drill Start Time/Date	-18-08- Drill	Finist	Time	e/Date	-18-	08
Dep	th to \	Nater	After	Drillir	ng (👿	): <sup>°</sup>	∽ Tim	ne/Dat	te: -	Well Completion Tim	e/Date: -					
Dep	th to d	other '	Water	Bear	ing Zo	ones:		-		Soil Boring Backfill T	ime/Date: 6-	8-0	18			
				lysis	ize					e de la companya de			Esti	mated	1%0	f
	irval	(in.)	s/61	r Anal	e & S	e		Type						Sand		
٨	e Inte	ered	Count	ed fo	J Typ	III SL	(Feel	Soil .					0	ε		
o/Dlo	sampl	lecov	Slow 0	Retain	asing	Innul	Depth	SOS	Soil Desc	ription		arave	Coars	Aediu	ine	silt/cla
L.L.,	N/					4		Cu	Cardin	0112 Fridad de	Vall St			<u> </u>		0)
	X	ŀΧ-		1820-22 19-1953				<b>_</b>	Sown.	woo Qualtz	foldsper	10	20	30	30	10
	ĺ						1 1 -		MERS, (	ock chils up to	$\lambda''$	17. je				
							2-		argula	r to ronally			e de la constante La constante de la constante La constante de la constante de			
							3 -		-surfa	ce hard/com	pactal,					
									w/107 5	une desert p	51k					
							4 -				<u>en en filoso de la composición de la co Esta de la composición de la composición</u>					
							5 -		GPS'D	00 - N240	06.033'					
										~115° C	7.1441	an Ala Ala				
							<b>.</b>									
							7 -						a di sha Ali sha Nasari sh			
							8 -		n and de Marine dae				i succesi.			
							9 -	1							in and and a second sec	
							10 -						10 - r			
							11 -									
							-									
1000004						1. 3.	1.25					1			13636	1

										Boring #25-21 MW#:	She	et	1	of	1
		MV	VH							Project: SAR Groundwater Recharg	e Opti	mizat	ion St	udy	
		and the second statement of the			Ry					Job #: 1343019	S	ite: Sa	510	Surn	<u>u</u>
/					X			λ	NNI	Logged By: A.Lear. ++ Revi	ewed	By:			
/					X ?	5-2		$\mathcal{V}$	(VV)	Drilling Contractor: Boart Longyoar	M	w	41	********	
ľ					$\times$			v		Drill Rig Type/Method: Senic Brilling	ha	nd.	Sai	mpl	ins
$\mathbb{N}$					X					Drillers Name: A. Lear ! +	-1_				
$\left  \right\rangle$					X					Borehole Diam./Drill Bit Type:	То	tal De	pth	1'	>
					X						Re	ef. Ele	v.	Quality and and	
				Site	Sketc	h Mac	,			Sampler Type: Mand Sau	n l	<sup>1</sup> ls n	4		
Dep	th to 1	ist Wa	ater (	Z):		~	Tim	e/Dat	e:	6-18-06 Drill Start Time/Date: Drill	Finist	Time	/Date	18-0	8
Dep	th to \	Nater	After	Drillir	na ( <b>W</b>	();	– Tim	e/Dat	e: 🖌	Well Completion Time/Date:	-		, Duit	·	
Dep	th to c	other	Water	Bear	ing Za	ones:				Soil Boring Backfill Time/Date: 6-1	8-0	28			
			i di	ysis	a s							Estir	nated	% Of	F I
	rval	in.)	s / 6 ir	Analy	& Si	2		ype					Sand		
A	e Inte	ered (	ounts	ed for	Type	s Fille	(Feet	Soil 7					E		~
ID/OI	ample	ecove	O M O	etaine	asing	nlnu	epth	scs	Soil Descr	iption	iravel	oarse	lediur	ine	ilt/cla
	$\sim$			<u> </u>	0	₹			Scada	ellender Margarel		0	2	<u> </u>	м м
	X	X						pω	dry voil	ovity bour 1. loose	15	25	25	25	10
		, ,				1	1-		Quartz	Ifeld spars/ mica, rock					
							2-		chips u	p to 3", angular to	1				
	<u>;</u>			 					rounded						
	j.						3-	4		E L C O Lul		 			
								•	- surfec	e hard (compacted)					
					<u> </u>		4		0.00	me alser plans					
<b> </b>								1	- notici	ns That @ 6"R" if					
							5-	]	Sets ex	Frendly hard and its	- 1 a				
							6-		di ffica	It to get through in					
						4			pidsax	د	-				
							7 -	-		SI A TRUE OF AGI				<u> </u>	
					<b> </b>				60:0	21 - NSY 00,011					
							8 –	1							
						1		1		· · · · · · · · · · · · · · · · · · ·	1				
<b>_</b>							9-	]		·····	-				
				ļ	·		10 -	4			ļ	ļ	ļ		
						<b> </b>					1				
<b> </b>							11 -	4							
<u> </u>															
ļ	l	<u> </u>	l	L	I	1	<sup> </sup> 12 –	<u> </u>	1		<u> </u>	L	L	L	

		MV	VH							Boring #:	She	et	1	of	
~	ν					44)				Project: SAR Groundwater Recharge	e Opti	mizat	ion St	udy	 
	$\bigcap$		•			X:55	-22	$\int$	(AA)	Job #: 1343019	Si	te:Sc	<u>s:16</u>	Seri	~~~~
1					1	x r				Logged By: A . Lear. 17 Revie	ewed	By:	11		
1					1	ŕ		v	Ĵ	Drilling Contractor: Boart Longyear	$\mathcal{N}$	14	4		 
					$\rangle$	٢				Drill Rig Type/Method: Sonic Brilling	ha	net	<u>5</u> £	mp	lin
					X					Drillers Name: M. LPavin				. 1	
Ϊ	/				-					Borehole Diam./Drill Bit Type:	Re	f. Ele	v.	1	
				Site	Sketcl	n Map				Sampler Type: handsa-	np	lin	5		
Dep	th to <sup>-</sup>	lst W	ater (	<b>ע</b> ):		-	Tim	e/Date	:	Drill Start Time/Date: Drill	7 Finish	Time	/Date	-18-	-08
Dep	th to \	Nater	After	Drillir	ng (👿	): -	- Time	e/Date	••••••••••••••••••••••••••••••••••••••	Well Completion Time/Date:					
Dep	th to c	other	Water	Bear	ing Zo	ones:				Soil Boring Backfill Time/Date: 6- 0	18-	- 05	<u> </u>		
			. <u>.</u>	lysis	lize							Estir	mated	1%0	f
	erval	(in.)	ts / 6	ır Ana	e & C	ler	ŧ	Type					Sand		
AVC	le Int	vered	Coun	ned fc	g Typ	US FI	ı (Fee	Soil			<b>*</b>	ě	Ę		ay
PID/O	Samp	Reco	Blow	Retair	Casin	Annul	Depth	nscs	Soil Desc	ription	Grave	Coars	Mediu	Fine	Silt/cl
	Х	Х							sandi May, Ye	ell 5 raled w or gravel, llowsh brown, loose,	15	26	25	25	15
							1		Quart:	z/foldspar/mica, rock					
							2 —		chips by	to 6" angular to					
									ronde	1 set state i se para					- - 194 - 194
						1.1 1.1		-					1 · · · ·	2011 - D.M.	1
							3								
							3		surfa	ce is hard/compacter,					
							3 —  4 —		surfa VM;	ce is hard/compacter, some desert plants					
							3		sunfa VM	ce is hard/compacter, some desert plants					
							3  4 5		surfa wm:	ce is hard/compacter, some desert plants					
							3 — 4 — 5 —		surfa WM; CPS:C	ce is hard/compacted, some desert plants 122- N34°06,150' WILP 07,142'					
······							3 — 4 — 5 — 6 —		surfa WM; GPS: C	cz is hard/compacted, some desert plants 122- N34°06,150' WILP07,142'					
							3 4 5 6 7		surfa VM	ce is hard/compacter, some desert plants 122-N34°06,150' WILP07,142'					
							3 4 5 6 7		surfa WM; GPS:C	ce is hard/compacted, some desert plants 122- N34°06,150' WILP 07,142'					
							3 4 5 6 7 8		surfa WM; Gls:c	cz is hard/compacted, some desert plants 122- N34°06;150' WILP07,142'					
							3 4 5 6 7 8		surfa WM; GPS: C	cz is hard/compacted, some desert plants 122- N34°06,150' WILP 07,192'					
							3 4 5 6 7 8 9		sunfa WM; Cr <i>PS</i> : C	ce is hard/compacted some desert plants 122- N34°06,150' WILP 07,142'					
							3 4 5 6 7 8 9		surfa WM	ce is hard/compacted, some desert plants 122-N34°06;150' WILP07,142'					
							3 4 5 6 7 8 9 10		surfr VM	cz is hard/compacted, some desert plants 22-N34°06,150' WILP07,142'					
							3 4 5 6 7 8 9 10		sunfr WM	ce is hard/compacted some desert plants 122- N34°06,150' WILP 07,192'					
							3 4 5 6 7 8 9 10 11		surfr wm	ce is hard/compacted, some desert plants 122-N34°06;150' WILP07,142'					

<b></b>										I				-	
		MV	VH							Boring #: 55-23 MW#:	She	et	1	of	[
	<b>W</b> .			Rí					~	Project: SAR Groundwater Recharge	e Opti	mizat	ion St	udy	
		NAMES OF TAXABLE PARTY.		X 55	-73		-	1	TAR )	Job #: 1343019	S	ite:50	:10	borra	~
/	,			X				(	)00 )	Logged By: A. Lean, + Previe	ewed	By:			
			Ś	Ċ						Drilling Contractor: Beart Longyear	N	IW	H		
			$\rangle$	C D						Drill Rig Type/Method: Senic Drilling	ha	nd.	59 -	201.	13
			. X	- 						Drillers Name: A. Leavin	4				
	State of the local division of the local div	No. of Concession, Name	X		the second s		Service State			Borehole Diam./Drill Bit Type:	То	tal De	pth	11	e -
										-	Re	ef. Ele	v.	•	
				Site	Sketc	h Map	5			Sampler Type: Wand Sam	pi	'în	9.		
Dep	th to 1	lst Wa	ater (	<b>Z</b> ):		galling,	Tim	e/Dat	e: 🔽	Drill Start Time/Date: Drill	Finish	n Time	/Date	-18-	-515-
Dep	th to V	Vater	After	Drillir	ng ( <b>T</b>	): `	— Tim	e/Dat	e: 🛰	Well Completion Time/Date:					
Dep	th to c	other	Water	Bear	ing Zo	ones:		_		Soil Boring Backfill Time/Date: 6-1	8-0	8			
			Ľ	lysis	Size							Estir	nated	1 % O1	F
	erval	(in.)	ts / 6	or Ane	be & S	ler	(j)	Type					Sand		
NA	le Int	/ered	Coun	ned fo	g Typ	II SN	I (Fee	Soil			-	e	Ę		ay
0/Olc	Samp	Jecov	Blow	Retair	Casin	Annul	Depth	SSS	Soil Desci	ription	Grave	Coars	Medit	Fine	Silt/cl
	$\overline{\mathbf{N}}$	$\overline{\mathbf{\nabla}}$						5	sand in	rellected gradelyin	1-	2	20	5-	10
	$\wedge$	$\wedge$							gravel	, dry, Yellowish brown,	D	25	2	CJ .	ũ
									10052	Quartz/feldspars/mita,					
L	ļ						2-		ruck chi	ips up to 4", angular					
		2							to rolling	ded					
<u> </u>							3 —		- C - C-	a la la calal				<u> </u>	
<u> </u>									esurfa.	ce hara l'angra fui,					
							4 —		<u></u>	me and periods					
					}·										
							5-		LPS'.0	23 - N 34° 06.1491					
										W117 07.214'					
							7-							<b> </b>	
 							•								
							8							<b> </b>	
							9-								
					·										
							10 —							<u> </u>	
					·										
							11			· · · · · · · · · · · · · · · · · · ·					
							1 Anno								DA/DC

		) I	viv	VH						a tha tha a tha tha tha tha tha tha tha	Boring #: SS-2 MW#: Project: SAR Groundwater Recharge	She Opti	et mizat	1 ion St	of udy	)
				****	$\overline{\mathbf{X}}$				~	(MM)	Job #: 1343019	Si	te:Se	511	Berri	~~~
	$\left( \right)$	-			X	55-2	Ч	0		$\left( \right) \left( \right) \left( \right) $	Logged By: A. Leavitt Bevie	ewed	Bv:		<u></u>	<u> </u>
	[				X		•	· (4	ĄV	VU	Drilling Contractor: Boart Longyear	Л	Λh	14		
					X				S.		Drill Rig Type/Method: Senie Drilling	han	d S	and	Plin	6
					X						Drillers Name: A. Leavet	-		¥		J
					X						Borehole Diam./Drill Bit Type:	To	tal De	pth	1.	
					<u></u>	NUMBER OF STREET, STREET, ST.	12 Contractor of the local division of the l					Re	ef. Ele	v.		
					Site S	Sketcl	h Man				Sampler Type: No ad Co	01	în s	-		
	enth	n to 1	st Wa	ater (N				Tim	e/Dat	e:	Drill Start Time/Date: Drill	-inish	Time	/Date	<b>23</b> -	ob
D	enth	to V	Vater	After	Drillin	na (👿	):	Tim	e/Dat	e: —	Well Completion Time/Date:	mon	1 1 11 10	, Date	· ·	
	enth	1 to 0	ther \	Nater	Bear	$\frac{1}{100}$	ones:				Soil Boring Backfill Time/Date:	3-0	58-			
F				-	/sis	en e					Con Doning Dackin Anno Dato.		Estir	nated	% O	
		val	in.)	./6 in	Analy	& Si	a.		ype					Sand		
₹		Inter	ered (	ounts	d for	Type	s Fille	(Feet)	Soil T							
NO/O		ample	ecove	O N O	etaine	asing	nInnr	epth (	SCS (	Soil Descr	iption	ravel	oarse	ediun	ne	lt/clay
	-	, vi	Ť,		ă.	Ö	Ā	ă	5	Carl	all a fal ional	G	0	Σ		S
		X	X							dall	ht alla abbana-	15	25	25	25	W
		<u> </u>	<u> </u>					1 —		IUYB 71	2 - Loose but compute	,				
<b> </b>						·		 0		rek a	ups whe I" angular	-				
								2	]	to rou-	nded,	20	30	30	20	0
								з —				u.				
	.					ļ			Sur	@ 10'	color Dto 104R9/2					
								4 —		6-10 GI	Rincharge Allss					
						<u></u>				1112 91	and a contra					
-								5 —	1	- rurfa	ce it hard lambacted	•	<u> </u>			
<u> </u>						·				some	desert plants					
								6	]							
								7	-	-6-PS!	024 N3406,106					
	-										W117 07.208					
<u> </u>								8 —						-		
<b> </b>									-							
<b> </b>	-+							9 —			a a cara a constructiva de mantener e presentar e presentar e presentar e presentar e presentar e presentar e p					-1-
<u> </u>																
⊢	-							10 —								
	-					<b> </b> -					· · · · · · · · · · · · · · · · ·					
						L	<b>_</b>									
<b></b>								12								04/00

4	۱ آ	MV	VН							Bori	ng #: <u>(5-25</u> MV	V#:	She	et	1	of	<u> </u>
V	יש		<b>8</b> 8 8							Proje	ect: SAR Groundw	vater Recharg	e Opti	mizat	ion St	udy	
	<u> </u>	2	6	<u>rí</u>	-				-	Job	#: 1343019		S	ite:Se	51	Bul	201
1				Х				$\int$	MAL	Logo	ged By: A. Lea	viff Revi	ewed	By:			
				X					$ \langle \rangle $	Drilli	ng Contractor: Bo	<del>art Longyea</del> r	M	W	H	-:	
				X	5-20	5		U	vv)	Drill	Rig Type/Method:	Senic Drilling	r h	snu	150	in	01:1
				X						Drille	ers Name: A.	leavit	-+				
				Х						Bore	hole Diam./Drill B	it Type:	То	tal De	pth	1	1
۱				X									Re	ef. Ele	v.		
	~	and the second secon		Site	Sketc	h Mar	)			Sam	pler Type: hs.	nd san	oli	19			
Der	oth to	1st W	ater (	 ⊽):			Time	e/Dat	e: ~	Drill	Start Time/Date:	-23-08	Finich	Time	6-	23-	0F
Der	oth to	Nater	After	Drillir	ia (🖤	- ۱۰	- Time	/Dat	o		Completion Time		FILISI	1 11110	Dale	•	
Der	oth to a	nther 1	Nator	Rear					<b>~</b> •	- Coll		no/Data:	-ر د د	. 05			
<u>net</u>			valel	- sis										Га <sup>11</sup>		0/ ~	•
	Val	(i-	/ 6 in	Analy:	& Siz			,be						⊏sur	Care	~₀ U	
×	Inter	red (i	ounts	d for ,	Type	Fille	reet)	iii T							Sand		
20/0	mple	cove	NO NO	taine	sing	sninu	pth (I	CS S	Soil Desc	ription			avel	arse	dium	e	Vclay
<u></u>	S N	<u>ل</u> ه /	ă	Å	ő	Å	മ്	S					5 -	ပိ	ž	Ϊ	ଞ
	-χ-	<u>X</u>						SU	Sgrdu	ell si	aded wor	srs.el,	15	25	25	25	10
	$\downarrow \land$	$\langle \rangle$					1		dry 1	<u>547</u>	brun -10	TR7/2		lister i T			
									loose.	but	confacted	, ack					
				<u>48.66.</u> 30.44			2 —		crips u	fti	g, my	ST TO					
				inizia d Materia Materia					Joan								
							3		- (0)00	- c 1	1 araine	42750					
									@ 1'5	ust	after se	-ple.					
							4										
							5 —		· Surf	ace	is hard e	relcompace.	fel				
									$[\mathcal{V},\mathcal{M}]$	sma	11# of ck	prer+pla.	15.				
					문문		6 —					7					
				1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.					6-PS:2	5 -	- N34° 06	.037'					
					2 22 - 2 - 2 2 - 2 - 2 2 - 2 - 2 2 - 2 -		7 —				w/1/0/	. 218					
	a servery	10.30 M	11 L					1.1.1.1					1				
											en e	n na manana na mana Manana na manana na manana na manana na manana na manana na mana na mana na mana na mana na mana na mana na man	1				
							8 —			· · · · · · · · · · · · · · · · · · ·							
							8 —										
							8 — 9 —										
							8 — 9 —										
							8 — 9 — 10 —										
							8 9 10										
							8 — 9 — 10 — 11 —										

П	N.	1/1/	<b>\/ L</b> i							Boring #: SS-2C MW#:	She	eet	1	of	-
W	יש	VIV	V M		61 <sup>-1</sup> -1					Project: SAR Groundwater Recharg	e Opt	imiza	tion S	tudy	، بىرىيە
				•					- A	Job #: 1343019	S	ite: <b>S</b> o	5:1	Bor	- <i>N</i> J
ſ				Ň				(	1(171)	Logged By: A. Lear: ++ Rev	ewed	By:			
1				X				l	jvvj	Drilling Contractor: Boart-Longycar	M	w	H		
				X						Drill Rig Type/Method: Sonic Drilling	rhe	and	S &	mp	»1;
				X	55-7	16				Drillers Name: A. Leavit	+				
										Borehole Diam./Drill Bit Type:	То	tal D	epth	1	
	$\swarrow$			$\sum_{i=1}^{n}$							R	əf. Ele	€V.		***
				Site	Sketc	h Map				Sampler Type: hand San	rpl	in	ĵ	<del></del>	
Dep	th to	lst W	ater (	<b>⊻</b> ):		9.00m,	Tim	e/Da	e:~	Drill Start Time/Date: Drill	Finisł	n Tim	e/Date		
Dep	th to \	Vater	After	Drillir	ng (👿	(): -	- Tim	e/Dat	e: `	Well Completion Time/Date:					
Dep	th to o	other \	Wate	r Bear	ing Zo	ones:		-		Soil Boring Backfill Time/Date:	23-	-08			
	<b>"</b>		6 in.	halysi	Size			<u>ē</u>			-	Esti	mated	1%0	۶f
	nterva	ed (in.	ints /	for Ar	ype &	Filler	eet)	il Typ					Sanc	l T	
NOVA	nple I	covere	v Col	ained	sing T	snjnu	oth (F	CS SC	Soil Desc	rintion	ivel	arse	dium	Ø	
표	Sar	Bec	B	Ret	Cas	Anr	Ğ	I S I			Gre	Ö	Š	i.	
	X	X						Sur	dry lig	ill graded with gravely	15	25	25	25	1
							-1, 2 ↓2 43, 13, 1 14, 13, 13, 14 14, 14, 14, 14 14, 14, 14, 14, 14, 14, 14, 14, 14, 14,		loose h	at compacted, rock clips					
							2 —		nport	", angular to counded	1				
									- er rea	-0 45 cd locan Dartal					
							3 –		vins	2mg dos at plants					•
							/								
									CPS:C	NZ6-N24°05,9841 W/17°07.228'					
							6 —								+
							7 —	1							
															1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -
							<u>8</u>								
							9 —								4
		andra National					10 —								+
				 						1771 (1997) - 1771 - 1997 (1997) 1873 - 1977 - 1977 (1977) - 1977 (1977)					
		er en antier de Les contrations					11						1		
		2223) 22222		1											1

Л	n,	N/I 1	\/LJ							Boring #: <u>く</u> らーシフ MW#:	She	eet	1	of	<u>Mark</u> î
V	יש	IVI V	W IT		-					Project: SAR Groundwater Recharg	e Opti	mizat	ion S	udy	
				F X	5					Job #: 1343019	S	ite: Se	511	Bo	rou
1				X					700)	Logged By: A. Leavit Revi	ewed	By:			
1				X						Drilling Contractor: Beart-Longyear	M	N	H		
				X					/	Drill Rig Type/Method: Sonic Drilling	·ha	nd.	San	npl:	ng
1				X	55-	22				Drillers Name: A. Loar It	-1-				
1	$\overline{\ }$			<u>×</u>						Borehole Diam./Drill Bit Type:	To Re	tal De ef. Ele	pth v.	1	
				Site	Sketc	h Mar	)			Sampler Type: hand Sa.	npl	ine			
Dep	oth to	1st Wa	ater ()	<b>∠</b> ):			Tim	e/Da	te: -	Drill Start Time/Date: Drill	Finist	n Time	/Date	23-	<b>1</b> 8-
Dep	oth to \	Water	After	Drillir	ng (👿	):	- Time	e/Da	ie: 🛥	Well Completion Time/Date:					
Dep	oth to o	other	Water	Bear	ing Zo	ones:		~		Soil Boring Backfill Time/Date: 6-2	23-0	28			in an airte Airtí Magailte
			. <u>c</u>	lysis	ize		ayaa a					Estir	natec	1%0	f
	erval	[(in.)	ts / 6	or Ana	be & G	ller	<b>F</b>	Type					Sand		
AVC	ole Int	vered	Coun	ned fc	ng Typ	lus Fl	h (Feé	S Soil				es	Ę		ay
PID/O	Samp	Reco	Blow	Retai	Casir	Annu	Depti	USC	Soil Desci	ription	Grav	Coar	Medi	Fine	Silt/cl
	X	X						s.	sond i	ellgraded with gravel	15	25	2,-	25	10
	1	<u> </u>			alesta a geografia		1-		10000 10	tran pacted rock chips					
									upto 2	" angular to rounded.					
_							<u>د</u>								
							3 —		- Surfe	ce hard/compacted,			- 1933) 		
-	-								VOS	ome desert plants					
			일하는 1일 2014년 2월				4 —		ARC 1	122 1/21005 923					
						-				W1170 U7.224"					
							5-								
							6								
							7			en de la desta de la desta Nota de la desta					
										<u> </u>					
							8 —								
	1						a —								
		1.1.1.1.1.1.	1.1.4.2	1			3								
								1.11.1	4 · · · · · · · · · · · · · · · · · · ·			• • • • • •			
							10								
							10								
							10 — 11 —								

	W I	WIN	VН							Boring #:SS-2C-MW#:		She	et	1	of	1
V	יש									Project: SAR Groundwater F	lecharge	Opti	mizat	ion S	tudy	
										Job #: 1343019		Si	te: <b>S</b>	0.1	Bor	102
1	•••••	*		X					NDO1	Logged By: A. Leavitt	Revie	ewed	By:			
1				X					VI	Drilling Contractor: Beart Ler	<del>igye</del> ar	M	wt	1		
				X					14	Drill Rig Type/Method: Sonic	<del>. Drillin</del> g	ha	nd	Sc.	mpl.	ing
				X				/	1	Drillers Name: A. Lea		2				
				X			/	al	A	Borehole Diam./Drill Bit Type	<b>):</b>	То	tal De	epth	× ×	
1				×3	5-2	8		-				Re	ef. Ele	V.	•	
				Site S	Sketcl	h Map	)			Sampler Type: hand	Sa.	npi	lin	5		
Dep	th to 1	lst Wa	ater (S	Z):		~	Tim	e/Da	te: -	Drill Start Time/Date:	を Drill F	-inish	Time	/Date	<b>23-</b> e:	0 <b>6</b> 7
Dep	th to V	Nater	After	Drillin	ng ( <b>V</b>	): 、	– Tim	e/Da	te: 🕆	Well Completion Time/Date:	gennes.					
Dep	th to c	other \	Water	Bear	ing Zo	ones:			in airthe	Soil Boring Backfill Time/Dat	e: 6-2	-3-	08			
			<u>.</u>	ulysis	Size								Esti	nated	1%0	f
	erval	] (In.)	ts / 6	or Ane	be & S	ller	<b>*</b>	Type				internet Sectors Sectors		Sanc		
NA	ole Int	vered	Coun	ned fo	g Typ	lus Fi	) (Fee	S Soil				70	e	Ę		ay
PID/O	Samp	Reco	Blow	Retai	Casir	Annu	Depti	USC(	Soil Desc	ription		Grave	Coars	Mediu	Fine	Silt/cl
	$\overline{\mathbf{N}}$	$\nabla$						ç.	Sandur	ell graded with sta	rely	1-	2-		35	I al
	$\bigwedge$	$\square$					1		dry, li	shit brown -104R)	1/2,	12	25	15	C3	10
									10050	but compacted,	rock.					
							2		crips	upto 2ª, angul	57					
						ر بر ا المحمد الراب الراب المحمد الم			tu rou-	rd al						
							3-		Faile	co had I compac	feel		n gy			
									1 . Ma	Some desert Pla	rk					
							4									
							5									
									GPS:0	28-13405,0	8551					
							6			W1170 67.2	24'					
											ilian e Constanti Constant					
							7									
							8 —			<u>na se anna an Anna.</u> Bha la sa Anna Anna Anna						
										1999 - Standard Brits, 1997 - Standard Brits, 1997 - Standard Brits, 1997 - Standard Brits, 1997 - Standard Bri 1997 - Standard Brits, 1997 - Standard Brits, 1997 - Standard Brits, 1997 - Standard Brits, 1997 - Standard Brits 1997 - Standard Brits, 1997 - Standard Brits, 1997 - Standard Brits, 1997 - Standard Brits, 1997 - Standard Brits						
							9-									
							10 -									
							11 -									
	and the second second	A.					2.1				and the second					

		WV	VH	re X X	ng diyati Deki Pang Langg		<u>.</u>	NC	10/	Boring #: 55-29 MW#: Project: SAR Groundwater Recharge Job #: 1343019 Logged By: A. Lean' FF Revi Drilling Contractor: Boart Longve ar	She e Opti S ewed	eet imizat ite: <b>S</b> By:	1 ion St	of udy 3 <i>ar i</i>	(
			~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	X X < <b>&amp;</b> <i>S</i> S	-29					Drill Rig Type/Method: <del>Sonic Drilling</del> Drillers Name: A. Leari Borehole Diam./Drill Bit Type:	he ff To Be	tal De	epth	1	ing 1
Dep Dep	th to 1 th to V	st Water	ater ( <u>)</u> After	Site : ☑): Drillir	Sketcl	h Map	) Tim	e/Dat	e: -	Sampler Type: hand Samp Drill Start Time/Date: Drill Well Completion Time/Date: -	<i>Oli n</i> Finist	n Time	<b>6</b> - e/Date	::	-28
Dep	th to c	other \	Nater .⊆.	Bear	ing Zo	ones:		-		Soil Boring Backfill Time/Date: 0 -	25-	Esti	mated	% O	f
AVO/DIc	Sample Interva	Recovered (in.	3low Counts /	Retained for Ar	Casing Type &	Annulus Filler	Jepth (Feet)	JSCS Soil Typ	Soil Descr	iption	Gravel	Coarse	Sand	ine	Silt/clay
	X	$\overline{\lambda}$						SI-	Sand w dry, lig	eil graded w. M. gravel, ht brown - lurnel,	15	25	25	25	10
							2		Chips in to ron	nler.					
					     		3 4		-surfa	ce hard/compactal, desert plants.					
							 5 — 		GPS: C	29-N34°05.852'					
							6 —  7 —								
							8				-				
							9								
								-							
							12	1		· · · · · · · · · · · · · · · · · · ·					QA/QC

	M I	MN	VL							Boring #: 55-36 MW#:	She	et	1	of	<u> </u>
V	יש									Project: SAR Groundwater Recharg	je Opti	mizat	ion St	udy	
	-		energen statistics of the second	V		Contraction of the local sector	CONCERNING & CONCERNING & CONCERNING			Job #: 1343019	Si	te:Se	,11	Sa rr	ขน
	/			∧ x				T	(DD)	Logged By: A.Leavitt Rev	riewed	By:			
								V	00	Drilling Contractor: Boart Longyear	M	we	4		
			ļ	r x					/	Drill Rig Type/Method: Sonic Drillin	ha	nd	san	yl;	15
			)	4 50	5-70			1		Drillers Name: A. Lean.	7				
				x	- )0		/			Borehole Diam./Drill Bit Type:	То	tal De	epth	11	ić Nate
	$\checkmark$		, 	~			-				Re	ef. Ele	v.	••••••	
				Site	Sketc	h Map				Sampler Type: hand San	pl.	13			
Dep	th to <sup>-</sup>	lst W	ater ()	☑):		Querran and	Tin	ne/Da	ie: 🔶	6-23-08 Drill Start Time/Date: Dril	/ I Finish	Time	e/Date	-23	-01
Dep	th to \	Nater	After	Drillir	ng (👿	):	– Tirr	ne/Da	e: -	Well Completion Time/Date:					
Dep	th to c	other '	Water	Bear	ing Zo	ones:				Soil Boring Backfill Time/Date: 6-	23-	-08			
			<u> </u>	lysis	ize							Estir	nated	% 0	f
	erval	(in.)	ts / 6	r Ana	e &	ler	t	Type					Sand		
NA	le Int	/ered	Coun	ned fo	g Typ	us Fil	ı (Fee	Soil			-	e	Ę		AV VE
PID/C	Samp	Reco	Blow	Retair	Casin	Annul	Depth	ISCE	Soil Desc	ription	Grave	Coars	Mediu	Fine	Silt/cl
	$\overline{\mathbf{X}}$	$\overline{\mathbf{V}}$						a	Candu	ellarciled , John spavel,					
	X	Ň						-	day, 113	ht RROWN-WYR 7/2, to	29	25	120	20	
	1						1	_	Loose 50	at compacted, rock chi	25				
							2 -		upto 10	", angular to roundal					
							2				-				
							3 -		alucter	a hard la sasated.					
			12 12 12 12 12 12 12 12 12 12 12 12 12 1						1-14	Plants					
							4 -								
							5 -	1							
									LPS:0	30 - 1/37 05,924"					
							6 -			W11/07.126					
							<u>.</u>								
					and a lange of		7 -		- 100 (0)	h mare and al make					
									other	areas -> still vizible					
							8-		Silt a.	nd large amount of					
							- م –		fine/	medium grainedsand,					
							-		but 1	nighor gravel 2.					
			1814 1914 - 1914 1914 - 1914 1914 - 1914				10 -								
					1.1.1.1	1							<ul> <li>• • • • • •</li> </ul>		
							11 -								

1	<i>w</i> '									Boring #: 55-31 MW#:	She	eet	1	of	
W	יש	WIW	V (1							Project: SAR Groundwater Recharge	Opti	mizat	ion S	tudy	
	<u> </u>		~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~						2022	Job #: 1343019	S	ite:S	o:1	Bur	10
1			X						100)	Logged By: A. Leavit+ Revie	ewed	By:			
			X							Drilling Contractor: Beart Longyear	N	w	4		
1			$\hat{\mathbf{x}}$	55-	31					Drill Rig Type/Method: Sonie Drilling	-h	and	'S e	rny	212
(			X	as	Ŷ					Drillers Name: A. Leau't	L				
$\mathcal{L}$			$\frac{\chi}{2}$							Borehole Diam./Drill Bit Type:	To Re	tal De ef. Ele	pth v.	1	
				Site \$	Sketc	h Map				Sampler Type: hand sam	PI.	15			
Dep	th to 1	st Wa	ater ( <u>)</u>	<b>Z</b> ):		~	Tirr	ne/Dat	ie: -	Drill Start Time/Date: Drill I	• Finish	n Time	/Date	- <u>کرک</u> :	-08
Dep	th to V	Vater	After	Drillin	ng (👿	·):	- Tim	ie/Dat	e: -	Well Completion Time/Date:					<u></u>
Dep	th to c	other \	Nater	Bear	ing Zo	ones:			in an antista da seconda da second Seconda da seconda da se Seconda da seconda da s	Soil Boring Backfill Time/Date: 6-2	.3-	48			<u></u>
			<u>.</u>	lysis	lize						<u></u>	Estir	nated	1%0	f
	erval	Ĵ.	ts / 6	or Ana	6 & C	ler	<b>(</b>	Type		이는 가지 있는 것은 것은 것이 있는 것이다. 같이 같이 같은 것을 많은 것은 것은 것을			Sanc		
ID/OVA	Sample Int	lecovered	llow Coun	Retained fo	Casing Typ	Vnnulus Fi	Jepth (Fee	JSCS Soil	Soil Desc	ription	àravel	Coarse	Aedium	ine	2ilt/clav
	Ň	$\overline{\mathbf{n}}$	ш					su	sandue	11 Graded, dry toslighty		2.	2	2	
	$\square$	$\bigwedge$					1 -		damp,	ofive brun, loose	<i>د</i>	30	<u> </u>		
									but co.	mpacted, some rock					
							2 –		chips.	afto 1.5", routs dawy					
			****						VUN 10	led;					
							у		a da anti-arte da anti- National da anti-arte da anti-arte da anti- National da anti-arte da anti-arte da anti-arte da anti-arte da anti- National da anti-arte						
							4 –		-Swf	rep compacted but					
				adri. Seret s					not a	hard as mo others,					
							5 -		10 73 00	plant material.					
										en en en ser en el ser de ser ser ser ser ser ser ser ser ser se					
							6 —		apsi	031 - N34" UC. 985'					
							7			~117 07.299'					
							1								
							8								
			ा सन्दर्भ भूषाः जन्म	Berne Merez Merez			9 -	1							
					·										
							10								
							11 -	]							
314S															

												glander.			
a	D I	ww	VH							Boring #:55-32 MW#:	She	eet	1	of	1
V	Ψ "			~1						Project: SAR Groundwater Recharge	e Opt	imizat	ion St	udy	
	-	*****		6					<u> </u>	Job #: 1343019	S	ite: S	oil	Born	rur
1	1		(	X					NDI	Logged By: A. Leavitt Revi	ewed	By:	: 		
1			į	X	7	2			vvv	Drilling Contractor: Beart Longyear	M	wf	1	n an Sainte	
			3	R 5>	-3 A 20	<u>~</u>				Drill Rig Type/Method: Senie Drilling	h	end	Sa	mp	11.
1			đ,	ינצי	452	٤				Drillers Name: A. Leavit	4				
			X	<b>V</b>				and the second s		Borehole Diam./Drill Bit Type:	Тс	tal De	epth	<u>)</u> `	
	$\overline{}$			-		eenseenseenseelen.					R	əf. Ele	v.		
				Site	Sketc	h Mar	<b>,</b>			Sampler Type: hand San	nP	lin			
Den	th to 1	st W:	ater ()				Tin	ne/Da	te <sup>.</sup>	Drill Start Time/Date:	Finiel	Tim		23	-08
Dop	th to V	Vator	Aftor	Drillir	) ( <b>1</b>	<u>.</u>	Tin	ne/Da	lo: <b></b>	Wall Completion Time/Date:			Date		
Dep	th to c	ther V	Motor	Dian		.). <u>-</u>				Coll Devine Deckfill Time/Date:	2	<u>.</u>		<u> </u>	<u></u>
		omer	valer	Bear		ones:				Soli Boring Backfill Time/Date: 6 -	2-1	<u> </u>			
	କ୍ଷା	(.1	/ 6 in.	Analys	& Siz			be				ESU	nateo	1%0	л Т
4	Inten	ed (ir	unts	d for /	Lype	Fille	Feet)	oil T)					Sand		
NO	nple	cover	°C ×	ainec	sing	snInc	pth (F	CS S	Soil Descr	intion	avel	arse	dium	θ	/clav
	Sai	, Å	Bo	Bel	0 G	Ani	ð	SU			ษั	8	Re	Ë	1
	$\mathbb{N}$	-X-						50	Sanda	-ellgraded with	21	25	25	25	•
	$\langle \Delta \rangle$	l	<b>\</b>				- 1 -		Gravel	, moist, Olive brann,					
									6 1005e	but compacted, rock					
							2 -		Cuip	ip to s, anguar ro					-
									Yourac						
							3 -		surfa	ce hard / compacted,					
							-	]	some d	esert plants					
							4	_							
							5 -								
									aps:	032 - N34° 06.044'					
							6 -			w117° 07.304'					
								2							
				1933) 1933 1933			7 -								
							8 -								
						1									
							9-								
							10 -								
							<sup>10</sup> -								
							11 -								
		1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	La contra de la co	1.1 1.1 1.2 6	1.1	5 S S S	1 -	a da la Alfreda a	<ul> <li>A statistics</li> </ul>			4	1.0	1 A A A A A A A A A A A A A A A A A A A	

æ		MV	VH							Boring #:55-33 MW#:	She	et	1	of	1
V	y									Project: SAR Groundwater Recharge	e Opti	mizati	on St		
		the party for the second	<u>rb</u>				NAMES OF TAXABLE PARTY OF TAXABLE PARTY.	nn <del>a vrzego</del> tnek		Job #: 1343019	Si	ite:20	011	Sor	พบ
$\left( \right)$			Č.					$\uparrow$	$(\overline{1})$	Logged By: A. Leavitt Revie	ewed	By:			
			<u>A</u> s	ر - کر -	53			ν	$\vee \vee )$	Drilling Contractor: Boart Longyear	N	1w	<del>t/</del>		
			<b>AR</b> A	KSI.	BZ					Drill Rig Type/Method: Sonic Drilling	he	and	Sei	mpl	ing
			$\overset{\sim}{\times}$				/			Drillers Name: A. Leav. 41	4				
		1	X			/				Borehole Diam./Drill Bit Type:	То	tal De	pth	1'	
		and the state of the				_					Re	ef. Ele	v.	•	
				Site	Sketc	h Map	<b>)</b>			Sampler Type: hand Sam	p1:	15			
Dep	th to 1	lst Wa	ater (	<b>Z</b> ):		8700ma	- Time	e/Da	te:	Drill Start Time/Date: Drill	Finish	n Time	/Date	27-	8
Dep	th to \	Nater	After	Drillir	ng (🔽	·): -	— Time	e/Dal	e: —	Well Completion Time/Date:					
Dep	th to c	other \	Nater	Bear	ing Zo	ones:				Soil Boring Backfill Time/Date: 6-2	3	8			
			.s	alysis	Size							Estir	natec	0 % 0	f
	terval	(j.)	its / 6	or An:	pe &	ller	et)	Type					Sand		
AN OVA	ple In	verec	Cour	ined f	Ty Ty	Ilus F	h (Fe	S Soi			<u>–</u>	e Se	Ę		lay
<u>Š</u>	Sam	Reco	Blow	Reta	Casil	Ann	Dept	nsc	Soil Desc	cription	Grav	Coar	Medi	Fine	Silt/c
	$\mathbb{N}$	$\mathbf{N}$						Su	sandu	ellgraded in gravel,	10	20	30	3-	6
	$\square$	$\square$							slight	lydamp, darebran,		20	23		
									10052	but compacted, rock					
							2 —		Chips n	pto 2", angular to rando					
									<u>, 2823 등 11</u> 1993 - 19	rradi (1965) a se de la casa (1967) r Se se definidad de la casa (1967) r					
							3-		Surfa	CS land lampertel					
					<u> </u>				1 ith	sune desertelante					
							4					n Ny Sec Ny Sec			
							5_								
									GPS:0	33 - N34° 06,103'					
							6-			W117°07,299					
							7 -								
				- 1999 - 19 1922 - 19						ented eacht - Loopla Affectering Loopla - California - California - California					
							8 —				Ngerou Logia di				
							9-								
							10								
					200		11								
	1									ha 4 - Alta Masaa ay kaladi					
				1	1	1	All the second	11.1.1.1.1	a second the state of the		10 C 1 C 1 C	1.1.1.1.1.1.1.1	이 같아 있어?	1.000	

1	<u>n</u> .	<b>\/I \/</b>					946) 		Boring #: <u></u> MW#:		She	et	1	of	<b>\</b>
V	עי								Project: SAR Groundwater Re	charge	Opti	mizat	ion St	udy	
				6		<u>.</u>			Job #: 1343019		Si	te: So	0111	Bu ri	านเ
1	Γ		1	<b>X</b> -	5-3	4			()()) Logged By: A. Lear. 7+	Review	ved	By:			
1				Ŷ					Drilling Contractor: Beart Long	yoas 🖌	n	w t	1		
				Ŷ					Drill Rig Type/Method: Sonic E	hilling	ha	nd	Sen	vel.	ing
				X					Drillers Name: A. L.ea.	.:++	<u> </u>				
1	$\checkmark$			× —					Borehole Diam./Drill Bit Type:		To Re	tal De	epth v.	1	ľ.
				Sito	Skotel	h Man			Sampler Tyrne: MGall	6	P	1:00		0. As	
Den		let W/	ator (1	7).			Tim	a/D a	Campier Type. 6-23- 4		- <u> </u>	9 11 6 	)6	-23-	<i>6</i> 8
 		Mator	After		a /	۱۰ ۰					nisn		Jate	•	
	un 10 V	valer	Aller		ing ( 🗶	<i>}</i> .	<u> </u>	JUal			2	pC			
роер		omer \	vater	ьear		ones:			Soil Boring Backfill Time/Date:		~ >-	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~		0/ ~	
	al	1)	/ 6 in.	Analys	& Siz		신다. 지역 개립은 1일	be				⊨stii	nated	% Ut	
4	Inten	red (ir	unts	1 for /	Lype	Filler	-eet)	oil Ty					Sand		
	Sample	Recover	Blow Co	Retained	Casing 7	Annulus	Depth (F	nscs s	Soil Description		Gravel	Coarse	Medium	Fine	Silt/clay
	X	$\mathbf{X}$				2.55		Ji	Scudwell, raded with gra	vel,	15	25	25	U	le
	$\overline{)}$	$\left  \left( \right) \right $					1		dey, 13h Fbrown - 10YR >/2	$\rangle$	10				
									louise but compacted, rock	ohB					
						an Anna An An Anna An An Anna Anna	2 —		up to Lis, angular to row	219/1					
									en 1977 en esta de la companya de l Companya de la companya de la company						
			lanen eta di Territaria				3 —		a face band ampacter	1				ana da serie Maria	
									with some desert plants	-/					
							4 —								
							5								
							<b>.</b>		LPS: 034 - N34°06.13	<u>13'</u>					
							6 —		wi/>° 67.	96.					
							1143-43 		<u>1997 - Andrea Andrea, and an </u>						
							7 —								
				n ng dig Marina di		<u>in de co</u> rrectiones Constantes	8 —								
							9 —							in an the transfer the second se the second se	
							10								
		1													
							11	1999							
							- 11								

G		MV	VH							Boring #: 55 - 35 MW#: Project: SAR Groundwater Rechard	Sh je Opt	eet imiza	1 tion S	of tudv	
			;2						1000	Job # 1343019	ç	site: S	a.1	Ber	row
	1		×	55	35-			1	101101	Logged By: A. LEGriff Bey	iowod	I Rv	011		
	1		X					l	/0- ]	Drilling Contractor: Beart-Longvear	N	1 W	14	 1. 1	
			$\hat{\mathbf{x}}$							Drill Big Type/Method: Sopio-Dritting	i he	rd	Sau	el:	Ac
			ý.							Drillers Name: A LOG	, , }-			4	<u>~_</u>
	1		$\lambda$							Borehole Diam./Drill Bit Type:	Тс	otal D	epth	1	1
											R	ef. Ele	ev.		
				Site	Sketc	h Mar				Sampler Type: Maad Sa	201	ine		L	
Den	th to 1	st W	ater (N				Tim	e/Da	te. 🥆	Generation of the second secon	Finic	h Tim	/Dati	23	38
Dep	th to V	Vater	After	Drillir	ים (ש	·)•	- Timi	e/Da	te: 🤊	Well Completion Time/Date:				J.	
Den	th to c	other \	Water	Bear	ina 70	ones:		-, 4	<u> </u>	Soil Boring Backfill Time/Date: 6-	23-	-05	entraji Stanaca Stanaca	yer de lej. Alexandre de legender Alexandre de legender	
P				/sis	ez ze							Esti	mater	1%0	f
	rval	(in.)	s / 6 ii	Anal	a & Si	e	<b>_</b>	ype					Sanc		<u> </u>
٨A	e Inte	ered	Count:	ed for	J Type	IS Fill	(Feèt	Soil					Ε	Γ	
0/016	Sampl	Recov	3low C	Setain	Casing	Annulu	Depth	JSCS	Soil Desc	ription	aravel	Coarse	Aediu	line	Silt/cla
	X	$\overline{\mathbf{X}}$	/ <b></b>					Su	sande	-ellsraded withsrever,	15	25	25	25	10
	$ \Delta$	/					1 -		dry, 113	Sht brown -10PR 1/2,					
									TOUSR 5	ALIZ " AND LOCK					
							2 —		rund	les, right ro					
							3		surfa	ce is hard loom preter	6				
							4		with 5	une desert plants'					
							5 —		8.01.						
									GPS- C	135 - N34 064/56					
							6 —			- [1/ 0/03/3					
							7 -								
					[		Q								
							°								
							9 —								
							10 —								
2222										<u>er en </u>					
							11			<u>n seen stende steld in de b</u>					
<u>e nalidet</u> analidade	La sectoria del Marca de com	ng sang sang Segting sang	Lana taki Asilari	n na star Na star	L	T in the second	<sup>1</sup> 12 —	L	L		1	1	1	L	L QA/Q

G		MV	٧H							Boring #:55-76 MW#:	She	eet	1	of	<u> </u>
•	<i>V</i>	1	7							Project: SAR Groundwater Recharg	le Opt	imizal	tion S	tudy	
		1	X					r	INAL	Job #: 1343019	S	ite:Se	>11	Barr	w/
1		1	XS	5-3	6			$\left( \right)$	///	Logged By: A . Lean' H Rev	iewed	By:			
		7	X					, v		Drilling Contractor: Boart Longycar	M	.w	<u>H_</u>		
		)	X							Drill Rig Type/Method: Senie Drilling	h	end	56	mp	lin
		)	X							Drillers Name: A. Leavin	4				
'	$\overline{\ }$	)	<u>&lt;</u>							Borehole Diam./Drill Bit Type:	To Re	otal De	epth ev.	<u> </u> _	
				Site	Sketc	h Mar	)			Sampler Type: hand Sa.	npl	; ne	\$		
Dep	oth to 1	1st Wa	ater (	<b>Z</b> ):			Tim	e/Dai	te: 🗕	Drill Start Time/Date: Drill	Finist	n Tim	6- a	23-6	8
Dep	th to \	Nater	After	Drillir	ng (🛡	): •	- Tim	e/Dat	e: 🔪	Well Completion Time/Date:					
Der	oth to c	other	Water	Bear	ing Zo	ones:				Soil Boring Backfill Time/Date: 6-	23.	. 05		<u>e sarit</u> Se te s Sati	
			 	ysis						1		Esti	matec	1%0	f
	rval	in.)	s / 6 iı	Anal	s & Si	P		ype		요즘에는 것 같은 바람이 있었다. 동네는 것 이 바람은 것같은 것이		Ī	Sand	   ::	
/A	e Inte	ered (	ounts	ed for	Type	s Fill	(Feet	Soil 1					<u>ء</u>		\
	Sample	Recove	Blow C	Retaine	Casing	Annulu	Depth	USCS	Soil Desc	pription	Gravel	Coarse	Mediur	Fine	Silt/cla
	X	ſχ						5~	Sanda	ellsraded with srally	15	25	SL	25	10
	$\vdash$	$\bigvee$					1-		dry 1	SAT GOV A-TOTICIZ,					
									COUSE 7	at longacter for	. : 				
							2 —		On al	al in q , signing 10					
									1,000,10						
							3-		surfac	ce hard/compacted,					
							<u> </u>		wh	some desert plants.					
							4								
							5								ļ
							1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		615:	036 - N34°06,094'					
		anana Marina Nganana					6 -			W112-01.380					
						e sta Narse	7 -			<u>ne na sector e tradición de la consecta de la conse</u> Esta de la consecta d					
						11272 		la de la constante Constante de la constante de la Constante de la constante de la							
							8 -								
							9-								
							<u>-</u>								
							0								
							11								
		Sec. Sec.	and the second second	- 北京 一部 たちから		1 S. S. S. S.	<ul> <li>A second sec second second sec</li></ul>	Lange Contractor	A state of the		a se	(4) S. S. S. S. S. S.	and the second	1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	4

[										11.55					,
	M I	VIV	VН							Boring #: 3> 3/MW#:	She	et	1	of	ļ
	<b>W</b> .									Project: SAR Groundwater Recharge	e Opti	nizati	on St	udy	
										Job #: 1343019	Si	te: S	oil	Bor	·70~
		X				an the second		$\mathcal{D}$	nm)	Logged By: A. Leav. ++ Revie	ewed	By:			
		X						[/(	$\mathcal{N}$	Drilling Contractor: Boart-Longycar	M	WK	1		
		X	\$\$-7	37				V		Drill Rig Type/Method: Senic Drilling	ha	1.	Se-	Pl	ing
		X								Drillers Name: A. Lear.'t	4				
$  \setminus$		X								Borehole Diam./Drill Bit Type:	Tot	al De	pth	1	/
	and the second s		- Martine Company of P	NO. MARKED CONTRACTOR OF	and the construction of the	Constitution of the local division of the lo				-	Re	f. Ele	v.	_	
				Site	Sketc	h Map	)			Sampler Type: Mand Sa.	nP	10			
Dep	th to 1	ist W	ater (	<b>∑</b> ):		entites	Tim	e/Dat	ie: ~	6-23- % Drill Start Time/Date: Drill	Finish	Time	/Date	23-	S.
Dep	th to \	Vater	After	Drillir	ng ( <b>V</b>	(): ~	- Tim	e/Dat	e: 🖌	Well Completion Time/Date:					
Dep	th to c	other	Water	Bear	ing Zo	ones:				Soil Boring Backfill Time/Date: 6-2	2.3-0	58			
		i. V	. <u></u>	lysis	Size							Estir	nated	% O1	
	erval	(in.)	ts / 6	or Ana	e & O	ler	(j)	Type					Sand		
A V	le Int	/ered	Coun	ned fc	g Typ	us Fil	(Fee	Soil			-	ė,	Ę		ay
0/OIe	Samp	Recor	Blow	Retair	Casin	Annul	Depth	SSS	Soil Descr	iption	Grave	Coars	Mediu	Fine	Silt/cla
	Ŵ	$\overline{\checkmark}$	<u> </u>			$\uparrow$		5/	Candwo	112 for in scare (	15		55		15
								<b>Sv</b> .	dry lig	htolme brown, 10032	æ	20	×.	39	*
							] — 		but com	Prot, rock elips up to			,		
					ļ		2	-	2, 6.15	uls to rounded.					
							3 —		<u> </u>	and real at					
									Surface	2 hard conflacted					
							4 —	ł							2
}		- <u></u>								· · · · · · · · · · · · · · · · · · ·					
							5 —		LPS.C	137 - N34°06,035'					
										W117° 07.381'					
[							6								
						ļ	7-								
							8-								
<b> </b>				<b> </b>											
				<b> </b>	<u> </u>		9 —								
					<u> </u>										
				<u> </u>	<u> </u>		10								
						1				·····					
							11								
							12 -								
1															QA/QC
G		VIV	VH				že,			Boring #: \$5-35 MW#: Project: SAR Groundwater Recharge	She e Opti	et mizati	1 ion St	of I udy	
----------	---------	------------------------	-------------	---------	------------------------------------------------------------------------------------------------------------------	------------------------------------------------------------------------------------------------------------------	-----------------------------	--------	------------	------------------------------------------------------------	---------------	--------------	-------------	-------------	----------
			V		and the second	and the state of the			)	Job #: 1343019	Si	te: >	SA [ 1,	2017	
	/	í	X					ŀ	SANI	Logged By: H.LEQVIT-Revie	ewed	By:			
		(	∕_ Maria						) ( $V$ )	Drilling Contractor: Beart Longyear	· /	M	JH		
			n De S	5-3	Ø			U		Drill Rig Type/Method: Sonic Drilling	ha.	nd S	مم	plin	19
			X		0					Drillers Name: A. Learl	++	pa		4 2	
			X						¢	Borehole Diam./Drill Bit Type:	To	tal De	pth	1	
					*****	and the second	- Contraction - Contraction				Re	ef. Ele	V.		
				Site	Sketc	h Map	)			Sampler Type: hand Sa	np	lin	5		
Dep	th to 1	lst Wa	ater (	☑):		Com	Tim	e/Dat	e: C	Drill Start Time/Date: Drill	Finish	Time	/Date	-23- :	-08
Dep	th to V	Vater	After	Drillir	ng ( 👿	): 、	– Tim	e/Dat	e: —	Well Completion Time/Date:					
Dep	th to c	other \	Nater	Bear	ing Zo	ones:		~	/	Soil Boring Backfill Time/Date: 6-	23-	צט			
			Ŀ.	lysis	ize		;			••••••••••••••••••••••••••••••••••••••		Estir	nated	% Of	ŧ
	erval	(in.)	s/6	r Ana	8	er	(1	Type					Sand		
A V	e Inte	ered	Sount	ed fo	Typ	lli st	(Fee	Soil .					Е		۲.
0/OI	ampl	lecov	low 0	tetain	asing	unut	epth	SCS	Soil Descr	iption	irave	oars	fediu	ine	ilt/cla
	00	$\frac{\pi}{\sqrt{2}}$	<u>, ш</u>			◄		5	all a	lacelal, Anereial	0			LL.	0)
	Х	X		<b></b>			'	سلا	dry lis	ht alve Sonn. loose but	15	20	25	?5	15
	ř	-1					1		(umpace	t, rock chips w to 2"					
									angular	to roundal					
					L										
							3-		······						
									-Surfac	e hard/compacted, w.M					
							4		Sme o	esert flants					
<b> </b>															
<u> </u>							5 —		C. P.C.	179. 1/34° AE 951					
									<u> </u>	W1120 07 279'					
							6-								
							8-								
							9-								
<b> </b>					<b> </b>		-,-								
<b> </b>							10 —								<u> </u>
							11 —		• ·····						
<u> </u>					<b> </b>							н -			
		·			•		· 12 —	•							QA/QC

	M,	<b>\/I \/</b>	VIJ							Boring #: 55-3  MW#:		She	et	1	of	
V	<b>V</b> •	VIV								Project: SAR Groundwate	er Recharge	e Opti	mizati	on St	udy	
			r						<u> </u>	Job #: 1343019		Si	te:So	:16	3011	کال
	1		Č					T	MAY	Logged By: A. Lean	++ Revie	ewed	By:			
			X					1/	00/	Drilling Contractor: Boart	<del>Longyear-</del>	1	nn	1H		
			Ŷ					V	/	Drill Rig Type/Method: Se	nic Drilling	ha	nd &	<i>s</i> am	plin	is
	1		$\sum$	,						Drillers Name: A.L	ear. 71	L				-
	$\Box$		£	>>-	<i>5</i> 1					Borehole Diam./Drill Bit T	уре:	To Re	tal De ef. Ele	pth v.	1	8
											<u> </u>	<u></u>		inan.		
				Site §	sketc	n Map	nga téngén nga ka			Sampler Type: NG	1 J E M	<i>pl.</i>	ng	6	-22	-08
Dep	th to 1	lst Wa	ater (	<b>Z</b> ):		<u> </u>	Time	e/Dat	e: -	Drill Start Time/Date:	Drill I	Finish	Time	/Date	:-0	0
Dep	th to V	Nater	After	Drillin	ig (👿	):	- ` Time	/Dat	e:	Well Completion Time/Da	ate: 🔪					
Dep	th to c	other V	Vater	Bear	ing Zo	ones:	•			Soil Boring Backfill Time/	Date: 6-2	3-07	8			
			ü.	ıalysik	Size			9					Estir	nated	% O	
	Iterva	d (în.)	nts / t	for Ar	pe &	filler	et)	ll Typ						Sand		
PID/OVA	sample Ir	Recovere	slow Cou	Retained	Casing T <sub>3</sub>	Annulus F	Jepth (Fe	JSCS So	Soil Desc	ription		Gravel	Coarse	Medium	ine	Silt/clay
<u></u>	$\overset{\circ}{\bigtriangledown}$	$\overline{\mathbf{v}}$	ш			4		ŚW	sandu	-ellgraled with	gravel,			<u>ح</u>	<u> </u>	0,
	X	$\wedge$							dry, li	gh Folke brown, li	resebut	ts	20	25	25	15
									Compac	t, ruckchipsul	ω2 <u>′</u>					
							2 —		angula	r to rounded.						
							3—				1.1					
									- Surfac	e hard / compac	490 ) L					
							4 —		W/104 )	sime deser the	173					
					1.1.1			1.1.1.1.1.1.1.1								
							5 —		1 85.0	39 - N 240 pr	9181					1944 A
							5 —		<u>L15:0</u>	39 - N34° OF W1170 07	<u>.9(8'</u> 1376					
							5 —  6 —		<u>L15:0</u>	39 - N340 OF WILZO OZ	<u>.9(8'</u> .376					
							5		<u>L.15: 0</u>	39 - N34° OG WIL70 07	<u>.9(8'</u> 1376					
							5 6 7		6.15:0	39 - N340 00 WIL70 07	.918' 1376					
							5 6 7 8		<u>L.(5:0</u>	39 - N34° OG WIL70 07	<u>.9(8'</u> .376					
							5 6 7 8		<u>L. (5: 0</u>	39 - N340 05 WIL70 07	<u>.9(8'</u> 1376					
							5 — 6 — 7 — 8 — 9 —		<u>l. (5: 0</u>	39 - N340 00 WIL70 07	<u>.918'</u> .376					
							5 6 7 8 9		<u>L. (5: 0</u>	39 - N 34° OG WIL70 07	<u>.9(8'</u> .376					
							5 — 6 — 7 — 8 — 9 — 10 —		<u>. (5: 0</u>	39 - N34° OC WIL70 07	<u>.9(8'</u> .376					
							5 6 7 8 9 10		<u>. (5: 0</u>	39 - N340 00 WIL70 07	<u>.918'</u> .376					
							5 6 7 8 9 10 11			39 - N 34° OG WIL70 07	<u>.9(8'</u> .376					

	n.									Boring #: \$ -40 MW#:	She	eet	1	of	
U	עו	VI V	VH							Project: SAR Groundwater Recharg	e Opti	mizat	ion St	udy	
	/		<u>v</u>							Job #: 1343019	S	ite: <b>S</b> a	oil.	Burr	مسان
1			x					Ń	NA	Logged By: A. Leavitt Rev	iewed	By:			
1			X					V	VVI	Drilling Contractor: Boart Longyoar	M	wi	4		
			X						1	Drill Rig Type/Method: Sonic Drilling	ha	nd	5ar	npl,	ng
			X					-		Drillers Name: A. Leavin	14				'
			×.	\$5-	40					Borehole Diam./Drill Bit Type:	То	tal De	pth	1'	
	$\searrow$		$\triangle$		No. of Concession, Name						Re	ef. Ele	V.		
				Site	Sketc	h Map				Sampler Type: hand Sam	npli	ing			
Dep	th to 1	st Wa	ater (	<b>Z</b> ):		~	Tim	e/Da	te: 🔸	Drill Start Time/Date: Drill	Finish	n Time	e/Date	<b>24</b> -c ::	8
Dep	th to V	Vater	After	Drillir	ng ( <b>T</b>	(): <i>•</i>	- Tim	e/Dat	ie: -	Well Completion Time/Date: -					
Dep	th to c	ther \	Nater	Bear	ing Zo	ones:				Soil Boring Backfill Time/Date: 6-	24	-8-			
		(	12. 0	alysis	Size			Û				Estii	nated	l % Ol	r :
	nterva	d (in.)	nts / {	for An	/pe &	-iller	et)	il Typ					Sand		
	Sample Ir	Recovere	Blow Cou	Retained	Casing T <sub>3</sub>	Annulus F	Depth (Fe	USCS So	Soil Des	scription Silty sand	Gravel	Coarse	Medium	Fine	Silt/clay
	$\mathbf{N}$	$\Sigma$						P	Sandp	graded, dry, olive.	-			28	- 7 4
	$\square$	$\underline{\wedge}$					1 1 -	SM	545/3-5	74, 100se put comprote	110	D	$\mathcal{D}$	23	9-4
									roda c	hips upt 1, subangular					
							2 —		to rou	rder.					n in sin La second
									esurfa	a not as herdand					
							3 —		compa	cted as the other hold.					
							1		1030	f algert plants.					
							5 —		100						
									611.	040 - NJ 05,81) W117'05,81)					
					NARAN M		6 —								
							/-								
							8 —								
			د سر کار کار د سر کار کار کار												
							9 —								
										2					
				ाः सन्दः अनुसुर्वेष			10 —								
							11								
							10								

	V.	VIV	VL							Boring #: <i>\$\$-4</i> / MW#:	She	et	1	of	
V	<b>w</b> "	WI W								Project: SAR Groundwater Recharg	je Opti	mizat	ion St	udy	
	مند مند المراجع	K.8							TAIT	Job #: 1343019	S	ite:	oil	Bor	ren
	(	X							1)00)	Logged By: A. Lear. HA Rev	iewed	By:			
		X	a A						0-/	Drilling Contractor: Boart Longycar	N	14	H		
	6	5	<u>ر</u>						1	Drill Rig Type/Method: Sonic Brilling	h.	end	55	pl	ing
		>	(					,		Drillers Name: A. Lear.	++				
	J	<u>}</u>	(S) (	-41			/			Borehole Diam./Drill Bit Type:	To Re	tal De ef. Ele	epth v.	1.	<b>,</b>
				Sito	Skotel	h Man				Sampler Type: brad So	, el	• A e			
Don	th to 1	ct Wa	ator (N			<u>n map</u>	Tim	o/Da	<b>o</b> .	Drill Start Time/Data:	Finick	Time	5- 20	4-0	8
Dep	th to \	Nator	Aftor	Drillir	) a ( 🜌	<u>)</u> . •	∽ Tim	o/Dat		Wall Completion Time/Date:	FILISI				
Dep	th to c	thor V	Nator	Bear		). DDAS.	1100		ю <b>.</b>	Soil Boring Backfill Time/Date:	24-0	30			
<u>neh</u>				-sis	l g				en el gres d'a del Maria			Fetin	nated	% 0	f
	val	n.)	/ 6 in	Analy	& Siz	<b>.</b>		ype					Sand	<i>7</i> 0 <b>C</b>	Ī
/A	ș Intei	) pəred	ounts	ad for	Type	s Fille	(Feet)	Soil T						se ed by ui	,
	Sample	Recove	Blow C	Retaine	Casing	Annulu	Depth (	uscs:	Soil Desc	cription	Gravel	Coarse	Mediun	Eine	Silt/clay
110	X	X						Su	Sandw dry, 1	ellgraded withgravel, isht hown-104R7/2,	- 15	20	20	30	15
		,							Ra lous	e but compacted, rock	-				
							2-		chils.	pto 2", angular to					
							1994 (1995) 1995 - 1995 1995 - 1995 - 1995		punde	<u>A</u> ,					
							3-								
							 4		ollsert	plants					
							5 —		D P VI		,				
									6131	1,117 07 445					
			ingen ( Maria				6 -								
			gardi.				/-	]							
							8-								
							9-								
						<b> </b>									
							10 -			angene ganger og en standerskinger. Na angene som en som en standerskinger og en som					
			1.1.1			<ul> <li>A 1992 A</li> </ul>	1 1 10 10 10 10 10	1. 00	▲ しいかん ひとう かんせん		[8] M. C. M.	1 4 8 8 8	1	<ol> <li>March</li> </ol>	1 1 1

	۱ آ	MN	VL							Borin	g #: <i>55-43</i>			She	eet	1	of	1
U	יש									Proje	ct: SAR Gro	undwate	r Recharg	e Opt	imizat	ion St	udy	
	سىر		18							Job #	: 1343019			S	ite: So	oil 1	Bori	-01
1	1	,	Х					1	)/()/_	Logge	ed By: A.	Lear.	H Revi	ewed	By:			
1		7	X					ι	$\mathcal{W}$	Drillin	g Contracto	r: <del>Boart</del> -	Longyear	- /	ni	vb	1	
1		)	٢						/	Drill F	Rig Type/Me	thod: Se	<del>nie Drillin</del> g	h	and	l Sa	.m/	olia
		X	$\langle \rangle$							Driller	rs Name: 🖍	4. Li	cavi's	1+				
	<	) Z	<b>(</b> 55- {	-42						Boreh	nole Diam./[	Drill Bit T	ype:	To Re	otal De ef. Ele	epth v.	1	1
	$\sim$			Site	Sketc	h Man	, ,			Sam	ler Type	hene	1Sa-	20	lin	4		
Den	th to '	1st W	ater (N				Tim	e/Da	ter 🛩		Start Time/D	6-24-	Je Drill	Finich		C=	87-	58
	th to V	Nator	Aftor	v∠)• Drillir	na ( <b>W</b>	)	- Tim	o/Da	to.	Woll		Timo/Da		<u> </u>	1 11118			
Dep	th to 4	other	Wator	Boor	ing 7	/·				Reil D			Data: 6-2	24	25	<u>ang 1999</u> Ang 1919		
<u>––––</u>			vaiel	sis	e e				<u>neroza alerene a</u> 11. organizatu			m mne/t			e Ecti	matod	0/ 0	f
	val	Î	/ 6 in	Analy	& Siz			/pe							su	Sand	/	•
¥	Inter	red (i	ounts	d for ,	Type	; Fille	Feet)	Soil T								Ganu	e tel Jacobs	
PID/OV	Sample	Recover	Blow Co	Retained	Casing <sup>-</sup>	Annulus	Depth (I	uscs s	Soil Desc	cription				Gravel	Coarse	Medium	Fine	Silt/clay
	X	X						sv	Sand u	ell sra ahtba	aded w	, th 3	rally	15	20	20	30	15
	·						1		but com	n pact	ed ruc	Krh	: PS WP					
									to 2",	angu	Jar to	round	lel.					
							2-											
							3_											
									Sartie Some a	e he deser	rd/con	-pact 18.	el; Lit					
							5-											
									<u>015:0</u>	> 72 -	- N34°	05.	922					
				na presi an an an an an an an an			6	1		lik titat 14 Milan ar	<u>~11/</u>	07,0	14/1					
							7							ingen Stillen				
											. La cacarata da ci Mangang Alban	rational de la constant Secondador de la constante Secondador de la constante de la constante de la constante de la constante de la cons						
							8											
							0											
							у — 				4							
							10											
												د میں بیار ہے۔ دیکھی کا ایک میں ایک میں ایک میں ایک						
<u>1999</u> 1999							11				in en anterna de la composición de la c En anterna de la composición de la compo							
				2000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 		1				ing (1997) Lagan Laga Shing Shing								
		<u> </u>		L	L		<sup>1</sup> 12 —	1	1.2-4.2	e di serie de la serie de Tenera de la serie de la ser	an an an an Anna Anna Anna Anna Anna Anna			1	1			QA/Q

	N/N	N/LB							Bori	ng #:55-43 MW#:		She	eet	1	of	(
W	IVIV								Proje	ect: SAR Groundwate	r Recharge	e Opt	mizat	ion St	udy	
1		18				<u>s</u>			Job	#: 1343019		S	ite:Se	111	Bør.	row
1		$\hat{\mathbf{x}}$					1	(DD)	Logg	jed By: A. Lean	'FF Revie	ewed	By:			
1		X					U		Drilli	ng Contractor: B <del>oart I</del>	<del>.ongyo</del> ar ,	M	$\overline{\mathcal{V}}_{i}$	H		
	Ś	\$ 55	-43					/	Drill	Rig Type/Method: <del>So</del>	nic Brilling	50	.1	Bui	ro	$\checkmark$
	X	\$	Ĩ			~			Drille	ers Name: A. L.	<i>earit</i>	-1				
(	× X								Bore	hole Diam./Drill Bit T	/pe:	To Re	tal De ef. Ele	epth v.	<u> '</u>	
			Site	Sketcl	h Map				Sam	pler Type: hanc	1 Sam	eli	ng			
Depth to	o 1st W	ater (	☑):		~	Tim	e/Da	te: 🖌	Drill	Start Time/Date	-US Drill	Finist	) Time	C-	247-	uf
Depth to	o Water	After	 Drillir	na (🖤	): ~	- Time	e/Da	te: -	Well	Completion Time/Da	te: -					
Depth to	o other '	Water	Bear	ing Zo	ones:				Soil	Boring Backfill Time/[	Date: 6-8	\$-0	8			
			lysis	ize									Esti	matec	1%0	f
Brval	(iu)	ls / 6 i	r Anal	e & S	-er	Ð	Type							Sand		
PID/OVA	Recovered	3low Count	Retained fo	Casing Typ	Annulus Fil	Jepth (Fee	JSCS Soil	Soil Des	cription			Gravel	Coarse	Vedium	line	Silt/clay
	$\langle X$						su	sande Ohy.li	-e1/5 gr + 1	raded w.m.	iravel,	5	20	20	30	B
						1 —  2 —		1 0132	rock	sular forma	ndel.					
						3		- surf	sce.	shard from	Pacted					
						4 —		W1795	and	areset pisno	7					
				11-111 (20-12)												
						5		CPS: 0	243	- N 34°05, WII70 07,0	976' 150'					
						 7			64 7 - 77 - 1 842 - 75 - 1							
						 8 —										
						 9 —										
						 10 —										
요즘은 요구?										<u>en de la constante de la const Esta de la constante de la const</u>	المتعادة كالأبالية					
						11 —										

		NV XXXXXXXX	<b>VH</b> \$ \$5-9	9				0	00)	Boring #: 15-49 MW#:       Sheet       1       of         Project: SAR Groundwater Recharge Optimization Study         Job #: 1343019       Site: So,'/ Borrow         Logged By: A. Lear PF       Reviewed By:         Drilling Contractor: Boart Longyear       M W H         Drill Rig Type/Method: Sonic Drilling       h & nd Sampling         Drillers Name:       A. Lear H         Borehole Diam./Drill Bit Type:       Total Depth          Ref. Elev.	<u> </u>
Dep	th to 1	st W	ater (N			up	Tim	e/Dat	e: 🗝	Drill Start Time/Date: Drill Einich Time/Date:	
Dep	th to V	Vater	After	 Drillir	ng (V	): *	Tim	e/Dat	e:	Well Completion Time/Date:	
Dep	th to c	other \	Nater	Bear	ing Zo	ones:			/	Soil Boring Backfill Time/Date: (,-20-08	
ID/OVA	ample Interval	ecovered (in.)	low Counts / 6 in.	etained for Analysis	asing Type & Size	nnulus Filler	epth (Feet)	SCS Soil Type	Soil Descr	ription	illuciay
	X					/	1 2 3	a	Sand w Oly, ver lase, ra Cugula Swife,	ell graded with gravel, 15 35 25 25 10 y pale brown - 104R 7/4, 15 35 25 25 10 pock chips up to 4", v to rounded,	
							4 5 6 7		aps;	0814- N34°06,038' W1/7° 07.448'	
							8 9 10				
							11 —  12 —				

	n.	<u>ал</u> 1/	VLI							Boring #: 55-45 MW#:	She	eet	1	of	<u> </u>
U	עי	WIV	₩∏							Project: SAR Groundwater Recharge	e Opt	imizat	ion St	tudy	
										Job #: 1343019	S	ite:Se	sill	Bori	rdw
	<u>_</u>	<u> </u>				1999-18-1999-1999-1993-1999-1999-1999-19			]	Logged By: A. Leaulth Revi	ewed	By:			
1	1	ŝ	'sc	4-						Drilling Contractor: Boart Longyear	M	WI	4		
1		X	11-	Þ						Drill Rig Type/Method: Senic Drilling	ine	ind	Sa	mp	l:n
		X						1		Drillers Name: A. Leavit	F				
		'X X X								Borehole Diam./Drill Bit Type:	Tc Re	otal De ef. Ele	epth ev.	1	•
				Site	Sketcl	h Man				Sampler Type: hand same	1:0	٢			
Den	th to <sup>-</sup>	1st W	ater (N	<u></u>			Tim	e/Dat	e: 🖛	Drill Start Time/Date: Drill	Finieł		G-2	3-0	8-
)en	th to \	Nater	After	Drillir	na ( <b>V</b>	):	Time	e/Dat	e: -	Well Completion Time/Date:	, 1101			•	
Den	th to c	other \	Nater	Bear	ina 70	ones:				Soil Boring Backfill Time/Date:	24-	as-			
P				- Sal	ez ez							Esti	mater	1%0	f
	rval	(in.)	s/6 li	Anal	s & Si	ъ	)	ype					Sand		
¥	e Inte	ered	Sounts	ed for	Type	III St	(Feet	Soil 7					E		۲ ۸
PID/O	Sampl	Recov	Blow (	Retain	Casing	Annulı	Depth	nscs	Soil Dese	cription	Gravel	Coarse	Mediu	Fine	Silt/cla
	$\mathbf{N}$	$ \mathbf{V} $						a	Sand L	ell graded vitn gravely	15	25	25	25	10
	$ \Delta\rangle$	$  \wedge$					1		dy, ve	ry pale brown -10YR 74,					
									lucise,	ruck Chips up to 4"					
							2 —		Angulas	to rounded.		na sing Sena sing Sena sing			
										<u>en en contra destructura.</u> Referencia					
							3 —		cur fac	ex hard I man Bactal					
									with	some desert plants					
							4 —								
							5 —								
							•		OPS: C	245 - N34° 06, 096					
							6 —			M17°07,449'					
							7 —								
							8 —		<u>ela di standi i s</u> sagin di sandi i						
										, energia esta esta esta esta esta en altera. En altera de la companya esta esta esta esta esta esta esta est					
							9 —		n an						
				<mark>11. militaria</mark> Manggala											
							10 —								
								1			1		1.1.1.1	12011	122.2
							्र भूष =								
							11			harran da an an Arian an Ariante. Arran harran an an Arian Ariante.					

r										1					
	M .	N <i>I</i> II N A	N/L							Boring #: 5 - YC MW#:	She	eet	1	of	ĺ
	ע	IVI V	w M							Project: SAR Groundwater Recharge	e Opti	mizati	on St	udy	
		15		engendetter Serverk Ser		ور ۲۰۰۰ ویکی و در	a the state of the			Job #: 1343019	S	ite:Sc	sil i	Burr	とし
	$\bigcap$	X	55-0	16			1997. 	1	1001	Logged By: A. Leavitt Revi	ewed	By:			
		X						U		Drilling Contractor: Boart Longyear	$\mathcal{N}$	24	·H	×	
		X								Drill Rig Type/Method: Senie-Drilling	ha	N.	sa.	npl,	19
		X								Drillers Name: A. Leav,	17			r.	
		$\hat{\mathbf{x}}$								Borehole Diam./Drill Bit Type:	То	tal De	pth	1'	,
		X			and the second secon	and the second secon				Sec.	Re	ef. Ele	v.	<b>F</b> 2444	
		**************************************		Site S	Sketc	h Map	<b>b</b> .			Sampler Type: hand Sam	pl.	ng			
Dep	th to 1	lst Wa	ater ( <u>1</u>	<b>Z</b> ):		theorem and the	Tim	ne/Da	te: -	Drill Start Time/Date: Drill	<b>f</b> Finist	n Time	/Date	24-	-05/
Dep	th to V	Vater	After	Drillir	ng ( 👿	·): —	- Tim	e/Dat	e: -	Well Completion Time/Date:					
Dep	th to c	other \	Water	Bear	ing Zo	ones:	-	~		Soil Boring Backfill Time/Date: 6-2	4 -	·v8			
			. <u></u>	ulysis	Size							Estir	nated	% Of	
	erval	(in.)	ts / 6	or Ana	e & S	ler	(t)	Type					Sand		
AV0	le Int	vered	Coun	ned fo	g Typ	us Fl	i (Fee	Soil			-	e	Ę		ay
PID/C	Samp	Heco	Blow	Retair	Casin	Annul	Depth	USCS	Soil Desci	ription	Grave	Coars	Mediu	Fine	Silt/cl
<u> </u>	- V	$\overline{\mathbf{V}}$			<u> </u>			Si	Sandu	ell graded with gravel.			2	S2	10
	$\square$	$\overline{\Lambda}$							dy ve	ale bro-n-104R7/4,	15	45		29	
									louse,	rock chips up to 4"					
							2 -	-	G-1gula	r to rounded.					
<b> </b>										·					
							3	-	Chiffie	0 Thread Icourds ctal					
									50000	Osert plante.					
							4 -	-							
							5-			· · · · · · · · · · · · · · · · · · ·					
					 				6-85:	046 - N34° 66,161"					
							6-	-		W117° 07.451°					]
						<b> </b>									
						,	7-	-							
<b> </b>				*-											
							8-								
										· · · · · · · · · · · · · · · · · · ·					
					ļ		9								
<u> </u>							10 -	-							
							11 -			·					
	L	L	L		I .		<sup>J</sup> 12 –	1	1			L		L	04/00

G		MV	NH							Boring #:	5-47M	W#:	Sh	eet		of	L
				<u>C6</u>						Project: SP		water Recharg	je Opi	imizai		<i>a</i>	
1				Ŷ				1	MAD)	JOD #: 134	3019	(4) -	<u> </u>	lte:e	211 6	2011	ッン
1				8				l	100/	Logged By	14.120	WIT Rev	iewed	By:	1		
1				X						Drilling Col	ntractor: B	<del>sart Longyca</del> r	MI	NE	Y /-		
			,	X					/	Drill Rig Ty	pe/Method	1: <del>Sonio Drillin</del> (	+ h	and	<u>&gt;a</u>	m./~	ling
				X X ss	117	,				Drillers Na	ne: 🔏 .	Leavi	r + •	stal D	an th		1
	$\leq$			M	<u> </u>					Borenole L	iam./Drill I	Bit Type:	R	ef. Ele	ev.	<u> </u> ~	
				Site	Sketc	h Mac				Sampler T	vpe: ha	and sam	01	ing			
Dept	th to 1	1st W	ater (`	▽):		····	Tim	e/Da	te:	Drill Start T	ime/Date	24-08 Drill	Finis	n Tim	6-7	2.4-	S
Dept	th to \	Nater	After	Drillir	na (👿	): .	_ Tim	e/Da	te:	Well Comp	letion Tim	e/Date: -			<i>.,</i> Date		
Depl	h to c	other	Water	Bear	ina Za	ones:				Soil Boring	Backfill Ti	me/Date: 6-	24.0	V			
٣				/sis	ez s					1				e Esti	matec	1%0	f
	rval	j.)	s / 6 ir	Analy	& Si	r.		ype							Sand		
٨	e Inte	ered (	ounts	ad for	Type	IS Fille	(Feet	Soil T						<b></b>	<u>F</u>		<b>۲</b>
PID/O/	Sample	Recove	Blow O	Retaine	Casing	Annulu	Depth	USCS	Soil Des	cription			Gravel	Coarse	Mediur	Eine	Silt/cla
	χ	X						s	Sand w Yells	ellgrade	d, dry	y light	10	25	30	30	5
							1		nekch	ips up to	1" 2	25 Arenker	(				
							· · ·		to rou	nded.							
	14 14 - 14 - 14						~ 										
							з-										
			 						Sw fac	e hard	but.	rot to					
							4 —		compa	ctel, u	4775	oml					
									oneser	+ plan	3						
							5 —		1000	117-	- N/211	UAE 799	• •	n de la set Se se se			
									<u>-12.</u>	<u> </u>	W1170	07.297'					
							6 —										
							/-										
							8										
									<u>Contractor</u>								
							9 —										
										یندا کی در از این از بردی داد داد داد از دیکار در داد د							
							10 —										addi 1949) 1994)
							11				n Angleri Statistica		<u>: 1994)</u> 1997 - 1997				
			-														
			<u> </u>		[]		12 —	1	<u>  </u>				A START	<u>Lanadi.</u> Katar		1000	

Boring #: 55-45-MW#: Sheet 1 of (III) MWH Project: SAR Groundwater Recharge Optimization Study Job #: 1343019 Site: Soil Borrow Logged By: A. Leavitt Reviewed By: Drilling Contractor: Beart-Longycar  $\mathcal{M}\mathcal{W}\mathcal{H}$ Drill Rig Type/Method: Sonis Drilling hand sampling Drillers Name: A. Lear. ++ Total Depth Borehole Diam./Drill Bit Type: Ref. Elev. Sampler Type: Uand Sampling G-23-08 Drill Start Time/Date: Drill Finish Time/Date: Site Sketch Map Depth to 1st Water ( $\nabla$ ): Time/Date: ~ Time/Date: Depth to Water After Drilling ( $\mathbf{\nabla}$ ): Well Completion Time/Date: Soil Boring Backfill Time/Date: 8-24-08 Depth to other Water Bearing Zones: Retained for Analysis Casing Type & Size Estimated % Of Blow Counts / 6 in. USCS Soil Type Sample Interval Recovered (in.) Sand Annulus Filler Depth (Feet) PID/OVA Medium Silt/clay Coarse Gravel Soil Description Fine Sandwell graded with gravel, ely, light brown - 104R 7/2, house, rock chips yet 3", angular to rounded. \$U 13 25 25 25 15 1 -. . . 2 -22 3 --Surface is hard and compacted and some desert plants. > ns 11 4 -Sample was taken in a diainspe Matis dry. 10ts of rocks mat seem mildly comented on the 5 -22 surface, I removed thom fist 6 -7 -C-PS. 048 - N34°05,801' W117°07,374 8 -9 -10 -11 -12 QA/Q

G	Ð	<b>NN</b> <u>18</u>	VH		****			Â	(A)	Boring #: 55-49 MW#: Project: SAR Groundwater Recharge Job #: 1343019	She Opti Si	eet mizat ite:Se	1 ion St	of udy 3 <i>s</i> //	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
(		x					ł	$\bigcup$	OO)	Logged By: A.Leavith Revie	ewed	By:	1.1		
		X X								Drill Rig Type/Method: Senie Drilling	ha	nd.	San	-DI	'ng
		×								Drillers Name: A. Leav. 7	Ļ			¥	
		X V ~	< 1	100						Borehole Diam./Drill Bit Type:	То	tal De	pth	1'	۲
	$\checkmark$		27	7							Re	ef. Ele	ν.	-	
				Site	Sketc	h Map	) '			Sampler Type: Mand Samp	plin	5			
Dep	th to 1	st Wa	ater ( <u>1</u>	<b>Z</b> ):			Tim	e/Dat	ie:	Drill Start Time/Date: Drill	Finish	n Time	6/Date	-24.	-08
Dep	th to V	Vater	After	Drillir	ng (👿	):	- Time	e/Dat	e: 🐂	Well Completion Time/Date: -					
Dep	th to c	other \	Nater	Bear	ing Zo	ones:			I	Soil Boring Backfill Time/Date: 6-2	4-0	8			
	7	<u> </u>	6 in.	nalysi	Size			e				Estir	nated	% Of	ť
	nterva	ed (in	unts /	for A	ype 8	Filler	eet)	oil Typ					Sand	<u> </u>	-
470/0	mple	cover	w Co	tained	sing T	ulus	pth (F	CS S	Soil Descr	iption	avel	arse	dium	e	t/clay
JIA	Sa	Re L	B	В.	Ca	An	å	SN 1		· · · · · · · · · · · · · · · · · · ·	ö	8	₩ ₩	Ē	Sit
	IX.	X···					-4	Ser.	Sandre	11 graded with glavel	15	25	25	25	10
	لك	<u> </u>					1		10030	rock chips up to 3"					
							2-		Gazala	to rounded,					
	-				<u> </u>		3 —		C. C.	New Line 1 1					
									fritte	nactal plante nos					
			·				4 —		by but	not on sample point					
						1	5 —			······					
							· · ·		6-p5:0	149 - N34°05. SR					
					i .		6 —			W1[7007.995					
<b> </b>															
							7-								
							8								
<b> </b>															
							9 —							<u> </u>	<b>  </b>
<b> </b>											-				
<b>.</b>				<b></b>			10		· · · · · · · · · · · · · · · · · · ·						
							11								ļ
<b> </b>															
	l			<u> </u>	<u>I</u>		<sub>12</sub> —		<u> </u>		L	l		<u> </u>	

А	M.	<b>1 1 1</b>	\/ L₿						Boring #:	5° MW#:	She	et	1	of	<u> </u>
V	יש	IVI V	w F1						Project: SAR G	roundwater Recharge	e Opti	mizati	on St	udy	
									Job #: 1343019	)	S	ite: <b>So</b>	.:1	Born	202
	1							1		Leavith Revie	ewed	By:			
	/							$\mathcal{O}$	Drilling Contrac	tor: <del>Boart Longyeat</del>	M	wł	4		
								55-5 2-52	Drill Rig Type/N	lethod: Sonie Drilling	he	ind	Sa.	np	lin
								/	Drillers Name:	A. Lear, 'to	F				
	$\int$								Borehole Diam.	/Drill Bit Type: -	To Re	tal De ef. Ele	pth v.	<u> '</u>	/
				Site	Sketc	h Map			Sampler Type:	handsan	P	ling			
Dep	th to <sup>-</sup>	lst Wa	ater ()	☑):			Tim	e/Da	e: Drill Start Time/	<b>6-24-08</b> Date: Drill	Finish	n Time	/Date	24-	US
Dep	th to \	Nater	After	Drillir	ng ( <b>V</b>	): -	— Tim	e/Dat	e: - Well Completion	n Time/Date: 🗧					
Dep	th to d	other	Water	Bear	ing Zo	ones:			Soil Boring Bac	kfill Time/Date: 6-2	4-0	8			
			.in	alysis	Size			n				Estir	nated	% Of	
	iterva.	d (in.)	nts / 6	for An	/pe &	filler	iet)	il Typ					Sand		
PID/OVA	Sample Ir	Recovere	Blow Cou	Retained	Casing T <sub>3</sub>	Annulus F	Depth"(Fe	USCS So	Soil Description		Gravel	Coarse	Medium	Fine	Silt/clay
	Х	X						ŝv	Sand well grada	Jury, Pala	5	30	30	30	5
		$ \rightarrow $					- 1		Chifs apto subri	unded-runde	1				
							2 —	]							
							 3 —		surface is sin. itis@ 1', no pla	o Corrans					
									5 had						
							4 — 		6.PS: 050 -1	V34° OJ 198	1				
							5 —			11/206.859					
							6 —	1							
							7 —								
							24 								
							8 —						alaten 2013-lat		
									<u>- Alexandra ang kana kana</u> Alekandra kana kana kana kana kana kana kana ka						
							9 —			<u>na sente da secono</u> Segui de 1999 - Secono Altaria					
							 10 —								
							IV 								
							11 —								
									terretra pitetin de la composition. La composition de la composition						
<u></u>				<u>l</u>			12 —	1.40							04/01

А	۸ <b>п</b>								Boring #: ディーント MW#:	St	ieet	1	of	1
W	y "	VI V							Project: SAR Groundwater Rech	arge Op	timizat	ion St	udy	
								A	Job #: 1343019		Site:Sc	sil	Bur	102
(								1	Logged By: A. Leavitt F	eviewe	d By:			
								$\mathcal{C}$	Drilling Contractor: Boart Longye	ar /	nw	H		
									Drill Rig Type/Method: <del>Conic Dri</del>	ing he	and	Saw	pli	19
							/		Drillers Name: A. Lean	++				
Χ	$\overline{\ }$								Borehole Diam./Drill Bit Type:	T	otal De Ref. Ele	epth ev.		<u>r:</u>
				Site	Sketc	h Man			Sampler Type: hand C	and	lina			
Depth	to 1	st Wa	ater (	 ⊽):			Tim	e/Da	Drill Start Time/Date:	rill Finis	h Time	/Date	•2 Y	-08-
Depth	to V	Vater	After	/ Drillin	na (🛡	):	- Tim	e/Dat	Well Completion Time/Date:	- -	41 I III I			
, Depth	to o	ther \	Nater	Bear	ina Za	ones:			Soil Boring Backfill Time/Date: (	-261	-08			
			ċ	ysis	e			Τ			Estii	nated	% OI	f
	erval	Ĵ.	s/6i	r Anal	e &	er	<b>(</b> )	Type				Sand		
VA	le Inte	/ered	Count	led fo	g Typ	us Fill	(Fee	Soil .			Ð	Ε		۲.
0/016	Samp	Jecor	3low (	Retair	Casin	Annul	Jepth	ISCS	Soil Description	Grave	Coars	Mediu	e ine	Silt/cla
	$\overline{\mathbf{V}}$	$\overline{\mathbf{X}}$				/		a	and well graded, dry, fe	10	2	~	2.	- -
/	$\Delta$	$\square$					- -		own-101/ 43, Louise, sub-	2	50	10	20	5
									under to rounded p					
							2 —							
									resend no plante	<u>97</u>				
							3							
							4							
									PS: 051 - N34005,975 W11706.769					
							<b>.</b>			사망 가는 작품 사망				
							6 —							
							7 —							
							8							
							9 —							
									ang sang sa sang sa sang sang sang sang					
							10 —			<u>e (</u> 1995) 1981 - Angel				
									en en ser de la composition de la comp En la composition de l					
그는 아이지 않는 것이 같다.			100 <u>- 100 -</u> 100 -			<u>, a contra d</u>	11	4			4	1.25		
													1996	

		<b>₩</b> ₩	VH 						1007	Bor Pro Job Log Drill Drill Drill Bor	ing #: ect: SAR G #: 134301 ged By: A ing Contrac Rig Type/f ers Name: ehole Diam	S MM Broundw 9 . Leca Ctor: Boo Method: A . ./Drill Bi	/#: ater Rech 	Review	She Opti Si wed V A A To Re	eet mizat ite: Sa By: I W NJ S tal De of. Ele	1 ion St il i i i i i i i v.	of   udy Burn Plin 	 
Den	th to 1	st W/	ater (N		JNEIU	n wap	, Tim	e/De	te• •		Ctort Time	6-	25-08		r 61	· · · · · ·	106-	25-	08
Dep	th to V	Vater	After	Drillin	na (🐨	): -	Tim	e/Dat	 ie: -		I Completic	n Time	L Date:		INSC		ະ/ມaເe	<b>.</b>	
Dep	th to c	other \	Vater	Bear	ing Zo	ones:	,	<i></i>		Soil	Boring Bac	ckfill Tim	e/Date:	6-2	5-	08			
				lysis	ize					1						Esti	nated	% Of	
	erval	(in.)	ts / 6	ır Ana	Je & S	ller	<b>(</b> )	Type									Sand		
PID/OVA	Sample Int	Recovered	Blow Coun	Retained fo	Casing Tyr	Annulus Fi	Depth (Fee	USCS Soil	Soil Desc	riptior					Gravel	Coarse	Medium	Fine	Silt/clay
		X					1 2 3 3 4 5 6 7 8 9		Sandu dry, fo loose j -surfa Mo va GPS: 0	2	iraded brown y angu 13 Jus mbo 1 - N3 W11	wit - 10y 1ar J + 55	h grav 19 43, 10 700-1 1 10 - 20 1 10 - 20 1 10 - 20 5 , 841 2 5 - 841	· <u>e</u> [, , , , , , , , , , , , , , , , , , ,		25	30	30	
							 10 —  11 — 												

<b>MWH</b>			Boring #: 55-53 MW#: Project: SAR Groundwater Recha Job #: 1343019 Logged By: A. Least H R Drilling Contractor: Boart Longyed Drill Rig Type/Method: Senic Drill Drillers Name: A. Least Borehole Diam./Drill Bit Type:	Sh rge Op sviewed # <i>N</i> H H H H R R	eet timiza Site:Se I By: I W M M Stal D ef. Ele	1 tion S S(1) S an epth ev.	of Bori Veli	         
Site Si	ketch Map		Sampler Type: hand Sam	olang		P	- 2	11
Depth to 1st Water (☑):	► Tir	me/Date	Drill Start Time/Date: D	ill Finis	h Tim	e/Dat	e:	
Depth to Water After Drilling	i ( <b>y</b> ): 🔶 Tir	ne/Date	Well Completion Time/Date:					
Depth to other Water Bearin	ig Zones:	-	Soil Boring Backfill Time/Date:	-25-	¥			
al ) 6 in. 1alysi	Size	e			Esti	mate	d % C	nf T
nterve ad (in. Ints / for Au	ype & Filler eet)	oil Typ				Sand	k 	
PID/OVA Sample I Recovers Blow Cou Retained	Casing T Annulus Depth (F	USCS Sc	IDescription	Gravel	Coarse	Medium	Fine	Silt/clay
	1 - 2 - 3 - 4 - 3 - 4 - 3 - 4 - 3 - 4 - 3 - 4 - 3 - 5 - 3 - 4 - 3 - 5 - 3 - 5 - 5 - 5 - 5 - 5 - 5 - 5		ysand/sand-s.1t mixtur , pale o)ive -546/3, se, weak ly comented. iface was muderack i little to no plant life : 053-N34° 05,926 ~117° 07,520				50	

G		MV	VH							Boring #: <i>SS〜</i> ケツMW#: Project: SAR Groundwater Recharge	She e Opti	eet imizat	1 ion S	of tudy	<u>]</u>
										Job #: 1343019	s	ite: Se	J1د	Bur	ruc
	/							ſ	()()	Logged By: A. Lean 17 Bevi	ewed	Bv:			
1	1							J		Drilling Contractor: Boart Longyear	$\mathcal{N}$	24	H		
	X								/	Drill Rig Type/Method: Senic Drilling	ha	nds	59~	Pli	15
	×	22	7.4						/	Drillers Name: A.Leava	·+				
			· /					/		Borehole Diam./Drill Bit Type:	То	tal De	pth	1	1
1											Re	ef. Ele	v.		
				Site S	Sketcl	h Map				Sampler Type: Maril Sam	1:1	19			
Dep	th to 1	lst Wa	ater (	Z):			Tim	e/Da	ate: 🗕	Drill Start Time/Date: Drill	Finist	n Time	/Date	26-	-08
Dep	th to V	Vater	After	Drillin	ig (🔽	): -	- Tim	e/Da	ite: 丶	Well Completion Time/Date:					
Dep	th to c	other \	Nater	Beari	ing Zo	ones:				Soil Boring Backfill Time/Date: 6. 2	5-	08			
			<u> </u>	lysis	ize							Estir	natec	1%0	f
	erval	(ji)	s/6	r Ana	9 8 8	ler	t)	Type					Sand	line di	T
AVC	ole Inte	vered	Count	ned fo	I Typ	lus Fil	ר (Fee	S Soil			10	es	Ę	T	ay
PID/O	Samp	Reco	Blow	Retai	Casir	Annu	Deptl	USC:	Soil Desc	ription	Grav	Coar	Medi	Fine	Silt/cl
	Х	<u>X</u> .					 1 2	SA	silty sa light o loose.	nd, dry to stightly drag, of ive brown - 2.545/3,	5	+1	5	50	40
							 3		- mud c plants	racks on surface no					
							 A								
							4  5		GPS:0	254- N34° 05.981' W117° 07.539'					
				2021.2 2021.2 2021.2		unt l									
							6 —								
				annad Raiste											
							7 -	1							
							8								
							9 —							1	
			<u>alinii</u> Nationali	Statis Statis Park			10 —			<u>na, de la secteta de Constan</u> Regiones de constantes de constantes de			an shiri 1997 - Shiri Shiriban		
										<u> Angeler, eta esteri di bandette ina</u> Manda - den politiko di dan antista da					
							11			en na en la constante de sera de la constante de la constante de la constante de la constante de la constante En la manda de la constante de l	nin die Angelee	<u>n ten fel</u> int Tim nete	ng transfil Transfil		
				1.00				100.00	김 말로 있는 것 이 가지 않는 것 같아요.	그는 것을 설치되었다. 그는 것 같아요? 그는 것 것이 나라 가지 않는 것 같아요?					

А	n.	N/1 1/	\/LI							Boring #: 55-55 MW#:	She	eet	1	of	1
W	שי									Project: SAR Groundwater Recharg	e Opt	imizat	ion S	tudy	
										Job #: 1343019	S	ite <b>:S</b> e	>:1 1	Bo rr	y U
/	/							111	71)/	Logged By: A. Lean, H Revi	ewed	By:			
1								$\mathcal{O}\iota$		Drilling Contractor: Boart Longyear	M	wb	1		
k	-22	55								Drill Rig Type/Method: Senie Drilling	ha	.nd.	san	<i>Apli</i>	ns
	X						an a			Drillers Name: A. Lear +	4				
										Borehole Diam./Drill Bit Type:	Тс	tal De	epth	1	1
$\checkmark$											R	əf. Ele	₽V.		
				Site	Sketcl	h Map	)		<u>+</u>	Sampler Type: Nandsam	plu	٤			
Dep	th to	lst Wa	ater ( <u>1</u>	<b>Z</b> ):			Tim	e/Dat	e: 🗕	Drill Start Time/Date: Drill	Finisł	n Time	e/Date	-25 9:	-08
Dep	th to \	Nater	After	Drillir	ng ( <b>T</b>	): •	<b>~</b> Time	e/Dat	e: 🖊	Well Completion Time/Date:					
Dep	th to c	other \	Nater	Bear	ing Zo	ones:		<u> </u>		Soil Boring Backfill Time/Date: 6-	25-	58			
			0 in	nalysis	Size			Φ				Esti	mateo	1%0	[] [
	nterva	ja (in.	Ints /	for Ar	ype &	Filler	eet)	ii Typ					Sand		
AVO/GI	ample I	lecovere	low Col	letained	asing T	unun	)epth (F	SCS Sc	Soil Desc	ription	iravel	oarse	ledium	ine	ilt/clay
LL	$\overline{\lambda}$	X				4		M	silty s	and, dry tosjightly	5	tr	5	50	20
		1					1		anp,	light olive 5rown - 20173	<u>,</u>				
									10050.						
							2-		-mude	cracks unsurface-no					
							3_		plan	<i>h</i> .					
							4		100.	NE ATUON ACUS					
									<u>~13 ·</u>	USS - 10 37 06:03 7 W/170 (1) 5 549'					
							5								
							7-								
				Sec. 12.22	111111111111111		1. The test set	1 · · · · · · · · · · · · · · · · · · ·					1 1 1 1 1		1.555
										<u>19 Ben kan dan berkeran di kerdapat kerden dari</u> Berkeran					
							 8 —								
							8-								
	 						8 9								
							 8 9  10								
							 8 9 10								
							8 —  9 —  10 —  11 —								

	n,								Boring #: SS-56 MW#:	Sh	eet	1	of	1
U	יש	WI W		l					Project: SAR Groundwater Recha	rge Op	limiza	tion S	tudy	
							i i i i i i i i i i i i i i i i i i i		Job #: 1343019	٤	Site:	0:11	3051	-01
	1	٢						$\bigwedge$	()) Logged By: A. Leavitt R	eviewed	By:			
	×	55-7	-					$\mathcal{O}$	Drilling Contractor:-Beart Longyer	TM	wf	J		
1	1-	~ \	0						Drill Rig Type/Method: Senic Drill	ng-h	and	Sau	Pl;	14
1								/	Drillers Name: A. Leavi	11				
l							Ζ	/	Borehole Diam./Drill Bit Type:	T(	otal D ef. Ele	epth ev.		
				Site	Sketc	h Mar	<b>)</b>		Sampler Type: hand san	PIN	16			
Dep	th to	lst W	ater (	<b>∑</b> ):			Tim	ne/Da	Brill Start Time/Date:	ill Finis	 h Tim	e/Dat	•25- =:	S
Dep	th to \	Nater	After	Drillir	ng (🛡	r): —	– Tim	e/Da	Well Completion Time/Date:					
Dep	th to c	other	Wate	r Bear	ing Z	ones:			Soil Boring Backfill Time/Date:	-25-0	-8			
			Ŀ	ysis	ize						Esti	mated	1%0	f
	ərval	(in.)	s / 6 i	r Anal	e & S	ler	()	Type	1996년 2017년 1월 2017년 1월 1997년 1997년 - 1996년 - 1997년 1월 19			Sanc	1	Γ
PID/OVA	Sample Inte	Recovered	3low Count	Retained fo	Casing Typ	Annulus Fil	Jepth (Fee	JSCS Soil	Soil Description	Gravel	Coarse	Aedium	eui	
	$\overline{\mathbf{N}}$	$\overline{\mathbf{V}}$				<b>4</b>		En	silty sand slightly damp.					
	X	$\Lambda$							1154+ 01 Ne brown-2.545/2,	\$	3	12	50	-
									1005e,	••••••••••••••••••••••••••••••••••••••				
							2		mund cracks orsurfaco-no					
				[ 			3-							
							4 -		- LOCLARY ALZIERAL DEC	- 10 - 10 - 10				
									WILT 07.565	<del>.</del>				
					<u></u> %									
		1999) 1999					6 -					<u>, 1997</u> 1997 1997		
							7-							
							8-							
														1000
							9 -							+
									an a					
							10 -							
		1212-222 1313-334		1		400088	1000							
				1000			4.4	100.000	<u>ender de la presentación de la compañía de la presentación de presentación de la presentación de la presentación</u>	<u>(* 1917)</u> - 1917		1.1	1	
				 			- 11							

	р I	WW	VH							Boring #:5-5-7 MW#:	She	et	<b>1</b>	of	1
V	י ש	1 WO BI 199	1997 1951 195							Project: SAR Groundwater Recharg	e Opti	mizat	ion St	tudy	
								4		Job #: 1343019	S	te:So	oil	Bon	n
1	X	55-s	57					$\bigwedge$	DDT	Logged By: Alean, H Revi	ewed	By:			
1								V	~/	Drilling Contractor: Boart Longyear	M	V	H		
										Drill Rig Type/Method: Sonie Drilling	-ha	.nd	Sau	npl	ing
								/		Drillers Name: A . Leavit.	$F_{-}$				
ſ										Borehole Diam./Drill Bit Type:	То	tal De	epth	1'	•
											Re	ef. Ele	V.		
				Site	Sketc	h Map				Sampler Type: hand sar	201.	ng			
Dep	th to <sup>-</sup>	lst W	ater ()	<b>∑</b> ):		<	Tim	ne/Da	ite: 🗕	Drill Start Time/Date: Drill	Finish	Time	e/Date	6-28	3
Dep	th to \	Nater	After	Drillir	ng (👿	y): ~	– Tim	ie/Da	te: -	Well Completion Time/Date:					
Dep	th to c	other	Water	Bear	ing Z	ones:				Soil Boring Backfill Time/Date: 6-2	5-0	18			
			. <u>ċ</u>	alysis	Size						ing ang sa Tang sa t	Esti	mated	1%0	
	erval	l (in.)	its / 6	or Ané	De & (	ller	<b>3</b> ť)	Type					Sand		
AVC	ole Ini	verec	Cour	ned fo	IVT BI	lus Fi	л (Fee	S Soil			0	Se	Ę		ay
PID/0	Samp	Reco	Blow	Retai	Casir	Annu	DeptI	nsc	Soil Des	cription	Grav	Coan	Medi	Fine	Silt/c
	$\mathbf{\nabla}$	X						SM	silty se	and, dry, light of ive bran				C	2
	$  \land  $	1/1					- 1		2.5451	14, loust, ruck chips up h	2				
									2", 194	bargalar to rounded.					
							2 —								
									- mine	docer folget	•				
							3								
			-		<b> </b>		,								
	 						4		LPS:	057 - 1 340 06. 136'					
							5			w11707.522'					
							6 -					an an a' saoi an an an a' saoi			
•									<b></b>						
							7-								
							8								
							9 —	-							
										nen og han skriver for det skriver af som					
					aaguste Salyaste		10 -								
	100 C 10 C 10 C 10 C	and a start of the	+initiation in the second seco	+		1	11 -	100	1	titististi aasta kan kan kan kan kan kan kan kan kan ka					

1	n.	<b>N<i>A</i> 1</b> 4	r i Li							Boring #: 55-376/MW#:	She	eet	1	of	1
W	יש	WIW								Project: SAR Groundwater Recharge	e Opt	imizat	ion St	udy	
				1	21					Job #: 1343019	S	ite: Oq	ster	PS	nıl.
				n'	1				}	Logged By: A. lear, H Revie	ewed	By:			
			1	11			. • \		/	Drilling Contractor: Beart Longycar	N	<i>w</i>	4		
				$\backslash I$		5	011	,		Drill Rig Type/Method: Sonic Drilling	h	and	San	npla	
		e		21		yo	cou	. /	/	Drillers Name: A.Leauth	•				
		25.	58 7	$\left( \right)$				<b>/</b> (		Borehole Diam./Drill Bit Type:	Tc Re	otal De ef. Ele	epth v.	<u>/'</u> ~	
			L	Site	Sketc	h Mar	)			Sampler Type: hand sam	pl	ing			
Dep	th to <sup>-</sup>	lst W	ater ('	▽):			Tim	e/Dat	•	Carll Start Time/Date:	-v Finisł	) Time	6. Date	26	46
Dep	th to \	Nater	After	 Drillir	na ( <b>V</b>	): -	– Tim	e/Date	ə: •	Well Completion Time/Date: -			<i>,, Da</i> te		
Dep	th to o	other	Water	Bear	ing Zo	ones:				Soil Boring Backfill Time/Date: 6-2	.6-9	8			
				lysis	ize (							Estir	nated	% 0	f
	arval	(in.)	s / 6 i	r Anal	e & S	er	t)	Type					Sand		
ID/OVA	ample Inte	ecovered	low Count	letained fo	asing Typ	nnulus Fil	lepth (Fee	SCS Soil	Soil Desc	ription	iravel	oarse	ledium	e.	ilt/clay
<u>0</u>	$\sim$		B	۳.	0	A		⊃ .cw	Coad	Oll are day i sty and	U		2	LL.	S S
	ľ	$\wedge$							moist	0/ive brown (2.5-44/4)	30	30	25	5	1
							1-		loose,	rockschips up to 3"					
							2 -		suban	gular to angular Asc					
									round	lecl.					
							3 -		ecurt	are hand mudareciss					
							1993		andd	riod algae.					
							4								
							5-						na tau Mang		
							6		-6PS;	058 - N34005,847' W 117007.597					
							7-								
							8 –								
							9 –								
				na di Kaler			10 -		<u>elentis di</u> Recultos						
										ka di mana di 1999 yang di 1999 y Mana di kang di					
							11 -		<u>er une en l. 1.</u> Afriktiskel						

A	n (	ЛV	VH							Boring #: 51-51MW#:	She	et	1	of	
y	<b>v</b> •				•	<b>e</b>				Project: SAR Groundwater Recharg	e Opti	mizati	on St	udy	
				,K	1				<del>)</del>	Job #: 1343019	S	ite:	• • • /	- 1200	10.
			1	/}/					1	Logged By: A. Leavitt Rev	iewed	By:			
			1	//	(	Soil				Drilling Contractor: Boart Longycar-	Л	14	1H		
	Ó	• • • •	L		1	000	row			Drill Rig Type/Method: Sonie Drilling	<u>r ha</u>	.nd.	sa	mpl	ine
	13	-37	ľ(	1			~	Ζ		Drillers Name: A. Leavi	17				
			1,	$\mathcal{I}$			~			Borehole Diam./Drill Bit Type:	То	tal De	pth	1	•
			1	(							Re	ef. Ele	v.	-	
	$\begin{array}{c} \textbf{WWH} \\ \hline \textbf{Ss.59} & ( \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c} \begin{array}{c}$		<u>.</u>	Sampler Type: handsan	~Pli	15									
Dep	MWH         S5-59       Soil         Site Sketch Map         In to 1st Water (\[mathbf{D}]:       Time/Date:         In to ther Water After Drilling (\[mathbf{T}]:       Time/Date:         In to other Water Bearing Zones:       Image: Control of the set			te: –	Drill Start Time/Date: Drill	Finisł	n Time	/Date	-a 1	;-4					
Dep	th to V	Vater	After	Drillir	ng (👿	):	- Tim	e/Dat	le: -	Well Completion Time/Date:	-				
Dep	th to c	other \	Water	Bear	ing Zo	ones:		-	1	Soil Boring Backfill Time/Date: 6	26-	98			
	_		Ľ	alysis	Size			l o				Estir	nated	% Of	
	iterva	d (in.)	nts / (	for An	/pe &	iller	et)	II Typ					Sand		
OVA	Iple Ir	overe	V Cou	lined	T gui	ulus F	th Fe	S So	0 10		ē	Ise	<u>E</u>		clay
<u>Ö</u>	Sarr	Rec	Blov	Reta	Cas	Ann	Dep	nsc	Soli Desc	ription	Gra	Coa	Mec	Ц. Ц.	Silt/
	$\mathbf{V}$	X						SW	Siltys	and I sand well grades,	15	24	24	2 <	22
	$\Lambda$	$\square$					1 —	SM	moisd	-, olivebrown (2,54				10	
									974),1	loose 1so Ft, weekly					
							2 —		Cemer	<i>тей</i> .					
				-											
							 		-surf	ace had dried algae.					
							4 —								
									- GPS	059 - N 390					
							5 —			<u> </u>					
							6	]							
							7								
							2- -								
							8 —								
							9 —								
							10								
							11 -								
	<u> </u>		<u>haddidd</u> Marthura	<u>la dela de</u>	<u>La la la la</u>		12	1	1				nter isl Geografie		QA/QI

П	n.	<b>1</b> 1 1							Boring #: <b>55 - 64</b> MW#:		She	et	1	of	1
M	יע	VIV		0	1				Project: SAR Groundwater Re	echarge	Opti	mizati	on S	tudy	
				Λ,		and the second secon	A PARTY OF THE OWNER	and the second secon	Job #: 1343019		Si	ite: <b>ov</b>	cter	· pon	d!
	64	55-6	ه (	1					Logged By: A.Leavit	7 Revie	wed	By:			
			p*1				•		Drilling Contractor: Beart Lon	<del>gyoar</del>	N	14	H		
			12	>		Soi	,1		Drill Rig Type/Method: Senic	Drilling	ha	nd.	59.	npl	ing
			14	\$1		50	ron	)	Drillers Name: A. Lea	いけ	F				
				$\left\{ \right\}$	~				Borehole Diam./Drill Bit Type:		To Re	tal De of. Ele	pth v.	<u> </u> ' _	
				Sito (	Skotel	n Man			Sampler Type: Mg al	Ca		dia	) c		
Den	th to 1	Ist Me	ater / <b>N</b>	<b>7</b> ).		-	Tim	e/Det	Drill Start Time/Data		inich	Time	J VDati	-26	-95-
Dep	th to V	Nator	After		ua (🕊	<u>،</u>			Well Completion Time/Date:		nnsr	<u>i inne</u>		3.	
De-	th to	water	Motor	Boor	ing <b>7</b>	<u></u>	11111		Cell Device Declafil Time/Date:		6	or			
р		Juner V	vater	⊡ear . <u>%</u>	mg ∠c	ones:			Soil Boring Backtill Time/Date	: 0-L	<b>v</b> -(	-0		1 0/ ~	•
	ଜ୍ଞା	(:	/ 6 in.	Inalys	& Siz			be				Estir	nated	1%0	T T
4	Inter	ed (ir	unts ,	1 for /	Lype	Filler	Teet)	oil Ty					Sanc		
PID/0/1	Sample	Recover	Blow Co	Retained	Casing 1	Annulus	Depth (F	nscs s	Soil Description		Gravel	Coarse	Medium	Fine	Silt/clay
	X	X					-	Si/	Sand well graded, moist	5 <u>)</u>	5	25	25-	35-	18
		· · · · · ·					1		looge.						
			<u>Electropicae</u> Statisticae References				2 —		- surface had this silt						
							3-		muderack layer and alg	iae					
							4		6PS:060 - N34°06,0	78'					
					en sin South		5 —		<u> </u>	99					
			_				1943 <u>- 1</u> 1948 - 1								
							6 —				<u>11 (184)</u> 11 (1944)				
							8-								
		Strategicker			3433							Store Star Starte			
										1		1	11.011.01	1.1	4.5555
							9-				alli shi Magazi				
							9								
							9								
							9								
							9 10 11								

	<b>V</b> '			~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	· ٢	)			_	Project: SAR Groundwater Recharg	e Opt	imizat	tion SI	tudy	
				5,	//r		and a star point of the second se	Change and a second		Job #: 1343019	S	ite: o	nte	r Pe	rnd
		- a 1	. []	' /	Π	1	e Î			Logged By: A. Lean H+ Rev	iewed	By:			
		5-6	'Ŋ			5	011			Drilling Contractor: Boart Longyear	M	W	14		
			$\prod$	11	/	6	03510			Drill Rig Type/Method: Sonie Drilling	r ha	rol	59 -	pl.	ng
			15							Drillers Name: A.Leav	1+	·F			
				$ \rangle$		and the second secon				Borehole Diam./Drill Bit Type:	To	otal De	epth	1	/
			i l							A	R	ef. Ele	ev.	<b>~~</b> ~	
				Site	Sketcl	n Map				Sampler Type: hand Sa-	npl	ing			<i>a</i>
Dep	th to	1st W	ater (	<b>▽</b> ):			Tim	e/Dat	e: _	Drill Start Time/Date: Drill	Finisł	n Time	e/Date	-28 ::	5-0
Dep	th to V	Water	After	Drillir	ng ( 🔽	): `	- Time	e/Dat	e: 🖌	Well Completion Time/Date:					
Dep	th to o	other '	Water	Bear	ing Zo	ones:	<b>~~</b>	-		Soil Boring Backfill Time/Date: 6-2	26-	08			
	_		3 in.	alysis	Size			0				Esti	matec	1%0	f
	iterva	d (in.)	nts / (	for An	/pe &	iller	et)	I Typ					Sand		
OVA	ple Ir	overe	/ Cou	uned	ng T)	ulus F	th (Fe	S So			/el	lse	ium		lav
PID/	Sam	Rec	Blow	Reta	Casi	Anni	Depi	nsc	Soll Desci	ription	Grav	Coa	Med	Fine	Silt/c
	X	<u>\</u>						s.	Sand w	ellgraded with	10	2	2-	2-	14
	1	$\Lambda$					1 —		Sparel	dry, palebrown		C 1	-3		
	<b> </b>								CloyR	(13), rock chips upt	2				
	<u> </u>						2 —		1. 5 )-	subangular to round	<u>er</u>				
					<b> </b>					· · · · · · · · · · · · · · · · · · ·					
					[		3	1	esurf	are had this layer					1
	ļ	·		ļ	ļ		4 —		of 5:11	ty mudcracks and			ļ	ļ	ļ
				l	<b> </b>				algae.						
							5 —		( De :	n/1 Alawanimiai					
		<b> </b> -							<u>rn[] []</u>	WI17 07 749'					
							6 —					1			
							7 —					ļ	<u> </u>		
		<b> </b>			 										
							8 —				-				
		<b> </b>									-				
	· · · · · ·	1	ł				9 —	1				1			
	· · · · · · · ·			1		. 1		1				1			ł
							40			· · · · · · · · · · · · · · · · · · ·					
							10 —								
							10 — 		· · · · · · · · · · · · · · · · · · ·						

	n,	лл 1	n/l						Boring #: 55-6 2MW#:	SI	ieet	1	of	1
y	יש	IAI A		<i>/</i> /7	ę	1			Project: SAR Groundwater Rec	harge Op	timiza	ition S	tudy	
				. –	À-				Job #: 1343019	(	Site: C	riter	Po	20
			//	' /	<i>}</i> [				) Logged By: A.Lear Ht	Reviewe	d By:			
			K	$\parallel$	1	50	1:0		Drilling Contractor: Beart Long	ear /	11	vt,	4	
			K	/}	1	bi	.crou	ノ	Drill Rig Type/Method: Sonic D	illing h	a nd	sa	-Ai	1
			}	12		ýc			Drillers Name: A. Lear	37				•
	S	5-62	PS	$\mathbb{N}$	<u> </u>				Borehole Diam./Drill Bit Type:	Т	otal D	epth	1	1
										F	lef. El	ev.		
				Site	Sketc	h Map			Sampler Type: Mand	anp	'lin	5	~ -	
Dep	th to	1st W	ater ()	<b>⊻</b> ):		~	Tin	ne/Da	e: Drill Start Time/Date:	Drill Finis	h Tim	e/Date	e:	Ø
Dep	th to \	Nater	r After	Drillir	ng (👿	'): <u> </u>	— Tirr	ne/Da	Well Completion Time/Date:					
Dep	th to o	other	Water	r Bear	ing Zo	ones:		-	Soil Boring Backfill Time/Date:	6-20	5-0	8		
	<b>"</b>		e in.	nalysi	Size			Ō			Est	imateo	1%0	f
	nterve	d (in.	ints /	for Ar	ype &	Tiller	set)	il Typ				Sand	l	4
NOVA	nple I	overe	w Cot	ained	sing T	Inlus	oth (F	CS Sc	Soil Description	Vel	arse	nin	0	
문	Sar	Rec	B	Ret	Ca:	Anr	Del	- NSI		- D 20	<u> </u>	Å	Ē	
	X	X		<b> </b>			-	S.	Sand well graded w. Tr gra	el 15	- 20	20	25	
	()						1 -		and silt, dry, lellowish	0				+
								-	rock chips up to 2". sub	<u></u>				
							2 -		angular to rounded.					t
							3 -							
							-		-surface had a thin la	101				
							4 -		ot silty muderacks e.	<u>d</u>				+
							-							
							5-		. CPS:062-N34° 05.8	76'				T
							6 -		W117°07.71	5-1				
								-						
							7 -							
				1										
							8 -							t
							9 -							
				<b> </b>				-						
							10 -							+
			1											
							11 -							t
		1		1				-1						

									···· ···	Boring #: CS=6 3 MW#	She	oot	1	of	(
		٧N	VH			A	0			Project: SAR Groundwater Recharge	e Onti	imizat	ion St	udv	·
			C	4	13	n	(			.loh #: 1343019	s opt	ite: 🔊	ater	· Por	ret 4
			R	r O	$\langle / \rangle$	8/	1			Logged By: A. Les with Bevi	ewed	By:			10 1
				$ \rangle$		] /	(  .		1	Drilling Contractor: Board ongyear	٨٨	by.	4		
		ł	Iζ	K	1/			>0 '	crow	Drill Rig Type/Method: Serie Brilling	ha	rh 6		AL.	1 <
		(	5		{		$\mathcal{A}$	6	)1	Drillers Name: A LEGUE +	4		. 84		)
			12	5	[(					Borehole Diam./Drill Bit Type:	To	otal De	pth	1	,
		5.	<i>3</i> 2 2	5		V	J				Re	ef. Ele	v.	- 1 Africa	No.
				Site	Sketc	h Mar	'n			Sampler Type: how Sam	-01	. A 6	4		
Dep	th to 1	st Wa	ater (N	$\nabla$ :		<u> </u>	 Tim	e/Dat	e. 🗢	Calificit Type. 6-26-05	Finish	Time	<b>6</b> Date	- 26	-95-
Dep	th to V	Vater	After	Drillir	na ( <b>V</b>	(): ~	 Tim	e/Dat	e: "	Well Completion Time/Date:	1 11 11 51	1 1 11110	Dale	•	
Dep	th to c	other \	Nater	Bear	ina 70	ones:				Soil Boring Backfill Time/Date:	26-0	28			
				/sis	e N			Ī				Estir	nated	% Of	;
	rval	in.)	s / 6 ir	Analy	& Si	er l		ype				1	Sand		
A/	e Inte	ered (	counts	ed for	Type	s Fille	(Feet)	Soil T					5		>
	ample	ecove	low C	etaine	asing	nnulu	epth	scs	Soil Desc	ription	ravel	oarse	lediun	ne	ilt/clay
	S S	ш Г\ /	<u> </u>	<u> </u>	0	<			- 00 1	sil ele tal la Vella	U L	0	Σ	LE .	S
ŀ	X	X						1.~	barran	( INTR \$6) LOOSE OCK	10	25	52	25	15
		<u> </u>					1-		Chips .	apto 1.5" Subargular		1			
							2-	]	to rou	ndel.					
							3	4.							
					<b> </b>				- Surfa	ce had thin layer					
							4		0 7 311	ty muderachs and					
					 			1		·					
							5-		- CPS !	063-N34°05,777	,				
							6-			W11707,760	r				
					<b> </b>	l									
<b> </b>							7 —								
							8 —								
							 0								
							9								
<b> </b>							10								
<b> </b>					<b> </b>										
<b> </b>							11 -								
					<u> </u>										
<u> </u>	I		L	I	I	ł	' 12	I	L		L	L			QA/QC

	n.									Boring #: <b>\$1-64</b> MW#:	She	eet	1	of	1
G	יש	VI V	V H	e	Ч	0	5 P	١		Project: SAR Groundwater Recharg	je Opti	imizat	ion S	tudy	
				$\hat{\mathcal{O}}$	> 0,	, Y	7)			Job #: 1343019	S	ite: 🔿	2+61	r pa	14
				15	$\hat{\Lambda}$	'/{	7{[	1.	<i .<="" td=""><td>Logged By: A. Learitt Rev</td><td>riewed</td><td>By:</td><td></td><td></td><td></td></i>	Logged By: A. Learitt Rev	riewed	By:			
			[	5	2	1/1	'À	>0		Drilling Contractor: Boart Longyear	- /	n	Nt	1	
		55	-67/3	ζ [	$\left\{ \right\}$	31	$\mathbb{N}$	Vi	>110	Drill Rig Type/Method: Sonic Drilling	z ho	and	sa,	mf	ling
				11		{	$\langle \rangle$			Drillers Name: A. Lear	++				
				C	S	7 /	L			Borehole Diam./Drill Bit Type:	To	tal De	pth v	1	•
				~	~ .					1 I co.				<u> </u>	<u>41.</u> 31. 51. 1
		-1 167		Site	Sketc	n iviap		- (D - 1		Sampler Type: 1 2 1 . 2 6 - 0 8	7711	">	6	5-26	-08
				<u>v):</u>		<u>~</u>		ie/Dai	e: -	Drill Start Time/Date: Dril	Finish	n Time	/Date	<b>):</b>	
Dep		vater	Atter	Drillir	ng ( 🔽	): -	- IIM	e/Dat	e: -	Well Completion Time/Date: -	21	nC/			
	tn to c	otner v	vater	Bear		ones:				Soil Boring Backfill Time/Date: 0 -	-6-				
	kal	(;:	/ 6 in.	Analy	& Siz			ed,				ESUI	nated	u 7⁄0 U	
¥	Inter	red (ii	ounts	d for /	Type	Fille	Feet)	Soil T						l T	
PID/0/	Sample	Recove	Blow Co	Retaine	Casing	Annulus	Depth (	nscs s	Soil Desc	cription	Gravel	Coarse	Medium	Fine	Silt/clay
	X	V.						se	Sand u	-oll graded, most, dark	s la	25	25	25	10
	//	()					1 -		Yella	34 brown (104R 3/6),					
									loose,	rodschips nph 2'					
				<u>antana</u> Setu			2 –		545 97	sain to rounded.					
							2								
							з 		-Surfi	ace had thin layer					
							4		ofsi	Ity muderack's and					
						-			algae		-				
							5 —		- CPS .	Phy Alguent and	<u> </u>			<u>1197</u> 1988	
									- ~,/ .	W117'07 820'					
							6	1							
							7 -								
							8 –								
										<u>na stop o na stala ta di na solo di di</u> La kao di sa sua stala di sa stala se s					
				ni nen Nga sé			9 -						ning de Proposition		
							10 -								
					+	أشتصيفها	11	- 10 Carl	àn an de contra contra de contr					1	

G		viv	VH			-	0	•		Boring #: <b>\$\$-65</b> MW#: Project: SAR Groundwater Recharge	She e Opti	eet mizat	1 ion Sl	of tudy	1
	$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		Job #: 1343019	 s	ite: <b>Ø</b> v	ater	Por	124							
			(	SF	>~:		: K		an a	Logged By: A LOGN HT Bevi		Bv <sup>,</sup>			
			55'	Ŕ	Π	11	'\\/	Sc	251	Drilling Contractor: Boart Longvear	Λ	<u>-,.</u> 1 L	11+	!	
			1	17	$\left \right\rangle$		\$   {	10	3.000	Drill Big Type/Method: Sonic Drilling	he	nd .	san	noli	16
				2	$\mathbb{N}$		N	. У	0110-	Drillers Name: A. Learit	4	42 488		<u> </u>	9
			1	く	В		/	$\overline{)}$		Borehole Diam./Drill Bit Type:	То	tal De	pth		, '
					Ž			{			Re	ef. Ele	v.	and a second	
				Sito	Skoto	h Mar				Sampler Type: hand can	~~~	lin	4		
	th to 1	et Ma	ator (S		SKeic		, Tim	o/Dat	a	Duill Ctart Time/Data:	- 1/	Time		5-26	-08
Dep	th to V	Notor	After			·)•	Tim	e/Dai		Drill Start Time/Date: Drill	Finisr		Date	): 	
Dep	th to c	ther	Motor	Dillin	iy (y	)				Coll Basian Bastelli Time/Date.	6-13	v			
Dep		other V	vater	Bear .g		ones:				Soil Boring Backfill Time/Date:		8			
	al	(;	/ 6 in.	Inalys	& Siz			be				Estil	natec	1 % Of	
	nterv	ed (ir	unts /	for A	ype	Filler	eet)	oil Ty					Sand	T	
1 NO/	nple	over	w Co	ained	T guis	snIn	oth (F	S S S S	Soil Descr	intion	wel	arse	dium	0	'clay
	Sar	Rec	Blo	Ret	Cas	Anr	Det	NS N			Gra	Ö	Me	ii.	Silt
	$  \rangle$	$\mathbf{N}$				 		5~	Sand .	-ell graded, moist,	10	25	20	25	10
	$ $ $\land$	$ \land $					1-	1.	dark,	Yellow 34 brown Cloyne		<u> </u>			
									1005R,1	rock chips up to 2"					
							2		Sub ang	alar to roundly,					
							3		- surf	ace had Thin layer				<u> </u>	
						1			0f5.74	y underactes and					
							4-	]	algar	· · · · · · · · · · · · · · · · · · ·					
L						 	5-			, 					
									· 6PS -	065 - N34°06:105'					
							6-			W117°07.882					
							7 —	-							
							8								
}															
							9-	]		· · · · · · · · · · · · · · · · · · ·					
						L	10-					ļ		<u> </u>	
<b> </b>					<b> </b>	<b> </b>									
							11 -					<b> </b>	<b> </b>	ļ	<b>  </b>
<b> </b>															
						L	<sup>]</sup> 12 –		l			I	L	<u> </u>	OA/OC

		WW	νщ							Boring #: <b>\$ \$-6 \$</b> MW#:	She	eet	1	of	1
W	<i>V</i> •	<b>88 8</b>				0	Q	0	);	Project: SAR Groundwater Recharg	e Opt	imizat	ion St	udy	
				es.		2	5	. /	1	Job #: 1343019	S	ite: 🔎	ute	r pa	2.70
				R	> / ſ	~~?	/ζ,	5		Logged By: A.Lewitt Revi	ewed	By:			
				12	'   {	$\left[\right]$	131	1/		Drilling Contractor: Beart Longyear	M	wŀ	4		
				13	RI	{	)(	$\prod$	Soil	Drill Rig Type/Method: Sonic-Drilling	- hi	hnd	اجد	npl	in
		55-6	6	r}	}	}	$\langle \langle \rangle$	)	borrou	Drillers Name: A. Leav	22				
			/		$\mathbb{N}$		$\langle \rangle$	L		Borehole Diam./Drill Bit Type:	Тс	otal De	epth	<u>)'</u>	
			C			S'	5	C	フ	-	R	ef. Ele	v.		
				Site S	Sketcl	h Map	•	-		Sampler Type: hand Sa	np	ling	Ĩ		
Dep	th to 1	st Wa	ater ( <u>S</u>	<b>Z</b> ):		-	Tim	e/Dat	e: 🖌	Drill Start Time/Date: Drill	Finisł	n Time	e/Date	5-2 ::	6~~
Dep	th to \	Vater	After	Drillin	ig (👿	):	<b>`</b> Tim	e/Dat	e: 🛩	Well Completion Time/Date:					
Dep	th to c	other \	Nater	Bear	ing Zo	ones:	1000-			Soil Boring Backfill Time/Date: 6-	26	- 08	~		
			Ŀ.	alysis	Size							Estir	mated	80	f
	terval	1 (in.)	its / 6	or Ani	pe &	iller	et)	Type					Sand		
AVC	ole Int	verec	Cour	ned f(	IV Br	lus Fi	h (Fe	S Soil			e	se	Ę		av
PID/0	Sam	Reco	Blow	Retai	Casir	Annu	Dept	nsc	Soil Descr	iption	Grav	Coar	Medi	Fine	Silt/C
	$\bigvee$	X						su	Sandw	ell graded, must, dark	la			3	
	$ \Delta $	$ \Delta$					1	4	Yellow,	34 brown (10 4R 4/4),	10	45	25	12.5	
									loose,	rock chips up to 2",					
							2 —	-	Sub any	gular to sounded					
											-				
							3 —	1	= Surf	ece had min la pr	+				
									v + si	lty muderacks and					
							4		algae	( 					
							5 —								-
					<b> </b>				· GPS+	068 - N34°06°143'					
				 			6 -	-		Wi1708.036					<u> </u>
								-							
							7	-	······				<u> </u>	<u> </u>	
					<i>.</i>					·					
							8-	]							
							<u>9</u> -	_							<b>_</b>
				<b> </b>	<b> </b>										
	ļ				<b></b>	<u> </u>	10	-							
					<b> </b>			-							
							11 -	-							
								1			4	1			1

(	ÿ ı	МV	VH							Boring #: 55-67 MW#:	She	et	1	of	1
V	<b>"</b>	s a W		~				<u>,</u>		Project: SAR Groundwater Recharg	je Opti	mizat	ion St	udy	
			,	rs 7	P	1	12	17		Job #: 1343019	S	ite: 🔿	~+&1	po	nd;
			1	{	R	P	タノ	1		Logged By: A. LearIH Rev	iewed	By:			
		Ss-	67	XI	151	21	513	1	1	Drilling Contractor: Beart Longyear	M	wh	<b>L</b>		
				$\langle  $	31	317	15	1:	2011	Drill Rig Type/Method: Sonic Drillin	zha	rds	6-0	pli	ng
				$\langle  $	$\{  $	$\langle   \rangle$	$\left\{ \right\}$	1	001100	Drillers Name: A. Learit	F				
			l	5	$ \langle  $	3 K	$\langle   \langle$	L >		Borehole Diam./Drill Bit Type:	То	tal De	pth	1'	
			ļ	U	C	ינ	$\sum$				R	ef. Ele	v.	-	
				Site	Sketcl	n Map				Sampler Type: Nandsan	pli	19			
Dep	th to 1	st Wa	ater (	☑):		~	Time	e/Da	le: 🗖	Drill Start Time/Date: Dril	Finisł	n Time	/Date	-26	-08
Dep	th to V	Vater	After	Drillir	ng (👿	): -	Time	e/Dat	e: 🖌	Well Completion Time/Date:					
Dep	th to c	other \	Water	Bear	ing Zo	ones:		-		Soil Boring Backfill Time/Date: 6-	26-0	8			
			. <u> </u>	Ilysis	Size							Estir	nated	%0	ſ
	erval	l (in.)	ts / 6	Jr Ane	be & S	ller	¥)	Type					Sand		
AVC	ole Int	vered	Coun	ned fc	Ig Tyr	Ius FI	ר) (Feé	3 Soil			6	3e	Ę		ay
PID/O	Samp	Reco	Blow	Retai	Casir	Annu	Deptl	USC(	Soil Desc	ription	Gravi	Coar	Medi	Fine	Silt/cl
	$\nabla$	$\overline{\mathbf{V}}$						su	Sanda	ell graded, dry, light					
	$ \Lambda $	X							Yellari	subrown (INR 6/4),	10	25	25	25	15
									loose, 1	ruck chips up to 1.5 "	-				
							2—		suban	sular to rounded.					
					10-1219 10-1219		3 —		- Surcfa	cold thrula pr	s stati Posta	gingi Mariji Mariji			
									of SAL	ty mud cracks and					
							4		algae						
							5 —								
									- GPS:	067-1/3406.261	<b>/</b>				
							6 —			WIL 1008.018					
									Service of the servic						
							7 —								
											-				
							8 —								
							o —								
							10 —								
							11 —					<u>landi</u> Sigar			
	1	<b>I</b>	<u>I</u>	<u>1 191166</u>	<u>n astali</u> Generali	1	12 —	1 <u>el el .</u>	taa ahaa ahaa		etteria. Secondaria	I Alan and A	L	1	QA/QC

6	n.	<b>\//</b> 1/	VL							Boring #: S5-65 MW#:	She	et	1	of	
<u></u>	יע	IAI A		P5	Pe	1 。	Pz	P	1	Project: SAR Groundwater Recharg	e Opt	mizati	ion St	udy	
		53		P	B	(3 ~	D /	5		Job #: 1343019	S	ite: 🕐	rte	r Pa	one
			er l	1	15	[}	RI	ſ,		Logged By: A. Leavitt Revi	ewed	By:			
				}	18	11	171	$\langle /$	50:1	Drilling Contractor: Boart Longyear	Λ	14	14		
			/	$\langle  $	31	5	1515	1	Barcourt	Drill Rig Type/Method: Senic Drilling	ha	nd	Sa	mp	lin
			1	/	31	(	115	L	POTTOS	Drillers Name: A . Lear . +	ン				
			( {		$\mathcal{N}$	$\mathcal{L}$	ا کی	2		Borehole Diam./Drill Bit Type:	То	tal De	pth	1	P
			~	>		ے		$\cup$	-		Re	ef. Ele	v.		-
				Site S	Sketc	h Map	)			Sampler Type: hand sa.	npi	'in	<u>g</u>		a
Dep	th to 1	lst Wa	ater (	Z):		~	、 Tim	e/Dat	:e: 👝	Drill Start Time/Date: Drill	Finisł	n Time	/Date	- 28	5-08
Dep	th to \	Nater	After	Drillir	ng (🔽	):	∽ Time	e/Dat	e: —	Well Completion Time/Date: -					
Dep	th to c	other \	Nater	Bear	ing Zo	ones:	<b>`</b>	-		Soil Boring Backfill Time/Date: 6-2	6-	28			
	<b>1</b>	)	6 in.	nalysi	Size			ð				Estir	nated	% O1	
	nterve	ed (in.	/ stur	for AI	ype &	Filler	eet)	il Typ					Sand		
NO/	nple I	covere	w Col	ained	sing T	snint	oth (F	CS SC	Soil Descri	ntion	Ivel	arse	dium	Ð	/clay
문	Sar	Нес	Blo	Ret	Cat	Anr	Dei	SU			0 20	Ő	Š.	Ē	ଞ୍ଚି
	X	$\mathbf{V}$						sv	Sardne	llg caded, dry, yellansh	5	25	25	70	15
	1)	$\sim$					1		brown (	1071(379), LUOSE, TOCK					
									Lu con	ded :					
							2-								
							3-								
									-Surfac	e had Minlayer of					
							4 —		Silty ~	nucleas and					
									( 9 - 4						
							5 —		. aps. c	068 - N340 06.054					
							6			W(12° 07,931'					
							7-							y de la Genera	
							8 —								
							9 								
	1.1.1.1	1.44.66													
							10		and the state of the		1.1.1.12	Sec. 1	ng si k	dia de	1.1.1.1.1
							10 —								
							10 —  11 —								

6		<u>y</u> y	ун		C.	4	1			Borir Proje	ng #: <i>\$</i> ect: SAI	<b>5-69</b> R Grou	MW#	: er Rech	SI arge Op	neet otimiza	1 tion S	of tudy	1
		Ιζ	>	P5	П	Pz	12	PI		Job #	1343	019				Site: 🖍	nte	o r	en.
55	-69	xr	. /	1	IS ,	กไ	11	7.			ed By:	A.L	par	HF F	eviewe	d By:		ŗ	
			K		) [	X	[] ?	?/		Drillin	na Con	ractor	Bear	Lonave	ar 'A	14	iH		
		10		> [	\$/{	$\langle 1 \rangle$	<i>\</i>  (	1	2011	Drill	3 Ria Tvr	e/Met	hod: 🕰	onic Dril	ling- U	and	Ca.	no	In
			15		۶le	5	\   <b>}</b>		parar	Drille	re Nam	<u>م</u> ،	$\overline{\Lambda}$	1.0a	14		<u></u>	<u> </u>	•••••
			$\left( \right)$	·   (	(1)		<u>} ا (</u>	1		Bore	hole Di	am./Di	ill Bit	Type:	1	otal D	əpth	1'	•
			U	/	Ľ		5 (	Ś						<b></b>	F	Ref. Ele	ev.		
				Site 9	Sketc	) h Map				Sam	oler Tvi	be: V	101	d sa	MP	ins		<u>serierieri</u> Alter Lasy Rassi (Ne	
Den	th to 1	lst Wa	ater (N				Tim	ne/Dat	ie:	Drill	Start Ti	no/Da	6-2	6-08	rill Finis	h Tim	o/Date	-26	200
Dep	th to \	Vater	After	Drillir	na (👿	<u></u>	<b>~</b> Tim	e/Dat	e: -	Well	Comple	ation T	ime/D	ate: =	-			•	
Dep	th to c	other \	Nater	Bear	ina Za	ones:		-		Soil	Borina I	Backfil	l Time	Date:	5-7.6	-08			
			Ŀ	ysis	eze									•		Esti	matec	1%0	f
	erval	(in.)	s/61	r Anal	e &	er	t)	Type									Sand		
VA	e Inte	ered	Count	ed fo	g Typ	III sr	(Feel	Soil -								Ð	E		٦.
PID/O	Sampl	Recov	Blow (	Retain	Casin	Annuli	Depth	SSSL	Soil Desci	ription					Grave	Coars	Mediu	Fine	Silt/cla
	$\mathbf{V}$	$\overline{V}$						5	Sandu	-e11	stad	led ,	dry	,					
	Δ.	Δ					1 1		Yellowis	46	rown	Ċı	04	25/4)	1	25	25	50	15
									[vorse,	rock	sch	'Ps	ur t	<u>e 65</u>	4				
							2 –		Suban	igula	<u>~ +</u>	ro	nne	led.		li de la co Referencia			
							3 -		-Surf	5.0	ha	Jr	hin	laver					
							· · ·		of sil	$\overline{f_{Y}}$	mna	ler	acks	Q-10	1				
							4		algap.										
							5 -					-		0					
									- apsi o	69 -	-N	340	05	, 767	;				
							6 -				<u>u</u>	2	0/.	10/					
												-							
							7 -												
							8 -												
							9 -					9264) - 14			가장 같이 이는 않으				
							-												
							10 -												

G		٧V	уң	55.	-26	8			<u>Bo</u> Pr	oring #: <b>\$\$-7</b> 2 MW#:	Sh arge Op	eet timizat	1 on S	of ludv	<u> </u>
		1		)						5,500 01 01 01 01 01 01 01 01 01 01 01 01 0	<u></u>		. 20	( D.	
		1	1		PS	Pч			<u> </u>	D#. 1343019			~7~	<u>' p</u> e	2026
		ļ	de la	1	]	21		2		gged By: /+ Ceal/17 F		1 By: ハム	14	1	<u></u>
				11	1	1 p	15	$\Lambda$		ill Dia Tuna (Mathad: Sania Dri			5		1.
				11	K	Λ	131	11			the ne	ina.	20	mp	11
				Ц	R	r	151	V		rebole Diam /Drill Bit Type:	, у с П	otal De	nth	1.	,
				U	15	ζ	$  \{   \}$	1	porror		R	ef. Ele	v.		
						y	01	5		- Nor of C	2	ha.			
		-1 1 4/		Site	SKetC	n Map		- /D -	Sa	6-26-08	<u>i mpi</u>	1.79		1-26	:-q
Jep			ater (	<u>v):</u>					le: Dr	III Start Time/Date:	Drill Finis	h lime	/Date	<b>)</b> ;	
Dep	th to v	vater	Atter	Drillir	ng ( 👿	): -		e/Dai	e: W	ell Completion Time/Date:		111			
Jep	in to c	otner \	vater	ьear	ing ∠o ∎ ₀	ones:			Sc	II Boring Backfill Time/Date: (	>-2 0-	- <u>8</u>			
	al	(·L	/6 in.	Analys	& Siz			be					natec	ı % U	
A	Inten	red (ii	ounts	d for /	Type	Fille	Feet)	ioil T <sub>y</sub>						territaria Literation	
NO/O	ample	есоле	ow Cc	etaine	asing	sninuc	epth (I	scs s	Soil Descriptio	n	ravel	oarse	edium	ne	lit/clay
<u> </u>	<u></u> М	ř V	Ē	Ĕ	Ö	Ā	ڡٞ	) )		and the Kall	0		Σ	Ē	Ŭ
	I-X-1	-X						s~	band well	stash) lourse mile	<u>5</u> 5	25	25	30	10
		<u>/                                    </u>					1-		ALOS LA	to 1,5" subarsul	-				
							 2		to rome	lel.					
							<b>-</b> 								
							3 —								
									-surface	> had Thinlaye					
							4 —		61660	mullans in	4				
							5-		- CPS: 07	0 - N34°05.78	1				
							6 —			W117007.84	5'				
							1000 - 1000 1000 - 1000 1000 - 1000								
							7 —								-
							8 —								
							9 —								
							10								
							10								
							11								
					l		12							1	1

4	Ŋ I	WN	VН						Boring #: 55-71 MW#:	Sh	eet	1	of	1
W	V "	***	<b>W 4 8</b>						Project: SAR Groundwater Rech	arge Opl	imizat	ion St	udy	
	1								Job #: 1343019 -0 0	5	Site: B	orro	w F	<u>'if</u>
			۲					NI	Logged By:	Reviewec	By:			
				1				1)[	Drilling Contractor: Beart Longye	Har-				
				1				VI	Drill Rig Type/Method: Senic Dri	ting-				
						VE			Drillers Name:					
		$\langle$	<u> </u>	$\bot$	~				Borehole Diam./Drill Bit Type:	T( R	otal De ef. Ele	epth ev.		
				Site	Sketcl	h Map			Sampler Type:					
Dep	h to 1	st W	ater ()	<b>Z</b> ):			Tim	e/Dat	Drill Start Time/Date:	<b>Drill Finis</b>	h Time	e/Date	»:	
Dep	h to V	Vater	After	Drillir	ng (👿	):	Tim	e/Dat	Well Completion Time/Date:					
Dep	h to c	ther '	Water	Bear	ing Zo	ones:			Soil Boring Backfill Time/Date:					
			i.	alysis	Size			0			Esti	matec	1%0	f
	terva	d (in.)	nts / 6	or An	pe &	iller	et)	I Typ				Sand	I	
ID/OVA	ample In	ecovere	low Coul	etained f	asing Ty	nnulus F	epth (Fe	SCS Sol	Soil Description	iravel	oarse	fedium	eu	ült/clav
ז	S	E .		н	0	A		SW	GRAVZUY SAND 10YD-5-2			4		
								Í.	SUBROUNDED TO WHL BOOM	DED 50	25	20	20	5
							1		OBBLES, BOULDERS UP 1	50				
							2 –		O-INCH DIAMETER					
							3-		GREN 1/2110 NG 8221	<u>1118</u> 동안은 1118 - 1118 - 1118 - 1118 - 1118 - 1118 - 1118 - 1118 - 1118 - 1118 - 1118 - 1118 - 1118 - 1118 - 1118 - 1118 - 1	<u>i teta.</u> Nationa			
									6)1170 07.222'					
							4					i en di si		
							5 —							
							6 -							
							7 -							
				[			8							
							9 -							1
					<b> </b>									
							10 -			<u>ar i pari</u> Generation				
	ender The sec													
				1.1.1.1.1.1.1.1	12240124	1.124.52	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	S. S		a second and the second second			and the second second	
							(* <b>11</b> –							

Л	<b>N</b>								Boring #: \$5-72 MW#:	She	et	1	of	1
U	اً ل	VI V	v F1						Project: SAR Groundwater Rechar	je Opti	mizat	on St	udy	
	1-								Job #: 1343019	Si	ite: <i>F</i>	, 70VV	ou?	Pit_
								AP	Logged By: Re	iewed	By:			
			1	$\left[ \right]$			1	$\{ \}$	Drilling Contractor: Boart Eongyear					
								) V	Drill Rig Type/Method: Sobie Drillin	₹				
				X			v		Drillers Name:					
	.	$\overline{}$					/		Borehole Diam./Drill Bit Type:	To Re	tal De ef. Ele	epth v.	<u> </u>	11
				Site S	Sketcl	h Map			Sampler Type:					
Dep	th to 1	st Wa	ater (	<b>Z</b> ):			Tim	e/Dat	Drill Start Time/Date: Dri	l Finisł	n Time	e/Date	:	
Dep	th to V	Vater	After	Drillin	ng ( <b>T</b>	):	Tim	e/Dat	Well Completion Time/Date:					
Dep	th to c	other '	Water	Bear	ing Zo	ones:			Soil Boring Backfill Time/Date:					
			ċ	lysis	ize						Esti	nated	1%0	f
	eval	(in.)	s / 6	r Ana	8 8 8	Ŀ	¢	Type				Sand		
ID/OVA	ample Inte	ecovered	low Count	etained fo	asing Typ	nnulus Fil	lepth (Fee	SCS Soil	Soil Description	åravel	Coarse	Aedium	ine	Silt/clay
<u>a</u>	S	Ē	8	Ω.	0	∢		Gul	TAN (REPUIX/ DDY CUPPODUD			<		
								L'W	Y GRASS. AT THE VET SURFACE	20	20	20	20	20
							1 -		YR-6-3 45/15/15/15/10	_				
							2 –		TONE (1) FOOT 10YR-5-2					
										-				
							3 -		PG : NZILODI GIZILI	일 (12)2종 전 (15)2종				
									() - 10 7 + 05 . 7 - 4 () 1120 07 310	-				
							4 —							
							5							
							<b>.</b>							
							6 —							
							<b>-</b> -			-				
							7 -							
							•			-				
							ð —							
							9 —							
					l de site la contra la contra		10 -							
							11 -							
				12-22										
Appendix E Photographic Logs















Customer: SBVWCD-SB	/MWD	Project Number:	1343019
Site Name: Santa Ana Riv	er Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: GEI-7			
Location:			
Direction:			
<b>Comments:</b> Water level measurement system consisted of PVC pipe, transducer, datalogger, modem, software and computer.			
Photograph ID: GEI-8			
Date: Sept. 30, 2008			
Location: Borrow Pit Pond 1 Trench	-		
Direction:	and a state of the second	THE CONTRACTOR OF STREET	
<b>Comments:</b> Linear trench of 6 feet wide by 4 feet deep in Pond 1 to test percolation rate.			



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: GEI-9 Date: March 27, 2008			
Location:	Wester and the second		
Diversion Structure	The Market Market	State of the second	TIME
Comments:			
Diversion Structure			
north gates, to the	a starting water		and the second s
existing westerly spreading ponds. Photograph ID: Date:			
Location:			
Direction:			
Comments:			











Customer: SBVWCD-SI	BVMWD	Project Number:	1343019
Site Name: Santa Ana R	iver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 5		Rive an	
Date: June 19, 2008	California and and a const	and the first state	Avenue and a state of the second
Location:	A CARLEN AND A CARLEN		and the state of the
Trench 1 in Pond 1	a state	and the second second	and and the second of the
Direction:		A The Art	Markey
Looking west	e and		
Comments:			and the second states
North-south trench,		1 4 8	Same the second second
materials at the top			
one foot, followed by	the states of the		
boulder, cobble,		and .	
gravel and coarse			
sand mixture, dry.	Can all		A CALL AND A CALL AND A CALL
Stake for scale.			
			the way in the first the
			A State of the sta
	- Carto	in the second	
Photograph ID: 6	and the second second		
Date: June 19, 2008		A Const.	A Contraction of the State
l ocation:		A LAND	AND
Trench 2 in Pond 1	the second second	A CAR	
Direction:			
Looking west		Enci	
Comments:			
North-south trench,			and the state of the
fine dark colored	the second second	CHART IN THE REAL	A CHARLES AND CONTRACT
one foot. followed by	A DECEMBER OF	Cart and A	ent to the 2
boulder, cobble,		The states	A REAL PROPERTY AND A REAL
gravel and coarse	and the second second	and the	The second second
sand mixture, dry.		and the last	10 200
Stake for scale	mut the second		
olate for soule.			
	the second second		
	the second second		a start to a start to
	and the second	Baller and a star	the second second



Customer: SBVWCD-SBVMWD	Project Number:	1343019
Site Name: Santa Ana River Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 7	A DULT A DECA	
Date: June 19, 2008	- the fi	the start and the
Location:	Hard and the	A- COMPANY
Trench 3 in Pond 2		A. A. A. A.
Direction:		
Looking west	The Carton	
Comments:	and the second	
fine dark colored	A 7 1 - 2	
materials at the top	Alt A	and the second
one foot, followed by		the survey of th
dravel and coarse	a the same a	A CONTRACT OF
sand mixture, dry.		
Stoke for coole		a strange to a start of
	Contraction of the second	and the second second second
1 Standard		
and the first state	and the Constant	A Start Start
Photograph ID: 8	01 10 - W 1 4 4	
	A Marine	
Trench 5		
Direction:		
Looking west		A CAR A CONTRACT
Comments:	NO NO	
North-south trench,		
boulder, cobble,	A STATIST	A MARCHAN TO SHARE SHARE
gravel and coarse	ALC: NO	
sand mixture, dry.	ALL YOUNG	A Carlos A Sta
Stake for scale.		A LE CALLER
and the same of the second		
Pha and a start of the start of		
		the the ment of the
		a water the second and the second sec



Customer: SBVWCD-S	BVMWD	Project Number:	1343019
Site Name: Santa Ana F	River Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 9		States 1	and the second second
Date: June 19, 2008		and the second s	and the second s
Location:		AT STATE	
Trench 6		18 2.19	
Direction:			
Looking west	and the second sec		
Comments:			
Light brown boulder,			
cobble, gravel and		all the main the	
Coarse sanu mixture, drv		Red K	
ury.		1200	
Stake for scale.		Coming Street and	
		the state	
1		All A	and the second second
1		1 pape	
1		AND CONT	
1			
1	and the start		
Bhotograph ID: 10			
	- Contraction and a	an lin	
	- Starter	State State	
Location:			Carla Alla
		and the second of	
Direction:		A Carl Land	State of the second sec
Looking west		AS 19 -	Alta -
Comments:	the first and the first of the	AND	The second second
fine dark colored		100 1 1 3 M	
materials with	S & adam 22		
grass at the top,			
followed by	1918 - 6-3		and the second s
boulder, cobble,	and the second	and the second second	and where the state of the
gravel and coarse	The second second		the state of the second
sand mixture, dry.			
Stake for scale.		A. S.	
1	The second second second	a starter and the	
1			



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 11	An and all	A MAN	
Date: June 22, 2008		and his Down	
Location:	Sale Partie	Canada Anna	
Trench 8	STATISTICS S	Land and the second	A CORDENSION OF
Direction:	and the set		
Looking north		Maria St.	
Comments:		Contraction of the	AN LOS SEL
West-east trench,	- TADA BANK		
brown boulder,			
coarse sand mixture,			
dry.		a in the	State and a state
	ATT I A PARTY		
Stake for scale.	AL- IN CONTRACT		March March March
	Real of the	A CARLER	The second se
			- Caller - Caller
	2000 91 91	in the second second	and the state of the second
Photograph ID: 12		The second se	
<b>Date:</b> June 22, 2008		and the	179
Location:	and the second	ight bet	And the second s
Trench 9	an read		Not the the state
Direction:			1 Contraction of the second se
Looking east			
Comments:	A CALLER A	A Charles	
North-south trench,		the second	
light gray, brown			A State of the
aravel and coarse		in the	the tall is the
sand mixture, dry.		Star 1	A MARKER AND A REAL
-	the state of the s		ATT 18 VILL AND
Stake for scale.		The Asian	A CALLER AND A
		Contraction of the second second	A CARLES AND A CAR
		( Contra and	A CONTRACTOR OF A CONTRACTOR OFTA CONTRACTOR O
	and the second second	ATT IN	
	Designed and	and the second second	



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 13	No. 1 Perf		AVA A
Date: June 19, 2008	A Start Start	and the second second	
Location:			
Trench 10 in Pond 10		Con Miller	North States
Direction:			
Looking north		and a state of the	
Comments:		A De Martin Part	
East-west trench,		A CONTRACT	
few fine materials		Some Sale	
at the top, followed by light brown	- Parada Ale		
boulder. cobble,			
gravel and coarse			
sand mixture, dry.		300	
		and the second	
Stake for scale.		· The same	
		A THATA	
Photograph ID: 14	Name Statistics IN THE	1. Contraption	Mary Mary Construction of the State
Date: June 19, 2008		and the state of the second	
Location:	and the second second	The present of the second s	and the state of the second
Trench 11A near wet spot	and the state of the	Proventing the second	- ALTER Y ANT
Direction:	5- 5+ 5 + · · ·	State Maria	and the same fragment
Looking west	the state of the	and the second sec	
Comments:			
East-west trench	and the state of the state	San But	A CONTRACTOR OF THE OWNER
Well sorted coarse		A CONTRACT	
followed by boulder		J'A'	
cobble. gravel and	State Street		i cherte in
sand mixture,	A CARLER OF COMPANY		and the second second
water encountered	and a figure of the	S. S. S. S.	
at a depth of less		ントレート	AND THE PARTY OF THE PARTY
than 3 feet below	and a strate		A CALLER AND
ground surface.	Ast op Real March	a second part and the second a	
Stake for scale		The light	R WESSER & S
Oldice for Sould.		a alteres	A CARLE AND CONTRACTOR
		and the second se	



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 15			
Date: June 19, 2008			
Location:			
Trench 11D		4	it
Direction:	Martin and	it was the sector	
Looking north	NY is the second	the start was	
Comments:		and the second second	A PARTICIPACITY OF THE PARTICI
East-west trench,	and the second second	ALE.	
Well sorted coarse		y to w	Mar
sand in the top layer,	Au Carlos		
followed by boulder,			
cobble, gravel and	The Card and	The Standard	
sand mixture,	The second s	The second	A REAL
water encountered	the state of the state of the	E .	the set of
at a depth of	a the second and	Participation of the second	
approximately 3 leet	Part The Part of Part		The second start
below ground surface.	Add the state		
Stake for scale.		a restance	
Photograph ID: 16			A Participation
Date: June 19, 2008		1 - A	
Location:		Reput in the	A CONTRACTOR
Trench 12		Contract of the second	
Direction:	2 August 1979 Aug		
Looking north		A CARLER OF	
Comments:			
At the western end of		Contraction and	
the Borrow Pit,		and the second second	
fine dark colored			
materials at the top			
and soil compaction	10 M 10 M 44		
is observed. Close to			And a start of the second start of the
surface ponding, no			A CALL AND A CALL
water was encountered		Carl Carl Carl	
to a depth of 6 feet			and the second s
below ground surface.	and the state of the	and the second second	
			The second s
Stake for scale.	Maria and Anna and An	T PARA MISER	
		March 1999 March 1999	



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Ri	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 17	- Martingarou !	and the second	
Date: June 18, 2008		the second	K CONTRACTOR
Location:		and the state of t	
Surface Sampling SS-13		1 Kr	
Direction:	1	-13	· · · ·
<b>Comments:</b> Hard dry cobble, gravel and sand to a depth of approximately one foot below ground surface. Stake for scale.			
Photograph ID: 18		the section of the	
Date: June 18, 2008	and they are		
Location:		a fut the contract	And the second and the
Surface Sampling SS-14		and the second second	
Direction:	A A		法在 经资金公司
<b>Comments:</b> Hard dry cobble, gravel and sand to a depth of approximately one foot below ground surface.			
Stake for scale.			



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 19	the first the second		
Date: June 18, 2008	No Reality		
Location:		A THE A	
Surface Sampling SS-15			
Direction:		Seat of the seat o	
	Star Barris	55-15	
Comments:	1	the A . they	Martin and the second
Hard dry cobble,			and a second
gravel and sand	A ARE A ARE	1 A Cate	
io a depin or approximately one			
foot below around		and the second	
surface.	The start is	and the second	
	the second of		
Stake for scale.	all the second	1.1.1.1	
	bert the states of		a the states
	ENCER STATE	Grand - Com	a surger and a set
		Edit in the	and the second second
	The stand	Hard Jun 2 1	and the second and the second se
Photograph ID: 20			the second state
Date: June 18, 2008		a Variation of the second	Shite - Charles
Location:		55-16-12-12-2-2-2-2-2-2-2-2-2-2-2-2-2-2-2-	the second second
Surface Sampling SS-16	A second se		the provide the
Direction:	and the set of the set	Current Car to	1 Alexandre
	and the state of the state		
Comments:	and the second second	10000	
malu uly cobble,	the second	1 grade to	No water and the second
to a depth of	and the the	a Vistic	the state of the
approximately one	the state of the	1 Vier Vier	
foot below ground	A CARLES CONTRACTOR	1 13 1 1	
surface.	the state of the		
Otalia far applo	and the set	i trat	and the second second
Stake for scale.	and the state of the		
	to South to the second	The same	
	A BASSING	A Provention	and a start of the second
		the the	
	and the second s	· Atta	



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 21	A Liter L		the second second
Date: June 18, 2008	and the states		
Location:	the state of the state	and the second second	
Surface Sampling SS-17	The state of the state	- Later -	
Direction:		the second	in the state
<b>Comments:</b> Hard dry cobble, gravel and sand to a depth of approximately one foot below ground surface. Stake for scale.			
Photograph ID: 22			8 the france of the
<b>Date:</b> June 18, 2008		and the second second	
Location:	States Catheren		
Surface Sampling SS-18			The second s
Direction:	the state of the		
Comments:			H
Hard dry cobble,		and the state	and the second second
gravel and sand	and the second	1: mat	
to a depth of	Jan a real		
approximately one	1- And I am	and a star	
surface			- A
Suilace.	The Section		in the state of the
Stake for scale.			















Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Ri	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 29			
Date: June 21, 2008			
Location:	ANTE SEC	Stor Partie	YELL & CAREER
Surface Sampling SS-25	ac man	de santisé	
Direction:			A DECEMBER OF CALL
		51.25 2 10 10 12	The second second second
Comments:			
<b>D</b> ry cobble, gravel			
and sand	Water and the state		
to a depth of			CARLEY ALL AND AND
approximately one			The state of the s
foot below ground			
sunace.			
Stake for scale.	A Anti-		
	Cotto 44		
		a very in the	
Photograph ID: 30		1 Tommer	
Date: June 21, 2008		A Trank	
Location:		the second second	and the second of
Surface Sampling SS-26			
Direction:	an and a section	1 43	and the second
			The second states
Comments:			
Dry cobble,	EF FLITS		
gravel and sand			
to a depth of			
approximately one	A A A A A A A A A A A A A A A A A A A		and the second of the second o
surface	and the second		
surface.			and a set of the
Stake for scale.	and the second second		
	and in the		
	and the second second		the second second
		A TON	
	and the second second		the first of the state of the s



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Ri	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 31		and the second	
Date: June 21, 2008			
Location:			
Surface Sampling SS-27			
Direction:			and the second sec
<b>Comments:</b> Dry cobble, gravel and sand			
to a depth of approximately one foot below ground surface.			
Stake for scale.		A A A A A A A A A A A A A A A A A A A	
Photograph ID: 32			
Date: June 21, 2008	2	A CAR LAN	
Location: Surface Sampling SS-28			
Direction:			
<b>Comments:</b> <b>D</b> ry cobble, gravel and sand to a depth of approximately one foot below ground surface.			
Stake for scale.			







Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 35			
Date: June 21, 2008		A Constant of the	
Location:			
Surface Sampling SS-31	Contraction of the second		and the second second
Direction:			STREET TO SHE
	A Charles		ALC: MARKEN
Comments:			
Some fine material		the second second	A CALLER AND A CALLER AND A
at the surface,		Sr-g	A CALL CALL
followed by			
gravelly sand	With the star		
to a depth of			ALL AND AN AN ALL AND
foot below around	「「「「「「」」		
surface.	Print all in the second		A MARTINE A MARTINE
our acc.		CAR LANG	
Stake for scale.			
	- the second		
	Mall	- 10 miles	States and the second
	a list in a second	And	
Photograph ID: 36			the second second
Date: June 21, 2008		To an and the for	· · · · · · · · · · · · · · · · · · ·
Location:	A MARKEN AND	and the second sec	
Surface Sampling SS-32	A CARLER AND A MARKEN	and the starter	a
Direction:		the states	
		Part Cas	
Comments:		and an los	- its make - 2
Some fine material		A A A A A A A A A A A A A A A A A A A	
at the surface,	NAME AND LET		
followed by		A Start A	Carlos A Car
gravelly sand			
approximately one			
foot below around	and the second second		The second s
surface.			
	The second second		
Stake for scale.			
		and the stand of the	A A A A A A A A A A A A A A A A A A A
	A MARTINE A	The second second	
	acceptor with		



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 37		and the second	
Date: June 21, 2008			The second second
Location:	and a second second		A AND THE THE
Surface Sampling SS-33	and the second second	and the D	the line of
Direction:	Constant in		
	A The state of the state of the	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
Comments:	the second s		A State of the second
Hard dry cobble,	A state of state of state of state	图 新生命 1	
to a denth of		and the second	Reality of the street of
approximately one	1 And a start		
foot below ground		The second	
surface.		The same of the	Sol
Stake for scale.	AT A PARA		
		P Propage	All ALESS
		prida Maria	A DEC DECT
		State Series	and the state of the
	a find the parties of the		
Photograph ID: 38			
Date: June 21, 2008		The fails	
Location:		- Line -	The second second
		A state Leak	
Direction:			
Comments:			a function of the second
Hard dry cobble,			
gravel and sand		STATISTICS AND	and the second second
to a depth of	and the street street a state		
approximately one		AN CAR	a state of the second second
surface	The section of the se	ALL S.	
Sundee.	A set of a set	THE WAY	The second states and second
Stake for scale.	A A A A A A A A A A A A A A A A A A A		E A COLOR
	and the last	a la	A REAL MARK STREET
	Contraction of the	16	
			4
	A A Test	and a start	The state of the state of
	and the second second	The Bring of the	and the second sec



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 39			AN MERICANTA 3
Date: June 21, 2008	A Charles Charles	all of the second second	
Location:		Kan	the second second second
Surface Sampling SS-35			the second second
Direction:			the second
<b>Comments:</b> Hard dry cobble, gravel and sand to a depth of approximately one foot below ground surface. Stake for scale.			
		A STA	
Photograph ID: 40	教会 化内心不定	Product of the	
Location: Surface Sampling SS-36			O. C. Maria
Direction:		236	
<b>Comments:</b> Dry cobble, gravel and sand to a depth of approximately one foot below ground surface. Stake for scale.			
		Con the	







Customer: SBVWCD-SBVMWD		Project Number:	1343019	
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA	
Photograph ID: 43				
Date: June 21, 2008				
Location:				
Surface Sampling SS-39				
Direction:	State States			
Comments:	A set of the			
Dry cobble,			NY SELLET STATES	
gravel and sand		55-39		
to a depth of			and the second second	
approximately one				
eurface			TRACK STOLEN	
Sunace.				
Stake for scale.				
		Series And		
		S. C. C.		
		Children and		
Photograph ID: 44			and the second	
Date: June 22, 2008			A CONTRACT OF A CONTRACT. CONTRACT OF A CONTRACT. CONTRACT OF A CONTRACT. CONTRACT OF A CONTRACT. CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT OF A CONTRACT. CONTRACT OF A CONTRACT OF A CONTRACT	
Location:	A Charles All Tak			
Surface Sampling SS-40	A Constant	X X X X X X X	L. Star 72 Marks	
Direction:			C SPACE AND AND THE	
Commontos		State Contractor		
Dry cobble				
dravel and sand				
to a depth of				
approximately one				
foot below ground				
surface.	CAN KARAKAN			
Otalia far applo			A State of the second state of the	
Stake for scale.	State And	Carlos and		
	A 14 14		A A A A A A A A A A A A A A A A A A A	
	The state of the state	ANY CONTRACT		
	A CAR ALE AND		Provide States of States o	
	A CALL AND A CALL AND A CALL AND A			







Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	/er Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 47	The second second second	los	L. A . L . A . A .
Date: June 22, 2008	and and the second of the		
Location:	Ant P Julie	And Street	and the second second
Surface Sampling SS-43		2 . S. /	Les an
Direction:			A State Share Share
Comments:		Stra Sta	
Some fine material	the part of the		
at the surface,			
followed by			and the second second
to a depth of			
approximately one			
foot below ground		and the second second	
surface.			
Stake for scale.		Contraction of the second	
		the second second	
			and the second second
Photograph ID: 48	and the second		10 - 10 - 10 - 10 - 10 - 10 - 10 - 10 -
Date: June 22, 2008			
Location:			
Surface Sampling SS-44			
Direction:	1 Partie	and the second second	A REAL PROPERTY AND INCOMENTS
Comments:			
Some fine material		and the second	and the second
at the surface,		1	
followed by		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
gravelly sallu			
approximately one		C. S. C.	
foot below around	Call Carl Carl		
surface.	R.L.	and the second	Mr. Carlos
		In the	
Stake for scale.			
		Constant of the	
	1 Martin	and the second	



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Ri	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 49	and the second second	the second	
Date: June 22, 2008			
Location:			The second se
Surface Sampling SS-45	and the seat of		
Direction:	STATION STORE	2 Mar Der	
	B		Carlos and the second
Comments:	a satisfield of a	14 3 A T V	
Hard dry cobble, gravel and sand	1	The second se	1 432
to a depth of	and the second sec	A CAR AN A	A CALL STATING
approximately one	· · · · · · · · · · · · · · · · · · ·	A CONTRACTOR	All as a second and a second
foot below ground			P. A.
surface.		San And I am	Real Bar
Stake for scale.		AL.	
		· · · · · · · · · · · · · · · · · · ·	North Contraction of the second
	A DAMA AND	11	
	and the second second	018 (S. 17)	and a start of the start of the start of the
	States and the states		
Photograph ID: 50			
Date: June 22, 2008		These August	
Location:	and the second sec		And and the second
Surface Sampling SS-46	and the second s		
Direction:	133 A HERE LOAD		
Commonts:	a- h		
Hard drv cobble.			the last the state of the state
gravel and sand	L S SPECE		A CONTRACTOR OF THE REAL OF TH
to a depth of			
approximately one	A ALL		
foot below ground	D		
surface.	ALC LAS MA		all the second
Stake for scale.			
		the second second	
			A CALLER AND A CAL
		ALL DUE TO BE	A TANK A AND A AND
	and a fair of the		
	Terra M. C. States	ATTEN AND AND A	



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 51			
Date: June 22, 2008		Contract in the	Note that the second
Location:			
Surface Sampling SS-47			
Direction:			The Part Real
	6.25		
Comments:	Prove and the		
Hard dry cobble,	and the second second		
gravel and sand			
to a depth of			and the second sec
approximately one			
foot below ground	Les the state	Talle And	
surface.		A start	ALT PAR STATIST
Stake for scale.			
			and the second second second
	a lateral and to		
Photograph ID: 52		MAN DEAL	N ANTER CONTRACTOR
Date: June 22, 2008			
Location:			The second for the file
Surface Sampling SS-48	Aug to the total	OF A ANT	
Direction:			
	P. M. Marson	A BEAR	A REAL AND A
Comments:			
Drv boulder and		ST C BUT	
cobble. less		A seal	and a share of the size
gravel and sand			
to a depth of			
approximately one		L'AN	You want
foot below ground	Trank when	and the second s	and the second second
surface.		Sold States	The Alexander
	2	The second	
Stake for scale.	A A A A A A A A A A A A A A A A A A A		
	The I have	Company of the	PARA NENE
			FOR ADDER
		The PP	
	AL IN THE ALL	STREET STREET	
		P J Presto and	


















Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 61		Pert yo	
Date: June 23, 2008		and the second second	
Location:	stored to the	Spinster La	
Surface Sampling SS-57	The second is		
Direction:			
Comments: Hard dry cobble.	TE -		
gravel and sand			A LE CALLER AND A LE CALL
to a depth of	- A BARRAN		A PART CARDEN
approximately one			
foot below ground		LARE NO.	
Sundee.			
Stake for scale.		mar i	
		VAN DE	
	A CALL CONTRACTOR AS	不常会副学校	
		Alter Contractor	
Photograph ID: 62	NY CARE	STEL \$	
Date: June 24, 2008	YIT	人下令外在的	Martin Contraction
Location:		YAX	A State A State A
Surface Sampling SS-59	- All	T. T. C.	
Direction:	En la fili	. 233	A MARCH AND THE MARCH AND AND A
Comments:		A STATE OF	
Some fungas on	The Fall	CONTRACTOR	A CARLENS AND AL
the surface,			Start Curris - Set Der -
followed by gravelly			
sand to a depth of			PIR THE SHAPPING
approximately one	and the second	March 1	
foot below ground	C + F		A MARY MARY
surrace.	AND	- Caller Constant	A sale as a
Stake for scale.			のからないのですか
	A CARLON AND A CARLON	States and Street	A CARLE AND
		A CARLER AND	Charles -
		Varia Andre -	SHA CHERRY THE



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 63	Billion El Ma	And Mark	light light
Date: June 24, 2008	The second		
Location:	· Martin State		All and parts
Surface Sampling SS-60	R SKOV	and Merel	
Direction:	V BAN	「人」の一面	the section
	Y SE TH		
Comments:	1 1 A 38 1 5 1 20 1	Children Maria	
Some fungas on		A THE AND	The second second second
followed by gravelly		and the second	Mary Mary Mary
sand to a depth of	and the second second	A COLOR OF THE REAL OF THE	a the second
approximately one			
foot below ground		a station	All A Pro in its
Suilace.	El CAPATELON	L. C. Star	All had the
Stake for scale.	in the second second	A Man &	
	2 Martin Bury Do	1. 1965 ( \$ 9 %	Sand States and States
	A CONTRACTOR OF CONTRACTOR	is sold	
	CREW MA		A THINK BELLED
Photograph ID: 64	Principal &		
Date: June 24, 2008			
Location:	the second se	- Wester And Part	The second of the
Surface Sampling SS-62	and the second		
Direction:	Contraction of the second	A CONTRACT	
Comments:		1 Martin San	
Some fungas on		A PACKAGE	The Andrew States
the surface,		S APPER	
followed by gravelly			
sand to a depth of		The second second	
foot below ground	- The second second	1 / 2 × 4 × 2 0	and the company
surface.			A STATE OF STATE
	the state		A CALL AND
Stake for scale.	the strings and the	Seat 1	Mar Aller
	A A A	E E COME	
		ALL STORE	all she was
			and the second second



Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Photograph ID: 65		and the second	and the second states and the
Date: June 24, 2008	A Bad and		Contraction of the second
Location:	The set of the set	Soila	A the second sec
Surface Sampling SS-63	13-24 plassing	Seres 100 raw	and the second second
Direction:	ATT A MAN	06-70	The second second
Comments:			and the state of the state of the
Dry fine materials			and and a start of the
followed by gravelly		0000	
sand to a depth of	1 For Star The	ALC: NO	and the second second
approximately one			
surface.			A Charles and
	Fit The Direction	President and a second	K. C. A. L. S. C. S.
	AUX AND A LOC		
	and the state of t	K HAR HE WILL	· · · · · · · · · · · · · · · · · · ·
	lest of the	S MORE Y	
Photograph ID: 66	THE PARTY CONTRACT	ACHA-F-	AVAN AVAN
Date: June 24, 2008	AT CHINA	and the	C ALL
Location:	THAT AND	A ATAT	7 - 61 - 0 - 12 - 1
Surface Sampling SS-65	and and	- Cart	
Direction:			Att Link the
Comments:			
Dry fine materials	30 1 10 10 10		
at the surface,			
followed by gravelly sand to a depth of	アナイトアン		A AND AN AN A
approximately one	X-Y-T-LYH		A LANDER G
foot below ground	大人への強持		The state of the s
surface.	SATTA B	A State State	the state of the
	27120	CARLAND CONTRACTOR	
	1-DASCO (M		Contraction of the second
	sound (	A DEST	
		All Della - C	A A A
	ATT THE TYPE		and the set of the set



Site Name: Santa Ana River Borrow Pit       Site Location:       San Bernardino County, CA         Photograph ID: 67       Date: June 24, 2008       Image: Constraint of the surface sampling SS-66         Direction:       Comments:       Some fungas at the surface, followed by silty sand with gravel and cobble to a depth of approximately one foot below ground surface.       Image: Constraint of the surface sampling SI and surface.         Photograph ID: 68       Image: Date: June 24, 2008       Image: Constraint of the surface sampling SI and surface.	Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Photograph ID: 67         Date: June 24, 2008         Location:         Surface Sampling SS-66         Direction:         Comments:         Some fungas at the surface, followed by silty sand with gravel and cobble to a depth of approximately one foot below ground surface.         Photograph ID: 68         Date: June 24, 2008	Site Name: Santa Ana Ri	ver Borrow Pit	Site Location:	San Bernardino County, CA
Date: June 24, 2008         Location:         Surface Sampling SS-66         Direction:         Comments:         Some fungas at the surface, followed by silty sand with gravel and cobble to a depth of approximately one foot below ground surface.         Photograph ID: 68         Date: June 24, 2008	Photograph ID: 67	time		
Location: Surface Sampling SS-66 Direction: Comments: Some fungas at the surface, followed by silty sand with gravel and cobble to a depth of approximately one foot below ground surface. Photograph ID: 68 Date: June 24, 2008	Date: June 24, 2008			And the state
Surface Sampling SS-66         Direction:         Comments:         Some fungas at the surface, followed by silty sand with gravel and cobble to a depth of approximately one foot below ground surface.         Photograph ID: 68         Date: June 24, 2008	Location:	and the second		
Direction: Comments: Some fungas at the surface, followed by silty sand with gravel and cobble to a depth of approximately one foot below ground surface. Photograph ID: 68 Date: June 24, 2008	Surface Sampling SS-66		ALL ALL ALL	the state of the s
Comments:         Some fungas at the surface, followed by silty sand with gravel and cobble to a depth of approximately one foot below ground surface.         Photograph ID: 68 Date: June 24, 2008	Direction:	and the second	and the second second	a la la la la la
Comments:         Some fungas at the surface, followed by silty sand with gravel and cobble to a depth of approximately one foot below ground surface.         Photograph ID: 68         Date: June 24, 2008			and the state	
Some fungas at the surface, followed by silty sand with gravel and cobble to a depth of approximately one foot below ground surface. Photograph ID: 68 Date: June 24, 2008	Comments:		PARKER .	KLAPA ANA -
At the surface, followed by silty sand with gravel and cobble to a depth of approximately one foot below ground surface. Photograph ID: 68 Date: June 24, 2008	Some fungas		A PARTY AND	
silty sand with gravel and cobble to a depth of approximately one foot below ground surface. Photograph ID: 68 Date: June 24, 2008	at the surface, followed by			
gravel and cobble         to a depth of         approximately one         foot below ground         surface.         Photograph ID: 68         Date: June 24, 2008	silty sand with	A REALING		
to a depth of approximately one foot below ground surface. Photograph ID: 68 Date: June 24, 2008	gravel and cobble	a set dati per		and the second
approximately one       foot below ground         surface.       Image: Comparent of the second	to a depth of		A Mail	Sector ( )
surface.  Photograph ID: 68 Date: June 24, 2008	foot below ground		and in the	
Photograph ID: 68 Date: June 24, 2008	surface.		- North And	
Photograph ID: 68 Date: June 24, 2008			4.15 × 24	
Photograph ID: 68 Date: June 24, 2008		and the second second	Lat a size	
Photograph ID: 68           Date: June 24, 2008				
Date: June 24, 2008	Photograph ID: 68	ASP - A COM	in the first of	
	Date: June 24, 2008			The Alter and
Location:	Location:		SY LA	
Surface Sampling SS-67	Surface Sampling SS-67		AL MA	LAC- A Jan
Direction:	Direction:			
Comments:	Comments:		and the	
Dry gravelly sand	Dry gravelly sand	C Sylver a	The Hard	LISC & AL
with cobble to a depth of	with cobble	CONSIGNOUTH		the states
approximately one	approximately one			the second se
foot below ground	foot below ground	and the		
surface.	surface.	E FLANE	CALL REAL	
			the truck	
			四方"学"	
			100 - M	A A A A A A A A A A A A A A A A A A A
		St X Station	1 Shanne	AT A REAL FROM L







Site Name: Santa Ana River Borrow Pit       Site Location:       San Bernardino County, CA         Photograph ID: 71       Date: June 24, 2008       Image: Contract of the second sec	Customer: SBVWCD-SB	VMWD	Project Number:	1343019
Photograph ID: 71         Date: June 24, 2008         Location:         Surface Sampling SS-70         Direction:         Comments:         Some fungas at the surface, followed by gravelly sand with coblest to a depth of approximately one foot below ground surface.         Shavel for scale.         Photograph ID:         Date:	Site Name: Santa Ana Riv	ver Borrow Pit	Site Location:	San Bernardino County, CA
Date: June 24, 2008         Location:         Surface Sampling SS-70         Direction:         Comments:         Some fungas         at the surface,         followed by         gravelly sand         with cobblest         to a depth of         approximately one         foot below ground         surface.         Shavel for scale.	Photograph ID: 71			and the second second
Location: Surface Sampling SS-70 Direction: Comments: Some fungas at the surface, followed by gravelly sand with cobblest to a depth of approximately one foot below ground surface. Shavel for scale. Photograph ID: Date: In control of the state of th	Date: June 24, 2008	A LES		
Surface Sampling SS-70         Direction:         Comments:         Some fungas         at the surface,         followed by         gravelly sand         with cobblest         to a depth of         approximately one         foot below ground         surface.         Shavel for scale.	Location:	2 Ph A		and the second s
Direction:         Comments:         Some fungas         at the surface,         followed by         gravelly sand         with cobblest         to a depth of         approximately one         foot below ground         surface.         Shavel for scale.	Surface Sampling SS-70			and the second
Comments:         Some fungas         at the surface,         followed by         gravelly sand         with cobblest         to a depth of         approximately one         foot below ground         surface.         Shavel for scale.	Direction:	NON CON		Sitter 1
Comments:         Some fungas         at the surface,         followed by         gravelly sand         with cobblest         to a depth of         approximately one         foot below ground         surface.         Shavel for scale.				
Some fungas at the surface, followed by gravelly sand with cobblest to a depth of approximately one foot below ground surface. Shavel for scale. Photograph ID: Date:	Comments:	「「「「「「「」」		
at the surface, followed by gravelly sand with cobblest to a depth of approximately one foot below ground surface. Shavel for scale. Photograph ID: Date:	Some fungas		that the way of	a service of the serv
followed by gravelly sand with cobblest to a depth of approximately one foot below ground surface. Shavel for scale. Photograph ID: Date:	at the surface,			Mu Contraction of the second
gravelly sand with cobblest to a depth of approximately one foot below ground surface. Shavel for scale. Photograph ID: Date: Lease time	followed by	1 A REAL AND		to de the second second
with cobblest to a depth of approximately one foot below ground surface.          Shavel for scale.         Photograph ID:         Date:	gravelly sand	a state the last		
to a depth of approximately one foot below ground surface. Shavel for scale. Photograph ID: Date:	with cobblest	CARE TO THE	Mining of States	the second states
approximately one foot below ground surface. Shavel for scale. Photograph ID: Date:	to a depth of	1 - the set	Start and	
foot below ground surface. Shavel for scale. Photograph ID: Date:	approximately one	to the first in the	The stand	A SC AND A ST
surface. Shavel for scale. Photograph ID: Date:	foot below ground	The Heart Heart	Carlo State	the state of the s
Shavel for scale.  Photograph ID: Date: Leasting	surface.	1 on Start Ser		
Photograph ID: Date:	Shavel for scale.	BE VIE	a QC	Honoren -
Date:	Photograph ID:			
	Date:	]		
	Location:			
Direction:	Direction:	-		
Comments:	Comments:	-		

Appendix F Monitoring Well As-Built and Lithology Logs



#### Monitoring Well #2 As-Built Well Diagram Graphic Log Ground Level Ground Level 10 10 ()Bentonite Seal (10') $\odot$ 20 20 Sandy Gravel, subrounded to well Sand filter pack rounded, dry, brown $\odot$ 30 30 40 40 Casing (12'-102') Gravelly Sand, brown / gray. 50 50 60 60 Sandy Gravel with Boulders 70 70 and pockets of Clayey Silt 80 80 90 90 Gravelly Sand with Boulders, brown 100 100 Total depth: 102 feet Drilled by Boart Longyear by Sonic Drilling. Notes: Well installed June 26, 2008 Latitude: 34,0969 Longitude: -117.11825 🜐 MWH

WHITER CONSERVATION DISTRIC		Lithologic Log and As-Built Well Completion Diagram for MW-2	()) MWH
OUR NAME IS OUR MISSION	Document: Appendix_E- 2_SBVWCD_MW-2.pub	Appendix F-2	
alley with artic	Date: December 22, 2008	Santa Ana River Groundwater Recharge Optimization Study	GEI Cesultants

#### Monitoring Well #3

As-Built Well Diagram

Graphic Log



## Sonic Boring #1







Appendix G Hydrographs













# Appendix C

**Cost Estimate Details** 

#### Construction Costs for Recharge Facilities - 300 cfs

Item No.	Description	Units	Quantity	Unit Price	Total
	Common				
1	Mobilization (~ 10% of Construction Costs)	LS	1		\$ 257,190
	Intake/Cuttle Weir Modifications				
2	Modify Gate No. 3 on Intake	EA	1	\$ 1,000	\$ 1,000
3	Channel Cleaning and Re-Grading	CY	122	\$ 24	\$ 2,928
4	Gabion Wall	LF	27	\$ 238	\$ 6,426
5	Log boom	LS	1	\$ 2,032	\$ 2,032
	Main Canal Modifications				
6	Canal Excavation - From Diversion to End of Phase II	CY	50,643	\$ 4.75	\$ 240,554
7	10 ft drop structure	EA	5	\$ 26,101	\$ 130,505
8	Pond Turnout Structures (30")	EA	3	\$ 12,674	\$ 38,022
9	Pond Turnout Structures (36")	EA	6	\$ 14,131	\$ 84,786
10	30-inch RCP, plus Trench	LF	150	\$ 147	\$ 22,050
11	36-inch RCP, plus Trench	LF	300	\$ 199	\$ 59,700
	Recharge Ponds				
12	Recharge Pond Excavation - Phase II	CY	420,270	\$ 4.75	\$ 1,996,283
	Sub-total				\$ 2,841,476
13	Construction Contingency (25% of Construction)	\$	\$ 2,841,476	25%	\$ 710,000
	Total Construction Cost				\$ 3,551,476

Notes:

1. Construction Costs shown are based on coceptual level design drawings and analysis.

2. Construction Contingency applied is standard industry for conceptual level cost estimating. Contingency accounts for items not known at conceptual level design phase.

3. This cost estimate of the conceptual design is a Class 5 Cost Estimate as defined by the Association for the Advancement of Cost Engineering. The MWH team has no control over costs of construction labor, materials, competitive bidding environments and procedures, unidentified field conditions, financial and/or market conditions, or other factors likely to affect the Opinion of Probable Construction Cost developed, all of which are and will unavoidably remain in a state of change, especially in light of the high volatility of the market attributable to Acts of God and other market events beyond the control of the parties. This cost estimate prepared during the work is a "snapshot in time" and that the reliability of this Opinion of Probable Construction Cost will inherently degrade over time.

Construction Costs for Recharge Facilities - 500 cfs

Item No.	Description	Units	Quantity	Unit Price	Total
	0	-			
4	Common	1.0			<b>*</b> 700 750
- 1		LS	1		\$ 708,753
	Intaka/Cuttle Wair Madifications	-			
2	Modify Gate No. 3 on Intake	FΔ	1	\$ 1,000	\$ 1,000
3	Channel Cleaning and Re-Grading	CY	122	\$ 1,000	\$ 2,928
4	Gabion Wall	LF	27	\$ 238	\$ 6.426
5	Log boom	LS	1	\$ 2.032	\$ 2.032
6	Raise Cuttle Weir (rock masonry)	CY	29	\$ 532	\$ 15.428
7	Sluice Gate and Related Installation Work	LS	1	\$ 36,284	\$ 36,284
8	O&M Platform	LS	1	\$ 8,384	\$ 8,384
9	OSHA Improvements (Caged Ladder and Handrail)	LS	1	\$ 2,720	\$ 2,720
	Sandbox Modifications				
10	Remove existing gate	EA	2	\$ 500	\$ 1,000
11	Install 5 ft x 6 ft sluice gate	EA	2	\$ 30,000	\$ 60,000
	Main Canal Modifications				-
12	Canal Excavation	CY	194,039	\$ 4.75	\$ 921,685
13	5 ft drop structure	EA	1	\$ 19,576	\$ 19,576
14	10 ft drop structure	EA	35	\$ 26,101	\$ 913,535
15	Pond Turnout Structures (18")	EA	1	\$ 9,947	\$ 9,947
16	Pond Turnout Structures (24")	EA	2	\$ 11,298	\$ 22,596
17	Pond Turnout Structures (30")	EA	9	\$ 12,674	\$ 114,066
18	Pond Turnout Structures (36")	EA	7	\$ 14,131	\$ 98,917
19	18-Inch RCP, plus Trench		50	\$ 73	\$ 3,650
20	24-Inch RCP, plus Trench		100	C 6 6	\$ 8,500
21	26 inch RCP, plus Trench		430	φ 147 ¢ 100	\$ 60,130
22		LI	330	φ 199	\$ 09,030
	Greenspot Road Culvert Modification				
23	6 ft x 6 ft Precast Concrete Box Culvert, plus Trench	LF	100	\$ 707	\$ 70,700
24	Concrete transition structures	CY	48	\$ 1.200	\$ 57.600
	Diversion Structure Modification				
25	Excavation	CY	65	\$ 17	\$ 1,105
26	Concrete retaining walls	CY	36	\$ 1,200	\$ 43,200
27	Grouted rock floor	SY	130	\$ 123	\$ 15,990
28	North and South Gate concrete diverging cones	CY	3	\$ 1,200	\$ 3,600
29	South Gate concrete structure	CY	9	\$ 1,200	\$ 10,800
30	5 ft x 8 ft gate	EA	1	\$ 35,000	\$ 35,000
31	Existing Gate Modifications	EA	3	\$ 2,000	\$ 6,000
		-			
	Recharge Ponds				
32	Recharge Pond Excavation - Phase II	CY	526,140	\$ 4.75	\$ 2,499,165
33	Recharge Pond Excavation - Phase III	CY	412,610	ə 4.75	۵ 1,959,898 ¢
	Sub total	-			¢ 7,700,005
	อนมะเงเลเ				φ /,/96,285
3/	Construction Contingency (25% of Construction)	¢	\$ 7 796 295	25%	\$ 1 9/0 000
54		φ	ψ 1,150,200	23 /0	φ 1,549,000
	Total Construction Cost	1			\$ 9,745,285
		1			

Notes:

 Construction Costs shown are based on conceptual level design drawings and analysis.
 Construction Contingency applied is standard industry for conceptual level cost estimating. Contingency accounts for items not known at conceptual level design phase.

3. This cost estimate of the conceptual design is a Class 5 Cost Estimate as defined by the Association for the Advancement of Cost Engineering. The MWH team has no control over costs of construction labor, materials, competitive bidding environments and procedures, unidentified field conditions, financial and/or market conditions, or other factors likely to affect the Opinion of Probable Construction Cost developed, all of which are and will unavoidably remain in a state of change, especially in light of the high volatility of the market attributable to Acts of God and other market events beyond the control of the parties. This cost estimate prepared during the work is a "snapshot in time" and that the reliability of this Opinion of Probable Construction Cost will inherently degrade over time.

#### Construction Costs for Recharge Facilities - 500 cfs (Alternate)

Item No.	Description	Units	Quantity	Unit Price	Total
	Common				
1	Mobilization (~ 10% of Construction Costs)	LS	1		\$ 709,787
	Intake/Cuttle Weir Modifications (ALTERNATE MODIFICATIONS)				
2	ORM Diatform	15	1	¢ 4.538	¢ 4.538
3	Aiustable Weir and Related Improvements	LS	1	\$ 81,000	\$ 81,000
	Ajustable Well and Related improvements	10	1	φ 01,000	φ 01,000
	Sandbox Modifications				
4	Remove existing gate	EA	2	\$ 500	\$ 1,000
5	Install 5 ft x 6 ft sluice gate	EA	2	\$ 30,000	\$ 60,000
	-				
	Main Canal Modifications				
6	Canal Excavation	CY	194,039	\$ 4.75	\$ 921,685
7	5 ft drop structure	EA	1	\$ 19,576	\$ 19,576
8	10 ft drop structure	EA	35	\$ 26,101	\$ 913,535
9	Pond Turnout Structures (18")	EA	1	\$ 9,947	\$ 9,947
10	Pond Turnout Structures (24")	EA	2	\$ 11,298	\$ 22,596
11	Pond Turnout Structures (30")	EA	9	\$ 12,674	\$ 114,066
12	Pond Turnout Structures (36")	EA	7	\$ 14,131	\$ 98,917
13	18-inch RCP, plus Trench	LF	50	\$ 73	\$ 3,650
14	24-inch RCP, plus Trench	LF	100	\$ 85	\$ 8,500
15	30-inch RCP, plus Trench	LF	450	\$ 147	\$ 66,150
16	36-inch RCP, plus Trench	LF	350	\$ 199	\$ 69,650
	One service of Deep I Outpart Madification				
47	Greenspot Koad Culvert Modification		100	<b>* 707</b>	<b>*</b> 70,700
1/	6 ft x 6 ft Precast Concrete Box Culvert, plus Trench		100	\$ 707	\$ 70,700
18	Concrete transition structures	CY	48	\$ 1,200	\$ 57,600
	Diversion Structure Modification	+ -			
19	Excavation	CY	65	\$ 17	\$ 1,105
20	Concrete retaining walls	ĊY	36	\$ 1.200	\$ 43.200
21	Grouted rock floor	SY	130	\$ 123	\$ 15,990
22	North and South Gate concrete diverging cones	CY	3	\$ 1,200	\$ 3,600
23	South Gate concrete structure	CY	9	\$ 1,200	\$ 10,800
24	5 ft x 8 ft gate	EA	1	\$ 35,000	\$ 35,000
25	Existing Gate Modifications	EA	3	\$ 2,000	\$ 6,000
	Recharge Ponds				
26	Recharge Pond Excavation - Phase II	CY	526,140	\$ 4.75	\$ 2,499,165
27	Recharge Pond Excavation - Phase III	CY	412,610	\$ 4.75	\$ 1,959,898
	Sub-total				\$ 7,807,655
28	Construction Contingency (25% of Construction)	\$	¢ 7,807,655	25%	¢ 1,952,000
20		Ψ	φ 1,001,000	20 /0	φ 1,302,000
	Total Construction Cost				\$ 9,759,655
					.,,

Notes:

1. Construction Costs shown are based on conceptual level design drawings and analysis.

2. Construction Contingency applied is standard industry for conceptual level cost estimating. Contingency accounts for items not known at

conceptual level design phase.

3. This cost estimate of the conceptual design is a Class 5 Cost Estimate as defined by the Association for the Advancement of Cost Engineering. The MWH team has no control over costs of construction labor, materials, competitive bidding environments and procedures, unidentified field conditions, financial and/or market conditions, or other factors likely to affect the Opinion of Probable Construction Cost developed, all of which are and will unavoidably remain in a state of change, especially in light of the high volatility of the market attributable to Acts of God and other market events beyond the control of the parties. This cost estimate prepared during the work is a "snapshot in time" and that the reliability of this Opinion of Probable Construction Cost will inherently degrade over time.

**Construction Costs for Recharge Facilities - Optional Facilities** 

Item No.	Optional Facilities Description	Units		Quantity	ı	Jnit Price		Total
	Trashrack Structure							
1	Trashrack Structure	LS	-	1	\$	144,646	\$	144,646
-	Sub-total - Optional Facilities						\$	144,646
2	Construction Contingency (25% of Construction)	\$	\$	144,646		25%	\$	36,000
	Total Construction Cost						¢	180.646
							÷	100,040
	Sedimentation Ponds							
	Presid Everyweiting	07		400.044	•	4.75	¢	0.014.077
2	Canal Excavation - To and From Ponds	CY	-	400,311	¢ Q	4.75	ф Ф	2,214,977
3	Concrete - For Overflow and Gate Structures	CY	-	100	\$	1.200	\$	120.000
4	72" Canal Gates	EA	1	6	\$	26,000	\$	156,000
5	72" RCP	LF	1	360	Ŝ	752	\$	270,720
6	42" RCP	LF		440	\$	232	\$	102,080
7	Rip Rap, non-grouted	SY		150	\$	93	\$	13,950
	Purk total - Optional Excilizion						¢	0.004.050
	Sub-total - Optional Facilities		-				ð	2,901,952
8	Construction Contingency (25% of Construction)	\$	\$	2,901,952		25%	\$	725,000
	Total Construction Cost						\$	3,626,952
			-					
	Lower South Canal							
	Borrow Pit Ponds 1. 2. and 3							
1	Canal Excavation - From South Diversion to Pond 1	CY		6,912	\$	4.75	\$	32,832
2	Pond 1 Inlet Structure	EA		1	\$	316,292	\$	316,292
3	Pond Overflow Structure	EA		2	\$	205,467	\$	410,934
	Borrow Pit Canal		-					
4	Pond 3 Discharge Structure	LS	1	1	\$	218,360	\$	218,360
5	72-inch RCP	LF	1	730	\$	535	\$	390,550
6	Canal Excavation	CY		166,394	\$	4.75	\$	790,369
7	5 ft drop structure	EA		4	\$	19,576	\$	78,304
8	10 ft drop structure	EA		12	\$	26,101	\$	313,212
9	Pond Turnout Structures (18")	EA		1	\$	9,947	\$	9,947
10	Pond Turnout Structures (24")	EA		2	\$	11,298	\$	22,596
11	Pond Turnout Structures (30")	EA		9	\$	12,674	\$	114,066
12	Pond Turnout Structures (36")	EA		7	\$	14,131	\$	98,917
13	18-inch RCP, plus Trench	LF	-	50	\$	73	\$	3,650
14	24-inch RCP, plus Trench		_	100	\$	85	\$	8,500
15	30-Inch RCP, plus Trench		-	450	¢	147	¢ Q	60,150
10	So-Inch RCP, plus Hench	LF		350	¢	199	Ð	69,650
	Borrow Pit Diversion Structure		L					
17	Diversion Structure	LS		1	\$	381,952	\$	381,952
18	Rip-rap, grouted	CY	-	83	\$	123	\$	10,209
	Borrow Pit Overflow Structure		ŀ					
19	Excavation	CY	L	7,000	\$	17	\$	119,000
20	Rip-Rap, grouted	CY	1	1,100	\$	123	\$	135,300
21	Concrete weir	ĊY	L	34	\$	1,200	\$	40,800
	Sub-total		E				\$	3,631,590
22	Construction Contingency (25% of Construction)	\$	\$	3,631,590		25%	\$	908,000
	Total Construction Cost						\$	4,539,590
		1	t				Ť	.,000,000

Notes:

1. Construction Costs shown are based on conceptual level design drawings and analysis.
2. Construction Contingency applied is standard industry for conceptual level cost estimating. Contingency accounts for items not known at conceptual level design phase.
3. This cost estimate of the conceptual design is a Class 5 Cost Estimate as defined by the Association for the Advancement of Cost Engineering. The MWH team has no control over costs of construction labor, materials, competitive bidding environments and procedures, unidentified field conditions, financial and/or market conditions, or other factors likely to affect the Opinion of Probable Construction Cost developed, all of which are and will unavoidably management of the second of th inherently degrade over time.